

# UNIVERSITÀ DEGLI STUDI DI PARMA

### DOTTORATO DI RICERCA IN MATEMATICA

### CICLO XXXV

# Nonlinear problems with biological and physical insights

Coordinatore: Prof.ssa Alessandra Lunardi

Tutore: Prof.ssa Michela Eleuteri

Dottoranda: Erica Ipocoana

Anni Accademici 2019/2020 - 2021/2022

# Abstract

Nonlinear models are a fundamental tool related to mathematical analysis. This is due to the wideness of problems which might be formulated according to them and the corresponding questions which arise concerning e.g. existence, uniqueness and regularity of solutions. In fact, they are often tied hand in glove to nonlinear PDEs. Although they have been studied for quite a long time, an infinite number of different assumptions can unlock different problems, which require unknown strategies and are fascinating for theoretical purposes, but not only. Indeed, nonlinear models have largely been used as a tool to describe and validate multiple phenomena and theories about what surrounds us, for example physics, biology and medicine.

The aim of this thesis is to present three different problems in this framework, whose study involves different techniques and approaches. This work is therefore divided into three Parts.

In **Part I**, we present two non-isothermal Cahn-Hilliard models. The first one is a twodimensional PDE system describing the phase separation behaviour of a two-component fluid in a bounded domain. In particular, we are interested in studying the existence, uniqueness and regularity of solutions. This is a starting point to introduce a second model, namely a three-dimensional non-isothermal Cahn-Hilliard system describing tumor growth.

**Part II** is devoted to study the mechanism of breathing. In particular, we consider the lungs as a viscoelastic deformable porous medium and breathing as an isothermal periodic process, which also takes into account the phenomenon of hysteresis.

Eventually, in **Part III** we move to the field of Calculus of Variations and in particular to the study of obstacle problems. Namely, we focus on higher differentiability properties of solutions to obstacle problems with nonstandard growth conditions. Our analysis takes into account a particular class of double phase functionals. These are a useful tool to study the behaviour of strongly anisotropic materials whose hardening properties are strongly dependent on the point and connected to the exponent ruling the growth of the gradient variable. The main results contained in this thesis have been obtained during my Ph.D. studies. In particular, **Part I** includes the content of the two papers

▶ E. Ipocoana and A. Zafferi, *Further regularity and uniqueness results for a nonisothermal Cahn-Hilliard equation*, Comm. Pure Appl. Anal., **20**(2), 763–782 (2021)

► E. Ipocoana, On a non-isothermal Cahn-Hilliard model for tumor growth, J. Math. Anal. Appl **506**(2) 125665 (2022)

 ${\bf Part}~{\bf II}$  is based on the submitted paper

► E. Ipocoana and P. Krejčí, A model for assisted periodic breathing with degenerate permeability, submitted (2022)

which is a further development of my Master's thesis results, published in

▶ M. Eleuteri, E. Ipocoana, J. Kopfová, P. Krejčí, *Periodic solutions of a hysteresis model for breathing*, ESAIM: Mathematical Modelling and Numerical Analysis **54** 255–271, (2020)

Eventually, **Part III** gathers the main results obtained in

► A.G. Grimaldi and E. Ipocoana, *Higher differentiability results in the scale of Besov* spaces to a class of double-phase obstacle problems, ESAIM: COCV **28** 51, (2022)

► A.G. Grimaldi and E. Ipocoana, *Higher fractional differentiability for solutions to a class of obstacle problems with non-standard growth conditions*, Adv. Calc. Var. (2022)

In fact, these are not the only products of my research. Indeed, during my Ph.D. I also worked on regularity results for the Kolmogorov equation, namely

▶ E. Ipocoana and A. Rebucci, *Pointwise estimates for degenerate Kolmogorov equations with*  $L^{p}$ -source term, J. Evol. Equ. **22**(1), 2 (2022)

and other currently ongoing projects whose content is not part of this thesis.

# Contents

Ι	No	on-isothermal Cahn-Hilliard 1	nodels	1
1	Two	vo-component fluid in a 2D-domain		4
	1.1	Setting		6
	1.2 Derivation of the model $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$			7
		1.2.1 Conserved order parameter app	broach	8
		1.2.2 Gurtin's microforces approach		13
	1.3 Assumptions			17
1.4 Well posedness results			19	
		1.4.1 Main result $\ldots$		19
		$1.4.2$ Initial regularity $\ldots$		20
		1.4.3 Further regularity $\ldots$ $\ldots$		28
		1.4.4 Uniqueness		31
<b>2</b>	An		40	
	2.1	Derivation of the model		43
		2.1.1 Constitutive relations		46
	2.2	Assumptions		47
2.3 Existence of solutions			48	
		2.3.1 A priori estimates $\ldots$ $\ldots$		50
		2.3.2 Weak sequential stability		56
II	A	A model for assisted periodic	breathing	59
3	Viscoelastic porous medium model			
	3.1	Statement of the problem		62
	3.2	Comparison with previous model $\therefore$		66
	3.3	Preisach operator		66

4	Pro	of of Theorem 3.1.3	69		
	4.1	Approximation scheme	69		
		4.1.1 Galerkin approximations	70		
		4.1.2 Limit as $m \to \infty$	74		
	4.2	Estimates independent of $\delta$ and $\varepsilon$	75		
	4.3	$L^{\infty}$ -bounds	78		
	4.4	Uniform estimates of $p$	85		
II	I	Obstacle problems applied to anisotropic materials	94		
<b>5</b>	Not	ation and background	97		
	5.1	Besov-Lipschitz spaces	98		
	5.2	Difference quotient	100		
	5.3	Preliminary results on standard growth conditions	101		
6	Hig	her differentiability for lagrangians $\tilde{F}(x, Du)$	102		
	6.1	Approximation results	105		
	6.2	Proof of Theorem 6.0.1	108		
		6.2.1 A priori estimate	108		
		6.2.2 Passage to the limit	119		
7	Higher differentiability for double-phase lagrangians $F(x, u, Du)$ 122				
	7.1	Higher integrability	124		
	7.2	Higher differentiability for comparison maps	133		
	7.3	Comparison	143		
	7.4	Main result	148		
		7.4.1 Proof of Theorem 7.0.1	151		
A	ppe	ndix	156		
A	$\mathbf{Sob}$	olev embeddings and interpolation theory	156		
	A.1	2D results	156		
	A.2	General results	157		
В	Тор	ological degree	159		

# Part I

# Non-isothermal Cahn-Hilliard models

The first Part of this work is devoted to present models in the Cahn-Hilliard framework. The Cahn-Hilliard equation was first formulated by J.W Cahn and J.Hilliard in [18, 19] and it describes phase separation of multi-component mixtures, that is the segregation of the system into spatial domains.

The Cahn-Hilliard equation reads

$$\varphi_t = \operatorname{div}\left[M(\varphi)\nabla\left[\frac{1}{\varepsilon}F'(\varphi) - \varepsilon\Delta\varphi\right]\right], \quad (x,t) \in \Omega \times \mathbb{R}^+$$
 (I.1)

complemented with appropriate initial and boundary conditions.

The process of phase separation is described by the order parameter  $\varphi = \varphi(x, t)$ , which represents the concentration of one of the two components in a binary system occupying a volume  $\Omega$ .  $M = M(\varphi)$  is a mobility coefficient and  $F(\varphi)$  is the free homogeneous energy function. We remark that we here stand in the case of the so-called diffuse interface models. Namely, we allow a partial mixing of components in a narrow interfacial region, whose thickness is represented by a (small) parameter  $\varepsilon$ .

In the main part of our discussion, we assume the domain  $\Omega$  to be bounded and with a boundary  $\partial\Omega$  smooth enough. We will explain in details the needed regularity according to the different cases we consider in Chapters 1 and 2 respectively. Besides, in our analysis we suppose that the mobility coefficient M is constant. In this particular case, the Cahn-Hilliard equation (I.1) can be formulated as the system

$$\varphi_t = \Delta \mu \tag{I.2}$$

$$\mu = M\left(-\varepsilon\Delta\varphi + \frac{1}{\varepsilon}F'(\varphi)\right),\tag{I.3}$$

where we introduced the auxiliary variable  $\mu$ , known in the literature as *chemical poten*tial.

According to both the formulations (I.1) and system (I.2)-(I.3), we notice that the temperature does not appear in the Cahn-Hilliard equation. This is because in general this phase separation model is isothermal, so it does not consider the possible variation of temperature during the process.

However, following the spirit of [46, 47, 39] we here also take into account the effects of the (absolute) temperature  $\theta$ . Indeed, the analysis of non-isothermal models has been used to describe the evolution of several types of substances, such as, e.g., plastic materials, shape memory alloys and liquid cristals [41, 42, 50].

In particular, in Chapter 1 we model the phase separation occurring in a two-component fluid occupying a bounded spatial domain. The first issue we address is the derivation of the model. We present two different approaches. The first one follows the lead of a more general case, presented in [46], based on the strategy proposed in [55]. In the second one, we adapt the method based on the balance of microforces in [69] to the non-isothermal case. The existence and uniqueness of the solution to a more general system were established respectively in [47] and [39]. However, we here focus on proving further regularity results for the solution, which allows us to give a simpler proof of the uniqueness. The well-posedness results presented in Chapter 1 are published in [77].

In Chapter 2 we use Cahn-Hilliard equations to describe tumor growth. In the context of diffuse interface models, the tumor is seen as a mass of cells surrounded by healthy tissue, with a thin layer separating the tumoral and healthy regions. The main novelty of this model is that we here consider the effects of temperature variations on the tumor. After presenting a thermodynamically consistent derivation of our system, we prove the existence of a weak (entropy) solution. The main results of Chapter 2 are published in [75].

# Chapter 1

# Two-dimensional non-isothermal Cahn-Hilliard model

We aim to establish new regularity properties for a non-isothermal Cahn-Hilliard model describing the phase separation of a two-component fluid occupying a bounded domain  $\Omega \subseteq \mathbb{R}^2$ . The model we consider consists of a PDE system describing the evolution of the unknown variables, namely the order parameter  $\varphi$ , chemical potential  $\mu$  and absolute temperature  $\theta$ . That is

$$\varphi_t = \Delta \mu, \tag{1.1}$$

$$\mu = -\Delta \varphi + F'(\varphi) - \theta, \qquad (1.2)$$

$$\theta_t + \theta \varphi_t - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2, \tag{1.3}$$

and it corresponds to the Cahn-Hilliard system for phase separation (cnfr. system (I.2)-(I.3)) coupled with the internal energy equation describing the evolution of temperature. This is a nonlinear system whose main source of difficulty is directly related to the termodynamic consistency of the model. Namely, it is represented by the quadratic term in the right-hand side of (1.3). The analysis is carried out in the 2-dimensional torus  $\Omega = [0, 1] \times [0, 1]$ , therefore we choose periodic boundary conditions for all the unknowns. The function F, whose derivative appears in (1.2), is a possibly non-convex potential whose minima represent the least energy configuration of the phase variable. Here, we assume that F is smooth and with power-like growth at  $\infty$ . Moreover the function  $\kappa(\theta)$  in (1.3) denotes the heat conductivity coefficient, assumed to grow at  $\infty$  as a power of  $\theta$ , as it has been recently considered in several contributions, for instance [47].

Our system is part of the more general model

$$\operatorname{div}\mathbf{u} = 0, \tag{1.4}$$

$$\mathbf{u}_t + \mathbf{u} \cdot \nabla \mathbf{u} + \nabla p = \Delta \mathbf{u} - \operatorname{div}(\nabla \varphi \otimes \nabla \varphi), \tag{1.5}$$

$$\varphi_t + \mathbf{u} \cdot \nabla \varphi = \Delta \mu, \tag{1.6}$$

$$\mu = -\Delta \varphi + F'(\varphi) - \theta, \qquad (1.7)$$

$$\theta_t + \mathbf{u} \cdot \nabla \theta + \theta \big( \varphi_t + \mathbf{u} \cdot \nabla \varphi \big) - \operatorname{div}(\kappa(\theta) \nabla \theta) = |\nabla \mathbf{u}|^2 + |\nabla \mu|^2$$
(1.8)

where the Cahn-Hilliard equation and the internal energy equation are coupled with a Navier-Stokes equation. This model has been derived and studied in [46] where the existence of solutions was shown in the 3D case (under some slightly different assumptions on the coefficients) in a very general and weak formulation. Then in [47] also the 2D case was analyzed, obtaining the existence of strong solutions. Eventually in [39] the authors were able to improve the previous results by defining a class of slightly smoother solutions and by proving that uniqueness holds in that class (and therefore well-posedness results have been proved). The key point on which these well-posedness results are proved is the following. The right hand side of (1.8) lies exactly in  $L^2(0,T;L^2(\Omega))$  and this information apparently does not seem to be sufficient to get additional regularity for  $\theta$ , which is essential in order to be able, for instance, to test the equation for the temperature by  $\theta_t$ . In particular, a  $L^{\infty}$ -bound is lacking because Moser iterations do not work for  $L^2$  on the right hand side and this would be crucial in order to manage some coefficients growing like powers of  $\theta$ . Therefore much efforts have been adopted in the already mentioned papers [47, 39] to overcome this difficulty and be able anyway to get a control of the gradient of  $\theta$  in  $L^2(\Omega)$ , uniformly in time.

Motivated by these works, we aim to show that, assuming a null velocity vector field, the Moser iteration scheme works, so that the crucial  $L^{\infty}$ -estimate for  $\theta$  is now available. As a consequence, we are thus able to present a simplified proof of uniqueness for the solution to our non-isothermal Cahn-Hilliard model (1.1)-(1.3).

The main results concerning further regularity and uniqueness of solutions presented in this Chapter are contained in the published paper [77].

Chapter 1 is then structured as follows. After introducing the suitable notation in Section 1.1, we proceed showing a thermodynamically consistent derivation of the model. In particular, in Section 1.2 we show two different approaches on how to derive our system. Namely, on one hand in Section 1.2.1 we present a first strategy based on the fact that the spatial mean of the order parameter  $\varphi$  is conserved in time. On the other hand, in Section 1.2.2 we follow Gurtin's approach based on microforces, which does not need the

condition on mass conservation. Then we focus on proving the well-posedness results. Thus, Section 1.3 provides the assumptions for our mathematical problem. The core of this Chapter is presented in Section 1.4, where the main result, Theorem 1.4.1, namely existence and uniqueness of a solution for our problem (1.1)-(1.3), together with the additional regularity for  $\theta$ , is proved.

## 1.1 Setting

Let us introduce some notation which will be useful in Section 1.2 and then adapted and extended in Section 1.4 to study the well-posedness results.

We suppose that a two-component fluid occupies a bounded spatial domain  $\Omega \subset \mathbb{R}^2$ , with a sufficiently regular boundary  $\partial \Omega$ . We let **n** denote the outer normal unit vector to  $\partial \Omega$ . Moreover,  $\varphi(x,t)$  is the *order parameter*, representing the concentration difference of the fluid, or the concentration of one component, and  $\theta(x,t)$  is the *absolute temperature*.

The symbol  $\|\cdot\|_X$  will denote the norm in a generic Banach space. We set  $H := L^2(\Omega)$ and  $V := H^1(\Omega)$ ,  $(\cdot, \cdot)$  stands for the usual standard product in H. For any function  $v \in H$ , we set

$$v_{\Omega} = \frac{1}{|\Omega|} \int_{\Omega} v \mathrm{d}x = \int_{\Omega} v \mathrm{d}x,$$

to indicate the spatial mean of v, being  $|\Omega| = 1$ , where  $\langle \cdot, \cdot \rangle$  denotes the duality pairing between V' and V and  $|\Omega|$  stands for the Lebesgue measure of  $\Omega$ . We note as  $H_0, V_0$  and  $V'_0$  the closed subspaces of functions (or functionals) having zero mean value in H, V, and, respectively, in V'. If the integral is replaced with the duality, the above can be extended to  $v \in V'$ . We denote as  $H_0, V_0$  and  $V'_0$  the closed subspaces of functions (or functionals) having zero mean value in H, V, and, respectively, in V'. Then

$$\|v\|_{V_0} := \left(\int_{\Omega} |\nabla v|^2 \mathrm{d}x\right)^{1/2}$$

represents a norm on  $V_0$ , which is equivalent to the norm inherited from V by the subsequent Poincaré-Wirtinger inequality (A.1f). In particular  $\|\cdot\|_{V_0}$  is a Hilbert norm and we can introduce the associated Riesz isomorphism mapping  $J: V_0 \to V'_0$  by setting, for  $u, v \in V_0$ ,

$$\langle Ju, v \rangle := ((u, v))_{V_0} := \int_{\Omega} \nabla u \cdot \nabla v dx.$$
 (1.9)

For  $f \in H_0$  it is easy to check that  $u = J^{-1}f \in H^2(\Omega)$ . Actually, u is the (unique) solution

to the elliptic problem

$$u \in H_0, \quad -\Delta u = f, \quad \nabla u \cdot \mathbf{n}_{|\partial\Omega} = 0$$

Moreover, if u is as above, then

$$\langle J(u-u_{\Omega}), v \rangle = -\int_{\Omega} v \Delta u \mathrm{d}x$$

for all  $v \in V_0$ . Finally, we can identify  $H_0$  with  $H'_0$  by means of the scalar product on H obtaining the Hilbert triplet  $V_0 \subset H_0 \subset V'_0$ , where inclusions are continuous and dense. In particular, if  $z \in V$  and  $v \in V_0$ , it holds that

$$\int_{\Omega} \nabla z \cdot \nabla (J^{-1}v) dx = \int_{\Omega} (z - z_{\Omega}) v dx = \int_{\Omega} z v dx.$$
(1.10)

## **1.2** Derivation of the model

We suppose that a two-component fluid occupies an open spatial domain  $\Omega \subset \mathbb{R}^2$ . We denote by  $\varphi(x,t)$  the concentration of one of the components of the fluid and  $\theta(x,t)$  is the absolute temperature.

We present here two different thermodynamically consistent approaches to derive our model. The first one relies on the fact that the scalar function  $\varphi = \varphi(x, t)$  satisfies the mass conservation constraint. The second strategy is on the other hand based on a microforces balance and does not require the condition on the conservation of mass.

We recall and collect here some general considerations, which will be exploited in both approaches.

According to the Ginzburg-Landau theory for phase transitions, we postulate the free energy density  $\psi$  and the energy functional  $\Psi$  respectively in the form

$$\psi = \frac{\varepsilon}{2} |\nabla \varphi|^2 + \frac{1}{\varepsilon} F(\varphi) + f(\theta) - \theta \varphi$$
(1.11)

$$\Psi = \int_{\Omega} \psi \mathrm{d}x. \tag{1.12}$$

Here,  $\varepsilon$  is a positive constant depending on the interface thickness. The function F in (1.11) penalizes the deviation of the length  $|\varphi|$  from its natural value 1. We refer for instance to the double-well potential and the logarithmic potential. The term f in (1.11) describes the part of free energy which is purely caloric and is related to the specific heat

 $c_V(\theta) = Q'(\theta)$  through relation

$$Q(\theta) = f(\theta) - \theta f'(\theta),$$

where Q is the entropy flux. In this Chapter (and in Chapter 2 as well) we assume that the specific heat is such that  $c_V \equiv 1$ . Assuming it constant is reasonable since in many materials the specific heat has small fluctuations around a single value. Moreover, we recall that it holds

$$q = Q\theta, \tag{1.13}$$

where q denotes the heat flux. The internal energy density of the system is given by Gibbs's relation

$$e = \psi + \theta s. \tag{1.14}$$

Here, s denotes the entropy of the system, which has the following expression, according to (1.11)

$$s = -\frac{\partial \psi}{\partial \theta} = -f'(\theta) + \varphi.$$
(1.15)

Combining the expression of internal energy density (1.14) and entropy formula (1.15), we infer

$$\frac{\partial e}{\partial t} = \frac{\partial \psi}{\partial t} + \theta \frac{\partial s}{\partial t} + s \frac{\partial \theta}{\partial t} = \frac{\partial \psi}{\partial \varphi} \frac{\partial \varphi}{\partial t} + \frac{\partial \psi}{\partial \nabla \varphi} \frac{\partial \nabla \varphi}{\partial t} + \theta \frac{\partial s}{\partial t}$$
(1.16)

and consequently

$$\frac{\partial \psi}{\partial t} + s \frac{\partial \theta}{\partial t} = \frac{\partial \psi}{\partial \varphi} \frac{\partial \varphi}{\partial t} + \frac{\partial \psi}{\partial \nabla \varphi} \frac{\partial \nabla \varphi}{\partial t}.$$
(1.17)

Both of the approaches rely on the use of the thermodynamic principles. In particular, we exploit the second law of thermodynamics in the form of the Clausius-Duhem inequality

$$\theta\left(\frac{\partial s}{\partial t} + \operatorname{div} Q\right) \ge 0. \tag{1.18}$$

#### 1.2.1 Conserved order parameter approach

We here follow the general approach presented in [55]. We note that our case can be interpreted as a particular one of [46]. However, for sake of completeness, we report here all the details. We denote by

$$E = (\varphi, \nabla \varphi, \theta)$$

the set of the state variables, which characterize the configuration of the material. On the other hand, the path along which the system tends to dissipate energy is described by the set of dissipative variables

$$\delta E = (\varphi_t, \nabla \theta).$$

The free energy density  $\psi(E)$  and the energy functional  $\Psi(E)$  are given respectively in the form (1.11) and (1.12).

As already mentioned, the approach we present here is based on the mass conservation of the order parameter. In order to impose this constraint, we write

$$\varphi = \varphi^0 + m_0, \tag{1.19}$$

where  $m_0$  denotes the mean value of the initial datum  $\varphi_0$ . Then, the conservation of mass corresponds to prescribe  $\varphi^0$  to take its values in  $H_0$  during the whole evolution of the system. The conservation of mass  $\varphi(t, \cdot) = \varphi(0, \cdot)$  a.e. for  $t \in (0, T)$  implies

$$\frac{D}{Dt} \int_{\Omega} \varphi dx = \int_{\Omega} \varphi_t dx = 0, \qquad (1.20)$$

where we remark that in our case the material derivative and time derivative coincide because we assume the fluid to have zero velocity.

We also notice that condition (1.20) translates into  $\varphi_t$  having zero spatial mean. Supposing that  $\varphi$  has a suitable regularity such that  $\varphi_t \in V'_0$  a.e. in time, we define

$$\mu^0 := -J^{-1}\varphi_t, \quad \text{so that } \varphi_t = -J\mu^0 = \Delta\mu^0 \quad \text{in } V'_0, \tag{1.21}$$

which entails  $\mu^0 \in V_0$ .

Moreover, if  $\varphi_t \in H_0$ , then, by elliptic regularity, we have

$$\mu^0 \in V_0 \cap H^2(\Omega) \text{ and } \nabla \mu^0 \cdot \mathbf{n}_{|_{\partial \Omega}} = 0.$$

This setting allows us to introduce the pseudo-potential of dissipation  $\Phi$ . This functional characterizes the evolution of the system and it is supposed to be non-negative and convex

with respect to the dissipative variables. Its expression is given by

$$\Phi(\delta E, E) = \int_{\Omega} \phi(\delta E, E) dx + \langle \varphi_t, J^{-1} \varphi_t \rangle, \qquad (1.22)$$

where the "local component"  $\phi$  of this *dissipation density* is given by

$$\phi(\delta E, E) = \frac{\kappa(\theta)}{2\theta} |\nabla \theta|^2, \qquad (1.23)$$

where  $\kappa(\theta) > 0$  is the heat conductivity.

The last term in (1.22), which is linked to the mass conservation, is nonstandard. In fact, it corresponds to a squared  $V'_0$  norm of the partial derivative of  $\varphi$  and thus depends in a nonlocal way on the dissipative variable.

We notice that we can equivalently rewrite the pseudo-potential  $\Phi$  as follows

$$\Phi(\delta E, E) = \int_{\Omega} \tilde{\phi}(\delta E, E) dx, \quad \text{where } \tilde{\phi}(\delta E, E) = \phi(\delta E, E) + \frac{1}{2} |\nabla \mu^{0}|^{2}.$$
(1.24)

Indeed, integrating by parts in space and using the definition of J, it turns out that

$$\int_{\Omega} |\nabla \mu^{0}|^{2} \mathrm{d}x = -\int_{\Omega} \Delta \mu^{0} \mu^{0} \mathrm{d}x = \int_{\Omega} J(\mu^{0}) \mu^{0} \mathrm{d}x = \langle \varphi_{t}, J^{-1}(\varphi_{t}) \rangle.$$

We also remark that we define the functionals  $\Phi$  and  $\Psi$  for all sets of variables E and  $\delta E$  for which they make sense. Namely, the class of admissible state variables is given by the condition of finiteness of  $\Psi$  and  $\Phi$ .

#### Balance equations and constitutive relations.

According to [55], by the principle of virtual power it follows that

$$\operatorname{div}\mathbf{H} - B = 0, \tag{1.25}$$

where the energy density B and the energy flux **H** are assumed to decompose as their non-dissipative and dissipative components. That is

$$B = B^{nd} + B^d,$$
$$\mathbf{H} = \mathbf{H}^{nd} + \mathbf{H}^d$$

where, taking to account (1.11), we have

$$B^{nd} = \frac{\partial \psi}{\partial \varphi} = \frac{1}{\varepsilon} F'(\varphi) - \theta, \qquad (1.26)$$

$$B^d = \delta_{H_0,\varphi_t} \Phi = J^{-1}(\varphi_t), \qquad (1.27)$$

According to (1.27),  $B^d$  is defined as the sub-differential of  $\Phi$  with respect to  $\varphi_t$  in the space  $H_0$ . This coincides in particular with  $J^{-1}(\varphi_t)$ . Indeed, for any  $v \in H_0$ , we have

$$(J^{-1}(\varphi_t), v - \varphi_t) = \langle v - \varphi_t, J^{-1}(\varphi_t) \rangle = ((v - \varphi_t, \varphi_t))_{V'_0}$$
  
 
$$\leq \frac{1}{2} \|v\|_{V'_0}^2 - \frac{1}{2} \|\varphi_t\|_{V'_0}^2 = \frac{1}{2} \langle v, J^{-1}v \rangle - \frac{1}{2} \langle \varphi_t, J^{-1}\varphi_t \rangle.$$

Moreover,

$$\mathbf{H}^{nd} = \frac{\partial \psi}{\partial \nabla \varphi} = \varepsilon \nabla \varphi$$
$$\mathbf{H}^d \equiv 0.$$

Analogously, we decompose the heat flux q and the entropy flux Q as

$$q = q^d + q^{nd},$$
$$Q = Q^d + Q^{nd},$$

with  $q^d = \theta Q^d$  and  $q^{nd} = \theta Q^{nd}$ , where, in view of (1.23),

$$q^{d} = \theta Q^{d} = \theta \frac{\partial \phi}{\partial \nabla \theta} = \kappa(\theta) \nabla \theta, \qquad (1.28)$$

while the non-dissipative component is determined a posteriori in order to comply with the second law of themodynamics.

In the sequel we also ask the heat conductivity to grow as a power of the absolute temperature. This choice is mainly motivated by mathematical reasons, however it is also coherent with physical interpretations (see [108]).

The order parameter equation (1.1) is derived according to the virtual power principle (1.25). In fact, we need to include the boundary conditions and the conservation mass constraint. Therefore we first rewrite  $\Psi$ , according to (1.12), as

$$\Psi(E) = \int_{\Omega} \left( \frac{\varepsilon}{2} |\nabla \varphi^0|^2 + \frac{1}{\varepsilon} F(\varphi^0 + m_0) + f(\theta) - \theta(\varphi^0 + m_0) \right) \mathrm{d}x,$$

where we used the decomposition  $\varphi = \varphi^0 + m_0$  introduced in (1.19). Thus we express (1.25) as a generalized gradient flow problem in  $H_0$ , namely

$$B^{d} + \delta_{H_{0},\varphi^{0}}\Psi = \delta_{H_{0},\varphi_{t}}\Phi + \delta_{H_{0},\varphi^{0}}\Psi = 0.$$
(1.29)

We remark that requiring  $\varphi^0$  to lie in the domain of the differential  $\delta_{H_0,\varphi^0}\Psi$  means that there exists a (unique) function  $z \in H_0$  such that  $\delta_{H_0,\varphi^0}\Psi(\varphi^0)$  can be represented by zin the scalar product of  $H_0$  (and so of H). This leads to the fact that (1.29) contains both the homogeneous Neumann boundary conditions for  $\varphi$  and the mass conservation constraint.

Additionally, we observe that such a function z must be of the form

$$z = -\varepsilon \Delta \varphi^0 + \frac{1}{\varepsilon} \left( F'(\varphi^0 + m_0) - F'(\varphi^0 + m_0)_{\Omega} \right) - \theta + \theta_{\Omega}.$$
(1.30)

Now, putting together (1.29) and (1.30) with (1.27), we then obtain

$$J^{-1}(\varphi_t) = \varepsilon \Delta \varphi^0 - \frac{1}{\varepsilon} \left( F'(\varphi^0 + m_0) - F'(\varphi^0 + m_0)_{\Omega} \right) + \theta - \theta_{\Omega}.$$

Eventually, we apply the distributional Laplace operator to both hand sides, recalling that  $-\Delta J^{-1}v = v$  for any  $v \in H_0$ , as stated in Section 1.1. Then we introduce the auxiliary variable  $\mu$  such that  $\mu^0 = \mu - (\mu)_{\Omega}$ , in accordance with (1.21). This entails

$$\varphi_t = \Delta \mu, \tag{1.31}$$

$$\mu = -\varepsilon \Delta \varphi + \frac{1}{\varepsilon} F'(\varphi) - \theta.$$
(1.32)

Therefore, choosing  $\varepsilon = 1$ , we then recover system (1.1)-(1.2). By the first law of thermodynamics, it follows that

$$\frac{\partial e}{\partial t} + \operatorname{div} q = B\varphi_t + \mathbf{H} \cdot \nabla \varphi_t + \Pi.$$
(1.33)

The last term in (1.33) does not appear in the standard theory proposed in [55]. Indeed,  $\Pi$  is needed to balance the nonlocal dependence with respect to the dissipative variable  $\varphi_t$  of the last term appearing in the expression (1.22) of the psudopotential of dissipation  $\Phi$ .

We now aim to determine the expressions of  $\Pi$  and the nondissipative component of the heat flux  $q^{nd}$ . This is done exploiting the second law of thermodynamics in the form of Clausius-Duhem inequality (1.18).

Before doing so, we need to gain some useful relations. We first notice that, combining (1.11), (1.14) and (1.15) we infer

$$e = \psi + \theta s = \frac{1}{\varepsilon} F(\varphi) + \frac{\varepsilon}{2} |\nabla \varphi|^2 + Q(\theta).$$
(1.34)

We also remark that the latter term in (1.17) is given by

$$\frac{\partial \psi}{\partial \nabla \varphi} (\nabla \varphi)_t = \mathbf{H}^{nd} \cdot (\nabla \varphi_t) \,. \tag{1.35}$$

Besides, by (1.30)-(1.32), we get that

$$\varphi_t J^{-1}(\varphi_t) = -\Delta \mu (\mu - \mu_{\Omega}) = -\frac{1}{2} \Delta (\mu - \mu_{\Omega})^2 + |\nabla \mu|^2.$$
(1.36)

We are now ready to focus on the second law of thermodynamics. Developing the left hand side of (1.18), we get

$$\begin{aligned} \theta \left( s_t + \operatorname{div} Q \right) &\stackrel{(1.34),(1.13)}{=} e_t + \operatorname{div} q - \frac{\partial \psi}{\partial \varphi} - \theta_t s - Q \cdot \nabla \theta \\ &\stackrel{(1.15)}{=} e_t + \operatorname{div} q - \frac{\partial \psi}{\partial \varphi} \varphi_t - \frac{\partial \psi}{\partial \nabla \varphi} \cdot (\nabla \varphi)_t - Q \cdot \nabla \theta \\ &\stackrel{(1.33),(1.26),(1.28)}{=} B \varphi_t + \mathbf{H} \cdot (\nabla \varphi)_t + \Pi - \frac{\partial \psi}{\partial \nabla \varphi} \cdot (\nabla \varphi)_t \\ &\quad - B^{nd} \varphi_t + \frac{\kappa(\theta)}{\theta} |\nabla \theta|^2 - Q^{nd} \cdot \nabla \theta \\ &\stackrel{(1.27),(1.35),(1.36)}{=} |\nabla \mu|^2 - \frac{1}{2} \Delta (\mu - \mu_\Omega)^2 + \Pi + \frac{\kappa(\theta)}{\theta} |\nabla \theta|^2 - Q^{nd} \cdot \nabla \theta. \end{aligned}$$

Then, in order to obtain the non-negativity of the right hand side of (1.18), we can assume, e.g., the following constitutive relations

$$q^{nd} = 0, \quad \Pi = \frac{1}{2}\Delta(\mu - \mu_{\Omega})^2.$$
 (1.37)

With these choices, we get  $\int_{\Omega} \Pi(x) dx = 0$ , as expected. Moreover, rewriting the internal energy balance (1.33) as

$$(Q(\theta))_t + \theta \varphi_t - \operatorname{div}(\kappa(\theta) \nabla \theta) = |\nabla \mu|^2, \qquad (1.38)$$

and choosing the specific heat equal to one, we recover the temperature equation (1.3). Notice that the dissipation terms on the right hand side are perfectly in agreement with the expression (1.24) of the pseudo-potential of dissipation  $\Psi$ . Indeed, as already mentioned, one has  $\mu^0 = \mu - \mu_{\Omega}$  due to (1.31)-(1.32).

### 1.2.2 Gurtin's microforces approach

In this Section, we follow Gurtin's approach proposed in [69] in order to derive our system (1.1)-(1.3). Namely, we treat separately the balance laws and the constitutive relations,

moreover we consider the following balance law for internal microforces

$$\operatorname{div}\zeta + \pi = 0, \tag{1.39}$$

where  $\zeta$  is a vector representing the microstress and  $\pi$  is a scalar corresponding to the internal microforces. Microforces describe the forces associated with microscopic configurations of atoms, differently from standard forces, which are associated with macroscopic length scales. These different length scales are the reason why a separate balance law for microforces is needed. Besides, since in Cahn-Hilliard models the kinematics is associated with the order parameter, it is natural to infer that the working of microforces has effects on  $\varphi$ . Such interatomic forces may be mirrored on the macroscopic level by fields which perform work when the order parameter undergoes changes. Therefore this working can be described in terms of  $\varphi_t$ , which explains why the microforces are scalar rather than vector quantities.

We complement the balance law for microforces with the fundamental balance laws. Since we do not consider external mass sources, the mass balance law reads

$$\varphi_t = -\operatorname{div} h,\tag{1.40}$$

where h is the mass flux.

The derivation of our system is based on the first and second fundamental laws of thermodynamics. We proceed showing first the derivation of the Cahn-Hilliard system (1.1)-(1.2) and then the one of the temperature equation (1.3).

#### Cahn-Hilliard system.

According to [69], we write the first law in the form

$$\frac{d}{dt} \int_{R} e \mathrm{d}x = -\int_{\partial R} q \cdot \nu \mathrm{d}\eta + \mathcal{W}(R) + \mathcal{M}(R), \qquad (1.41)$$

where R is the control volume,  $\nu$  is the outward unit normal to  $\partial R$  and

$$\mathcal{W}(R) = \int_{\partial R} (\zeta \cdot \nu) \frac{\partial \varphi}{\partial t} \mathrm{d}\eta, \qquad (1.42)$$

$$\mathcal{M}(R) = -\int_{\partial R} \mu h \cdot \nu \mathrm{d}\eta \tag{1.43}$$

are the rate of working and the rate at which free energy is added to R (assuming no heat

supply) respectively. Using Green's formula, we can rewrite (1.41) as

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + \frac{\partial \varphi}{\partial t} \operatorname{div} \zeta + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h \cdot \nabla \mu - \mu \operatorname{div} h, \qquad (1.44)$$

where we introduced the heat flux q in (1.13). Since the control volume R is arbitrary, exploiting the mass balance (1.40) and the microforce balance (1.39), we infer

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + (\mu - \pi)\frac{\partial \varphi}{\partial t} + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h\nabla \mu.$$
(1.45)

We now impose the validity of the second law of thermodynamics in the form of the Clausius-Duhem inequality (1.18). We develop the left hand side of (1.18) as follows

$$\theta \left( \frac{\partial s}{\partial t} + \operatorname{div} Q \right) \stackrel{(1.14)}{=} \frac{\partial e}{\partial t} - \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} + \theta \operatorname{div} Q$$

$$\stackrel{(1.13)}{=} \frac{\partial e}{\partial t} - \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} + \operatorname{div} q - Q \cdot \nabla \theta$$

$$\stackrel{(1.45)}{=} (\mu - \pi) \frac{\partial \varphi}{\partial t} + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h \nabla \mu - \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} - Q \cdot \nabla \theta$$

$$\stackrel{(1.17)}{=} \left( \mu - \pi - \frac{\partial \psi}{\partial \varphi} \right) \frac{\partial \varphi}{\partial t} + \left( \zeta - \frac{\partial \psi}{\partial \nabla \varphi} \right) \frac{\partial \nabla \varphi}{\partial t} - h \nabla \mu - Q \cdot \nabla \theta.$$

In order to satisfy relation (1.18), we impose

$$\mu - \pi - \frac{\partial \psi}{\partial \varphi} = 0, \qquad (1.46)$$

$$\zeta = \frac{\partial \psi}{\partial \nabla \varphi},\tag{1.47}$$

$$h\nabla\mu + Q\cdot\nabla\theta \le 0,\tag{1.48}$$

where in particular in order for (1.48) to hold, we exploited Fourier's law

$$q = -\kappa(\theta)\nabla\theta,\tag{1.49}$$

with  $\kappa = \kappa(\theta) > 0$  heat conductivity.

The combination of (1.11) and (1.47) straightly gives

$$\zeta = \varepsilon \nabla \varphi, \tag{1.50}$$

which leads to, according to (1.11), (1.39) and (1.46),

$$\mu = -\varepsilon \Delta \varphi + \frac{1}{\varepsilon} F'(\varphi) - \theta.$$
(1.51)

Eventually, inequality (1.48) can be satisfied choosing  $h = -\nabla \mu$ , which is a suitable assumption according to [69]. Therefore equation (1.40) reads

$$\varphi_t = \Delta \mu. \tag{1.52}$$

#### Temperature equation.

We start from the internal energy equation (1.45), taking advantage of (1.47) and of the expression for the chemical potential (1.46), therefore

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + \frac{\partial \psi}{\partial \varphi} \frac{\partial \varphi}{\partial t} + \frac{\partial \psi}{\partial \nabla \varphi} \frac{\partial \nabla \varphi}{\partial t} - h \nabla \mu$$

Now, exploiting the assumption  $h = -\nabla \mu$  and Fourier's law (1.49), we infer

$$\frac{\partial e}{\partial t} = \operatorname{div}\left(\kappa(\theta)\nabla\theta\right) + \frac{\partial\psi}{\partial\varphi}\frac{\partial\varphi}{\partial t} + \frac{\partial\psi}{\partial\nabla\varphi}\frac{\partial\nabla\varphi}{\partial t} + |\nabla\mu|^2$$

and by identity (1.16),

$$\theta \frac{\partial s}{\partial t} - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2.$$
(1.53)

From (1.15), we might write

$$\theta \frac{\partial s}{\partial t} = \theta (-f'(\theta))_t + \theta \varphi_t$$

On the other hand, according to the definition of Q, it holds  $(Q(\theta))_t = (-f'(\theta))_t$ , with in particular  $(Q(\theta))_t = Q'(\theta)\theta_t$ . Since we supposed that we are considering the case in which the specific heat  $c_V = 1$ , it follows that  $Q'(\theta) = 1$ . This implies that

$$\theta s_t = \theta_t + \theta \varphi_t.$$

Thus, equation (1.53) reads

$$\theta_t + \theta \varphi_t - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2.$$
(1.54)

## **1.3** Assumptions

In this Section we expand and adapt the notation presented in Section 1.1, in order to study the well-posedness of our system (1.1)-(1.3).

We consider our PDE system taking place in the two-dimensional flat torus with periodic boundary conditions, namely  $\Omega = [0, 1] \times [0, 1]$  and  $\varphi|_{x_i=0} = \varphi|_{x_i=1}$   $i = 1, 2 \quad \forall t \in (0, T)$ .

We denote as  $H := L_{\text{per}}^2(\Omega)$  the space of functions in  $L^2(\mathbb{R}^2)$  which are  $\Omega$ -periodic (i.e., 1-periodic both in  $x_1$  and  $x_2$ ). Analogously, we set  $V := H_{\text{per}}^1(\Omega)$ . The spaces H and Vare endowed with the norms of  $L^2(\Omega)$  and  $H^1(\Omega)$ , respectively. For brevity, the norm of Hwill be simply indicated by  $\|\cdot\|$ . Still for brevity, we omit the variables of integration. We will specify them when there could be a misinterpretation. The symbol  $\langle \cdot, \cdot \rangle$  will indicate the duality between V' and V and  $(\cdot, \cdot)$  will stand for the standard product of H. We also write  $L^p(\Omega)$  instead of  $L_{\text{per}}^p(\Omega)$ , and the same for other spaces; indeed, no confusion should arise since periodic boundary conditions are assumed to hold for all unknowns. We denote  $H_{\text{per}}^m(\Omega)$  the space of functions which are  $H_{\text{loc}}^m(\Omega)$  and  $\Omega$ -periodic, for  $m \in \mathbb{R}, m \ge 0$ . In particular, for m = 0 we have  $H_{\text{per}}^0(\Omega) = L_{\text{per}}^2(\Omega)$ .

We now focus on the mathematical hypotheses needed on the nonlinear terms. We ask the configuration potential F to satisfy:

$$F \in C^{3}(\mathbb{R}; \mathbb{R}), \quad \liminf_{|r| \to \infty} \frac{F(r)}{|r|} > 0, \tag{1.55}$$

$$F''(r) \ge -\lambda$$
 for some  $\lambda \ge 0$ , and all  $r \in \mathbb{R}$ , (1.56)

$$|F'''(r)| \le \tilde{c}_F (1+|r|^{p_F-1})$$
 for some  $\tilde{c}_F \ge 0, p_F \ge 1$ , and all  $r \in \mathbb{R}$ . (1.57)

We remark that (1.57) implies

$$|F''(r)| \le c_F(1+|r|^{p_F})$$
 for some  $c_F \ge 0, p_F \ge 0$ , and all  $r \in \mathbb{R}$ .

Assumption (1.55) postulates regularity and coercivity of F, (1.56) is  $\lambda$ -convexity and (1.57) prescribes a polynomial growth at infinity. Note that (1.55) implies that

$$F(s) \ge -c_0 \quad \forall s \in \mathbb{R}$$

for some constant  $c_0 > 0$ . We assume moreover the heat conductivity to be given by

$$\kappa(r) = 1 + r^q, \quad q \in [2, \infty), \quad r \ge 0.$$
 (1.58)

Correspondingly, we define

$$K(r) := \int_0^r \kappa(s) ds = r + \frac{1}{1+q} r^{1+q}, \quad r \ge 0.$$
(1.59)

We aim to estimate the norm  $||K(\theta)||_V^2$ , since it will be needed in the sequel. We observe that, for some  $k_q > 0$ ,

$$\int_{\Omega} \kappa(\theta)^2 |\nabla \theta|^2 \,\mathrm{dx} = \|\nabla K(\theta)\|^2 \ge \|\nabla \theta\|^2 + k_q \left\|\nabla \theta^{q+1}\right\|^2.$$
(1.60)

Then, by (A.2) with p = 2, we infer

$$\|K(\theta)\|_{V}^{2} \leq c_{q} \left(\int_{\Omega} (\theta + \theta^{q+1}) \, dx\right)^{2} + c_{q} \left(\int_{\Omega} \left|\nabla \theta + \nabla \left(\frac{\theta^{q+1}}{q+1}\right)\right|^{2} \, dx\right) =: I + II \quad (1.61)$$

for some  $c_q > 0$ . From (A.2), choosing p = 2(q+1), it holds

$$I = c_q \left( \int_{\Omega} \theta + \int_{\Omega} \theta^{q+1} \right)^2 \le c_q \|\theta\|_{L^1(\Omega)}^2 + c_q \left( \|\theta\|_{L^1(\Omega)}^{2(q+1)} + \|\nabla\theta^{q+1}\|^2 \right)$$
  
$$\le c_q \left( 1 + \|\theta\|_{L^1(\Omega)}^{2(q+1)} + \|\nabla\theta^{q+1}\|^2 \right).$$
(1.62)

We then estimate II according to (1.60). Therefore we get

$$\|K(\theta)\|_{V}^{2} \leq C\left(1 + \|\theta\|_{L^{1}(\Omega)}^{2(q+1)} + \int_{\Omega} \kappa^{2}(\theta) |\nabla\theta|^{2}\right),$$
(1.63)

for some C depending on q.

Complying with the boundary conditions and the lack of external forces, we have the conservation of mass and of the total energy  $\mathcal{E}$  that will be rigourously defined in (1.69), namely

$$\varphi(t)_{\Omega} = \varphi(0)_{\Omega} \qquad \mathcal{E}(\varphi(t), \theta(t)) = \mathcal{E}(\varphi(0), \theta(0))$$

It comes natural to define the "energy-entropy space" of data as:

$$\mathcal{H} = \left\{ z = (\varphi, \theta) \in V \times L^1(\Omega) : \theta > 0 \text{ a.e. in } \Omega, \log \theta \in L^1(\Omega) \right\}.$$

In our space we omitted the chemical potential  $\mu$ , in view of the fact that  $\mu$  can be regarded as an auxiliary variable, and sometimes, depending on the situation, it will be more convenient to "exclude"  $\mu$ . This can be easily achieved rewriting the system (1.1)-(1.2) as a single equation where  $\mu$  no longer appears. Now, in agreement with [47] and [39], we define the set

$$\mathcal{V} := \left\{ z = (\varphi, \theta) \in \mathcal{H} \cap \left( H^3(\Omega) \times V \right) : K(\theta) \in V, 1/\theta \in L^1(\Omega), \theta > 0 \text{ a.e.} \right\},\$$

where the requirement  $K(\theta) \in V$  yields  $\theta \in V$ .

Eventually, we recall here a result which will be useful in order to reach regularity in Section 1.4.2. The proof of this Lemma can be found in [47].

Lemma 1.3.1. Let  $\mathcal{O}$  a smooth bounded domain in  $\mathbb{R}^2$ . Then, there exists c > 0 depending only on  $\mathcal{O}$  such that

$$\|\xi\|_{H^1(\mathcal{O})'} \le c \Big(1 + \|\xi\|_{L^1(\mathcal{O})} \log^{1/2} \big(e + \|\xi\|_{L^2(\mathcal{O})}\big)\Big)$$
(1.64)

for any  $\xi \in L^2(\mathcal{O})$ .

## 1.4 Well posedness results

#### 1.4.1 Main result

We are now ready to present the main result of this Chapter, namely

**Theorem 1.4.1.** Let us assume (1.55)-(1.57) and (1.58). Let also T > 0. Then given  $z_0 \in \mathcal{V}$  there exists a unique solution to our problem, namely a triple  $(\varphi, \mu, \theta)$  with the regularity

$$\varphi \in W^{1,\infty}(0,T;V') \cap H^1(0,T;V) \cap L^{\infty}(0,T;H^3(\Omega)),$$
(1.65)

$$\mu \in H^1(0,T;V') \cap L^{\infty}(0,T;V) \cap L^2(0,T;H^3(\Omega)),$$
(1.66)

$$\theta \in H^1(0,T;L^2(\Omega)) \cap L^\infty(0,T;L^\infty(\Omega)) \cap L^\infty(0,T;V),$$
(1.67)

$$\theta > 0 \ a.e. \ in \ (0,T) \times \Omega,$$
  

$$K(\theta) \in L^{\infty}(0,T; L^{\infty}(\Omega)) \cap L^{\infty}(0,T; V),$$
(1.68)

satisfying equations (1.1)-(1.3) a.e. in  $(0,T) \times \Omega$  and complying with the initial conditions

$$\varphi|_{t=0} = \varphi_0, \qquad \theta|_{t=0} = \theta_0$$

almost everywhere in  $\Omega$ .

An existence result for our model can be proved by means of the solution of an approximating problem and then on the application of Schauder's fixed point theorem. The complete proof can be found in [47]. Here, we focus o regularity and uniqueness results. Concerning the regularity, in Section 1.4.2, we recover the basic regularity already shown in [47] for the general model, while in Section 1.4.3 we obtain further regularity with respect to [39]. We then exploit the regularity obtained to prove the uniqueness of the solution in Section 1.4.4.

### 1.4.2 Initial regularity

In this section we recover the basic regularity already obtained for the general model in [47] and we sketch the main points. We observe that, considering here null velocity, as one may expect, also this part of the proof turns to be simplified. In particular this is evident in two points: in the complementary estimates (1.82) and (1.84), that can be easily derived one from the other (due to the absence of convective terms) and in the key estimate of the term  $(\theta_t, \varphi_t)$ . This last achievement is obtained by means of the control of two terms: the estimate of  $\int \kappa^2(\theta) |\nabla \theta|^2$  follows exactly as in [47] while the estimate of the term  $\int |\nabla \mu|^2 \varphi_t$  can be heavily simplified, even if the idea of relying on conjugate functions still is necessary, for the presence of the quadratic term in the right hand side of (1.3).

#### Energy and entropy estimates

The energy estimate is obtained by testing (1.1) by  $\mu$ , (1.2) by  $-\varphi_t$ , (1.3) by 1 and then integrating over  $\Omega$ . We then sum up all the obtained relations. Therefore we infer

$$\frac{d}{dt}\mathcal{E}(\varphi,\theta) = 0, \quad \text{where } \mathcal{E}(\varphi,\theta) := \int_{\Omega} \left(\frac{1}{2}|\nabla\varphi|^2 + F(\varphi) + \theta\right)$$
(1.69)

which is the *total energy* of the system, given by the sum of the *interfacial, configuration*, and *thermal* energies (the three terms in  $\mathcal{E}$ ). From relation (1.69) we infer the following a priori estimates

$$\|\varphi\|_{L^{\infty}(0,T;V)} \le c,$$
 (1.70)

$$\|\theta\|_{L^{\infty}(0,T;L^{1}(\Omega))} \le c,$$
 (1.71)

where we exploited (1.55) in order to obtain (1.70) and we used the nonnegativity of  $\theta$  to get (1.71) from (1.69). Moreover, from (1.70) and Sobolev's embeddings, we also have

$$\|\varphi\|_{L^{\infty}(0,T;L^{p}(\Omega))} \leq c_{p} \quad \text{for all } p \in [1,\infty).$$

$$(1.72)$$

On the other hand, integrating (1.1) over  $\Omega$ , and using the periodic boundary conditions, we observe

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \varphi = 0 \quad \text{a.e. in} \ (0,T).$$
(1.73)

The entropy estimate corresponds to the entropy production principle. In order to obtain it, we test (1.3) by  $-\theta^{-1}$  and integrate over  $\Omega$ , namely

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (-\log \theta - \varphi) + \int_{\Omega} \frac{1}{\theta} |\nabla \mu|^2 + \int_{\Omega} \frac{\mathrm{div}(\kappa(\theta) \nabla \theta)}{\theta} = 0.$$

The last term on the left hand side can be written as

$$\int_{\Omega} \frac{\operatorname{div}(\kappa(\theta)\nabla\theta)}{\theta} = \int_{\Omega} \frac{\operatorname{div}(\nabla\theta)}{\theta} + \int_{\Omega} \frac{\operatorname{div}(\theta^{q}\nabla\theta)}{\theta}$$
$$= \int_{\Omega} \frac{\Delta\theta}{\theta} + \int_{\Omega} \frac{q\theta^{q-1}|\nabla\theta|^{2} + \theta^{q}\Delta\theta}{\theta}$$
$$= \int_{\Omega} |\nabla\log\theta|^{2} + k_{q}|\nabla\theta^{q/2}|^{2}.$$

Therefore we infer

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (-\log\theta - \varphi) + \int_{\Omega} \frac{1}{\theta} |\nabla\mu|^2 + \int_{\Omega} \left( |\nabla\log\theta|^2 + k_q |\nabla\theta^{q/2}|^2 \right) = 0, \qquad (1.74)$$

with  $k_q > 0$  only depending on the exponent q introduced in (1.58). We now integrate in time. Recalling that  $|\log r| \leq r - \log r \quad \forall r > 0$  and owing to (1.70) and (1.71), we get the a priori bounds

$$\|\log \theta\|_{L^{\infty}(0,T;L^{1}(\Omega))} + \|\log \theta\|_{L^{2}(0,T;V)} \le c, \tag{1.75}$$

$$\|\nabla\theta^{q/2}\|_{L^2(0,T;H)} \le c. \tag{1.76}$$

In particular, from (1.75) we see that the strict positivity of  $\theta$  is preserved a.e. in  $(0, T) \times \Omega$ also in the limit. Moreover, the combination of inequality (A.2) with estimates (1.71) and (1.76) gives

$$\|\theta^{q/2}\|_{L^2(0,T;V)} \le c,$$

which implies in particular

$$\|\theta\|_{L^2(0,T;H)} \le c$$

and, being  $q \ge 2$ , according to (1.58),

$$\|\theta\|_{L^2(0,T;V)} \le c. \tag{1.77}$$

### First estimates for $\mu$ , $\varphi$ and $\varphi_t$

From equation (1.3) and periodic boundary conditions, we get

$$\int_{\Omega} |\nabla \mu|^2 = \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \theta + \int_{\Omega} \theta \varphi_t.$$
(1.78)

Our aim is to control the terms on the right hand side.

In order to do so, we first integrate in time, and then we estimate the first one thanks to (1.71). On the other hand, by using (1.1) and Hölder's and Young's inequalities, the second integral can be controlled as follows

$$\int_{\Omega} \theta \varphi_t = \int_{\Omega} \theta \Delta \mu = -\int_{\Omega} \nabla \theta \cdot \nabla \mu \leq \frac{1}{2} \left( \|\nabla \mu\|^2 + \|\nabla \theta\|^2 \right).$$
(1.79)

The first term on the right hand side is absorbed by the corresponding one on the left hand side of (1.78), while we use (1.77) to estimate the latter. Hence, we obtain

$$\|\nabla\mu\|_{L^2(0,T;H)} \le c. \tag{1.80}$$

We now integrate (1.2) in space, combine (1.57), (1.71) and (1.72) and then take the (essential) supremum with respect to time; we infer

$$\|\mu_{\Omega}\|_{L^{\infty}(0,T)} \le c. \tag{1.81}$$

This estimate, combined with (1.80), gives

$$\|\mu\|_{L^2(0,T;V)} \le c. \tag{1.82}$$

Now, testing (1.1) by nonzero  $v \in V$ , we can notice that

$$\langle \varphi_t, v \rangle = -\int_{\Omega} \nabla \mu \cdot \nabla v \le \|\nabla \mu\| \|\nabla v\| \le c \|\nabla \mu\| \|v\|_V.$$
(1.83)

Hence, dividing by  $||v||_V$ , passing to the supremum with respect to  $v \in V \setminus \{0\}$ , squaring, integrating in time, and using (1.82), we infer

$$\|\varphi_t\|_{L^2(0,T;V')} \le c. \tag{1.84}$$

On the other hand, we test equation (1.1) by  $-\mu$  and note that  $\langle \varphi_t, \mu_{\Omega} \rangle = 0$ , since  $\varphi_t$  has

zero (generalized) mean. Therefore the use of the Poincaré-Wirtinger inequality yields

$$\begin{aligned} \|\nabla\mu\|^2 &= -\int_{\Omega} \varphi_t \mu = -\int_{\Omega} \varphi_t (\mu - \mu_{\Omega}) \le \|\mu - \mu_{\Omega}\|_V \|\varphi_t\|_{V'} \\ &\le \frac{1}{2} \|\nabla\mu\|^2 + c \|\varphi_t\|_{V'}^2, \end{aligned}$$

which allows us to get

$$\|\nabla \mu\|^2 \le c \|\varphi_t\|_{V'}^2. \tag{1.85}$$

Finally, if we test (1.2) by  $\Delta^2 \varphi$  and integrate over  $\Omega$ , by recalling (1.57), we get

$$\|\nabla\Delta\varphi\|^2 \le c(1+\|\nabla\theta\|^2+\|\nabla\mu\|^2)$$

Integrating this inequality in time and using (1.77) and (1.80), we then obtain

$$\|\varphi\|_{L^2(0,T;H^3(\Omega))} \le c.$$
(1.86)

### Key estimate: control of the term $(\theta_t, \varphi_t)$

First of all we take (1.1), differentiate it with respect to time, and test the result by  $J\varphi_t$ , where J was first introduced in (1.9). Correspondingly, we differentiate (1.2) in time and test the result by  $-\varphi_t$ . Therefore we have

$$\langle \varphi_{tt}, J\varphi_t \rangle = \langle \Delta \mu_t, J\varphi_t \rangle \langle \mu_t, -\varphi_t \rangle = \langle -\Delta \varphi_t, -\varphi_t \rangle + \langle F''(\varphi)\varphi_t, -\varphi_t \rangle + (\theta_t, \varphi_t)$$

Summing the obtained relations, noting that a couple of terms cancel in view of

$$\langle \Delta \mu_t, J\varphi_t \rangle = -\langle (-\Delta)(\mu_t - (\mu_t)_{\Omega}), (-\Delta)^{-1}\varphi_t \rangle = -\langle \mu_t - (\mu_t)_{\Omega}, \varphi_t \rangle = -\langle \mu_t, \varphi_t \rangle, \quad (1.87)$$

by (A.3) and (1.73), we then get

$$\frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \|\varphi_t\|_{V'}^2 + \|\nabla\varphi_t\|^2 + \int_{\Omega} (F''(\varphi) + \lambda)|\varphi_t|^2 
= \lambda \|\varphi_t\|^2 + (\theta_t, \varphi_t) \le \frac{1}{8} \|\nabla\varphi_t\|^2 + c \|\varphi_t\|_{V'}^2 + (\theta_t, \varphi_t).$$
(1.88)

Reabsorbing, this is equivalent to

$$\frac{1}{2}\frac{\mathrm{d}}{\mathrm{d}t}\|\varphi_t\|_{V'}^2 + \frac{7}{8}\|\nabla\varphi_t\|^2 + \int_{\Omega} (F''(\varphi) + \lambda)|\varphi_t|^2 \le c\|\varphi_t\|_{V'}^2 + (\theta_t, \varphi_t).$$
(1.89)

On the other hand, testing (1.3) by  $\varphi_t$  yields

$$(\theta_t, \varphi_t) + \int_{\Omega} \theta \varphi_t^2 = -\int_{\Omega} \kappa(\theta) \nabla \theta \cdot \nabla \varphi_t + \int_{\Omega} |\nabla \mu|^2 \varphi_t$$

$$\leq \frac{1}{16} \|\nabla \varphi_t\|^2 + 4 \int_{\Omega} \kappa^2(\theta) |\nabla \theta|^2 + \int_{\Omega} |\nabla \mu|^2 \varphi_t.$$
(1.90)

To treat the term in  $\nabla \theta$ , we test (1.3) by  $6K(\theta)$  introduced in (1.59) (the coefficient 6 is suitable for reabsorbing the second term in the right hand side of the previous inequality by the left hand side) and working exactly as in [47] we deduce the following estimate

$$6\frac{\mathrm{d}}{\mathrm{d}t}\int_{\Omega}\mathcal{J}(\theta) + 5\int_{\Omega}\kappa^{2}(\theta)|\nabla\theta|^{2}$$
  
$$\leq c(1+\|\varphi_{t}\|_{V'}^{4}) + \frac{1}{8}\|\nabla\varphi_{t}\|^{2} + 6\int_{\Omega}K(\theta)|\nabla\mu|^{2}.$$
 (1.91)

where we set

$$\mathcal{J}(r) := \int_0^r K(s) \, \mathrm{d}s = \frac{r^2}{2} + \frac{1}{(q+1)(q+2)} r^{q+2}, \quad r \ge 0.$$
(1.92)

Summing (1.90) and (1.91) we get

$$(\theta_t, \varphi_t) + \int_{\Omega} \theta \varphi_t^2 + 6 \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \mathcal{J}(\theta) + \int_{\Omega} \kappa^2(\theta) |\nabla \theta|^2$$
  
$$\leq c \left(1 + \|\varphi_t\|_{V'}^4\right) + \frac{3}{16} \|\nabla \varphi_t\|^2 + \int_{\Omega} (6K(\theta) + \varphi_t) |\nabla \mu|^2.$$
(1.93)

Then, adding together (1.89) and (1.93) we obtain

$$\frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \|\varphi_t\|_{V'}^2 + 6 \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \mathcal{J}(\theta) + \frac{11}{16} \|\nabla\varphi_t\|^2 + \int_{\Omega} (F''(\varphi) + \lambda) |\varphi_t|^2 \\
+ \int_{\Omega} \theta\varphi_t^2 + \int_{\Omega} \kappa^2(\theta) |\nabla\theta|^2 \\
\leq c \|\varphi_t\|_{V'}^2 + c \left(1 + \|\varphi_t\|_{V'}^4\right) + \int_{\Omega} (6K(\theta) + \varphi_t) |\nabla\mu|^2.$$
(1.94)

Neglecting some positive terms in the left hand side and rearranging, we then arrive at

$$\frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \|\varphi_t\|_{V'}^2 + 6 \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \mathcal{J}(\theta) + \frac{11}{16} \|\nabla\varphi_t\|^2 + \int_{\Omega} \kappa^2(\theta) |\nabla\theta|^2 \\
\leq c(1 + \|\varphi_t\|_{V'}^2)^2 + \int_{\Omega} (6K(\theta) + \varphi_t) |\nabla\mu|^2.$$
(1.95)

We now focus on controlling of the last term in the right hand side of (1.95), which

represents the most difficult part of our argument. In order to do so, we use the embedding inequality (1.64) setting  $\xi = |\nabla \mu|^2$ . Then, exploiting (1.63), (1.73) and the Poincaré-Wirtinger inequality, we infer

$$\int_{\Omega} (6K(\theta) + \varphi_{t}) |\nabla \mu|^{2} \leq c \left( ||K(\theta)||_{V} + ||\nabla \varphi_{t}|| \right) |||\nabla \mu|^{2} ||_{V'} \\
\leq c + \frac{1}{2} \int_{\Omega} \kappa^{2}(\theta) |\nabla \theta|^{2} + \frac{1}{8} ||\nabla \varphi_{t}||^{2} + c |||\nabla \mu|^{2} ||_{V'}^{2} \\
\leq c + \frac{1}{2} \int_{\Omega} \kappa^{2}(\theta) |\nabla \theta|^{2} + \frac{1}{8} ||\nabla \varphi_{t}||^{2} \\
+ c |||\nabla \mu|^{2} ||_{L^{1}(\Omega)}^{2} \log \left( e + |||\nabla \mu|^{2} ||_{L^{2}(\Omega)} \right).$$
(1.96)

Next, we consider the functions  $\psi(r) = e^r$ ,  $r \in \mathbb{R}$  and  $\psi^*(s) = s(\log s - 1)$ , s > 0(extended by continuity to s = 0 by setting  $\psi^*(0) = 0$ ), which are convex conjugate. This means that  $\forall r \in \mathbb{R}, s \ge 0$ , it holds  $rs \le \psi(r) + \psi^*(s)$ , as we can see for example in [10, Sec. 1.4]. If we now set  $r = \log(e + ||\nabla \mu|^2 ||_{L^2(\Omega)})$  and  $s = c ||\nabla \mu|^2 ||_{L^1(\Omega)}^2$ , we can estimate the last term in (1.96) as follows

$$c \| |\nabla \mu|^{2} \|_{L^{1}(\Omega)}^{2} \log \left( e + \| |\nabla \mu|^{2} \|_{L^{2}(\Omega)} \right)$$
  

$$\leq c \| |\nabla \mu|^{2} \|_{L^{1}(\Omega)}^{2} \left( \log \left( c \| |\nabla \mu|^{2} \|_{L^{1}(\Omega)}^{2} \right) - 1 \right) + e + \| |\nabla \mu|^{2} \|_{L^{2}(\Omega)}$$
  

$$\leq c + c \| \nabla \mu \|^{4} \log \left( e + \| \nabla \mu \|^{2} \right) + \| \nabla \mu \|_{L^{4}(\Omega)}^{2}, \qquad (1.97)$$

where we used the fact that  $\||\nabla \mu|^2\|_{L^1(\Omega)}^2 = \|\nabla \mu\|^4$  and elementary inequalities concerning logarithms.

The first non-constant term on the right hand side of (1.97) can be estimated by using (1.85) as follows

$$c\|\nabla\mu\|^{4}\log\left(e+\|\nabla\mu\|^{2}\right) \leq c\left(1+\|\varphi_{t}\|_{V'}^{4}\right)\log\left(e+c\left(1+\|\varphi_{t}\|_{V'}^{2}\right)\right)$$
$$\leq c(1+\|\varphi_{t}\|_{V'}^{2})^{2}\log\left(e+\|\varphi_{t}\|_{V'}^{2}\right)$$
(1.98)

while the second one can be controlled by using equation (1.1) and inequalities (A.1a) and (A.3) as

$$\begin{aligned} \|\nabla\mu\|_{L^{4}(\Omega)}^{2} \leq c \|\nabla\mu\|\|\mu\|_{H^{2}(\Omega)} \leq c \|\nabla\mu\|\left(\|\mu\|_{V} + \|\Delta\mu\|\right) \\ \leq c \|\mu\|_{V}^{2} + c \|\varphi_{t}\|^{2} \leq c \|\mu\|_{V}^{2} + \frac{1}{8} \|\nabla\varphi_{t}\|^{2} + c \|\varphi_{t}\|_{V'}^{2}. \end{aligned}$$
(1.99)

Then, setting  $M(t) := c \|\mu\|_V^2$ , plugging (1.97)-(1.99) in (1.96) and in turn the result

into (1.95) we finally deduce

$$\frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \|\varphi_t\|_{V'}^2 + 6 \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \mathcal{J}(\theta) + \frac{7}{16} \|\nabla\varphi_t\|^2 + \frac{1}{2} \int_{\Omega} \kappa^2(\theta) |\nabla\theta|^2 \\
\leq c(e + \|\varphi_t\|_{V'}^2)^2 [1 + \log\left(e + \|\varphi_t\|_{V'}^2\right)] + M(t) \\
\leq c(e + \|\varphi_t\|_{V'}^2)^2 \log\left(e + \|\varphi_t\|_{V'}^2\right) + M(t).$$
(1.100)

Let us now set

$$\Phi(t) := e + \frac{1}{2} \|\varphi_t\|_{V'}^2, \qquad \Theta(t) := 6 \int_{\Omega} \mathcal{J}(\theta(t)).$$
(1.101)

Hence, (1.100) reads

$$\Phi'(t) + \Theta'(t) \le c \, [\Phi(t)]^2 \log(\Phi(t)) + M(t).$$
(1.102)

We define  $Z(t) := e + \Phi(t) + \Theta(t)$ , then we divide both hand sides of (1.102) by  $Z \log Z$ ,

$$\begin{aligned} \frac{\mathrm{d}}{\mathrm{d}t} \log \log Z(t) &= \frac{Z'(t)}{Z(t)\log Z(t)} \\ &\leq M(t) \frac{\Phi(t)}{(e + \Phi(t) + \Theta(t))\log(e + \Phi(t) + \Theta(t))} + \frac{M(t)}{Z(t)\log Z(t)} \\ &+ c\Phi \frac{\Phi\log(e + \Phi(t))}{(e + \Phi(t) + \Theta(t))\log(e + \Phi(t) + \Theta(t))} \\ &\leq M(t) + \frac{M(t)}{Z(t)\log Z(t)} + c\Phi(t). \end{aligned}$$

Thus we get

$$\frac{d}{dt}\log\log Z(t) = \frac{Z'(t)}{Z(t)\log Z(t)} \le \Phi(t) + \frac{M(t)}{Z(t)\log Z(t)},$$
(1.103)

where we recall that  $\|\Phi\|_{L^1(0,T)} + \|M(t)\|_{L^1(0,T)} \leq c$  in view of the a-priori estimates (1.82) and (1.84). Moreover, working in a similar way as in [47] in order to estimate the initial condition, we have  $Z(0) < \infty$ , hence we can integrate (1.103) over (0,T) to obtain

$$\|Z\|_{L^{\infty}(0,T)} \le c. \tag{1.104}$$

#### Consequences

From (1.92), (1.104) reads

$$\|\varphi_t\|_{L^{\infty}(0,T;V')} \le c, \tag{1.105}$$

$$\|\theta\|_{L^{\infty}(0,T;L^{q+2}(\Omega))} \le c.$$
(1.106)

Combining (1.1) with (1.81), we get

$$\|\mu\|_{L^{\infty}(0,T;V)} \le c. \tag{1.107}$$

According to the above relations and using (1.59), (1.73), after integrating (1.100) over (0,T), we infer

$$\|\varphi_t\|_{L^2(0,T;V)} \le c,\tag{1.108}$$

$$\|K(\theta)\|_{L^2(0,T;V)} \le c. \tag{1.109}$$

Now we read (1.1) as a time-dependent family of elliptic problems. Combining standard regularity results with (1.108), we have

$$\|\mu\|_{L^2(0,T;H^3(\Omega))} \le c. \tag{1.110}$$

We conclude by providing some estimates for the terms  $\mu_t$  and  $\varphi_t$ . We (formally) differentiate (1.2) with respect to time and use (1.3), therefore we infer

$$\mu_t = -\Delta\varphi_t + F''(\varphi)\varphi_t + \theta\varphi_t - \Delta K(\theta) - |\nabla\mu|^2.$$
(1.111)

We now test the above relation by nonzero  $v \in V$ . Recalling the boundary conditions, we obtain

$$\langle \mu_t, v \rangle = \int_{\Omega} \nabla(\varphi_t + K(\theta)) \cdot \nabla v + \left\langle F''(\varphi)\varphi_t + \theta\varphi_t - |\nabla \mu|^2, v \right\rangle$$
  
 
$$\leq \|v\|_V \Big( \|\nabla(\varphi_t + K(\theta))\| + \|F''(\varphi)\varphi_t + \theta\varphi_t - |\nabla \mu|^2\|_{L^{3/2}(\Omega)} \Big).$$
 (1.112)

Then, dividing by  $||v||_V$ , passing to the supremum with respect to  $v \in V \setminus \{0\}$ , squaring, and integrating in time, we get

$$\|\mu\|_{H^1(0,T;V')} \le c. \tag{1.113}$$

Indeed, according to (1.108)-(1.109), it holds

$$\left\|\nabla(\varphi_t + K(\theta))\right\|_{L^2(0,T;H)} \le c.$$
(1.114)

and moreover it holds

$$\left\|F''(\varphi)\varphi_t + \theta\varphi_t - |\nabla\mu|^2\right\|_{L^2(0,T;L^{3/2}(\Omega))} \le c \tag{1.115}$$

where the exponent 3/2 is chosen just for simplicity (any number strictly greater than 1 would be allowed, indeed).

This last inequality can be proved considering each term in the norm separately. We know that  $F''(\varphi)$  grows as a power of  $\varphi$ , whose regularity is given by (1.70) and (1.86). Hence, integrating in time and exploiting (1.70) and (1.108) we infer

$$||F''(\varphi)\varphi_t||_{L^2(0,T;L^{3/2}(\Omega))} \le c.$$

On the other hand, integrating in time, from (1.106) and (1.108), we obtain

$$\|\theta\varphi_t\|_{L^2(0,T;L^{3/2}(\Omega))} \le c.$$

For the last term, integrating once more in time and taking advantage of (1.110) we get

$$\left\| |\nabla \mu|^2 \right\|_{L^2(0,T;L^{3/2}(\Omega))} \le c.$$

Combining the previous bounds we eventually gain (1.115).

Eventually, testing (1.3) by a nonzero  $v \in V$  and proceeding similarly as above, it follows

$$\|\theta\|_{H^1(0,T;V')} \le c.$$

#### **1.4.3** Further regularity

Thanks to the estimates obtained in the previous section, we are now able to prove the regularity presented in Theorem 1.4.1.

First of all we focus our attention on the estimate

$$\theta \in L^{\infty}(0,T;L^{\infty}(\Omega)) \tag{1.116}$$

obtained by a Moser's iteration technique, as in [87].

We start multiplying (1.3) by  $\theta^p$ , where p is a positive exponent which will be specified later on, and then we integrate over  $\Omega$ . Therefore, we have, in view of (1.58),

$$\frac{1}{p+1} \frac{d}{dt} \int_{\Omega} \theta^{p+1} + \frac{4p}{(p+1)^2} \int_{\Omega} \left| \nabla \theta^{\frac{p+1}{2}} \right|^2 + \frac{4p}{(p+q+1)^2} \int_{\Omega} \left| \nabla \theta^{\frac{p+q+1}{2}} \right|^2 \\
\leq \int_{\Omega} |\Delta \mu| \theta^{p+1} + \int_{\Omega} |\nabla \mu|^2 \theta^p.$$
(1.117)

This entails, using (A.2),

$$\begin{aligned} &\frac{d}{dt} \int_{\Omega} \theta^{p+1} + \frac{4p}{c_p(p+1)} \left\| \theta^{\frac{p+1}{2}} \right\|_{V}^{2} \\ &\leq &\frac{4p}{(p+1)} \left\| \theta \right\|_{L^{1}(\Omega)}^{p+1} + (p+1) \int_{\Omega} |\Delta \mu| \theta^{p+1} + (p+1) \int_{\Omega} |\nabla \mu|^{2} \theta^{p} \end{aligned}$$

$$\leq (p+1) \int_{\Omega} (|\Delta \mu| + 1)\theta^{p+1} + (p+1) \int_{\Omega} |\nabla \mu|^2 \theta^p =: I_1 + I_2, \qquad (1.118)$$

where we observed that  $\frac{4p}{p+1} \le p+1$  and where  $c_p$  denotes the Poincaré constant in (A.2). Now,

$$\begin{split} I_1 := &(p+1) \int_{\Omega} (|\Delta \mu| + 1) \theta^{p+1} = (p+1) \int_{\Omega} \theta^{\frac{(p+1)}{2}} (|\Delta \mu| + 1) \theta^{\frac{(p+1)}{2}} \\ &\leq &(p+1) \left\| \theta^{\frac{p+1}{2}} \right\|_{V} \left\| \theta^{\frac{(p+1)}{2}} (|\Delta \mu| + 1) \right\|_{V'} \\ &\leq &\frac{2p}{c_p(p+1)} \left\| \theta^{\frac{p+1}{2}} \right\|_{V}^2 + C(p+1)^2 \left\| \theta^{\frac{p+1}{2}} (|\Delta \mu| + 1) \right\|_{L^{6/5}(\Omega)}^2 \end{split}$$

where the positive constant C is allowed to vary from line to line.

At this point we use Hölder's inequality with exponents 5 and 5/4, therefore we get

$$\begin{split} \left\|\theta^{\frac{p+1}{2}}(|\Delta\mu|+1)\right\|_{L^{6/5}(\Omega)}^{2} &= \left(\int_{\Omega}\theta^{\frac{3}{5}(p+1)}(|\Delta\mu|+1)^{\frac{6}{5}}\right)^{\frac{5}{3}} \\ &\leq \||\Delta\mu|+1\|_{L^{6}(\Omega)}^{2}\left(\int_{\Omega}\theta^{\frac{3}{4}(p+1)}\right)^{\frac{4}{3}} \\ &\leq c(1+\|\mu\|_{H^{3}(\Omega)}^{2})\left(\int_{\Omega}\theta^{\frac{3}{4}(p+1)}\right)^{\frac{4}{3}}. \end{split}$$

Eventually we deduce

$$I_1 \le \frac{2p}{c_p(p+1)} \left\| \theta^{\frac{p+1}{2}} \right\|_V^2 + C(p+1)^2 (1 + \|\mu\|_{H^3(\Omega)}^2) \left( \int_\Omega \theta^{\frac{3}{4}(p+1)} \right)^{\frac{4}{3}},$$
(1.119)

On the other hand, observing that  $\theta^p \leq \theta^{p+1} + 1$  and recalling Sobolev's embedding theorem, thanks to (1.107) we are led to

$$\begin{split} I_2 := &(p+1) \int_{\Omega} |\nabla \mu|^2 \theta^p \le (p+1) \int_{\Omega} |\nabla \mu|^2 \theta^{p+1} + (p+1) \|\nabla \mu\|_{L^2(\Omega)}^2 \\ \le &c(p+1) \left\| \theta^{\frac{p+1}{2}} \right\|_V \left\| |\nabla \mu|^2 \theta^{\frac{p+1}{2}} \right\|_{V'} + C(p+1) \\ \le &\frac{2p}{c_p(p+1)} \left\| \theta^{\frac{p+1}{2}} \right\|_V^2 + C(p+1)^2 \left\| |\nabla \mu|^2 \theta^{\frac{p+1}{2}} \right\|_{L^{5/4}(\Omega)}^2 + C(p+1). \end{split}$$

Now, applying Hölder's inequality with exponents 6/5, 6 and interpolation inequality (A.1e) we obtain

$$\left\| |\nabla \mu|^2 \theta^{\frac{p+1}{2}} \right\|_{L^{5/4}(\Omega)}^2 = \left( \int_{\Omega} \theta^{\frac{5}{8}(p+1)} |\nabla \mu|^{\frac{5}{2}} \right)^{\frac{8}{5}} \le \|\nabla \mu\|_{L^{15}(\Omega)}^4 \left( \int_{\Omega} \theta^{\frac{3}{4}(p+1)} \right)^{\frac{4}{3}}$$

$$\leq C \|\mu\|_{H^3(\Omega)}^2 \left(\int_{\Omega} \theta^{\frac{3}{4}(p+1)}\right)^{\frac{4}{3}}.$$

Therefore, we have the following inequality for  $I_2$ :

$$I_2 \le \frac{2p}{c_p(p+1)} \left\| \theta^{\frac{p+1}{2}} \right\|_V^2 + C(p+1)^2 \left\| \mu \right\|_{H^3(\Omega)}^2 \left( \int_{\Omega} \theta^{\frac{3}{4}(p+1)} \right)^{\frac{4}{3}} + C(p+1).$$
(1.120)

Using (1.118), (1.119) and (1.120) yields

$$\frac{d}{dt} \int_{\Omega} \theta^{p+1} \le C(p+1)^2 \left( 1 + \|\mu\|_{H^3(\Omega)}^2 \right) \left( \int_{\Omega} \theta^{\frac{3}{4}(p+1)} \right)^{\frac{4}{3}} + C(p+1),$$

then, by a further integration over  $(0, t), t \in (0, T]$ ,

$$\int_{\Omega} \theta^{p+1}(t) \le C \, (p+1)^2 \left( \sup_{[0,T]} \left( \int_{\Omega} \theta^{\frac{3}{4}(p+1)} \right)^{\frac{4}{3}} + K_{p,T} \right), \tag{1.121}$$

where we took advantage of (1.110) and where with  $K_{p,T}$  we denote a term cointaining the information on the initial datum  $\theta^{p+1}(0)$  which possibly depends on T. This term depends exponentially on p, but this difficulty is later overcome taking the (1/p)-power. In order to apply Moser's iteration, we consider the sequence  $(p_k)_k$  of real numbers defined by

$$p_0 = 3, \quad p_{k+1} = \frac{4}{3}p_k, \quad k \in \mathbb{N}.$$

Let us take  $p = p_{k+1} - 1$  in (1.121). We then have

$$\int_{\Omega} \theta^{p_{k+1}}(t) \le C p_{k+1}^2 \left( \sup_{[0,T]} \left( \int_{\Omega} \theta^{p_k} \right)^{\frac{4}{3}} + K_{p,T} \right),$$

hence

$$\sup_{[0,T]} \int_{\Omega} \theta^{p_{k+1}} \le C p_{k+1}^2 \max\left\{ \sup_{[0,T]} \left( \int_{\Omega} \theta^{p_k} \right)^{\frac{4}{3}}, K_{p,T} \right\}.$$

Thanks to (1.106), we already have  $\theta \in L^{\infty}(0,T; L^{q+2}(\Omega))$ , where q was introduced in (1.58). Therefore, we can apply the Moser lemma and get

$$\forall k \in \mathbb{N}, \qquad \sup_{[0,T]} \|\theta\|_{L^{p_k}(\Omega)} \le C.$$

Taking the limit as k goes to infinity leads to (1.116). This also immediately entails that  $K(\theta) \in L^{\infty}(0,T; L^{\infty}(\Omega)).$ 

We are now able to prove that  $\theta \in L^{\infty}(0,T;V)$ . In order to do so, we formally multiply
(1.3) by  $\partial_t K(\theta) = \kappa(\theta) \theta_t$ . We obtain

$$\int_{\Omega} \left| \sqrt{\kappa(\theta)} \theta_t \right|^2 + \frac{1}{2} \frac{d}{dt} \int_{\Omega} |\nabla K(\theta)|^2$$
$$= -\int_{\Omega} \theta \Delta \mu \kappa(\theta) \theta_t + \int_{\Omega} |\nabla \mu|^2 \kappa(\theta) \theta_t = \int_{\Omega} \kappa(\theta) \theta_t [|\nabla \mu|^2 - \theta \Delta \mu] \, \mathrm{dx}$$

Owing to (1.116), (1.107) and (A.1a), the right hand side can be controlled as

$$\begin{split} &\int_{\Omega} \kappa(\theta)\theta_t [|\nabla\mu|^2 - \theta\,\Delta\mu] \,\mathrm{dx} \\ &= \int_{\Omega} \sqrt{\kappa(\theta)} \sqrt{\kappa(\theta)} \theta_t \left(\theta\Delta\mu + |\nabla\mu|^2\right) \\ &\leq \|\sqrt{\kappa(\theta)}\|_{L^{\infty}(\Omega)} \|\sqrt{\kappa(\theta)} \theta_t\| \left( \|\theta\|_{L^{\infty}(\Omega)} \|\Delta\mu\| + \|\nabla\mu\|_{L^4(\Omega)}^2 \right) \\ &\leq c \|\sqrt{\kappa(\theta)} \theta_t\| \|\mu\|_{H^2(\Omega)} \leq \frac{1}{2} \|\sqrt{\kappa(\theta)} \theta_t\|^2 + c \|\mu\|_{H^2(\Omega)}^2, \end{split}$$

where in the last row we controlled  $\|\nabla \mu\|_{L^4(\Omega)}^2$  by (A.1a) and (1.107). Then, on account of (1.110), we get

$$\theta \in H^1(0,T;L^2(\Omega)) \qquad K(\theta) \in L^{\infty}(0,T;V)$$

and this last estimate entails the desired result  $\theta \in L^{\infty}(0,T;V)$ . Finally, by reading (1.2) as

$$-\Delta\varphi + F'(\varphi) = \theta + \mu \in L^{\infty}(0,T;V),$$

we deduce the thesis using (1.65), that is  $\varphi \in L^{\infty}(0,T; H^3(\Omega))$ .

#### 1.4.4 Uniqueness

We now address the uniqueness of solution in Theorem 1.4.1. Let  $z_0 \in \mathcal{V}$  and let  $(\varphi_i, \mu_i, \theta_i), i = 1, 2$ , be a couple of (stable) solutions both emanating from  $z_0$  over the interval (0, T). Taking  $(\varphi, \mu, \theta) := (\varphi_1 - \varphi_2, \mu_1 - \mu_2, \theta_1 - \theta_2)$ , we can readily obtain

$$\varphi_t = \Delta \mu, \tag{1.122}$$

$$\mu = -\Delta\varphi + F'(\varphi_1) - F'(\varphi_2) - \theta, \qquad (1.123)$$

$$\theta_t + \theta_1 \Delta \mu + \theta \Delta \mu_2 - \Delta [K(\theta_1) - K(\theta_2)] = (\nabla \mu_1 + \nabla \mu_2) \cdot \nabla \mu$$
(1.124)

coupled with null initial data. This guarantees e.g.  $\varphi_{\Omega}(t) = 0 \quad \forall t \ge 0$ . By the regularity (1.65)-(1.67), we observe

$$\|\varphi_i(t)\|_{H^3(\Omega)} + \|\mu_i(t)\|_V + \|\theta_i(t)\|_V \le c, \quad t \in (0,T)$$

$$\|\mu_i\|_{L^2(0,T;H^3(\Omega))} + \|\theta_i\|_{L^\infty(0,T;L^\infty(\Omega))} \le c$$
(1.125)

for some positive constant c depending on T and on the initial data. These properties will be frequently used in the following.

The proof we show is based on the application of Gronwall's Lemma to the functional

$$\mathcal{Z}(t) := \varepsilon \left\| \nabla \varphi(t) \right\|^2 - 2\varepsilon \left\langle \theta(t) - \theta_{\Omega}(t), \varphi(t) \right\rangle + \left\| \theta(t) - \theta_{\Omega}(t) \right\|_{V'}^2 + \theta_{\Omega}(t)^2, \quad (1.126)$$

which is zero for t = 0. We notice that for  $\varepsilon > 0$  small enough,

$$\mathcal{Z}(t) \ge c_{\varepsilon} (\|\nabla\varphi(t)\|^2 + \|\theta(t) - \theta_{\Omega}(t)\|_{V'}^2 + \theta_{\Omega}(t)^2).$$
(1.127)

Therefore we need to estimate the terms  $\frac{d}{dt} \|\nabla\varphi(t)\|^2$ ,  $\frac{d}{dt} \|\theta(t) - \theta_{\Omega}(t)\|_{V'}^2$  and  $\frac{d}{dt}\theta_{\Omega}(t)^2$ . The first one will be addressed in Section 1.4.4, the second in Section 1.4.4 and the third in Section 1.4.4.

#### **Preliminary estimates**

First of all we control  $\Delta \varphi$  by testing (1.123) by  $-\Delta \varphi$ :

$$\begin{split} \|\Delta\varphi\|^2 &= -\langle \mu - \mu_{\Omega}, \Delta\varphi \rangle - \langle \theta - \theta_{\Omega}, \Delta\varphi \rangle + \langle F'(\varphi_1) - F'(\varphi_2), \Delta\varphi \rangle \\ &\leq c \|\Delta\varphi\| \left( \|\nabla\mu\| + \|\theta - \theta_{\Omega}\| + \|F'(\varphi_1) - F'(\varphi_2)\| \right) \end{split}$$

Exploiting Hölder's inequality with exponents 3 and 3/2 and recalling (1.57), (1.125), we obtain

$$\|F'(\varphi_1) - F'(\varphi_2)\|^2 \le c \left\langle (1 + |\varphi_1|^{2p_F} + |\varphi_2|^{2p_F}), \varphi^2 \right\rangle$$
  
$$\le c \|\varphi\|_{L^3(\Omega)}^2 \le c \|\varphi\|_V^2.$$

Eventually, according to (1.85), we conclude that

$$\|\Delta\varphi\|^{2} \leq c(\|\nabla\mu\|^{2} + \|\theta - \theta_{\Omega}\|^{2} + \|\varphi\|_{V}^{2}) \leq c(\|\varphi_{t}\|_{V'}^{2} + \|\theta - \theta_{\Omega}\|^{2} + \|\varphi\|_{V}^{2}).$$
(1.128)

#### Difference in order parameters for Gronwall's argument (1.126)

Testing (1.122) by  $J^{-1}\varphi_t$ , we get

$$\left\|\varphi_t\right\|_{V'}^2 + \left\langle\mu, \varphi_t\right\rangle = 0,$$

where, multiplying (1.123) by  $\varphi_t$ , we can write the second term as

$$\langle \mu, \varphi_t \rangle = \frac{1}{2} \frac{d}{dt} (\|\nabla \varphi\|^2 - 2 \langle \theta - \theta_\Omega, \varphi \rangle) + \langle F'(\varphi_1) - F'(\varphi_2), \varphi_t \rangle + \langle \theta_t, \varphi \rangle.$$

Combining the previous relations we infer

$$\|\varphi_t\|_{V'}^2 + \frac{1}{2} \frac{d}{dt} (\|\nabla\varphi\|^2 - 2\langle\theta - \theta_\Omega, \varphi\rangle) = -\langle F'(\varphi_1) - F'(\varphi_2), \varphi_t \rangle - \langle\theta_t, \varphi\rangle.$$
(1.129)

We take care of second term on the right hand side by multiplying (1.124) by  $\varphi$ :

$$\begin{split} \langle \theta_t, \varphi \rangle &= \langle K(\theta_1) - K(\theta_2), \Delta \varphi \rangle + \langle \nabla \mu, \nabla \theta_1 \varphi \rangle + \langle \nabla \mu, \theta_1 \nabla \varphi \rangle \\ &- \langle \theta \Delta \mu_2, \varphi \rangle + \langle (\nabla \mu_1 + \nabla \mu_2) \cdot \nabla \mu, \varphi \rangle \,. \end{split}$$

We notice that a direct estimate of the first term in the right hand side could have been provided due to (1.128) and the regularity achieved on  $K(\theta)$ . However in this way, it turns to be difficult to reabsorb the term  $C \|\varphi_t\|_{V'}$  in the left hand side. Therefore the gain of regularity in  $\theta$  apparently does not simplify this part of the proof and we need to proceed as in [39].

We notice that, exploiting (1.123) two times, the first term on the right hand side reads

$$\langle K(\theta_1) - K(\theta_2), \Delta \varphi \rangle$$

$$= \langle K(\theta_1) - K(\theta_2), -\mu + \mu_{\Omega} + F'(\varphi_1) - F'(\varphi_2) - \theta + \theta_{\Omega} \rangle$$

$$- (\mu_{\Omega} + \theta_{\Omega}) \int_{\Omega} [K(\theta_1) - K(\theta_2)]$$

$$= \langle K(\theta_1) - K(\theta_2), -\mu + \mu_{\Omega} + F'(\varphi_1) - F'(\varphi_2) - (F'(\varphi_1) - F'(\varphi_2))_{\Omega} \rangle$$

$$- \langle K(\theta_1) - K(\theta_2), \theta - \theta_{\Omega} \rangle .$$

Therefore we have,

$$\begin{aligned} &\langle \theta_t, \varphi \rangle \\ &= \langle K(\theta_1) - K(\theta_2), -\mu + \mu_\Omega + F'(\varphi_1) - F'(\varphi_2) - (F'(\varphi_1) - F'(\varphi_2))_\Omega \rangle \\ &- \langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega \rangle + \langle \nabla \mu, \nabla \theta_1 \varphi \rangle + \langle \nabla \mu, \theta_1 \nabla \varphi \rangle - \langle \theta \Delta \mu_2, \varphi \rangle \\ &+ \langle (\nabla \mu_1 + \nabla \mu_2) \cdot \nabla \mu, \varphi \rangle \,. \end{aligned}$$

Owing (1.129) and combining the above relations, we obtain

$$\left\|\varphi_{t}\right\|_{V'}^{2} + \frac{1}{2}\frac{d}{dt}\left(\left\|\nabla\varphi\right\|^{2} - 2\left\langle\theta - \theta_{\Omega}, \varphi\right\rangle\right) - \left\langle K(\theta_{1}) - K(\theta_{2}), \theta - \theta_{\Omega}\right\rangle$$

$$= - \langle F'(\varphi_1) - F'(\varphi_2), \varphi_t \rangle$$
  

$$- \langle K(\theta_1) - K(\theta_2), -\mu + \mu_{\Omega} + F'(\varphi_1) - F'(\varphi_2) - (F'(\varphi_1) - F'(\varphi_2))_{\Omega} \rangle$$
  

$$- \langle \nabla \mu, \nabla \theta_1 \varphi + \theta_1 \nabla \varphi \rangle + \langle \theta \Delta \mu_2, \varphi \rangle$$
  

$$- \langle (\nabla \mu_1 + \nabla \mu_2) \cdot \nabla \mu, \varphi \rangle =: I_3 + I_4 + I_5 + I_6 + I_7.$$
(1.130)

First of all using (1.57), we have

$$\begin{aligned} \|\nabla(F'(\varphi_1) - F'(\varphi_2))\| & (1.131) \\ \leq c \left\| |\nabla\varphi|(1 + |\varphi_1^{p_F}|) + |\varphi|(1 + |\varphi_1|^{p_F-1} + |\varphi_2|^{p_F-1})|\nabla\varphi_2| \right\| \leq c \left\|\varphi\right\|_V. \end{aligned}$$

Therefore, as  $\langle \varphi_t, 1 \rangle = 0$ , we deduce

$$I_3 := - \langle F'(\varphi_1) - F'(\varphi_2), \varphi_t \rangle$$
  
$$\leq C \|\nabla (F'(\varphi_1) - F'(\varphi_2))\| \|\varphi_t\|_{V'} \leq c \|\varphi\|_V \|\varphi_t\|_{V'}.$$

On the other hand, by (1.116),

$$\|K(\theta_1) - K(\theta_2)\|_{3/2} \le c(\|\theta - \theta_\Omega\| + |\theta_\Omega|),$$

and, according to (1.131), it follows

$$I_{4} := - \langle K(\theta_{1}) - K(\theta_{2}), \mu - \mu_{\Omega} - F'(\varphi_{1}) - F'(\varphi_{2}) - (F'(\varphi_{1}) - F'(\varphi_{2}))_{\Omega} \rangle$$
  

$$\leq \|K(\theta_{1}) - K(\theta_{2})\|_{L^{3/2}(\Omega)} (\|\mu - \mu_{\Omega}\|_{L^{3}(\Omega)} + \|F'(\varphi_{1}) - F'(\varphi_{2}) - (F'(\varphi_{1}) - F'(\varphi_{2}))_{\Omega}\|_{L^{3}(\Omega)})$$
  

$$\leq c(\|\theta - \theta_{\Omega}\| + |\theta_{\Omega}|)(\|\nabla\mu\| + \|\varphi\|_{V}).$$

Owing to (A.1a), (A.1b) and (A.1f)

$$I_{5} := - \langle \nabla \mu, \nabla \theta_{1} \varphi + \theta_{1} \nabla \varphi \rangle$$
  

$$\leq c \|\nabla \mu\| \left( \|\nabla \theta_{1}\| \|\varphi\|_{L^{\infty}(\Omega)} + \|\nabla \varphi\|_{L^{4}(\Omega)} \|\theta_{1}\|_{L^{4}(\Omega)} \right)$$
  

$$\leq c \|\nabla \mu\| \left( \|\varphi\|^{1/2} \|\varphi\|_{H^{2}(\Omega)}^{1/2} + \|\nabla \varphi\|^{1/2} \|\varphi\|_{H^{2}(\Omega)}^{1/2} \right)$$
  

$$\leq c \|\nabla \mu\| \|\nabla \varphi\|^{1/2} \|\varphi\|_{H^{2}(\Omega)}^{1/2}.$$

Now, using (A.1f) and the injection  $V \subset L^p(\Omega)$ , for  $p \ge 1$ ,

$$I_{6} := \langle \theta \Delta \mu_{2}, \varphi \rangle = \theta_{\Omega} \langle \Delta \mu_{2}, \varphi \rangle + \langle (\theta - \theta_{\Omega}) \Delta \mu_{2}, \varphi \rangle$$
$$\leq c |\theta_{\Omega}| \|\nabla \mu_{2}\| \|\nabla \varphi\| + c \|\theta - \theta_{\Omega}\| \|\Delta \mu_{2}\|_{L^{4}(\Omega)} \|\varphi\|_{L^{4}(\Omega)}$$

$$\leq c|\theta_{\Omega}| \|\nabla \varphi\| + c \|\theta - \theta_{\Omega}\| \|\mu_{2}\|_{H^{3}(\Omega)} \|\nabla \varphi\|.$$

Finally combining the previous strategy with (A.1a), we get

$$I_{7} := - \left\langle (\nabla \mu_{1} + \nabla \mu_{2}) \cdot \nabla \mu, \varphi \right\rangle \leq \|\nabla \mu_{1} + \nabla \mu_{2}\|_{L^{4}(\Omega)} \|\nabla \mu\| \|\varphi\|_{L^{4}(\Omega)} \\ \leq \left( \|\mu_{1}\|_{H^{2}(\Omega)}^{1/2} + \|\mu_{2}\|_{H^{2}(\Omega)}^{1/2} \right) \|\nabla \mu\| \|\nabla \varphi\|.$$

Eventually, the above computations and Young's inequality yield

$$I_{3} + I_{4} + I_{5} + I_{6} + I_{7}$$

$$\leq c \|\varphi_{t}\|_{V'} \|\varphi\|_{V} + c(\|\theta - \theta_{\Omega}\| + |\theta_{\Omega}|)(1 + \|\mu_{2}\|_{H^{3}(\Omega)}) \|\varphi\|_{V}$$

$$+ c \|\nabla\mu\| \|\nabla\varphi\|^{1/2} \|\varphi\|^{1/2}_{H^{2}(\Omega)} + c(\|\theta - \theta_{\Omega}\| + |\theta_{\Omega}|) \|\nabla\mu\|$$

$$+ c \left(1 + \|\mu_{1}\|^{1/2}_{H^{2}(\Omega)} + \|\mu_{2}\|^{1/2}_{H^{2}(\Omega)}\right) \|\nabla\mu\| \|\nabla\varphi\|$$

$$\leq \frac{1}{2} \|\varphi_{t}\|^{2}_{V'} + \frac{\alpha}{2} \|\nabla\mu\|^{2} + \frac{c}{\alpha} \|\theta - \theta_{\Omega}\|^{2} + \frac{\alpha}{2} \|\Delta\varphi\|^{2} + g(t)(\|\varphi\|^{2}_{V} + \theta^{2}_{\Omega})$$

$$\leq \left(\frac{1}{2} + c\alpha\right) \|\varphi_{t}\|^{2}_{V'} + c \left(\frac{1}{\alpha} + \alpha\right) \|\theta - \theta_{\Omega}\|^{2} + g(t)(\|\varphi\|^{2}_{V} + \theta^{2}_{\Omega}).$$

Where in the last passage we took advantage of (1.128) to estimate the term depending on  $\Delta \varphi$ . Moreover, we have defined

$$g(t) := c[1 + \|\mu_1\|_{H^3(\Omega)}^2 + \|\mu_2\|_{H^3(\Omega)}^2], \qquad (1.132)$$

with (large) constant c > 0 also depending on the choice of the small constant  $\alpha > 0$ Combining the previous estimates with (1.130), we finally get

$$\begin{aligned} \|\varphi_t\|_{V'}^2 + \frac{d}{dt}(\|\nabla\varphi\|^2 - 2\langle\theta - \theta_\Omega, \varphi\rangle) - 2\langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega\rangle \\ \leq \left(\frac{1}{2} + c\alpha\right)\|\varphi_t\|_{V'} + c\left(\frac{1}{\alpha} + \alpha\right)\|\theta - \theta_\Omega\|^2 + g(t)(\|\varphi\|_V^2 + \theta_\Omega^2). \end{aligned}$$
(1.133)

As mentioned before, the aim of these calculations is to apply Gronwall's Lemma to a specific functional already introduced. In order to do that, we are trying to obtain the derivative of such functional on the left-hand side and the functional itself on the right. Thus all terms arising from that must be either integrable over (0, T) (which is the g(t)) or they must be balanced with some term on the left as  $\|\varphi_t\|_{V'}^2$  or  $\|\theta - \theta_{\Omega}^2\|$  (which will arise from  $\langle K(\theta_1) - K(\theta_2), \theta - \theta_{\Omega} \rangle$ ).

#### Difference of temperatures for Gronwall's argument (1.126).

We test (1.124) by  $J^{-1}(\theta - \theta_{\Omega})$  and integrate by parts, therefore we obtain

$$\frac{1}{2} \frac{d}{dt} \|\theta - \theta_{\Omega}\|_{V'}^{2} + \langle K(\theta_{1}) - K(\theta_{2}), \theta - \theta_{\Omega} \rangle$$
  
=  $\langle \nabla \theta_{1} \cdot \nabla \mu, J^{-1}(\theta - \theta_{\Omega}) \rangle + \langle \theta_{1} \nabla \mu, \nabla J^{-1}(\theta - \theta_{\Omega}) \rangle$   
-  $\langle (\theta - \theta_{\Omega}) \Delta \mu_{2}, J^{-1}(\theta - \theta_{\Omega}) \rangle + \theta_{\Omega} \langle \mu_{2} - (\mu_{2})_{\Omega}, \theta - \theta_{\Omega} \rangle$   
+  $\langle (\nabla \mu_{1} + \nabla \mu_{2}) \cdot \nabla \mu, J^{-1}(\theta - \theta_{\Omega}) \rangle =: I_{8} + I_{9} + I_{10}.$ 

First of all we have

$$I_8 := \left\langle \theta_1 \nabla \mu, \nabla J^{-1}(\theta - \theta_\Omega) \right\rangle$$
  
$$\leq c \|\nabla \mu\| \|\nabla J^{-1}(\theta - \theta_\Omega)\| \leq c \|\nabla \mu\| \|\theta - \theta_\Omega\|_{V'}.$$

From (A.1b),

$$\left\|J^{-1}(\theta-\theta_{\Omega})\right\|_{L^{\infty}(\Omega)} \leq c \left\|\theta-\theta_{\Omega}\right\|_{V'}^{1/2} \left\|\theta-\theta_{\Omega}\right\|^{1/2}.$$

Thus, we infer

$$I_{9} := -\left\langle \nabla \theta_{1} \cdot \nabla \mu - (\theta - \theta_{\Omega}) \Delta \mu_{2}, J^{-1}(\theta - \theta_{\Omega}) \right\rangle + \left\langle (\nabla \mu_{1} + \nabla \mu_{2}) \cdot \nabla \mu, J^{-1}(\theta - \theta_{\Omega}) \right\rangle \leq \left\| J^{-1}(\theta - \theta_{\Omega}) \right\|_{L^{\infty}} [\|\theta_{1}\|_{V} \|\nabla \mu\| + \|\Delta \mu_{2}\| \|\theta - \theta_{\Omega}\| + (\|\nabla \mu_{1}\| + \|\nabla \mu_{2}\|) \|\nabla \mu\|] \leq c \|\theta - \theta_{\Omega}\|_{V'}^{1/2} \|\theta - \theta_{\Omega}\|^{1/2} \|\nabla \mu\| + c \|\mu_{2}\|_{H^{2}(\Omega)} \|\theta - \theta_{\Omega}\|_{V'}^{1/2} \|\theta - \theta_{\Omega}\|^{3/2}.$$

Eventually,

$$I_{10} := \theta_{\Omega} \langle \mu_2 - (\mu_2)_{\Omega}, \theta - \theta_{\Omega} \rangle$$
  
$$\leq c |\theta_{\Omega}| \|\mu_2 - (\mu_2)_{\Omega}\|_V \|\theta - \theta_{\Omega}\|_{V'} \leq c |\theta_{\Omega}| \|\theta - \theta_{\Omega}\|_{V'}.$$

Combining the estimates of  $I_8, I_9, I_{10}$  and exploiting Young's inequality, we finally have

$$\frac{1}{2}\frac{d}{dt}\left\|\theta - \theta_{\Omega}\right\|_{V'}^{2} + \left\langle K(\theta_{1}) - K(\theta_{2}), \theta - \theta_{\Omega} \right\rangle \leq \delta\varepsilon \left\|\theta - \theta_{\Omega}\right\|^{2}$$

$$+ \alpha\varepsilon \left\|\nabla\mu\right\|^{2} + C_{*}(1 + \left\|\mu_{2}\right\|_{H^{2}(\Omega)}^{2})\left\|\theta - \theta_{\Omega}\right\|_{V'}^{2} + c|\theta_{\Omega}|^{2},$$

$$(1.134)$$

with the (large) constant  $C_*$  depending on the small constants  $\alpha, \delta, \varepsilon$  which will be specified at the end.

#### Difference of temperatures' means for Gronwall's argument (1.126)

Integrating (1.124) over  $\Omega$  we obtain

$$|\Omega|(\theta_{\Omega})_{t} = \langle \nabla \theta_{1}, \nabla \mu \rangle - \langle \theta - \theta_{\Omega}, \Delta \mu_{2} \rangle + \langle (\nabla \mu_{1} + \nabla \mu_{2}), \nabla \mu \rangle, \qquad (1.135)$$

and by (1.125) it yields

$$\left|(\theta_{\Omega})_{t}\right| \leq c(\left\|\nabla\mu\right\| + \left\|\mu_{2}\right\|_{H^{3}(\Omega)} \left\|\theta - \theta_{\Omega}\right\|_{V'}).$$

Moreover, multiplying (1.135) by  $\theta_{\Omega}$  we have, for (small)  $\alpha > 0$  we will choose later and corresponding (large) c > 0,

$$\frac{1}{2}\frac{d}{dt}\theta_{\Omega}^{2} \leq c|\theta_{\Omega}|(\|\nabla\mu\| + \|\mu_{2}\|_{H^{3}(\Omega)} \|\theta - \theta_{\Omega}\|_{V'}^{2})$$
(1.136)

$$\leq \alpha \varepsilon \|\nabla \mu\|^{2} + c(\theta_{\Omega}^{2} + \|\mu_{2}\|_{H^{3}(\Omega)}^{2} \|\theta - \theta_{\Omega}\|_{V'}^{2}).$$
(1.137)

#### Conclusion

We recall the definition of the functional we want to use

$$\mathcal{Z}(t) := \varepsilon \left\| \nabla \varphi \right\|^2 - 2\varepsilon \left\langle \theta - \theta_{\Omega}, \varphi \right\rangle + \left\| \theta - \theta_{\Omega} \right\|_{V'}^2 + \theta_{\Omega}^2,$$

Summing (1.134), (1.136) and  $\frac{\varepsilon}{2}$  times (1.133), we obtain

$$\frac{1}{2}\frac{d}{dt}\mathcal{Z} + (1-\varepsilon)\left\langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega \right\rangle + \frac{\varepsilon}{2} \left\| \varphi_t \right\|_{V'}^2$$
  
$$\leq \varepsilon (c + c\alpha^2 + \delta) \left\| \theta - \theta_\Omega \right\|^2 + 2\alpha\varepsilon \left\| \nabla \mu \right\|^2 + g(t)\mathcal{Z},$$

where g was introduced in (1.132).

Next, we take care of the second term in the left hand side. From (1.59), we obtain

$$\langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega \rangle = \|\theta - \theta_\Omega\|^2 + \frac{1}{q+1} \langle l(\theta_1) - l(\theta_2), \theta - \theta_\Omega \rangle$$

where  $l(\theta_i) = \theta_i^{q+1}, i = 1, 2$ . Now

$$l(\theta_1) - l(\theta_2) = \int_0^1 \frac{d}{ds} l(s\theta_1 + (1-s)\theta_2) ds$$
  
=  $\int_0^1 l'(s\theta_1 + (1-s)\theta_2)(\theta_1 - \theta_2) ds = \omega(\theta_1, \theta_2)\theta,$ 

where we set

$$\omega(\theta_1, \theta_2) := \int_0^1 l'(s\theta_1 + (1-s)\theta_2) ds.$$

We observe that it holds  $\omega(\theta_1, \theta_2) \ge 0$  almost everywhere. Moreover, we notice that (1.116) implies,

$$|\omega(\theta_1, \theta_2)| \le c \tag{1.138}$$

Therefore we infer

$$\begin{split} \langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega \rangle &= \|\theta - \theta_\Omega\|^2 + \frac{1}{1+q} \langle l(\theta_1) - l(\theta_2), \theta - \theta_\Omega \rangle \\ &= \|\theta - \theta_\Omega\|^2 + \frac{1}{q+1} \int_\Omega \omega(\theta_1, \theta_2) \theta(\theta - \theta_\Omega) \\ &= \|\theta - \theta_\Omega\|^2 + \frac{1}{q+1} \int_\Omega \omega(\theta_1, \theta_2) |\theta - \theta_\Omega|^2 \\ &+ \frac{1}{q+1} \int_\Omega \omega(\theta_1, \theta_2) \theta_\Omega(\theta - \theta_\Omega) \\ &\geq \|\theta - \theta_\Omega\|^2 + \frac{1}{q+1} \int_\Omega \omega(\theta_1, \theta_2) \theta_\Omega(\theta - \theta_\Omega). \end{split}$$

Hence, exploiting (1.138), we get

$$\left|\frac{1}{q+1}\int_{\Omega}\omega(\theta_{1},\theta_{2})\theta_{\Omega}(\theta-\theta_{\Omega})\right| \leq c|\theta_{\Omega}| \left\|\theta-\theta_{\Omega}\right\| \leq \frac{1}{2}\left\|\theta-\theta_{\Omega}\right\|^{2} + c\theta_{\Omega}^{2}$$

and moreover

$$\langle K(\theta_1) - K(\theta_2), \theta - \theta_\Omega \rangle \ge \frac{1}{2} \|\theta - \theta_\Omega\|^2 - c\theta_\Omega^2.$$

Putting everything together we finally have

$$\frac{1}{2}\frac{d}{dt}\mathcal{Z} + \frac{1-\varepsilon}{2} \|\theta - \theta_{\Omega}\|^{2} + \frac{\varepsilon}{2} \|\varphi_{t}\|_{V'}^{2}$$

$$\leq \varepsilon(c + c\alpha^{2} + \delta) \|\theta - \theta_{\Omega}\|^{2} + 3\alpha\varepsilon \|\nabla\mu\|^{2} + g(t)\mathcal{Z}$$

$$\stackrel{(1.85)}{\leq} \varepsilon(c + c\alpha^{2} + \delta) \|\theta - \theta_{\Omega}\|^{2} + 3\alpha\varepsilon c \|\varphi_{t}\|_{V'}^{2} + g(t)\mathcal{Z}.$$
(1.139)

The constant c on the right hand side of (1.139) only depends on the regularity properties of the solutions collected in (1.125). In particular, it is *independent* of the parameters  $\alpha, \delta, \varepsilon$ . Therefore, chosing  $\alpha > 0$  small enough, the second term on the right hand side of (1.139) can be absorbed the corresponding quantities in the left hand side. Moreover, taking  $\varepsilon$  sufficiently small (it might depend on other parameters), the first term on the right hand side can be absorbed, too. As a matter of fact, we are able to semplify (1.139) as follows

$$\frac{d}{dt}\mathcal{Z} + \kappa_0(\|\theta - \theta_\Omega\|^2 + \|\varphi_t\|_{V'}^2) \le g(t)\mathcal{Z},$$

where  $\kappa_0 > 0$  and g was defined in (1.132), hence, exploiting (1.125), it is summable over the interval (0, T). Since  $\mathcal{Z}(0) = 0$ , then by Gronwall's Lemma and (1.127) we eventually see that  $\mathcal{Z}$  is identically 0 over (0, T), which gives us the assert.

# Chapter 2

# Non-isothermal Cahn-Hilliard model for tumor growth

The study of tumor growth processes has become of great interest also for mathematicians in recent years [5, 9, 21, 28, 92, 104, 118]. Indeed, mathematical models might be able to give further insights in tumor growth behaviour. In particular, the framework of diffuse interface modeling with Cahn-Hilliard equations [19] has received increasing attention. In this context, the tumor is seen as an expanding mass surrounded by healthy tissues. Its evolution is assumed to be governed by mechanisms such as proliferation of cells via nutrient consumption, apoptosis [56, 78, 102] and, in more complex models like [58, 59, 60, 72], also chemotaxis and active transport of specific chemical species effects. Moreover it is possible to include the effects of fluid flow into the evolution of the tumor, which brings to the so-called Cahn-Hilliard-Darcy models (see [60, 78]). However, up to our knowledge it seems that even if the effects of variations of temperature have been studied for Cahn-Hilliard equations [39, 46, 77], they have been neglected so far in the analysis of tumor growth. From the medical point of view, the effects of temperature on tumor growth have not been completely understood yet, although they have been investigated since the very beginning of the 20th century [110]. The general tendency of the scientific community seems to support the thesis that hyperthermia can lead to partial or complete destruction of tumor cells [15, 98, 109, 115]. In fact, it has also been observed that low ambient temperature influences the production of particular nutrients for the tumor [88]. Nevertheless, we focus here on the case which does not take into account the production of a nutrient due to temperature. In this work we introduce a new diffuse interface model for tumor growth, taking into account proliferation of cells, nutrient consumption and apoptosis and moreover temperature effects. Our aim consists in proving an existence result for weak entropy solutions (cnfr. Definition 2.1) to our model. We remark that a rigorous mathematical theory of well-posedness results has been addressed in multiple

works, such as [56, 58, 102]. From the biological point of view, we assume that tumor cells only die by apoptosis, therefore we do not take into account the possibility of tumor necrosis (differently e.g. from [59]). We also suppose that the healthy cells surrounding the tumor do not interact with the tumor itself, neglecting the possible response of the immune system.

According to these considerations, we will derive the following PDE system, describing the behaviour of a two-component mixture consisting of healthy cells and tumor cells

$$\varphi_t = \Delta \mu + (\mathcal{P}\sigma - \mathcal{A})h(\varphi) \tag{2.1}$$

$$\mu = -\varepsilon \Delta \varphi + \frac{1}{\varepsilon} F'(\varphi) - \theta - \chi_{\varphi} \sigma$$
(2.2)

$$\theta_t + \theta \varphi_t - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2$$
(2.3)

$$\sigma_t = \Delta \sigma - \mathcal{C}\sigma h(\varphi) + \mathcal{B}(\sigma_B - \sigma). \tag{2.4}$$

We carry out our analysis in  $\Omega \times (0, \infty)$ , where  $\Omega \subset \mathbb{R}^3$  is a smooth domain. According to the derivation of the model shown in Section 2.1, we suppose that the system is isolated from the exterior. This condition translates in no-flux boundary conditions (i.e. homogeneous Neumann) for all the unknowns.

The evolution of the tumor is described by the order parameter  $\varphi$  which represents the local concentration of tumor cells,  $\varphi \in [-1, 1]$ , with  $\{\varphi = 1\}$  representing the tumor phase and  $\{\varphi = -1\}$  the healthy one. Moreover  $\mu$  denotes the chemical potential of phase transition from healthy to tumor cells,  $\theta$  is the absolute temperature,  $\kappa(\theta)$  represents the heat conductivity and  $\varepsilon$  is a small parameter related to the thickness of interfacial layers. We denote by  $\sigma$  the concentration of a nutrient consumed (only) by the tumor cells (e.g. oxygen and glucose). The parameter  $\chi_{\varphi} \geq 0$  is linked to transport mechanisms such as chemotaxis and active uptake. Although we will show in Section 2.1 how this parameter is included in the model, for sake of simplicity we will neglect it throughout the mathematical analysis, with the aim of including it in future works. The positive constant parameters  $\mathcal{P}, \mathcal{A}, \mathcal{C}$  and  $\mathcal{B}$  indicate respectively the tumor proliferation rate, apoptosis rate, nutrient consuption rate and nutrient supply rate. The function h is chosen as monotone increasing, nonnegative in [-1, 1] and such that h(-1) = 0 and h(1) = 1. The tumor growth is thus described by the term  $\mathcal{P}\sigma h(\varphi)$ , which reasonably increases proportionally to the concentration of tumor cells, while the death of tumor cells is modelled by the term  $\mathcal{A}h(\varphi)$ . Therefore, according to (2.1), if  $\mathcal{P}\sigma - \mathcal{A} > 0$ , then the tumor expands and it happens faster when the concentration of tumor cells is already high. If otherwise  $\mathcal{P}\sigma - \mathcal{A} < 0$  then the tumor reduces and the tumor cells die faster when the concentration of tumor cells is large. The term  $\mathcal{C}\sigma h(\varphi)$  represents the consumption of the nutrient by the tumor cells. The term  $\mathcal{B}(\sigma - \sigma_B)$  is due to the fact that we consider here the case where the tumor has its own vasculature (as in e.g. [16], [102]), where the threshold  $\sigma_B \in (0, 1)$  is the constant nutrient concentration in the pre-existing vasculature. In particular, if  $\sigma_B > \sigma$ ,  $\mathcal{B}(\sigma_B - \sigma)$  models the supply of nutrient from the blood vessels, on the other hand if  $\sigma_B < \sigma$ ,  $\mathcal{B}(\sigma_B - \sigma)$  represents the transport of nutrient away from the domain. Eventually, the function F(s) represents a polynomial potential having at least cubic growth at infinity, whose assumptions will be specified in Section 2.2. A simple choice might be a *double-well potential* with equal minima at  $s = \pm 1$  penalizing the deviation of the length  $|\varphi|$  from its natural value 1. This more general potential allows  $\varphi$  to take values also outside of the significance interval [-1, 1], therefore we will carry out our analysis also in the case  $|\varphi| > 1$  and correspondingly extend function h. We also remark that although among Cahn-Hilliard literature the singular potentials, such as logarithm type (see e.g. [20]), are very common, the growth conditions that the problem requires make them unsuitable for our case, as it will be clear in Section 2.3.1.

In this work we derive a new phase field model according to the laws of thermodynamics describing the tumor growth, published in [75]. The novelty of [75] with respect to the pre-existing literature is to include possible variations of temperature in the tumor growth model. The presence of nutrient concentration  $\sigma$  in the system implies that here the spatial mean of  $\varphi$  is not conserved in time (as we can see from equation (2.1)), therefore the derivation of the model cannot follow the techniques proposed in Section 1.2.1. However, we are able to gain enough regularity for the quadruple ( $\varphi, \mu, \theta, \sigma$ ) in order to prove the existence of weak (entropy) solutions to the initial-boundary value problem associated to (2.1)–(2.4).

The structure of this Chapter is the following. In Section 2.1 we derive system (2.1)–(2.4) according to the approach proposed by Gurtin in [69]. Then we proceed with the mathematical analysis of our problem in the case  $\varepsilon = 1$ ,  $\chi_{\varphi} = 0$ . In particular, Section 2.3 is devoted to give the setting and to present the main result of this Chapter (which is Theorem 2.3.1) concerning the existence of weak entropy solutions to our problem. The proof is carried out in two steps. In Section 2.3.1 we gain a priori bounds for ( $\varphi, \mu, \theta, \sigma$ ). In Section 2.3.2 we use the weak sequential stability argument to prove the existence of weak entropy solutions. Namely, we exploit the a priori bounds obtained for a sequence of weak entropy solutions together with standard compactness results to pass to the limit.

### 2.1 Derivation of the model

We suppose that a two-component mixture consisting of healthy cells and tumor cells occupies an open spatial domain  $\Omega \subset \mathbb{R}^3$ . We denote by  $\varphi(x,t)$  the tumor phase concentration,  $\theta(x,t)$  is the absolute temperature and  $\sigma(x,t)$  is the concentration of a nutrient for the tumor cells. According to the Ginzburg-Landau theory for phase transitions, we postulate the free energy density  $\psi$  in the form

$$\psi = \frac{\varepsilon}{2} |\nabla \varphi|^2 + \frac{1}{\varepsilon} F(\varphi) + f(\theta) - \theta \varphi + N(\varphi, \sigma).$$
(2.5)

We observe that (2.5) differs from (1.11), because of the presence of the term N. This latter term describes both the chemical energy of the nutrient and the energy contributions given by the interactions between the tumor tissues and the nutrient.

One of the main difficulties we have to face in the derivation of our model is that, differently from standard Cahn-Hilliard models, such as the one studied in Chapter 1, the spatial mean of the tumor phase concentration  $\varphi$  is not conserved. Indeed the tumor may grow or shrink according to the right hand side of (2.1). Because of this issue, we cannot derive this model according to strategy proposed in Section 1.2.1. Indeed, we follow the approach presented in Section 1.2.2, proposed by Gurtin in [69], which relies on the balance law for internal microforces (1.39). The pivotal concepts behind Gurtin's strategy are described in Section 1.2.2. However, the main difference between this case and the one presented in Chapter 1 occurs in the mass balance equation.

Indeed, we are here considering the case where the tumor grows or reduces according to a source term. Therefore relation (1.40) is not suitable to describe this phenomenon. In this case the mass balance law thus reads

$$\varphi_t = -\operatorname{div} h + m, \tag{2.6}$$

where h is the mass flux and m is the external mass supply. The derivation procedure follows the same steps as the one in Section 1.2.2 is based on the first and second laws of thermodynamics.

#### Cahn-Hilliard system.

According to [69], we might write the first law in the form

$$\frac{d}{dt} \int_{R} e \mathrm{d}x = -\int_{\partial R} q \cdot \nu \mathrm{d}\eta + \mathcal{W}(R) + \mathcal{M}(R), \qquad (2.7)$$

that coincides with (1.44), but we recall here for convenience. In particular

$$\mathcal{W}(R) = \int_{\partial R} (\zeta \cdot \nu) \frac{\partial \varphi}{\partial t} \mathrm{d}\eta, \qquad (2.8)$$

$$\mathcal{M}(R) = -\int_{\partial R} \mu h \cdot \nu \mathrm{d}\eta + \int_{R} \mu m \mathrm{d}x$$
(2.9)

are the rate of working and the rate at which free energy is added to R (assuming no heat supply) respectively. We remark that (2.9) presents an extra term with respect to (1.43), given by the external mass supply m. From Green's formula, we can rewrite (2.7) as

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + \frac{\partial \varphi}{\partial t} \operatorname{div} \zeta + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h \cdot \nabla \mu - \mu \operatorname{div} h + \mu m.$$
(2.10)

Because the control volume R is arbitrary, exploiting the mass balance (2.6) and the microforce balance (1.39), we infer

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + (\mu - \pi)\frac{\partial \varphi}{\partial t} + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h\nabla \mu.$$
(2.11)

Since we here consider the free energy density  $\psi$  in (2.5) to be possibly dependent also on the nutrients, the combination of (1.14) and (1.15) leads to

$$\frac{\partial e}{\partial t} = \frac{\partial \psi}{\partial t} + \theta \frac{\partial s}{\partial t} + s \frac{\partial \theta}{\partial t} = \frac{\partial \psi}{\partial \varphi} \frac{\partial \varphi}{\partial t} + \frac{\partial \psi}{\partial \nabla \varphi} \frac{\partial \nabla \varphi}{\partial t} + \frac{\partial \psi}{\partial \sigma} \frac{\partial \sigma}{\partial t} + \theta \frac{\partial s}{\partial t}, \qquad (2.12)$$

hence

$$\frac{\partial\psi}{\partial t} + s\frac{\partial\theta}{\partial t} = \frac{\partial\psi}{\partial\varphi}\frac{\partial\varphi}{\partial t} + \frac{\partial\psi}{\partial\nabla\varphi}\frac{\partial\nabla\varphi}{\partial t} + \frac{\partial\psi}{\partial\sigma}\frac{\partial\sigma}{\partial t}.$$
(2.13)

We now impose the validity of the second law of thermodynamics in the form of the Clausius-Duhem inequality (1.18). We develop the left hand side of (1.18) as follows

$$\begin{split} \theta \left( \frac{\partial s}{\partial t} + \operatorname{div} Q \right) \stackrel{(1.14)}{=} \frac{\partial e}{\partial t} &- \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} + \theta \operatorname{div} Q \\ \stackrel{(1.13)}{=} \frac{\partial e}{\partial t} - \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} + \operatorname{div} q - Q \cdot \nabla \theta \\ \stackrel{(1.45)}{=} (\mu - \pi) \frac{\partial \varphi}{\partial t} + \zeta \cdot \nabla \frac{\partial \varphi}{\partial t} - h \nabla \mu - \frac{\partial \psi}{\partial t} - s \frac{\partial \theta}{\partial t} - Q \cdot \nabla \theta \\ \stackrel{(2.13)}{=} \left( \mu - \pi - \frac{\partial \psi}{\partial \varphi} \right) \frac{\partial \varphi}{\partial t} + \left( \zeta - \frac{\partial \psi}{\partial \nabla \varphi} \right) \frac{\partial \nabla \varphi}{\partial t} - h \nabla \mu - Q \cdot \nabla \theta. \end{split}$$

In order to satisfy relation (1.18), we impose

$$\mu - \pi - \frac{\partial \psi}{\partial \varphi} = 0, \qquad (2.14)$$

$$\zeta = \frac{\partial \psi}{\partial \nabla \varphi},\tag{2.15}$$

$$\frac{\partial \psi}{\partial \sigma} = 0, \tag{2.16}$$

$$h\nabla\mu + Q \cdot \nabla\theta \le 0, \tag{2.17}$$

where in particular in order for (2.17) to hold, we exploited Fourier's law (1.49). The combination of (2.5) and (2.15) straightly gives

$$\zeta = \varepsilon \nabla \varphi, \tag{2.18}$$

which leads to, according to (2.5), (1.39) and (2.14),

$$\mu = -\varepsilon \Delta \varphi + \frac{1}{\varepsilon} F'(\varphi) - \theta + \frac{\partial N}{\partial \varphi}.$$
(2.19)

Eventually, inequality (2.17) can be satisfied choosing  $h = -\nabla \mu$ , which is a suitable assumption according to [69]. Therefore equation (2.6) reads

$$\varphi_t = \Delta \mu + m. \tag{2.20}$$

#### Temperature equation.

We start from the internal energy equation (2.11), taking advantage of (2.15) and of the expression for the chemical potential (2.14), therefore

$$\frac{\partial e}{\partial t} = -\operatorname{div} q + \frac{\partial \psi}{\partial \varphi} \frac{\partial \varphi}{\partial t} + \frac{\partial \psi}{\partial \nabla \varphi} \frac{\partial \nabla \varphi}{\partial t} - h \nabla \mu.$$

Now, exploiting the assumption  $h = -\nabla \mu$  and Fourier's law (1.49), we infer

$$\frac{\partial e}{\partial t} = \operatorname{div}\left(\kappa(\theta)\nabla\theta\right) + \frac{\partial\psi}{\partial\varphi}\frac{\partial\varphi}{\partial t} + \frac{\partial\psi}{\partial\nabla\varphi}\frac{\partial\nabla\varphi}{\partial t} + |\nabla\mu|^2$$

and by identity (2.12), taking into account (2.16),

$$\theta \frac{\partial s}{\partial t} - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2.$$
(2.21)

From (1.15), we might write

$$\theta \frac{\partial s}{\partial t} = \theta (-f'(\theta))_t + \theta \varphi_t.$$

On the other hand, according to the definition of Q, it holds  $(Q(\theta))_t = (-f'(\theta))_t$ , with in

particular  $(Q(\theta))_t = Q'(\theta)\theta_t$ . Since we suppose, as in Chapter 1, the specific heat to be  $c_V = 1$ , it follows that  $Q'(\theta) = 1$ . This implies that

$$\theta s_t = \theta_t + \theta \varphi_t.$$

Thus, equation (2.21) reads

$$\theta_t + \theta \varphi_t - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2.$$
(2.22)

We remark that, even though the temperature equation (2.22) reads exactly like (1.54), the expressions of  $\varphi_t$  and  $\mu$  are different.

#### Nutrient equation.

We postulate the nutrient balance equation in the form

$$\sigma_t = -\operatorname{div} J - \mathcal{S},\tag{2.23}$$

where J is the nutrient flux and  $\mathcal{S}$  denotes a source/sink term for the nutrient. Choosing  $J = -\nabla \sigma$ , equation (2.23) reads

$$\sigma_t = \Delta \sigma - \mathcal{S}. \tag{2.24}$$

#### 2.1.1 Constitutive relations

Owing to [16, 60, 102], we now make the following constitutive assumptions.

•  $m = (\mathcal{P}\sigma - \mathcal{A})h(\varphi),$ 

where  $h(\varphi)$  is a monotone increasing, nonnegative function in [-1, 1] and such that h(-1) = 0 and h(1) = 1. Hence this relation states that on one hand the tumor growth is proportional to the nutrient supply in the tumoral region. This assumption reflects the fact that it often happens that tumors bring mutations which switch off certain growth inhibiting proteins. Therefore the tumor cells increasing is limited only by the supply of nutrients, despite of healthy cells where the mitotic cycle regulates the growth. On the other hand, when we are in the healthy region, this equation shows that the proliferation rate of the tumor is greater than the one of healthy cells.

• 
$$\frac{\partial N}{\partial \varphi} = -\chi_{\varphi}\sigma$$
, in fact, we take  $\chi_{\varphi} = 0$ .

Indeed, this equation is due to the mechanism of chemotaxis, which we exclude in our analysis.

•  $\mathcal{S} = \mathcal{C}\sigma h(\varphi) - \mathcal{B}(\sigma_B - \sigma).$ 

We here assume that the sink/source of nutrient is regulated by consumption of nutrients and the term  $\mathcal{B}(\sigma_B - \sigma)$  which models the fact that we here consider the case in which the tumor has its own vasculature. In particular the threshold  $\sigma_B$  indicates whether the nutrient is supplied to the tumor or transported away.

## 2.2 Assumptions

In order to carry out a mathematical analysis of our problem, let us introduce some notation we will use in the sequel. We recall that  $\Omega$  is a smooth domain of  $\mathbb{R}^3$  and we denote by  $\partial\Omega$  its boundary. For sake of semplicity, let us assume  $|\Omega| = 1$ . We denote by (0,T) an assigned but otherwise arbitrary time interval. We set  $H := L^2(\Omega)$  and  $V := H^1(\Omega)$  and we will use these symbols also referring to vector valued functions. The symbol  $(\cdot, \cdot)$  will indicate the standard scalar product in H, while  $\langle \cdot, \cdot \rangle$  will stand for the duality between V' and V. We denote by  $\|\cdot\|_X$  the norm in the generic Banach space X. For brevity we will write  $\|\cdot\|$  instead of  $\|\cdot\|_H$ . Still for brevity, we omit the variables of integration. We will specify them when there could be a misinterpretation. For any function  $v \in V$ , we define

$$v_{\Omega} := \frac{1}{|\Omega|} \int_{\Omega} v = \int_{\Omega} v, \qquad (2.25)$$

where the last equality holds since we assumed  $|\Omega| = 1$ . We recall the Poincaré-Wirtinger inequality

$$\|v - v_{\Omega}\| \le c_{\Omega} \|\nabla v\| \quad \forall v \in V \tag{2.26}$$

and the non-linear Poincaré inequality

$$\|v^{\frac{p}{2}}\|_{V}^{2} \leq c_{p} \left(\|v\|_{L^{1}(\Omega)}^{p} + \|\nabla v^{\frac{p}{2}}\|^{2}\right), \qquad (2.27)$$

which holds  $\forall v \in L^1(\Omega)$  s.t.  $\nabla v^{\frac{p}{2}} \in L^2(\Omega)$  and  $\forall p \in [2, \infty)$ .

We assume the coefficients  $\mathcal{P}, \mathcal{A}, \mathcal{B}$  and  $\mathcal{C}$  to be strictly positive and  $\sigma_B \in (0, 1)$ . Next, we suppose that the derivative of potential  $F \in C^1_{loc}(\mathbb{R}, \mathbb{R})$  decomposes as a sum of a monotone increasing part  $\beta$  and a linear perturbation, namely

$$F'(r) = \beta(r) - \lambda r \quad \lambda \ge 0, \ r \in \mathbb{R}.$$
(2.28)

Moreover we normalize  $\beta$  s.t.  $\beta(0) = 0$  and we require

$$\exists c_{\beta} > 0 \text{ s.t. } |\beta(r)| \le c_{\beta}(1 + F(r)) \quad \forall r \in \mathbb{R},$$
(2.29)

$$|\beta(r)| \ge k|r| \text{ for some } k > 0, \tag{2.30}$$

where (2.29) means that F has at most an exponential growth at infinity, while (2.30) states that  $\beta$  has superlinear growth. Moreover, we assume potential F to be strictly positive.

Next, we assume  $h \in C^1(\mathbb{R})$  increasingly monotone s.t.

i) 
$$h(-1) = 0, \ h(r) \equiv 1 \quad \forall r \ge 1.$$

ii) 
$$\exists \underline{h} \ge 0$$
 and  $\varphi \le -1$  s.t.  $h(r) \equiv -\underline{h} \quad \forall r \le \varphi$ .

Therefore h is globally Lipschitz continuous and there exists a constant c > 0 s.t.

$$|h(r)| + |h'(r)| \le c \quad \forall r \in \mathbb{R}.$$

$$(2.31)$$

Moreover we assume the thermal conductivity to depend on the absolute temperature  $\theta$  as follows

$$\kappa(\theta) = 1 + \theta^q, \quad q \in [2, \infty), \quad \theta \ge 0.$$
 (2.32)

Eventually, we require the initial data to be such that

$$\begin{aligned} \varphi|_{t=0} &= \varphi_0, \quad \varphi_0 \in V, \quad F(\varphi_0) \in L^1(\Omega) \\ \theta|_{t=0} &= \theta_0, \quad \theta_0 \in L^1(\Omega), \ \theta_0 > 0 \text{ a.e. in } \Omega, \ \log \theta_0 \in L^1(\Omega) \\ \sigma|_{t=0} &= \sigma_0, \quad \sigma_0 \in L^\infty(\Omega), \ 0 \le \sigma_0 \le 1 \text{ a.e. in } \Omega \end{aligned}$$
(2.33)

where the last assumption on  $\sigma_0$  is due to the interpretation of  $\sigma$  as a nutrient concentration. We also recall that we couple our system with homogeneous Neumann boundary conditions for all the unknowns.

## 2.3 Existence of solutions

In this section we present the main result of this Chapter, concerning the existence of solutions for the tumor growth model (2.1)–(2.4) for  $\chi_{\varphi} = 0$  and  $\varepsilon = 1$ . Namely, we work on system

$$\varphi_t = \Delta \mu + (\mathcal{P}\sigma - \mathcal{A})h(\varphi) \tag{2.34}$$

$$\mu = -\Delta \varphi + F'(\varphi) - \theta \tag{2.35}$$

$$\theta_t + \theta \varphi_t - \operatorname{div} \left( \kappa(\theta) \nabla \theta \right) = |\nabla \mu|^2$$
(2.36)

$$\sigma_t = \Delta \sigma - \mathcal{C}\sigma h(\varphi) + \mathcal{B}(\sigma_B - \sigma). \tag{2.37}$$

We here present what will be called a *weak entropy solution*, already used for example in [97], which is in fact weaker than other corresponding notions appearing in related contexts. This is due to the fact that we do not get enough regularity to pass to the limit in some non-linear terms in the temperature equation (2.36). In particular, the definition stated here does not include (a weak formulation of) the conservation of internal energy, differently from e.g. [14, 46].

Multiplying (2.36) by  $\frac{1}{\theta}$ , we have

$$(\Lambda(\theta) + \varphi)_t - \operatorname{div}\left(\frac{\kappa(\theta)\nabla\theta}{\theta}\right) = \frac{\kappa(\theta)}{\theta^2} |\nabla\theta|^2 + \frac{|\nabla\mu|^2}{\theta}, \qquad (2.38)$$

with

$$\Lambda(\theta) := \int_{1}^{\theta} \frac{1}{s} \mathrm{d}s = \log \theta.$$
(2.39)

We remark that in our case  $\Lambda(\theta)$  is a very well-known function, but we stick with this notation in order to be coherent with the literature [46, 97], where  $\Lambda(\theta)$  might be a more generic function. Testing (2.38) by  $\xi \in C^{\infty}([0,T] \times \overline{\Omega}), \ \xi \geq 0, \ \xi(T, \cdot) = 0$  and integrating by parts we infer

$$\int_0^T \int_\Omega (\Lambda(\theta) + \varphi) \xi_t dx dt + \int_0^T \int_\Omega \frac{\kappa(\theta)}{\theta} \nabla \theta \cdot \nabla \xi dx dt$$
$$= -\int_0^T \int_\Omega \frac{|\nabla \mu|^2}{\theta} \xi dx dt - \int_0^T \int_\Omega \frac{\kappa(\theta)}{\theta^2} |\nabla \theta|^2 dx dt - \int_\Omega (\Lambda(\theta_0) + \varphi_0) \xi(\cdot, 0) dx.$$

Setting  $\delta(\theta) := \int_1^\theta \frac{\kappa(s)}{s} ds = \ln \theta + \frac{1}{q}(\theta^q - 1)$  according to (2.32), we get

$$\int_{0}^{T} \int_{\Omega} (\Lambda(\theta) + \varphi) \xi_{t} dx dt + \int_{=}^{T} \int_{\Omega} \delta(\theta) \Delta \xi dx dt$$
$$= -\int_{0}^{T} \int_{\Omega} \frac{|\nabla \mu|^{2}}{\theta} \xi dx dt - \int_{0}^{T} \int_{\Omega} \frac{\kappa(\theta)}{\theta^{2}} |\nabla \theta|^{2} dx dt - \int_{\Omega} (\Lambda(\theta_{0}) + \varphi_{0}) \xi(\cdot, 0) dx. \quad (2.40)$$

**Definition 2.1.** We say that  $(\varphi, \mu, \theta, \sigma)$  is a *weak entropy solution* to our non-isothermal Cahn-Hilliard model if it satisfies the following equations

$$\langle \varphi_t, \xi \rangle = -\int_{\Omega} \nabla \mu \cdot \nabla \xi dx + \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)\xi dx \quad \text{a.e. in } (0,T) \text{ and } \forall \xi \in V,$$
  
 
$$\mu = -\Delta \varphi + F'(\varphi) - \theta \quad \text{a.e. in } (0,T) \times \Omega,$$

$$\langle \sigma_t, \xi \rangle = -\int_{\Omega} \nabla \sigma \cdot \nabla \xi dx - \int_{\Omega} \mathcal{C}\sigma h(\varphi)\xi dx + \int_{\Omega} \mathcal{B}(\sigma_B - \sigma)\xi dx$$
  
a.e. in  $(0, T)$  and  $\forall \xi \in V$ ,

complying a.e. in  $\Omega$  with the initial conditions (2.33), homogeneous Neumann boundary conditions and the entropy production inequality

$$\int_{0}^{T} \int_{\Omega} (\Lambda(\theta) + \varphi) \xi_{t} dx dt + \int_{0}^{T} \int_{\Omega} \delta(\theta) \Delta \xi dx dt$$

$$\leq -\int_{0}^{T} \int_{\Omega} \frac{|\nabla \mu|^{2}}{\theta} \xi dx dt - \int_{0}^{T} \int_{\Omega} \frac{\kappa(\theta)}{\theta^{2}} |\nabla \theta|^{2} dx dt - \int_{\Omega} (\Lambda(\theta_{0}) + \varphi_{0}) \xi(\cdot, 0) dx \qquad (2.41)$$

 $\forall \xi \in C^{\infty}([0,T] \times \overline{\Omega}), \ \xi \geq 0, \ \xi(T, \cdot) = 0.$ 

**Theorem 2.3.1.** Suppose that the assumptions in Section 2.2 hold and let T > 0. Then there exists at least one weak solution to our model problem, namely a quadruple ( $\varphi, \mu, \theta, \sigma$ ) with regularity

$$\begin{split} \varphi &\in C([0,T];V) \cap H^{1}(0,T;V') \cap L^{2}(0,T;H^{2}(\Omega)) \\ \beta(\varphi) &\in L^{2}(0,T;H) \\ \mu &\in L^{2}(0,T;V) \\ \theta &\in L^{2}(0,T;V) \cap L^{\infty}(0,T;L^{1}(\Omega)) \cap L^{q}(0,T;L^{3q}(\Omega)), \ q \geq 2, \quad \theta > 0 \ a.e. \ in \ \Omega \\ \sigma &\in C([0,T];H) \cap H^{1}(0,T;V') \cap L^{\infty}((0,T) \times \Omega) \cap L^{2}(0,T;V) \end{split}$$

satisfying system (2.34)–(2.37) in the sense of Definition 2.1.

### 2.3.1 A priori estimates

This section is devoted to gain the suitable regularity for the quadruple  $(\varphi, \mu, \theta, \sigma)$  to prove the existence of solutions in Section 2.3.2. These a priori bounds are obtained formally, working directly on our system (2.34)-(2.37). We remark that the existence (of weak entropy solutions) argument might be made rigorous by the Faedo-Galerkin method that we decided not to detail here.

#### Nutrient estimate

We first search for a priori bounds for the nutrient following [102]. Therefore we give here only a sketch of the main steps.

Testing (2.37) by  $-\sigma_{-}$ , where  $\sigma_{-} \geq 0$  represents the negative part of the nutrient  $\sigma$ ,

exploting the initial conditions on  $\sigma$  and applying the Gronwall lemma, we gain

$$\sigma(t, x) \ge 0$$
 for a.e.  $t \ge 0, x \in \Omega$ .

Now, testing (2.37) by  $(\sigma - \bar{\sigma})_+$  (where  $\bar{\sigma} \ge 1$  is a suitable constant) using the Gronwall lemma and our assuptions on h and  $\sigma_B$ , it is possible to obtain

$$\|\sigma\|_{L^{\infty}((0,T)\times\Omega)} \le c_T,\tag{2.42}$$

where  $c_T$  is a constant depending on time.

#### **Energy** estimate

We test (2.34) by  $\mu$ , (2.35) by  $\varphi_t$  and (2.36) by 1 and then sum up. This yields, taking into account the boundary conditions,

$$\frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{1}{2} \|\nabla\varphi\|^2 + \int_{\Omega} F(\varphi) + \int_{\Omega} \theta \right) = \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A}) h(\varphi) \mu.$$
(2.43)

We take care of the right hand side, in particular

$$\begin{split} & \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)\mu \\ \stackrel{(2.35)}{=} & -\int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)\Delta\varphi + \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)F'(\varphi) - \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)\theta \\ \stackrel{(2.28)}{=} & \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h'(\varphi)|\nabla\varphi|^2 + \int_{\Omega} \mathcal{P}h(\varphi)\nabla\sigma \cdot \nabla\varphi + \int_{\Omega} \beta(\varphi)(\mathcal{P}\sigma - \mathcal{A})h(\varphi) \\ & + \int_{\Omega} \lambda\varphi(\mathcal{A} - \mathcal{P}\sigma)h(\varphi) + \int_{\Omega} (\mathcal{A} - \mathcal{P}\sigma)h(\varphi)\theta. \end{split}$$

Thus (2.43) reads

$$\frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{1}{2} \|\nabla\varphi\|^2 + \int_{\Omega} F(\varphi) + \int_{\Omega} \theta \right) = \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A}) h'(\varphi) |\nabla\varphi|^2 
+ \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A}) \beta(\varphi) h(\varphi) 
+ \int_{\Omega} \lambda(\mathcal{A} - \mathcal{P}\sigma) h(\varphi) \varphi + \mathcal{P} \int_{\Omega} h(\varphi) \nabla\sigma \cdot \nabla\varphi 
+ \int_{\Omega} (\mathcal{A} - \mathcal{P}\sigma) h(\varphi) \theta 
:= I + II + III + IV.$$
(2.44)

We now estimate each term on the right hand side separately. The estimate on the nutrient (2.42) is a key point for all these bounds. In particular this is where a time-

dependent constant  $c_T$  comes from. Exploting the assumption (2.31), we infer

$$I \le c_T \|\nabla \varphi\|^2. \tag{2.45}$$

According to (2.29) it is straightforward that

$$II \le c_T \left( 1 + \int_{\Omega} F(\varphi) \right).$$
(2.46)

Moreover, using once again the assumption (2.31) on h and Young's inequality, we get

$$III \le \frac{1}{2} \|\nabla \sigma\|^2 + c_T \left( 1 + \|\varphi\|_{L^1(\Omega)} + \|\nabla \varphi\|^2 \right).$$
(2.47)

Eventually, by the same tools used to estimate *III*, it holds

$$IV \le c_T \|\theta\|_{L^1(\Omega)}.\tag{2.48}$$

Combining estimates (2.45)-(2.48), (2.44) reads

$$\frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{1}{2} \| \nabla \varphi \|^2 + \int_{\Omega} F(\varphi) + \int_{\Omega} \theta \right)$$

$$\leq \frac{1}{2} \| \nabla \sigma \|^2 + c_T \left( 1 + \| \varphi \|_{L^1(\Omega)} + \| \nabla \varphi \|^2 + \int_{\Omega} F(\varphi) + \| \theta \|_{L^1(\Omega)} \right)$$
(2.49)

Our aim is to apply Gronwall's lemma in order to gain the energy estimate. Therefore we estimate and reabsorb the term  $\|\varphi\|_{L^1(\Omega)}$  according to (2.30). Moreover we test (2.37) by  $\sigma$  which yields

$$\frac{1}{2}\frac{\mathrm{d}}{\mathrm{d}t}\|\sigma\|^2 + \|\nabla\sigma\|^2 \le c(1+\|\sigma\|^2).$$
(2.50)

Hence, summing this last estimate to (2.49) we finally get

$$\frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{1}{2} \|\nabla\varphi\|^2 + \int_{\Omega} F(\varphi) + \int_{\Omega} \theta + \frac{1}{2} \|\sigma\|^2 \right) + \frac{1}{2} \|\nabla\sigma\|^2$$
$$\leq c_T \left( 1 + \|\nabla\varphi\|^2 + \int_{\Omega} F(\varphi) + \|\theta\|_{L^1(\Omega)} + \|\sigma\|^2 \right).$$
(2.51)

We are now able to apply Gronwall's lemma to (2.49), therefore we obtain the following a priori estimates

$$\|\nabla\varphi\|_{L^{\infty}(0,T;H)} \le c_T \tag{2.52}$$

$$||F(\varphi)||_{L^{\infty}(0,T;L^{1}(\Omega))} \le c_{T}$$
 (2.53)

$$\|\theta\|_{L^{\infty}(0,T;L^{1}(\Omega))} \le c_{T} \tag{2.54}$$

$$\|\sigma\|_{L^{\infty}(0,T;H)\cap L^{2}(0,T;V)} \le c_{T}.$$
(2.55)

In particular, combining (2.29) and (2.30) with (2.53), we gain

$$\|\varphi\|_{L^{\infty}(0,T;L^{1}(\Omega))} \le c_{T} \tag{2.56}$$

#### Entropy estimate

We now derive the entropy estimate testing (2.36) by  $-\frac{1}{\theta}$ . Therefore

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (-\log\theta - \varphi) + \int_{\Omega} \frac{1}{\theta} |\nabla\mu|^2 + \int_{\Omega} \left( |\nabla\log\theta|^2 + k_q |\nabla\theta^{q/2}|^2 \right) = 0, \qquad (2.57)$$

where  $k_q > 0$  is a suitable constant only depending on the exponent  $q \ge 2$ , introduced in (2.32).

Now, integrating in time, owing to (2.54) and (2.56) and recalling that  $|\log r| \leq r - \log r \quad \forall r > 0$ , we infer

$$\|\log \theta\|_{L^{\infty}(0,T;L^{1}(\Omega))} + \|\log \theta\|_{L^{2}(0,T;V)} \le c_{T},$$
(2.58)

$$\|\nabla \theta^{q/2}\|_{L^2(0,T;H)} \le c_T. \tag{2.59}$$

Then, combining (2.27) with (2.54) and (2.59), it holds

$$\|\theta^{\frac{q}{2}}\|_{L^2(0,T;V)} \le c_T \tag{2.60}$$

which implies in particular, since  $q\geq 2$ 

$$\|\theta\|_{L^2(0,T;V)} \le c_T. \tag{2.61}$$

On the other hand, using Sobolev embedding theorems, (2.60) also implies

$$\|\theta^{\frac{q}{2}}\|_{L^{2}(0,T;L^{6}(\Omega))} \leq c_{T}$$

and hence

$$\|\theta\|_{L^q(0,T;L^{3q}(\Omega))} \le c_T.$$
(2.62)

#### Chemical potential estimate

Integrating (2.36) over  $\Omega$  and exploiting boundary conditions together with Gauss-Green formula, we infer

$$\|\nabla\mu\|^2 = \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \theta + \int_{\Omega} \theta\varphi_t.$$
(2.63)

We now rewrite the latter term according to (2.34), then using (2.31) and (2.42), it follows that (2.63) reads

$$\frac{1}{2} \|\nabla \mu\|^2 \le \frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \theta + \frac{1}{2} \|\nabla \theta\|^2 + c_T \|\theta\|_{L^1(\Omega)}.$$
(2.64)

Thus from (2.54) and (2.61), we obtain

$$\|\nabla \mu\|_{L^2(0,T;H)} \le c_T. \tag{2.65}$$

Now we integrate (2.35) over  $\Omega$ , then

$$|\mu_{\Omega}| \stackrel{(2.25),(2.28)}{=} \left| \int_{\Omega} (\beta(\varphi) - \lambda\varphi) - \int_{\Omega} \theta \right|$$
(2.66)

$$\leq \int_{\Omega} |\beta(\varphi)| + \int_{\Omega} |\lambda\varphi| + \|\theta\|_{L^{1}(\Omega)}$$
(2.67)

$$\stackrel{(2.29),(2.54),(2.56)}{\leq} c_{\beta} \left( 1 + \int_{\Omega} F(\varphi) \right) + c_{T}.$$
(2.68)

Using now the bound (2.53), we get

$$\|\mu_{\Omega}\|_{L^{\infty}(0,T)} \le c_T.$$
(2.69)

Combining this last bound with the Poincaré inequality (2.26) and the previous estimate (2.65), we achieve

$$\|\mu\|_{L^2(0,T;V)} \le c_T. \tag{2.70}$$

#### $\varphi$ -dependent estimates

We start testing (2.34) by  $\varphi$ , which leads to

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} |\varphi|^2 = -\int_{\Omega} \nabla \mu \cdot \nabla \varphi + \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)\varphi.$$
(2.71)

Exploiting Young's inequality, the uniform bounds on h and (2.42) we infer

$$\frac{\mathrm{d}}{\mathrm{d}t}\|\varphi\|^2 \leq \frac{1}{2}\|\nabla\mu\|^2 + \frac{1}{2}\|\nabla\varphi\|^2 + c_T\|\varphi\|_{L^1(\Omega)}$$

Thus, integrating in time and using (2.65), (2.52) and (2.56) we get

$$\|\varphi\|_{L^{\infty}(0,T;H)} \le c_T,$$

whence estimate (2.52) gives

$$\|\varphi\|_{L^{\infty}(0,T;V)} \le c_T.$$
 (2.72)

Next we test (2.35) by  $\beta(\varphi)$  and we obtain

$$\int_{\Omega} |\beta(\varphi)|^2 + \int_{\Omega} \beta'(\varphi) |\nabla \varphi|^2 = \int_{\Omega} \mu \beta(\varphi) + \int_{\Omega} \lambda \varphi \beta(\varphi) + \int_{\Omega} \theta \beta(\varphi)$$

Now, from (2.70), (2.72), (2.61) and the monotonicity of  $\beta$ , it follows

$$\|\beta(\varphi)\|_{L^2(0,T;H)} \le c_T.$$
(2.73)

Taking advantage of this last estimate with (2.29) and again of (2.72) and (2.61), a direct comparison within equation (2.35) yields

$$\|\varphi\|_{L^2(0,T;H^2)} \le c_T. \tag{2.74}$$

#### Further regularity

We start testing (2.34) by a nonzero test function  $v \in V$  and we infer

$$\langle \varphi_t, v \rangle = -\int_{\Omega} \nabla \mu \cdot \nabla v + \int_{\Omega} (\mathcal{P}\sigma - \mathcal{A})h(\varphi)v.$$

Now, according to estimates (2.42), (2.65) and (2.72) it follows

$$\|\varphi_t\|_{L^2(0,T;V')} \le c_T. \tag{2.75}$$

Taking advantage of this last estimate and exploting (2.74) together with (2.72), we infer (for example from [90])

$$\varphi \in C([0,T];V). \tag{2.76}$$

Similarly, multiplying equation (2.37) by a nonzero test function  $v \in V$  and exploiting

the bound (2.55), it holds

$$\|\sigma_t\|_{L^2(0,T;V')} \le c_T. \tag{2.77}$$

From standard embedding results (see e.g. [12]), combining (2.77) and (2.55), we gain the additional regularity for the nutrient

$$\sigma \in C([0,T];H). \tag{2.78}$$

#### 2.3.2 Weak sequential stability

We assume to have a sequence of weak solutions  $(\varphi_n, \mu_n, \theta_n, \sigma_n)$  which satisfies the a priori estimates obtained in Section 2.3.1 uniformly with respect to  $n \in \mathbb{N}$ .

We then show that, by weak compactness arguments, up to the extraction of a subsequence,  $(\varphi_n, \mu_n, \theta_n, \sigma_n)$  converges in a suitable way to an entropy solution to our problem, i.e., to a limit quadruple  $(\varphi, \mu, \theta, \sigma)$  solving (2.34)–(2.37) in the sense of Theorem 2.3.1. Indeed, exploiting the above bounds (2.42), (2.54), (2.55), (2.61), (2.62), (2.70), (2.72), (2.74), (2.75) and (2.77), together with standard weak compactness results, it is possible to extract a nonrelabelled subsequence such that

$$\varphi_n \to \varphi$$
 weakly star in  $L^{\infty}(0,T;V) \cap L^2(0,T;H^2(\Omega)) \cap H^1(0,T;V')$  (2.79)

$$\mu_n \to \mu$$
 weakly in  $L^2(0,T;V)$  (2.80)

$$\theta_n \to \theta$$
 weakly star in  $L^2(0,T;V) \cap L^\infty(0,T;L^1(\Omega)) \cap L^q(0,T;L^{3q}(\Omega))$  (2.81)

$$\sigma_n \to \sigma$$
 weakly star in  $L^{\infty}(0,T) \times \Omega) \cap L^2(0,T;V) \cap H^1(0,T;V')$  (2.82)

Moreover, combining (2.75) and (2.77) with (2.79) and (2.82) respectively and applying the Aubin-Lions lemma, we infer that

$$\varphi_n \to \varphi \text{ and } \sigma_n \to \sigma \text{ strongly in } L^2(0,T;H).$$
 (2.83)

Furthermore, convergence (2.81) and interpolation theory for  $L^p$  spaces imply that

$$\theta_n \to \theta \text{ strongly in } L^p(0,T;L^p(\Omega)), \quad p \in \left[1,q+\frac{2}{3}\right).$$
(2.84)

Indeed, from Proposition A.2.1, with  $s = \infty$  and p = q, it holds

$$\frac{1}{r} = \frac{\gamma}{q}.\tag{2.85}$$

We then apply the general interpolation result to the space-spaces  $L^1$  and  $L^{3q}$ , from which

it follows

$$\frac{1}{r} = \frac{\gamma}{3q} + \frac{1-\gamma}{1}.$$
(2.86)

Since (2.85) and (2.86) must hold simultaneously, we infer that  $r = q + \frac{2}{3}$ . Now, according to Theorem A.2.6 with s = r = 0, it follows that

$$L^{\infty}(0,T;L^{1}(\Omega)) \cap L^{q}(0,T;L^{3q}(\Omega)) \hookrightarrow L^{p}(0,T;L^{p}(\Omega)) \quad p \in \left[1,q+\frac{2}{3}\right).$$

Therefore it is possible to pass to the limit also in the nonlinear terms, according to the continuity of  $\kappa, \beta$  and h. Indeed, by a generalized version of Lebesgue's dominated convergence theorem it holds

$$\kappa(\theta_n) \to \kappa(\theta) \text{ strongly in } L^p(0,T;L^p(\Omega)), \quad p \in \left[1,1+\frac{2}{3q}\right)$$
(2.87)

$$\beta(\varphi_n) \to \beta(\varphi)$$
 weakly in  $L^2(0,T;H)$ . (2.88)

We now want to pass to the limit in the balance of entropy. Namely let us assume that (2.38) is satisfied by the approximate solution  $(\varphi_n, \mu_n, \theta_n, \sigma_n), \forall n \in \mathbb{N}$ . Testing it by  $\xi \in C^{\infty}([0,T] \times \overline{\Omega}), \ \xi \geq 0, \ \xi(T, \cdot) = 0$  and integrating by parts we infer

$$\int_{0}^{T} \int_{\Omega} (\Lambda(\theta_{n}) + \varphi_{n}) \xi_{t} \mathrm{d}x \mathrm{d}t + \int_{0}^{T} \int_{\Omega} \delta(\theta_{n}) \Delta \xi \mathrm{d}x \mathrm{d}t$$
$$= -\int_{0}^{T} \int_{\Omega} \frac{|\nabla \mu_{n}|^{2}}{\theta_{n}} \xi \mathrm{d}x \mathrm{d}t - \int_{0}^{T} \int_{\Omega} \frac{\kappa(\theta_{n})}{\theta_{n}^{2}} |\nabla \theta_{n}|^{2} \mathrm{d}x \mathrm{d}t - \int_{\Omega} (\Lambda(\theta_{0}) + \varphi_{0}) \xi(\cdot, 0) \mathrm{d}x, \quad (2.89)$$

where  $\delta(\theta_n) := \int_1^{\theta_n} \frac{\kappa(s)}{s} \mathrm{d}s.$ 

We first take care of the terms on the left hand side. According to (2.39), by (2.58) and (2.84),

 $\Lambda(\theta_n) \to \Lambda(\theta_n)$  strongly in  $L^{1+}(0,T) \times \Omega$ ). (2.90)

Moreover, from (2.58) and (2.84), it follows that

$$\delta(\theta_n) \to \delta(\theta)$$
 strongly in  $L^{1+}((0,T) \times \Omega)$ , (2.91)

hence in particular

$$\int_0^T \int_\Omega \delta(\theta_n) \Delta \xi \to \int_0^T \int_\Omega \delta(\theta) \Delta \xi.$$

Then the first row of (2.89) passes to the desired limit not only as a supremum limit, but as a true limit. In order to deal with the first two terms in the right hand side we recall a useful lower semicontinuity result by Ioffe. **Theorem 2.3.2** (Ioffe). Let  $\mathcal{O} \subset \mathbb{R}^d$  be a smooth bounded open set and  $f : \mathcal{O} \times \mathbb{R}^n \times \mathbb{R}^m \to [0, +\infty], d, n, m \ge 1$ , be a measurable non-negative function such that

 $f(x,\cdot,\cdot)$  is lower semicontinuous on  $\mathbb{R}^n \times \mathbb{R}^m$  for every  $x \in \mathcal{O}$ ,

 $f(x, u, \cdot)$  is convex on  $\mathbb{R}^m$  for every  $(x, u) \in \mathcal{O} \times \mathbb{R}^n$ .

Let also  $(u_k, v_k), (u, v) : \mathcal{O} \to \mathbb{R}^n \times \mathbb{R}^m$  be measurable functions s. t.  $u_k(x) \to u(x)$  in measure in  $\mathcal{O}, \quad v_k \to v$  weakly in  $L^1(\mathcal{O}; \mathbb{R}^m)$ . Then,

$$\liminf_{k \to +\infty} \int_{\mathcal{O}} f(x, u_k(x), v_k(x)) \ge \int_{\mathcal{O}} f(x, u(x), v(x)).$$

We start considering the first term in the right hand side. We exploit this result setting  $\mathcal{O} = \Omega \times (0,T), f : \mathcal{O} \times \mathbb{R}^+ \times \mathbb{R}^3 \to [0,\infty]$  s.t  $(x,t) \times w \times v \mapsto w|v|^2$ . Such fsatisfies loffe's assumptions. putting  $w_n = \frac{\xi}{\theta}, v_n = \nabla \mu_n, \forall n \in \mathbb{N}$ . Hence, by (2.80) it holds  $\{\nabla \mu_n\}_n \to \nabla \mu$  in  $L^1(\mathcal{O})$ . Therefore by loffe's theorem,

$$\liminf_{n \to +\infty} \int_0^T \int_\Omega |\nabla \mu_n|^2 \xi \ge \int_0^T \int_\Omega |\nabla \mu|^2 \xi.$$
(2.92)

In a similar way, from (2.84) and (2.87),

$$\liminf_{n \to +\infty} \int_0^T \int_\Omega \frac{\xi}{\theta_n} \frac{\kappa(\theta_n)}{\theta_n} |\nabla \theta_n|^2 \ge \int_0^T \int_\Omega \frac{\xi}{\theta} \frac{\kappa(\theta)}{\theta} |\nabla \theta|^2.$$
(2.93)

Furthermore, assuming that  $\theta_n(0, \cdot)$  converges in a suitable way to  $\theta_0$ , putting together (2.83), (2.90), (2.91), (2.92) and (2.93), it follows that we eventually recover (2.40). It is worth noting that the inequality sign is due to the application of Ioffe's theorem. This concludes the procedure and so the proof of existence of weak entropy solutions.

Remark 1. We notice that we have assumed throughout the proof that the absolute temperature is a.e. positive. This is crucial in order for estimates in Section 2.3.1 to make sense. In particular it should be shown that the solution  $\theta_n$  of the discretized problem (for instance in a Faedo-Galerkin scheme) is positive. Anyway, taking the initial datum  $\theta_0 > 0$ , according to (2.58) the strict positivity of  $\theta_n$  will be preserved a.e. in  $(0, T) \times \Omega$  also in the limit.

# Part II

# A model for assisted periodic breathing

We represent the lungs as a nonhomogeneous deformable viscoelastic porous body and breathing as an isothermal time-periodic process modelled by a PDE system with hysteresis. In particular, Part III contains the results obtained in [76], where we extended the model proposed in [40] taking into account nonconstant and possibly degenerate permeability of the lungs.

As pointed out in [99], the first measurements which showed a hysteretic pressurevolume characteristic in mammalian lungs were obtained in [23] in 1913. There exist different hypotheses about the nature of the forces which originate the hysteresis behavior by opposing the lung distension, but up to now, there is no theory which could explain both small and large volume excursions, as reported in [117].

The process of breathing has been studied from a mathematical point of view first for cats [73] and dogs [91] and later also for humans [86]. Understanding the relationship between pressure and volume in lungs is important for controlling the ventilation of patients with lung diseases [4, 33, 71]. In order to set up the ventilator correctly and so to prevent the possibility of damaging the patient's lungs, it is crucial to test how the healthy lungs respond to mechanical ventilation. In [103] such a ventilation study simulating hysteretic pressure-volume relationships in the lungs of a healthy individual is shown. Moreover, the knowledge of the pressure-volume hysteresis loops were used to develop a new technique for treating patients with Acute Lung Injury (ALI). This procedure gave the same levels of oxygen but with lower airway pressures and less over-inflation than other techniques [3]. Furthermore, [3] also shows that this kind of strategy reduces trauma and increases the probability of survival with respect to methods which do not take into account pressure-volume curves. We also remark that such studies are useful not only for treating damaged lungs; indeed, e.g., cerebrospinal pressure-volume curve might be of interest for hydrocephalus symptoms [79].

Our aim is to further develop the model proposed in [40], motivated by the analysis carried out in [52], where the Preisach operator is shown to be an appropriate model for the pressure-volume hysteresis relationship in lungs. The process of breathing is assumed to be isothermal and driven by a periodic muscle activity and, possibly, by a periodic external ventilation. Our model is represented by a PDE system consisting of a momentum balance equation and a mass balance equation coupled with boundary conditions prescribing the mechanical reaction between lungs and their surroundings. The mathematical problem consists in proving that the PDE system with a degenerating pressure-mass content term under the time derivative admits a periodic solution for every periodic boundary forcing with a given regularity. The novelty of [76] with respect to [40] is that we expand the previous model considering the permeability of the lungs depending also on the pressure

and possibly degenerate. This is, on the one hand, a realistic hypothesis from the physical and biological point of view. On the other hand, this condition produces considerable difficulties in the mathematical problem. Indeed, the additional degenerate nonlinearity requires to choose a multi-parameter approximation technique involving several steps with a series of regularizing terms in the system which have to vanish in the limit. The main ingredient of the proof is a time-periodic parameter-dependent variant of the Moser iterations.

The structure of this Part is the following. In Section 3.1 we present our model and state the main result of this work, which is Theorem 3.1.3 stating the existence and regularity of the resulting PDE system with hysteresis. In Section 3.3 we recall the main properties of the Preisach operator which are needed in the analysis of the problem. The proof of Theorem 3.1.3 is given in several steps in Sections 4.1–4.4. In order to get the nonlinearities under control, we introduce in Section 4.1 a regularized problem with a large cut-off parameter R and with some higher order regularizing terms depending on small parameters  $\delta$  and  $\varepsilon$ . For the regularized system, we construct a periodic solution by Galerkin approximations and a fixed point argument. In Sections 4.2 and 4.3, we derive suitable a priori bounds independent of the parameters  $\delta$  and  $\varepsilon$ . These estimates are used in Section 4.4, where according to a Moser-type technique we obtain  $L^{\infty}$ -estimates of the pressure which allow us to remove the cut-off parameter and eventually let  $\delta$  and  $\varepsilon$  tend to zero and prove Theorem 3.1.3.

# Chapter 3

# Viscoelastic porous medium model

### 3.1 Statement of the problem

As proposed in [40], we represent the lungs as a deformable viscoelastic porous medium, and breathing as a periodic isothermal process of gas exchange between the inside and the outside of the body. The mathematical model is based on a momentum balance equation

$$\rho u_{tt} = \operatorname{div} \sigma, \tag{3.1}$$

where  $\rho$  is the solid mass density, u is the displacement vector in the solid and  $\sigma$  is the stress tensor, and a gas mass balance equation

$$s_t + \operatorname{div} q = 0, \tag{3.2}$$

where s is the gas mass content in the pores and q is the mass flux. We also consider three constitutive relations. The first one represents the mechanical interaction, namely

$$\sigma = \mathbf{B}(x)\nabla_{\!\!s} u_t + \mathbf{A}(x)\nabla_{\!\!s} u - p\mathbf{1}, \qquad (3.3)$$

where  $\mathbf{B}(x)$  (viscosity),  $\mathbf{A}(x)$  (elasticity) are symmetric positive definite space-dependent tensors of order 4, the symbol  $\nabla_s$  denotes the symmetric gradient, p is the air pressure, and 1 in the Kronecker tensor.

The second one assumes the pressure-volume relation in the form

$$f(x,p) + G[p] = \frac{1}{\rho_a}s - \operatorname{div} u,$$
 (3.4)

where  $\rho_a > 0$  is the referential air mass density at standard pressure,  $f : \Omega \times \mathbb{R} \to (0, \infty)$  is a function increasing in p, and G is a Preisach operator defined in Section 3.3.

We can interpret (3.4) as follows. Assuming that the volume of the solid phase remains constant during the process, the term div u represents the void volume difference with respect to the reference state. At constant pressure p, if div u increases, then  $s/\rho_a$  increases at the same rate. At constant void volume, the mass content s is an increasing function (with different inflation and deflation curves) of the pressure. At constant gas mass content, the pressure increases if the void volume decreases.

Note that the mass content s cannot be negative, so that, e.g., at constant volume, the pressure-dependent term f(x, p) + G[p] must be bounded from below. In particular, the partial derivative  $f_p(x, p)$  of f with respect to p has to degenerate as  $p \to -\infty$ .

As the third constitutive relation, we assume the validity of Darcy's law for the mass flux q in (3.2)

$$q = -\rho_a \mu(x, p) \nabla p \tag{3.5}$$

where we underline that, as a novelty with respect to [40], the permeability  $\mu(x, p)$  is allowed to depend both on the space variable and on the pressure, and degenerate as  $p \to \pm \infty$ .

The combination of (3.1)–(3.5) leads to the following PDE system

$$\rho u_{tt} = \operatorname{div} \left( \mathbf{B}(x) \nabla_{\!\!s} u_t + \mathbf{A}(x) \nabla_{\!\!s} u \right) - \nabla p \,, \qquad (3.6)$$

$$(f(x,p) + G[p])_t = -\operatorname{div} u_t + \operatorname{div} \mu(x,p)\nabla p,$$
 (3.7)

for x in a bounded connected Lipschitzian domain  $\Omega \subset \mathbb{R}^3$  and for  $t \geq 0$ . We couple system (3.6)–(3.7) with the following boundary conditions on  $\partial \Omega$ 

$$-\sigma \cdot n\Big|_{\partial\Omega} = \beta(x)(\mathbf{C}(x)u + \mathbf{D}(x)u_t - g) + pn, \qquad (3.8)$$

$$\frac{1}{\rho_a} q \cdot n \Big|_{\partial\Omega} = \alpha(x)(p - \bar{p}) - u_t \cdot n, \qquad (3.9)$$

where n is the unit outward normal vector, g = g(x, t) is a given external force acting on the body  $\Omega$ ,  $\bar{p} = \bar{p}(x, t)$  is the given outer air pressure possibly supplied by the ventilator,  $\mathbf{C}(x), \mathbf{D}(x)$  are symmetric positive definite  $3 \times 3$  matrices,  $\beta(x) \ge 0$  is the relative elasticity modulus of the boundary at the point  $x \in \partial\Omega$ , and  $\alpha(x) \ge 0$  is the boundary permeability at the point  $x \in \partial\Omega$ .

The first boundary condition (3.8) describes the mechanical interaction between the inside and the exterior of the body on the boundary, which is controlled by  $\beta$ . The second boundary condition (3.9), on the other hand, describes the gas exchange between the inside and the exterior of the body on the boundary, which is controlled by  $\alpha$  (see [40]). We denote

$$X_3 = W^{1,2}(\Omega; \mathbb{R}^3), \quad X = W^{1,2}(\Omega).$$
 (3.10)

Then, the full PDE system in variational form for all test functions  $\phi \in X_3$  and  $\psi \in X$  reads as follows:

$$\int_{\Omega} \left( \rho u_{tt} \phi + (\mathbf{B}(x) \nabla_{\!\!s} u_t + \mathbf{A}(x) \nabla_{\!\!s} u) : \nabla_{\!\!s} \phi + \nabla p \ \phi \right) \mathrm{d}x \\ + \int_{\partial\Omega} \beta(x) (\mathbf{C}(x) u + \mathbf{D}(x) u_t - g) \phi \mathrm{d}s(x) = 0 \,, \quad (3.11)$$

$$\int_{\Omega} \left( (f(x,p) + G[p])_t \psi + (\mu(x,p)\nabla p - u_t)\nabla \psi \right) \mathrm{d}x + \int_{\partial\Omega} \alpha(x)(p - \bar{p})\psi \mathrm{d}s(x) = 0. \quad (3.12)$$

We fix a period T > 0 and denote by  $L_T^q$  the  $L^q$ -space of T-periodic functions  $v : \mathbb{R} \to \mathbb{R}$ for  $q \ge 1$ , by  $W_T^{1,q}$  the associated Sobolev space, and by  $C_T$  the space of continuous real T-periodic functions on  $\mathbb{R}$ . Similarly, we deal with the spaces  $L_T^q(Y)$  of T-periodic  $L^q$ -functions  $v : \mathbb{R} \to Y$  with values in a Banach space Y, as well as with the spaces  $L^q(\Omega; C_T)$  and  $L^q(\Omega; W_T^{1,q})$ .

#### Hypothesis 3.1.1. We assume that

(i)  $f : \Omega \times \mathbb{R} \to \mathbb{R}$  is such that  $f(\cdot, p)$  is bounded and measurable for all  $p \in \mathbb{R}$ ,  $f(x, \cdot)$  is continuously differentiable in  $\mathbb{R}$  for a. e.  $x \in \Omega$ , and there exist constants  $0 < f^{\flat} < f^{\sharp}$  and  $\frac{1}{2} < \eta < 1$  with the property

$$f^{\flat}(1+p^2)^{-\eta} \leq \partial_p f(x,p) \leq f^{\sharp} \quad a. e. \quad \forall p \in \mathbb{R}.$$

(ii) The permeability coefficient  $\mu(x,p)$  is positive and continuous. Moreover, putting  $\mu_0(x) = \mu(x,0), \ \bar{\mu}(x,p) = \frac{\mu(x,p)}{\mu_0(x)}, \ and \ M(x,p) = \int_0^p \bar{\mu}(x,p') dp'$  we assume that there exist constants  $\mu^{\flat}, \mu^{\sharp}, \gamma$  such that

$$0 \le \gamma < \frac{1}{6} e^{-1}, \quad \gamma + \eta \le 1,$$
 (3.13)

and for a.e.  $(x,p) \in \Omega \times \mathbb{R}$  we have

$$\mu^{\sharp} \ge \mu(x, p) \ge \mu^{\flat} (1 + p^2)^{-\gamma}, \tag{3.14}$$

$$|\nabla_x \bar{\mu}(x,p)| \le \mu^{\sharp} (1+p^2)^{-\eta/2}, \qquad (3.15)$$

$$|\nabla_x M(x,p)| \le \mu^\sharp,\tag{3.16}$$

where the symbol  $\nabla_x$  denotes the partial gradient with respect to x.

(iii) The nonnegative functions  $\alpha$  and  $\beta$  belong to  $L^{\infty}(\partial\Omega)$  and we have  $\int_{\partial\Omega} \beta(x) ds(x) > 0$ 

0,  $\int_{\partial\Omega} \alpha(x) ds(x) > 0$ .

(iv) The functions  $g, g_t$  belong to  $L^2_T(L^2(\partial\Omega; \mathbb{R}^3)), \bar{p}, \bar{p}_t$  belong to  $L^2_T(L^2(\partial\Omega))$ , and there exists a constant  $p^{\sharp}$  such that

$$|\bar{p}(x,t)| \le p^{\sharp}$$
 for a.e.  $(x,t) \in \partial\Omega \times (0,T)$ .

(v) The symmetric positive definite tensors A, B ∈ L<sup>∞</sup>(Ω; R<sup>3×3</sup><sub>s</sub> × R<sup>3×3</sup>), where R<sup>3×3</sup> is the space of real symmetric tensors of order 3 × 3, and symmetric definite matrices C, D ∈ L<sup>∞</sup>(∂Ω; R<sup>3</sup> × R<sup>3</sup>) are given and there exists a constant c̄, independent of x, such that

$$\begin{cases} \mathbf{A}(x)\xi:\xi \geq \bar{c}(\xi:\xi), & \mathbf{B}(x)\xi:\xi \geq \bar{c}(\xi:\xi) & a. e. \ \forall \xi \in \mathbb{R}^{3\times 3}_{s} \\ \mathbf{C}(x)v\cdot v \geq \bar{c}|v|^{2}, & \mathbf{D}(x)v\cdot v \geq \bar{c}|v|^{2} & a. e. \ \forall v \in \mathbb{R}^{3}. \end{cases}$$

Note that in the case  $\mu(x,p) = \mu_1(x)\mu_2(p)$  we have  $\bar{\mu}(x,p) = \mu_2(p)/\mu_2(0)$ , hence  $\nabla_x \bar{\mu}(x,p) = 0$  and the conditions (3.15)–(3.16) are trivially satisfied.

The Preisach operator G is characterized by its *density function*  $\psi$ , see Definition 3.3.1 below. We suppose that it has the following properties.

**Hypothesis 3.1.2.** A function  $\psi \in L^{\infty}(\Omega \times (0, \infty) \times \mathbb{R})$  is given, and there exist constants  $\Psi^{\sharp} > 0, B > 0$  such that

$$0 \le \psi(x, r, v) \le \Psi^{\sharp}$$
 a.e.,  $\psi(x, r, v) = 0$  for  $r + |v| \ge B$ . (3.17)

We are then ready to state the main result of this Part, concerning the existence of a time-periodic solution to our model.

**Theorem 3.1.3.** Let Hypotheses 3.1.1, 3.1.2 hold. Then there exists a solution (u, p) to (3.11)–(3.12) such that  $u, u_t, \nabla p \in L^2_T(L^2(\Omega; \mathbb{R}^3)) \cap L^\infty_T(L^2(\Omega; \mathbb{R}^3)), u_{tt} \in L^2_T(L^2(\Omega; \mathbb{R}^3)), p_t \in L^2_T(L^2(\Omega)), p \in L^\infty(\Omega \times (T, 2T)), \nabla_s u, \nabla_s u_t \in L^2_T(L^2(\Omega; \mathbb{R}^{3\times 3})) \cap L^\infty_T(L^2(\Omega; \mathbb{R}^{3\times 3})).$ 

We remark that the reason why [T, 2T] is chosen to be the referential interval for T-periodic functions is related to the delayed periodicity of the operator G stated in Proposition 3.3.2.

The main issue to face in order to prove Theorem 3.1.3 is the degenerate character of the functions f and  $\mu$  and of the operator G, which makes the analysis difficult. The proof will therefore be carried out in several steps. In Section 4.1 we regularize the critical terms by means of small parameters  $\varepsilon$  and  $\delta$  and a large cut-off parameter R, propose a Galerkin discretization scheme in dimension  $m \in \mathbb{N}$ , and pass to the limit as  $m \to \infty$  to obtain a solution to the regularized problem. In Sections 4.2–4.4 we show the existence of estimates independent of  $\varepsilon$ ,  $\delta$ , and R which enable us to pass to the limit as  $\varepsilon \to 0$ ,  $\delta \to 0$ , and  $R \to \infty$ . In particular, the estimates independent of R follow from a time-periodic variant of the Moser iteration scheme.

# 3.2 Comparison with previous model

As already mentioned, the system presented in Section 3.1 is a further development from the modellistic point of view of the one in [40], namely

$$\rho u_{tt} = \operatorname{div} \left( \mathbf{B}(x)(x) \nabla_{\!\!s} u_t + \mathbf{A}(x)(x) \nabla_{\!\!s} u \right) - \nabla p \,, \qquad (3.18)$$

$$(f(p) + G[p])_t = -\operatorname{div} u_t + \operatorname{div} \mu(x)\nabla p,$$
 (3.19)

where we considered a permeability  $\mu$  only dependent on the position.

Although in [40] we were able to prove the existence of periodic solutions (u, p) with the same regularity stated in Theorem 3.1.3, the assumptions needed to be refined and so were several steps of the proof.

Concerning the hypotheses, the main focus is of course on  $\mu$ . Indeed, even though in [40] we only asked for the permeability coefficient to belong to  $L^{\infty}(\Omega)$  and to be positively bounded from below, the dependence on p in [76] introduces a nonlinearity. Therefore, we need stronger assumptions, namely Hypotheses 3.1.1 (ii). Moreover, as a consequence of this new nonlinearity, we cannot follow the Galerkin approximations proposed in [40]. As a matter of fact, we here want to test our PDE system by a nonlinear expression v of pressure p, which is its Kirchhoff transform introduced in Hypothesis 3.1.1. This leads to considering the approximations of u and v, since the Galerkin method allows testing only by linear functions and their derivatives. Furthermore we need to regularize the critical terms by means of small parameters  $\varepsilon$  and  $\delta$ . This adds an ulterior complexity to our proof, since the estimates must be independent of these parameters. According to these technicalities, also the Moser-type iteration scheme had to be revised.

## 3.3 Preisach operator

We recall here the basic theory of hysteresis operators, in particular of the Preisach operator G, which is needed in the sequel.

The detailed proofs of the statements of this Section can be found in [84].
We fix some  $\hat{T} > 2T$  and construct the Preisach operator G in terms of the play operator, which is defined as a solution to the following variational inequality

$$\begin{cases} |p(t) - \xi_r(t)| \le r & \forall t \in [0, \hat{T}], \\ (\xi_r(t))_t (p(t) - \xi_r(t) - z) \ge 0 & \text{a. e. } \forall z \in [-r, r], \\ p(0) - \xi_r(0) = \max\{-r, \min\{p(0), r\}\}. \end{cases}$$

$$(3.20)$$

It is well known (see [83, 116]) that for any given input function  $p \in W^{1,1}(0,\hat{T})$  and each parameter r > 0, there exists a unique solution  $\xi_r \in W^{1,1}(0,\hat{T})$  of the variational inequality (3.20). The mapping  $\mathfrak{f}_r : W^{1,1}(0,\hat{T}) \to W^{1,1}(0,\hat{T})$  which with each  $p \in W^{1,1}(0,\hat{T})$ associates the solution  $\xi_r = \mathfrak{f}_r[p] \in W^{1,1}(0,\hat{T})$  of (3.20) is called the *play operator* and the parameter r > 0 can be interpreted as a *memory parameter*.

**Proposition 3.3.1.** For each r > 0, the mapping  $\mathfrak{f}_r : W^{1,1}(0,\hat{T}) \to W^{1,1}(0,\hat{T})$  is Lipschitz continuous and admits a Lipschitz continuous extension to  $\mathfrak{f}_r : C[0,\hat{T}] \to C[0,\hat{T}]$ , in the sense that for every  $p_1, p_2 \in C[0,\hat{T}]$  and for every  $t \in [0,\hat{T}]$  we have

$$|\mathfrak{f}_r[p_1](t) - \mathfrak{f}_r[p_2](t)| \le ||p_1 - p_2||_{[0,t]} := \max_{\tau \in [0,t]} |p_1(\tau) - p_2(\tau)|.$$
(3.21)

Besides, for every  $p \in W^{1,1}(0,\hat{T})$ , the energy balance equation

$$\mathfrak{f}_r[p]_t p - \frac{1}{2} (\mathfrak{f}_r^2[p])_t = |r\mathfrak{f}_r[p]_t|$$
(3.22)

and the identity

$$\mathfrak{f}_r[p]_t p_t = (\mathfrak{f}_r[p]_t)^2 \ge 0 \tag{3.23}$$

hold almost everywhere in  $(0, \hat{T})$ .

**Proposition 3.3.2.** Let  $p \in W_{loc}^{1,1}(0,\infty)$  be periodic with period T > 0. Then  $\mathfrak{f}_r[p](t+T) = \mathfrak{f}_r[p](t)$  for all  $t \ge T$ , namely  $\mathfrak{f}_r[p]$  is periodic for  $t \ge T$  for all r > 0.

In what follows, we consider input functions p which depend on  $x \in \Omega$  and t > 0. For  $r > 0, q \ge 1$  and  $p \in L^q(\Omega; C_T)$ , we interpret the play  $\mathfrak{f}_r[p](x, t)$  as

$$\mathfrak{f}_r[p](x,t) = \mathfrak{f}_r[p(x,\cdot)](t).$$

**Definition 3.3.1.** Let  $\psi$  be given and satisfy Hypothesis 3.1.2. We define the Preisach operator

$$G[p](x,t) = \int_0^\infty \int_0^{\xi_r(x,t)} \psi(x,r,v) dv dr,$$
 (3.24)

where  $\xi_r(x,t) = \mathfrak{f}_r[p](x,t)$  is the output of the play operator applied to  $p(x,\cdot)$ .

From the properties of the play operator in Proposition 3.3.1, namely (3.21) and (3.23), we can prove the Lipschitz continuity and local monotonicity of the Preisach operator.

**Proposition 3.3.3.** Let  $\psi$  be a function fulfilling Hypothesis 3.1.2. Then, there exists a constant  $\Psi^{\sharp} > 0$  such that for  $q \in [1, \infty]$  and for all  $p_1, p_2 \in L^q(\Omega; C_T)$  and  $t \ge 0$ , it holds

$$|G[p_1](x,t) - G[p_2](x,t)| \le \Psi^{\sharp} B \max_{\tau \in [0,t]} |p_1(x,\tau) - p_2(x,\tau)| \quad a. e. in \quad \Omega.$$
(3.25)

In particular, the mapping G is Lipschitz continuous in  $L^q(\Omega; C_T)$  for every  $q \in [1, \infty]$ .

**Proposition 3.3.4.** Let the Preisach operator G from Definition 3.3.1 satisfy Hypothesis 3.1.2. Then G is locally monotone, i. e.  $\forall p \in L^q(\Omega; W_T^{1,1}), G[p]$  belongs to  $L^q(\Omega; W_T^{1,1}),$  and

$$G[p]_t(x,t)p_t(x,t) \ge 0 \ a. e.$$
 (3.26)

Moreover, as a consequence of the energy identity (3.22), we have  $\mathfrak{f}_r[p]_t(p-\mathfrak{f}_r[p]) \geq 0$ a.e. Hence the inequality  $\mathfrak{f}_r[p]_t(h(p) - h(\mathfrak{f}_r[p])) \geq 0$  holds almost everywhere for every nondecreasing function  $h : \mathbb{R} \to \mathbb{R}$ . We then easily conclude that the following Preisach energy inequality holds.

**Proposition 3.3.5.** Let  $h : \mathbb{R} \to \mathbb{R}$  be nondecreasing, h(0) = 0. Then for every  $p \in L^q(\Omega; W_T^{1,1})$  and a. e.  $(x,t) \in \Omega \times (0,\infty)$  it holds

$$\left(\frac{d}{dt}G[p]\right)h(p) - \frac{d}{dt}V_h[p] \ge 0 \tag{3.27}$$

where  $V_h[p](x,t) = \int_0^\infty \int_0^{\xi_r} \psi(x,r,v)h(v)dvdr$  is the h-energy potential. If moreover Hypothesis 3.1.2 holds, then

$$0 \le V_h[p](x,t) \le \Psi^{\sharp} B^2 \max\{h(B), -h(-B)\} \quad a. e.$$
(3.28)

# Chapter 4

# Proof of Theorem 3.1.3

### 4.1 Approximation scheme

We choose a cut-off parameter R > 1 and consider the truncations

$$f_{R}(x,p) = \begin{cases} f(x,-R) + \partial_{p}f(x,-R)(p+R) & \text{for } p \leq -R, \\ f(x,p) & \text{for } p \in (-R,R), \\ f(x,R) + \partial_{p}f(x,R)(p-R) & \text{for } p \geq R, \end{cases}$$

$$Q_{R}(p) = \max\{-R,\min\{p,R\}\}, \qquad (4.1)$$

$$\mu_{R}(x,p) = \mu(x,Q_{R}(p)),$$

$$\bar{\mu}_{R}(x,p) = \bar{\mu}(x,Q_{R}(p)) = \frac{\mu(x,Q_{R}(p))}{\mu_{0}(x)}.$$

Instead of (3.11)–(3.12), we further choose (small) parameters  $\delta > 0$ ,  $\varepsilon > 0$  and for all test functions  $\phi \in X_3$ ,  $\psi \in X$ , and  $\lambda \in L_T^{\infty}$  we consider the approximate system

$$\int_{T}^{2T} \int_{\Omega} \left( (\rho u_{tt} + \delta | u_t | u_t) \phi + (\mathbf{B}(x) \nabla_s u_t + \mathbf{A}(x) \nabla_s u_t + \varepsilon \mathbf{B}(x) \nabla_s u_{tt}) : \nabla_s \phi + \nabla p \ \phi \right) \lambda(t) \mathrm{d}x \mathrm{d}t \\ + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) (\mathbf{C}(x) u + \mathbf{D}(x) u_t - g) \phi \lambda(t) \mathrm{d}s(x) \mathrm{d}t = 0,$$
(4.2)

$$\int_{T}^{2T} \int_{\Omega} \left( \left( (f_R(x, p) + G[p])_t + \delta |p| p) \psi + (\mu_R(x, p) \nabla p - u_t) \nabla \psi \right) \lambda(t) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) (p - \bar{p}) \psi \lambda(t) \mathrm{d}s(x) \mathrm{d}t = 0.$$

$$(4.3)$$

The strategy is the following. We first prove the solvability of (4.2)–(4.3) for fixed  $\delta > 0$ ,  $\varepsilon > 0$ , R > 0 via a Galerkin scheme. Then, we derive estimates for the solution

to (4.2)–(4.3) independent of the regularizing parameters  $\delta, \varepsilon, R$ . Further, we apply a time-periodic Moser-Alikakos iteration scheme (see [1]) which yields an upper bound for  $\sup_{(x,t)\in\Omega\times\mathbb{R}} |p(x,t)|$  and will allow us to remove the *R*-truncation of *p* by choosing *R* sufficiently large. Finally, we prove that passing to the limit as  $\delta \to 0$  and  $\varepsilon \to 0$  we obtain a solution to the original problem (3.11)–(3.12).

The remaining part of this section is devoted to the proof of the following result.

**Proposition 4.1.1.** Let Hypotheses 3.1.1, 3.1.2 hold. Then there exists a solution (u, p)to (4.2)–(4.3) such that  $u, u_t, \nabla p \in L^{\infty}_T(L^2(\Omega; \mathbb{R}^3)), u, u_t, u_{tt} \in L^2_T(W^{1,2}(\Omega; \mathbb{R}^3)), p_t \in L^2_T(L^2(\Omega)).$ 

#### 4.1.1 Galerkin approximations

We choose orthonormal bases  $\{\phi_l\}_{l=0}^{\infty}$  in  $L^2(\Omega)$  and  $\{e_k\}_{k=-\infty}^{\infty}$  in  $L^2_T$  as

$$-\Delta \phi_l = \lambda_l \phi_l \text{ in } \Omega, \ \nabla \phi_l \cdot n = 0 \text{ on } \partial \Omega,$$

$$e_k(t) = \begin{cases} \frac{2}{T} \sin \frac{2\pi k}{T} t & \text{for } k \ge 1, \\ \frac{1}{T} & \text{for } k = 0, \\ \frac{2}{T} \cos \frac{2\pi k}{T} t & \text{for } k \le -1. \end{cases}$$

$$(4.4)$$

Note that for every  $k \in \mathbb{Z}$  we have

$$\dot{e}_k(t) = \frac{2\pi k}{T} e_{-k}(t), \tag{4.5}$$

where the dot denotes here and in the sequel the derivative with respect to t.

For a fixed  $m \in \mathbb{N}$  we consider Galerkin approximations

$$u_{j}^{(m)}(x,t) = \sum_{k=-m}^{m} \sum_{l=0}^{m} u_{jkl}\phi_{l}(x)e_{k}(t),$$
  

$$v^{(m)}(x,t) = \sum_{k=-m}^{m} \sum_{l=0}^{m} v_{kl}\phi_{l}(x)e_{k}(t).$$
(4.6)

We denote by

$$M_R(x,p) = \int_0^p \bar{\mu}_R(x,p') dp'$$
(4.7)

the Kirchhoff transform associated with  $\bar{\mu}_R$ , and by  $M_R^{-1}$  its partial inverse defined by the identity  $M_R^{-1}(x, M_R(x, p)) = p$  for  $(x, p) \in \Omega \times \mathbb{R}$ . We put

$$p^{(m)} = M_R^{-1}(x, v^{(m)})$$
(4.8)

and consider the scalars  $u_{jkl}$ ,  $v_{kl}$  the unknowns of the problem

$$\int_{T}^{2T} \int_{\Omega} \left( \left( \rho u_{tt}^{(m)} + \delta | u_{t}^{(m)} | u_{t}^{(m)} \right) \phi_{l} \eta_{j} + \left( \mathbf{B}(x) \nabla_{s} u_{t}^{(m)} + \mathbf{A}(x) \nabla_{s} u^{(m)} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt}^{(m)} \right) : \nabla_{s} (\phi_{l} \eta_{j})$$
$$+ \nabla p^{(m)} \phi_{l} \eta_{j} \right) e_{-k}(t) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial \Omega} \beta(x) (\mathbf{C}(x) u^{(m)} + \mathbf{D}(x) u_{t}^{(m)} - g) \phi_{l} \eta_{j} e_{-k}(t) \mathrm{d}s(x) \mathrm{d}t = 0.$$
(4.9)

$$\int_{T}^{2T} \int_{\Omega} \left( \left( (f_R(x, p^{(m)}) + G[p^{(m)}])_t + \delta | p^{(m)} | p^{(m)} \right) \phi_l + \left( \mu_0(x) \left( \nabla v^{(m)} - \nabla_x M_R(x, p^{(m)}) \right) - u_t^{(m)} \right) \nabla \phi_l \right) e_k(t) dx dt + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) (p^{(m)} - \bar{p}) \phi_l e_k(t) ds(x) dt = 0,$$
(4.10)

for j = 1, 2, 3, l = 0, ..., m, and k = -m, ..., m, where  $\eta_j$  are the vectors

$$\eta_1 = \begin{pmatrix} 1 \\ 0 \\ 0 \end{pmatrix}, \ \eta_2 = \begin{pmatrix} 0 \\ 1 \\ 0 \end{pmatrix}, \ \eta_3 = \begin{pmatrix} 0 \\ 0 \\ 1 \end{pmatrix}.$$

Notice that we used in (4.10) the identity  $\bar{\mu}_R(x, p^{(m)})\nabla p^{(m)} = \nabla v^{(m)} - \nabla_x M_R(x, p^{(m)})$ according to (4.8).

The existence of a solution to (4.9)-(4.10) will be proved by a topological degree argument (see Appendix B). The left-hand side of (4.9)-(4.10) defines a mapping H:  $\mathbb{R}^{4(m+1)(2m+1)} \to \mathbb{R}^{4(m+1)(2m+1)}$ , and we look for a solution  $z \in \mathbb{R}^{4(m+1)(2m+1)}$  to the equation H(z) = 0. We define a homotopy  $H_{\chi}$  for  $\chi \in [0,1]$  such that  $H_1 = H$  and  $H_0$  is an odd mapping, and find a sufficiently large ball  $\mathcal{B} \subset \mathbb{R}^{4(m+1)(2m+1)}$  such that the equation  $H_{\chi}(z) = 0$  has no solution on  $\partial \mathcal{B}$  for any  $\chi \in [0,1]$ . Since the topological degree of  $H_0$ with respect to  $\mathcal{B}$  is odd, we conclude that also the degree of  $H_1$  is odd, and this will enable us to conclude that the equation  $H_1(z) = 0$  has a solution.

The mapping  $H_{\chi}$  is defined as the left-hand side of the system

$$\int_{T}^{2T} \int_{\Omega} \left( (\rho u_{tt}^{(m)} + \delta | u_{t}^{(m)} | u_{t}^{(m)}) \phi_{l} \eta_{j} + (\mathbf{B}(x) \nabla_{s} u_{t}^{(m)} + \mathbf{A}(x) \nabla_{s} u^{(m)} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt}^{(m)}) : \nabla_{s} (\phi_{l} \eta_{j}) \right) \\
+ \chi \nabla p^{(m)} \phi_{l} \eta_{j} e_{-k}(t) dx dt \\
+ \int_{T}^{2T} \int_{\partial \Omega} \beta(x) (\mathbf{C}(x) u^{(m)} + \mathbf{D}(x) u_{t}^{(m)} - \chi g) \phi_{l} \eta_{j} e_{-k}(t) ds(x) dt = 0, \quad (4.11) \\
\int_{T}^{2T} \int_{\Omega} \left( \left( (\chi (f_{R}(x, p^{(m)}) + G[p^{(m)}]) + (1 - \chi) v^{(m)}) \right)_{t} + \delta |p^{(m)}| p^{(m)} \phi_{l} \right) dt$$

$$+ \left(\mu_0(x)(\nabla v^{(m)} - \chi \nabla_x M_R(x, p^{(m)})) - u_t^{(m)}\right) \nabla \phi_l \right) e_k(t) \mathrm{d}x \mathrm{d}t + \int_T^{2T} \int_{\partial\Omega} \alpha(x) \left(\chi(p^{(m)} - \bar{p}) + (1 - \chi)v^{(m)}\right) \phi_l e_k(t) \mathrm{d}s(x) \mathrm{d}t = 0,$$
(4.12)

for  $j = 1, 2, 3, l = 0, \dots, m$ , and  $k = -m, \dots, m$ .

Testing (4.11) by  $\frac{2k\pi}{T}u_{jkl}$  and (4.10) by  $v_{kl}$ , and using the fact that the integral over the period of the time derivative of a periodic function vanishes, we obtain

$$\int_{T}^{2T} \int_{\Omega} \left( \delta |u_{t}^{(m)}|^{3} + \mathbf{B}(x) \nabla_{s} u_{t}^{(m)} : \nabla_{s} u_{t}^{(m)} + \chi \nabla p^{(m)} u_{t}^{(m)} \right) \mathrm{d}x \mathrm{d}t \\
+ \int_{T}^{2T} \int_{\partial\Omega} \left( \beta(x) (\mathbf{D}(x) u_{t}^{(m)} \cdot u_{t}^{(m)} - \chi g u_{t}^{(m)} \right) \mathrm{d}s(x) \mathrm{d}t = 0,$$
(4.13)

$$\int_{T}^{2T} \int_{\Omega} \left( G[p^{(m)}])_{t} v^{(m)} + \delta |p^{(m)}| p^{(m)} v^{(m)} + \left( \mu_{0}(x) (\nabla v^{(m)} - \chi \nabla_{x} M_{R}(x, p^{(m)})) + u_{t}^{(m)} \right) \cdot \nabla v^{(m)} \right) \mathrm{d}x \mathrm{d}t$$

$$+ \int_{T}^{2T} \int_{\partial\Omega} (\alpha(x) \Big( \chi(p^{(m)} - \bar{p}) + (1 - \chi) v^{(m)} \Big) v^{(m)}) \mathrm{d}s(x) \mathrm{d}t = 0.$$
(4.14)

In order to deal with the term  $G[p^{(m)}]_t v^{(m)}$ , we introduce the modified Preisach potential

$$V_{M,R}[p] := \int_0^\infty \int_0^{\mathfrak{f}_r[p]} M_R(x,v)\psi(x,r,v) \mathrm{d}v \mathrm{d}r \ge 0,$$
(4.15)

which satisfies the inequality

$$G[p]_t M_R(p) - V_{M,R}[p]_t \ge 0 \quad a. e.$$
(4.16)

in agreement with Proposition 3.3.5. We also have

$$\frac{1}{C(R)}(|v^{(m)}|^2 + |p^{(m)}|^2) \le v^{(m)}p^{(m)} \le C(R)(|v^{(m)}|^2 + |p^{(m)}|^2)$$

with a constant C(R) depending only on R. More generally, in all the estimates below, c and C denote some non-specified constants, possibly different in different estimates, independent of  $\delta$ ,  $\varepsilon$ , m and R. Similarly, C(R) and  $C(R, \delta)$  denote constants which may possibly depend only on R, or only on R and  $\delta$ , respectively.

Summing up (4.13)–(4.14) and exploiting Hypothesis 3.1.1 (ii) we thus get

$$\int_{T}^{2T} \int_{\Omega} \left( |\nabla v^{(m)}|^2 + \mathbf{B}(x) \nabla_s u_t^{(m)} : \nabla_s u_t^{(m)} + \delta(|u_t^{(m)}|^3 + |p^{(m)}|^3) \right) \mathrm{d}x \mathrm{d}t \\ + \int_{T}^{2T} \int_{\partial\Omega} (\alpha(x)|v^{(m)}|^2 + \beta(x)\mathbf{D}(x)u_t^{(m)}u_t^{(m)}) \mathrm{d}s(x) \mathrm{d}t$$

$$\leq C(R) \left( 1 + \int_{T}^{2T} \int_{\Omega} (|u_{t}^{(m)}| + |p^{(m)}|) |\nabla v^{(m)}| \right) \mathrm{d}x \mathrm{d}t.$$
(4.17)

Using Hölder's inequality with exponents 3, 2 and 6 and then Young's inequality with exponents 3 and 3/2 we have

$$\begin{split} &\int_{T}^{2T} \int_{\Omega} |u_{t}^{(m)}| |\nabla v^{(m)}| \mathrm{d}x \mathrm{d}t \\ &\leq \delta^{\frac{1}{3}} \bigg( \int_{T}^{2T} \int_{\Omega} |u_{t}^{(m)}|^{3} \mathrm{d}x \mathrm{d}t \bigg)^{1/3} \delta^{-\frac{1}{3}} \bigg( \int_{T}^{2T} \int_{\Omega} |\nabla v^{(m)}|^{2} \mathrm{d}x \mathrm{d}t \bigg)^{1/2} \bigg( \int_{T}^{2T} \int_{\Omega} 1 \mathrm{d}x \mathrm{d}t \bigg)^{1/6} \\ &\leq \frac{1}{3} \delta \int_{T}^{2T} \int_{\Omega} |u_{t}^{(m)}|^{3} \mathrm{d}x \mathrm{d}t + \frac{2}{3} C^{3/2} \delta^{-\frac{1}{2}} \bigg( \int_{T}^{2T} \int_{\Omega} |\nabla v^{(m)}|^{2} \mathrm{d}x \mathrm{d}t \bigg)^{3/4} \end{split}$$

and similarly for the term  $\int_T^{2T} \int_{\Omega} |p^{(m)}| |\nabla v^{(m)}| dx dt$ . From (4.17) we thus deduce the inequality

$$\int_T^{2T} \int_{\Omega} |\nabla v^{(m)}|^2 \mathrm{d}x \mathrm{d}t \le C \left( 1 + \delta^{-\frac{1}{2}} \left( \int_T^{2T} \int_{\Omega} |\nabla v^{(m)}|^2 \mathrm{d}x \mathrm{d}t \right)^{3/4} \right),$$

which is an inequality of the form

$$V \le C(R) \left( 1 + \frac{1}{\sqrt{\delta}} V^{\frac{3}{4}} \right) \tag{4.18}$$

for  $V = \int_T^{2T} \int_{\Omega} |\nabla v^{(m)}|^2 dx dt$ . From Young's inequality (with exponents 4 and  $\frac{4}{3}$ ) it follows that

$$\frac{C(R)}{\sqrt{\delta}}V^{3/4} \le \frac{1}{4}\frac{C(R)^4}{\delta^2} + \frac{3}{4}V.$$

Then from (4.18), we infer

$$\frac{1}{4}V \le C(R) + \frac{1}{4}\frac{C(R)^4}{\delta^2},$$

therefore

$$V \le 4C(R) + \frac{C(R)^4}{\delta^2} =: C(R, \delta).$$

Putting everything together, (4.17) gives the following estimate

$$\int_{T}^{2T} \int_{\Omega} |\nabla v^{(m)}|^{2} + \mathbf{B}(x) \nabla_{s} u_{t}^{(m)} : \nabla_{s} u_{t}^{(m)} + \delta \left( |u_{t}^{(m)}|^{3} + |p^{(m)}|^{3} \right) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \left( \alpha(x) |v^{(m)}|^{2} + \beta(x) |u_{t}^{(m)}|^{2} \right) \mathrm{d}s(x) \mathrm{d}t \le C(R, \delta).$$
(4.19)

We see that all possible solutions of the equation  $H_{\chi}(z) = 0$  remain bounded independently of  $\chi \in [0, 1]$ . Hence, the topological degree of  $H_1$  with respect to a sufficiently large ball is the same as the topological degree of  $H_0$ , which is odd. We conclude that a solution to (4.9)–(4.10) exists, see [119, § 13.6]. In the next subsection, we derive further estimates which will enable us to pass to the limit as  $m \to \infty$ .

### 4.1.2 Limit as $m \to \infty$

We now test (4.9) by  $-\left(\frac{2\pi k}{T}\right)^2 u_{j(-k)l}$  and get

$$\int_{T}^{2T} \int_{\Omega} \left( \rho |u_{tt}^{(m)}|^2 + \varepsilon \mathbf{B}(x) |\nabla_s u_{tt}^{(m)}|^2 \mathrm{d}x \mathrm{d}t \right)$$
$$= \int_{T}^{2T} \int_{\Omega} (\mathbf{A}(x) \nabla_s u_t^{(m)} : \nabla_s u_t^{(m)} - \nabla p^{(m)} u_{tt}^{(m)}) \mathrm{d}x \mathrm{d}t$$
$$+ \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \left( \mathbf{C}(x) u_t^{(m)} \cdot u_t^{(m)} + g_t u_t^{(m)} \right) \mathrm{d}s(x) \mathrm{d}t$$

which leads to

$$\int_{T}^{2T} \int_{\Omega} \left( \rho |u_{tt}^{(m)}|^2 + \varepsilon \mathbf{B}(x) \nabla_s u_{tt}^{(m)} : \nabla_s u_{tt}^{(m)} \mathrm{d}x \mathrm{d}t \le C(R, \delta), \right)$$
(4.20)

where we used Hypothesis 3.1.1 and the estimate (4.19). Testing (4.9) by  $u_{j(-k)l}$  we infer

$$\int_{T}^{2T} \int_{\Omega} \left( \delta |u_t^{(m)}|^2 u^{(m)} + \mathbf{A}(x) \nabla_s u^{(m)} : \nabla_s u^{(m)} + u^{(m)} \nabla p^{(m)} \right) \mathrm{d}x \mathrm{d}t$$
$$- \int_{T}^{2T} \int_{\Omega} \left( \rho |u_t^{(m)}|^2 + \varepsilon \mathbf{B}(x) \nabla_s u_t^{(m)} : \nabla_s u_t^{(m)} \right) \mathrm{d}x \mathrm{d}t$$
$$+ \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \left( \mathbf{C}(x) u^{(m)} \cdot u^{(m)} - g \ u_t^{(m)} \right) \mathrm{d}s \mathrm{d}t = 0.$$

The terms  $\rho |u_t^{(m)}|^2$  and  $\varepsilon B \nabla_s u_t^{(m)} : \nabla_s u_t^{(m)}$  are under control because of (4.19), Hypothesis 3.1.1 and the Poincaré inequality. Moreover, we can handle the term  $\delta |u_t^{(m)}|^2 u^{(m)}$  thanks to the embedding inequality

$$\left(\int_{T}^{2T} \int_{\Omega} |u_t^{(m)}|^4 \mathrm{d}x \mathrm{d}t\right)^{1/4} \leq C \left[ \left(\int_{T}^{2T} \int_{\Omega} |u_{tt}|^2 \mathrm{d}x \mathrm{d}t\right)^{1/2} + \left(\int_{T}^{2T} \int_{\Omega} |\nabla_s u_t^{(m)}|^2 \mathrm{d}x \mathrm{d}t\right)^{1/2} \right]$$
(4.21)

Thus it holds

$$\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u^{(m)} : \nabla_{s} u^{(m)} \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \mathbf{C}(x) u^{(m)} \cdot u^{(m)} \mathrm{d}s(x) \mathrm{d}t \le C(R, \delta).$$
(4.22)

Finally, we test (4.10) by  $-\frac{2\pi k}{T}v_{-kl}$  and obtain

$$\int_{T}^{2T} \int_{\Omega} f_{R}(p^{(m)})_{t} v_{t}^{(m)} \mathrm{d}x \mathrm{d}t$$
  

$$\leq C(R, \delta) \left( \int_{T}^{2T} \int_{\Omega} (|p_{t}^{(m)}| + |u_{tt}^{(m)}|) |\nabla v^{(m)}| \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) |\bar{p}_{t}| |v^{(m)}| \mathrm{d}s(x) \mathrm{d}t \right).$$

From the previous estimates we get

$$\int_{T}^{2T} \int_{\Omega} |v_t^{(m)}|^2 \mathrm{d}x \mathrm{d}t \le C(R,\delta).$$
(4.23)

The estimates (4.19), (4.20), (4.22), and (4.23) are sufficient for using the compactness argument and pass to the limit in (4.9)–(4.10) as  $m \to \infty$ , and get a solution to the regularized system (4.2)–(4.3), completing thus the proof of Proposition 4.1.1.

### 4.2 Estimates independent of $\delta$ and $\varepsilon$

System (4.2)–(4.3) can be equivalently written as

$$\int_{\Omega} \left( (\rho u_{tt} + \delta | u_t | u_t) \phi + (\mathbf{B}(x) \nabla_s u_t + \mathbf{A}(x) \nabla_s u + \varepsilon \mathbf{B}(x) \nabla_s u_{tt}) : \nabla_s \phi + \nabla p \ \phi \right) \mathrm{d}x \\
+ \int_{\partial \Omega} \beta(x) (\mathbf{C}(x) u + \mathbf{D}(x) u_t - g) \phi \mathrm{d}s(x) = 0, \qquad (4.24) \\
\int_{\Omega} \left( \left( (f_R(x, p) + G[p])_t + \delta |p| p \right) \psi + (\mu_R(x, p) \nabla p - u_t) \nabla \psi \right) \mathrm{d}x \\
+ \int_{\partial \Omega} \alpha(x) (p - \bar{p}) \psi \mathrm{d}s(x) = 0 \qquad (4.25)$$

for almost all  $t \in (T, 2T)$ , and therefore admits a solution with the regularity stated in Proposition 4.1.1. In order to derive estimates independent of  $\delta$  and  $\varepsilon$ , We test (4.24) by  $\phi = u_t$  and (4.25) by  $\psi = p$  and obtain, using (3.27) with h(p) = p, that

$$\int_{T}^{2T} \int_{\Omega} \left( \delta |u_t|^3 + \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t + \nabla p \cdot u_t \right) \mathrm{d}x \mathrm{d}t$$

$$+ \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \left( \mathbf{D}(x)u_{t} \cdot u_{t} - gu_{t} \right) \mathrm{d}s(x) \mathrm{d}t = 0,$$
$$\int_{T}^{2T} \int_{\Omega} \left( \mu_{R}(x,p) |\nabla p|^{2} + \delta |p|^{3} - \nabla p \cdot u_{t} \right) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \left( \alpha(x)p^{2} - \alpha(x)\bar{p}p \right) \mathrm{d}s(x) \mathrm{d}t \le 0.$$

Summing up the two above relations we get

$$\int_{T}^{2T} \int_{\Omega} \left( \delta(|u_t|^3 + |p|^3) + \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t + \mu_R(x, p) |\nabla p|^2 \right) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \left( \beta(x) \mathbf{D}(x) u_t \cdot u_t + \alpha(x) p^2 \right) \mathrm{d}s(x) \mathrm{d}t \le C$$
(4.26)

as a consequence of Hypothesis 3.1.1 and Hölder's inequality. As a next step, we test (4.24) by  $\phi = u_{tt}$ . Integrating by parts we infer

$$\int_{T}^{2T} \int_{\Omega} \left( \rho |u_{tt}|^{2} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt} : \nabla_{s} u_{tt} \right) dx dt$$

$$= \int_{T}^{2T} \int_{\Omega} \left( \mathbf{A}(x) \nabla_{s} u_{t} : \nabla_{s} u_{t} - u_{tt} \nabla p \right) dx dt + \int_{T}^{2T} \int_{\partial \Omega} \left( \beta(x) \mathbf{C}(x) u_{t} \cdot u_{t} - \beta(x) g_{t} u_{t} \right) ds(x) dt$$

$$(4.27)$$

 $\operatorname{Put}$ 

$$\Gamma_R := (1+R^2)^{\gamma}. \tag{4.28}$$

We have by (4.1), (4.26), and Hypothesis 3.1.1 (ii) that

$$\int_{T}^{2T} \int_{\Omega} |\nabla p|^2 \mathrm{d}x \mathrm{d}t \le \frac{1}{\mu^{\flat}} \Gamma_R \int_{T}^{2T} \int_{\Omega} \mu_R(x, p) |\nabla p|^2 \mathrm{d}x \mathrm{d}t \le C \Gamma_R, \tag{4.29}$$

and (4.26)-(4.27) yield that

$$\int_{T}^{2T} \int_{\Omega} \left( \rho |u_{tt}|^{2} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt} : \nabla_{s} u_{tt} \right) \mathrm{d}x \mathrm{d}t \le C \Gamma_{R} \,. \tag{4.30}$$

The next estimate is obtained by testing (4.24) by  $\phi = u$  and integrating by parts in time:

$$\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u dx dt + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \mathbf{C}(x) u \cdot u ds(x) dt$$

$$\leq \int_{T}^{2T} \int_{\Omega} \left( \delta |u_{t}|^{2} |u| + |u| |\nabla p| + \rho |u_{t}|^{2} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{t} : \nabla_{s} u_{t} \right) dx dt + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) g u ds(x) dt$$

$$(4.31)$$

and from (4.26) and Hölder's inequality we get

$$\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u dx dt + \frac{1}{2} \int_{T}^{2T} \int_{\partial \Omega} \beta(x) \mathbf{C}(x) u \cdot u ds(x) dt$$
$$\leq \int_{T}^{2T} \int_{\Omega} \left( \delta |u_{t}|^{2} |u| + |u| |\nabla p| \right) dx dt + C.$$
(4.32)

To estimate the right hand side of (4.32), we rewrite  $\delta |u_t|^2 |u|$  as  $\delta^{2/3} |u_t|^2 \delta^{1/3} |u|$ , and use Hölder's inequality to obtain

$$\int_{T}^{2T} \int_{\Omega} \delta |u_t|^2 |u| \mathrm{d}x \mathrm{d}t \le \left( \int_{T}^{2T} \int_{\Omega} \delta |u_t|^3 \mathrm{d}x \mathrm{d}t \right)^{2/3} \left( \int_{T}^{2T} \int_{\Omega} \delta |u|^3 \mathrm{d}x \mathrm{d}t \right)^{1/3}$$
$$\le C \left( \int_{T}^{2T} \int_{\Omega} \delta |u|^3 \mathrm{d}x \mathrm{d}t \right)^{1/3} \tag{4.33}$$

by virtue of (4.26). In order to estimate the remaining term on the right-hand side of (4.32), we exploit Hypothesis 3.1.1 and Hölder's inequality, namely

$$\begin{split} \int_{T}^{2T} \int_{\Omega} |u| \, |\nabla p| \mathrm{d}x \mathrm{d}t &\leq C \int_{T}^{2T} \int_{\Omega} \sqrt{\mu_R(x,p)} |\nabla p| |u| (1+p^2)^{\gamma/2} \mathrm{d}x \mathrm{d}t \\ &\leq C \left( \int_{T}^{2T} \int_{\Omega} \mu_R(x,p) |\nabla p|^2 \mathrm{d}x \mathrm{d}t \right)^{\frac{1}{2}} \left( \int_{T}^{2T} \int_{\Omega} |u|^2 (1+p^2)^{\gamma} \mathrm{d}x \mathrm{d}t \right)^{1/2} \\ &\leq C \left( \int_{T}^{2T} \int_{\Omega} |u|^4 \mathrm{d}x \mathrm{d}t \right)^{1/4} \left( \int_{T}^{2T} \int_{\Omega} (1+p^2)^{2\gamma} \mathrm{d}x \mathrm{d}t \right)^{1/4} \end{split}$$

where the term  $\int_T^{2T} \int_{\Omega} \mu_R(x, p) |\nabla p|^2 dx dt$  was estimated according to (4.26). By the Sobolev embedding theorem, we have

$$\left(\int_{T}^{2T} \int_{\Omega} |u|^{4} \mathrm{d}x \mathrm{d}t\right)^{1/4} \leq \left(\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u \mathrm{d}x \mathrm{d}t + \frac{1}{2} \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \mathbf{C}(x) u \cdot u \mathrm{d}s(x) \mathrm{d}t\right)^{1/2} + \left(\int_{T}^{2T} \int_{\Omega} |u_{t}|^{2} \mathrm{d}x \mathrm{d}t\right)^{1/2},$$

and a similar estimate holds for the right-hand side of (4.33). From the above computations, Young's inequality, and (4.32) we thus get

$$\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \beta \mathbf{C}(x) u \cdot u \mathrm{d}s \mathrm{d}t \le C \left(1 + \int_{T}^{2T} \int_{\Omega} (1 + p^{2})^{2\gamma} \mathrm{d}x \mathrm{d}t\right)^{1/2}.$$
(4.34)

To estimate the right-hand side of (4.34), we define an auxiliary function

$$h(p) := p(1+p^2)^{-\gamma/2}.$$
(4.35)

We have  $0 \le h'(p) \le (1+\gamma)(1+p^2)^{-\gamma/2} \le C\sqrt{\mu_R(x,p)}$  for all  $p \in \mathbb{R}$ . Hence, by Sobolev embedding and by (4.26) we have

$$\begin{split} \int_{T}^{2T} &\int_{\Omega} |h(p)|^{2} \mathrm{d}x \mathrm{d}t \leq C \left( \int_{T}^{2T} &\int_{\Omega} |\nabla h(p)|^{2} \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} &\int_{\partial \Omega} \alpha(x) |h(p)|^{2} \mathrm{d}s(x) \mathrm{d}t \right) \\ &\leq C \left( \int_{T}^{2T} &\int_{\Omega} \mu_{R}(x,p) |\nabla p|^{2} \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} &\int_{\partial \Omega} \alpha(x) |p|^{2} \mathrm{d}s(x) \mathrm{d}t \right) \leq C. \end{split}$$

We thus have

$$\int_{T}^{2T} \int_{\Omega} (1+p^2)^{1-\gamma} \mathrm{d}x \mathrm{d}t \le \int_{T}^{2T} \int_{\Omega} (1+|h(p)|^2) \mathrm{d}x \mathrm{d}t \le C.$$

Since  $(1 - \gamma) > 2\gamma$  by (3.13), we conclude from (4.34) that

$$\int_{T}^{2T} \int_{\Omega} \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \beta \mathbf{C}(x) u \cdot u \mathrm{d}s \mathrm{d}t \le C.$$
(4.36)

### 4.3 $L^{\infty}$ -bounds

Testing (4.24) again by  $\phi = u_t$  we get

$$\int_{\Omega} \left( \rho u_{tt} \cdot u_t + \delta |u_t|^3 + \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t + \mathbf{A}(x) \nabla_s u : \nabla_s u_t + \varepsilon \mathbf{B}(x) \nabla_s u_{tt} : \nabla_s u_t + \nabla p \cdot u_t \right) dx$$
$$+ \int_{\partial \Omega} \left( \beta(x) \mathbf{C}(x) u \cdot u_t + \beta(x) \mathbf{D}(x) u_t \cdot u_t - \beta(x) g \cdot u_t \right) ds(x) = 0$$

which can be rewritten as

$$\frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \left( \int_{\Omega} \left( \rho |u_t|^2 + \mathbf{A}(x) \nabla_s u : \nabla_s u + \varepsilon \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t \right) \mathrm{d}x + \int_{\partial\Omega} \beta(x) \mathbf{C}(x) u \cdot u \mathrm{d}s(x) \right) \\
+ \int_{\Omega} \left( \delta |u_t|^3 + \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t \right) \mathrm{d}x + \int_{\partial\Omega} \beta(x) \left( \mathbf{D}(x) u_t \cdot u_t \right) \mathrm{d}s(x) \\
= \int_{\Omega} -\nabla p \cdot u_t \mathrm{d}x + \int_{\partial\Omega} \beta(x) g \cdot u_t \mathrm{d}s(x) \quad \text{a.e.} \quad \text{in} \quad (T, 2T).$$
(4.37)

Let y(t) denote the expression under the time derivative in (4.37), and let z(t) denote the remaining terms. According to (4.26), (4.36), and Hypothesis 3.1.1, Eq. (4.37) is of the

type  $\dot{y}(t) = z(t)$  with  $y, z \in L_T^1$ , and

$$\int_{T}^{2T} |z| \mathrm{d}t \le C + \left(\int_{T}^{2T} \int_{\Omega} |\nabla p|^2 \mathrm{d}x \mathrm{d}t\right)^{1/2} \left(\int_{T}^{2T} \int_{\Omega} |u_t|^2 \mathrm{d}x \mathrm{d}t\right)^{1/2} \le C\sqrt{\Gamma_R}.$$

by virtue of (4.26) and (4.29). From the identity

$$y(t) = \frac{1}{T} \int_{t-T}^{t} (y(s) + (T+s-t)z(s)) ds$$

it follows that the  $L^1\text{-norm}$  of y and z control the  $L^\infty\text{-norm}$  of y and so it follows in particular that

$$\sup_{t \in (T,2T)} \int_{\Omega} \left( |u_t|^2 + |\nabla_s u|^2 + \varepsilon |\nabla_s u_t|^2 \right) \mathrm{d}x + \int_{\partial \Omega} \beta(x) |u|^2 \mathrm{d}s(x) \le C\sqrt{\Gamma_R}.$$
(4.38)

Then we test (4.24) by  $\phi = u_{tt}$ . The term  $\sqrt{\varepsilon}u_{tt}$  belongs to  $L^2_T(W^{1,2}(\Omega))$ , so that it is an admissible choice. This yields for a.e.  $t \in (T, 2T)$  that

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} \left( \frac{1}{2} \mathbf{B}(x) \nabla_s u_t : \nabla_s u_t + \mathbf{A}(x) \nabla_s u : \nabla_s u_t + \frac{\delta}{3} |u_t|^3 \right) \mathrm{d}x \\ + \frac{\mathrm{d}}{\mathrm{d}t} \int_{\partial\Omega} \beta(x) \left( \mathbf{C}(x) u \cdot u_t + \frac{1}{2} \mathbf{D}(x) u_t \cdot u_t \right) \mathrm{d}s(x) + \int_{\Omega} \left( \rho u_{tt}^2 + \varepsilon \mathbf{B}(x) \nabla_s u_{tt} : \nabla_s u_{tt} \right) \mathrm{d}x \\ = \int_{\Omega} \left( -\nabla p \cdot u_{tt} - \mathbf{A}(x) \nabla_s u_t : \nabla_s u_t \right) \mathrm{d}x + \int_{\partial\Omega} \beta(x) \left( -g_t \cdot u_t + \mathbf{C}(x) u_t \cdot u_t \right) \mathrm{d}s(x)$$

Arguing similarly as above and using (4.26), (4.29)–(4.30) and (4.38), we get

$$\sup_{t \in (T,2T)} \int_{\Omega} |\nabla_s u_t|^2 \mathrm{d}x + \int_{\partial \Omega} \beta(x) |u_t|^2 \mathrm{d}s(x) \le C\Gamma_R.$$
(4.39)

By Sobolev embedding of  $W^{1,2}(\Omega)$  into  $L^6(\Omega)$  we obtain from (4.39) that

$$\sup_{t \in (0,T)} \exp|u_t(t)|_6^2 \le C\Gamma_R.$$
(4.40)

The next strategy is to test (4.25) by  $v_t$  with

$$v = M_R(x, p), \tag{4.41}$$

see (4.7). However, there is no evidence that  $v_t(\cdot, t) \in W^{1,2}(\Omega)$ , so that this is not an admissible choice. Instead, we therefore test (4.25) by  $\hat{v}(x,t) = \frac{1}{\tau}(v(x,t) - v(x,t-\tau))\lambda(t)$ , where  $\tau > 0$  is a small time shift, and  $\lambda$  is a positive Lipschitz continuous *T*-periodic

function. We get

$$\int_{\Omega} \left( (f_R(x,p) + G[p])_t + \delta|p|p \right) \frac{1}{\tau} (v(x,t) - v(x,t-\tau)\lambda(t) dx \\
+ \int_{\Omega} \mu_R(x,p) \nabla p \cdot \nabla \left( \frac{v(x,t) - v(x,t-\tau)}{\tau} \lambda(t) \right) dx \\
- \int_{\Omega} u_t \cdot \nabla \left( \frac{v(x,t) - v(x,t-\tau)}{\tau} \lambda(t) \right) dx + \int_{\partial\Omega} \alpha(x)(p-\bar{p}) \frac{v(x,t) - v(x,t-\tau)}{\tau} \lambda(t) = 0 \\$$
(4.42)

We now treat separately each of the four integrals on the left hand side of (4.42), which we write for simplicity as I + II + III + IV. In order to evaluate the term I, we note that for a general function  $v_0 \in W_T^{1,2}$  we have

$$\begin{split} \int_{T}^{2T} \left| \dot{v}_{0}(t) - \frac{1}{\tau} (v_{0}(t) - v_{0}(t - \tau)) \right|^{2} \mathrm{d}t &= \int_{T}^{2T} \left| \dot{v}_{0}(t) - \frac{1}{\tau} \int_{t - \tau}^{t} \dot{v}_{0}(s) \mathrm{d}s \right|^{2} \mathrm{d}t \\ &= \frac{1}{\tau^{2}} \int_{T}^{2T} \int_{t - \tau}^{t} (|\dot{v}_{0}(t) - \dot{v}_{0}(s)| \mathrm{d}s)^{2} \mathrm{d}t \leq \frac{1}{\tau} \int_{T}^{2T} \int_{t - \tau}^{t} |\dot{v}_{0}(t) - \dot{v}_{0}(s)|^{2} \mathrm{d}s \mathrm{d}t \\ &\leq \frac{1}{\tau} \int_{T}^{2T} \int_{0}^{\tau} |\dot{v}_{0}(t) - \dot{v}_{0}(t - h)|^{2} \mathrm{d}h \mathrm{d}t = \frac{1}{\tau} \int_{0}^{\tau} \int_{T}^{2T} |\dot{v}_{0}(t) - \dot{v}_{0}(t - h)|^{2} \mathrm{d}t \mathrm{d}h \end{split}$$

where  $\dot{v}_0(t)$  is defined as the time derivative of  $v_0(t)$ . From Lusin's Theorem it follows that  $\frac{1}{\tau}(v_0(t) - v_0(t-\tau))$  converge strongly to  $\dot{v}_0$  in  $L_T^2$  as  $\tau \to 0$ . Hence,  $\hat{v}$  converge strongly in  $L_T^2(L^2(\Omega))$  to  $v_t\lambda(t)$ , with  $v_t = \bar{\mu}(x, p)p_t$ , and we get

$$\lim_{\tau \to 0^+} \int_T^{2T} I dt = \lim_{\tau \to 0^+} \int_T^{2T} \int_\Omega \left( (f_R(x, p) + G[p])_t + \delta |p| p \right) \frac{1}{\tau} (v(x, t) - v(x, t - \tau)) \lambda(t) dx dt$$
$$= \int_T^{2T} \int_\Omega \left( (f_R(x, p) + G[p])_t + \delta |p| p \right) v_t \lambda(t) dx dt.$$
(4.43)

Put

$$M_R^*(x,p) = \int_0^p \bar{\mu}(x,s) |s| s \mathrm{d}s.$$
(4.44)

Then (4.43) can be rewritten by integration by parts as

$$\lim_{\tau \to 0^+} \int_T^{2T} I dt = \int_T^{2T} \int_\Omega (f_R(x, p) + G[p])_t v_t \lambda(t) dx dt - \delta \int_T^{2T} \int_\Omega M_R^*(x, p) \dot{\lambda}(t) dx dt.$$
(4.45)

Let us pass to the integral II. We have by definition of v that

$$\mu_R(x,p)\nabla p = \mu_0(x)\left(\nabla v - \nabla_x M_R(x,p)\right).$$

We split the integral into two parts. On the one hand we have

$$\begin{split} &\frac{1}{\tau} \int_{T}^{2T} \int_{\Omega} \mu_{0}(x) \nabla v(x,t) \cdot \left( \nabla v(x,t) - \nabla v(x,t-\tau) \right) \lambda(t) \mathrm{d}x \mathrm{d}t \\ &= \frac{1}{2\tau} \int_{T}^{2T} \int_{\Omega} \mu_{0}(x) \left( |\nabla v(x,t)|^{2} - |\nabla v(x,t-\tau)|^{2} + |\nabla v(x,t) - \nabla v(x,t-\tau)|^{2} \right) \lambda(t) \mathrm{d}x \mathrm{d}t \\ &\geq \frac{1}{2\tau} \int_{T}^{2T} \int_{\Omega} \mu_{0}(x) |\nabla v(x,t)|^{2} (\lambda(t) - \lambda(t+\tau)) \mathrm{d}x \mathrm{d}t, \end{split}$$

hence,

$$\liminf_{\tau \to 0+} \frac{1}{\tau} \int_{T}^{2T} \int_{\Omega} \mu_0(x) \nabla v(x,t) \cdot \left( \nabla v(x,t) - \nabla v(x,t-\tau) \right) \lambda(t) dx dt$$
  
$$\geq - \int_{T}^{2T} \int_{\Omega} \frac{\mu_0(x)}{2} |\nabla v(x,t)|^2 \dot{\lambda}(t) dx dt.$$
(4.46)

On the other hand,

$$\int_{T}^{2T} \int_{\Omega} \mu_0(x) \nabla_x M_R(x, p) \cdot \left( \nabla v(x, t) - \nabla v(x, t - \tau) \right) \lambda(t) dx dt$$
$$= -\int_{T}^{2T} \int_{\Omega} \mu_0(x) (\nabla_x M_R(x, p(x, t + \tau)) \lambda(t + \tau) - \nabla_x M_R(x, p(x, t)) \lambda(t)) \cdot \nabla v(x, t) dx dt,$$

hence,

$$\lim_{\tau \to 0+} \frac{1}{\tau} \int_{T}^{2T} \int_{\Omega} \mu_0(x) \nabla_x M_R(x, p) \cdot \left( \nabla v(x, t) - \nabla v(x, t - \tau) \right) \lambda(t) dx dt$$
$$= -\int_{T}^{2T} \int_{\Omega} \mu_0(x) \left( \nabla_x \bar{\mu}_R(x, p) p_t \lambda(t) + \nabla_x M_R(x, p) \dot{\lambda}(t) \right) \nabla v dx dt.$$
(4.47)

Combining (4.46) with (4.47) we obtain

$$\liminf_{\tau \to 0+} \int_{T}^{2T} II dt \geq -\int_{T}^{2T} \int_{\Omega} \frac{\mu_0(x)}{2} |\nabla v(x,t)|^2 \dot{\lambda}(t) dx dt \\
+ \int_{T}^{2T} \int_{\Omega} \mu_0(x) \left( \nabla_x \bar{\mu}_R(x,p) p_t \lambda(t) + \nabla_x M_R(x,p) \dot{\lambda}(t) \right) \nabla v dx dt.$$
(4.48)

We integrate III in time and get

$$-\frac{1}{\tau} \int_{T}^{2T} \int_{\Omega} u_t(x,t) \left( \nabla v(x,t) - \nabla v(x,t-\tau) \right) \lambda(t) \mathrm{d}x \mathrm{d}t$$
$$= -\frac{1}{\tau} \int_{T}^{2T} \int_{\Omega} \left( \nabla v(x,t) u(x,t) \lambda(t) - \nabla (x,t) u_t(x,t+\tau) \lambda(t+\tau) \right) \mathrm{d}x \mathrm{d}t$$

$$-\frac{1}{\tau}\int_{T}^{2T}\int_{\Omega}\nabla v(x,t)\big(u_t(x,t)\lambda(t)-u_t(x,t+\tau)\lambda(t+\tau)\big)\mathrm{d}x\mathrm{d}t.$$

Hence,

$$\lim_{\tau \to 0} \int_{T}^{2T} III dt = \int_{T}^{2T} \int_{\Omega} \nabla v(x,t) \left( u_{tt}(x,t)\lambda(t) + u_t(x,t)\dot{\lambda}(t) \right) dx dt.$$
(4.49)

Finally, in order to handle the term IV, we consider the function  $\hat{M}_R(x,v) = \int_0^v M_R^{-1}(x,v') dv'$ , where  $M_R^{-1}$  is as in (4.8). Then  $p = M_R^{-1}(x,v)$ . The function  $\hat{M}_R$  is convex in v because  $M_R^{-1}$  is an increasing function of v. This means that the inequality  $\hat{M}_R(x,v) - \hat{M}_R(x,\tilde{v}) \leq M_R^{-1}(v)(v-\tilde{v})$  holds for all  $v, \tilde{v} \in \mathbb{R}$ , in particular

$$\frac{1}{\tau} \left( \hat{M}_R(x, v(x, t)) - \hat{M}_R(x, v(x, t - \tau)) \right) \le \frac{1}{\tau} M_R^{-1}(x, v(x, t))(v(x, t) - v(x, t - \tau)).$$

We thus have

$$\frac{1}{\tau} \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) \left( \hat{M}_{R}(x, v(x, t)) - \hat{M}_{R}(x, v(x, t - \tau)) \right) \lambda(t) \mathrm{d}s(x) \mathrm{d}t \\
\leq \frac{1}{\tau} \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) p(x, t) \left( v(x, t) - v(x, t - \tau) \right) \lambda(t) \mathrm{d}s(x) \mathrm{d}t,$$
(4.50)

hence

$$\liminf_{\tau \to 0+} \frac{1}{\tau} \int_{T}^{2T} \int_{\partial \Omega} \alpha(x) p(x,t) \big( v(x,t) - v(x,t-\tau) \big) \lambda(t) \mathrm{d}s(x) \mathrm{d}t \\ \geq - \int_{T}^{2T} \int_{\partial \Omega} \alpha(x) \hat{M}_{R}(x,v(x,t)) \dot{\lambda}(t) \mathrm{d}s(x) \mathrm{d}t.$$

In a similar way,

$$\lim_{\tau \to 0+} \frac{1}{\tau} \int_{T}^{2T} \int_{\partial \Omega} \alpha(x) \bar{p}(v(x,t) - v(x,t-\tau))\lambda(t) \mathrm{d}s(x) \mathrm{d}t$$
$$= -\int_{T}^{2T} \int_{\partial \Omega} \alpha(x) (\bar{p}_t \lambda(t) + \bar{p} \dot{\lambda}(t)) v(x,t) \mathrm{d}s(x) \mathrm{d}t.$$

Subtracting these two limits we get

$$\liminf_{\tau \to 0+} \int_{T}^{2T} IV dt \ge \int_{T}^{2T} \int_{\partial \Omega} \alpha(x) \Big( \big(\bar{p}_t \lambda(t) + \bar{p} \dot{\lambda}(t)\big) v(x,t) - \hat{M}_R(x, v(x,t)) \dot{\lambda}(t) \Big) ds(x) dt.$$
(4.51)

Combining (4.42) with (4.45), (4.48), (4.49), and (4.51) yields

$$\int_{T}^{2T} \int_{\Omega} (f_R(x,p) + G[p])_t v_t \lambda(t) \mathrm{d}x \mathrm{d}t - \delta \int_{T}^{2T} \int_{\Omega} M_R^*(x,p) \dot{\lambda}(t) \mathrm{d}x \mathrm{d}t$$

$$-\int_{T}^{2T} \int_{\Omega} \frac{\mu_{0}(x)}{2} |\nabla v|^{2} \dot{\lambda}(t) \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\Omega} \mu_{0}(x) \left( \nabla_{x} \bar{\mu}_{R}(x,p) p_{t} \lambda(t) + \nabla_{x} M_{R}(x,p) \dot{\lambda}(t) \right) \cdot \nabla v \mathrm{d}x \mathrm{d}t \\ + \int_{T}^{2T} \int_{\Omega} \left( u_{tt}(x,t) \lambda(t) + u_{t}(x,t) \dot{\lambda}(t) \right) \cdot \nabla v \mathrm{d}x \mathrm{d}t \\ + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) \left( \left( \bar{p}_{t} \lambda(t) + \bar{p} \dot{\lambda}(t) \right) v - \hat{M}_{R}(x,v) \dot{\lambda}(t) \right) \mathrm{d}s(x) \mathrm{d}t \leq 0.$$

This is an inequality of the form

$$-\int_{T}^{2T} A(t)\dot{\lambda}(t)\mathrm{d}t + \int_{T}^{2T} B(t)\lambda(t)\mathrm{d}t \le 0$$
(4.52)

with

$$\begin{aligned} A(t) &= \int_{\Omega} \left( \frac{\mu_0(x)}{2} |\nabla v|^2 + \delta M_R^*(x, p) - \mu_0(x) \nabla_x M_R(x, p) \cdot \nabla v - u_t \cdot \nabla v \right) (x, t) \mathrm{d}x \\ &+ \int_{\partial \Omega} \alpha(x) \left( \hat{M}_R(x, v) - \bar{p}v \right) (x, t) \mathrm{d}s(x), \\ B(t) &= \int_{\Omega} \left( (f_R(x, p) + G[p])_t v_t + (u_{tt} + \mu_0(x) p_t \nabla_x \bar{\mu}_R(x, p)) \cdot \nabla v \right) \mathrm{d}x + \int_{\partial \Omega} \alpha(x) \bar{p}_t v \mathrm{d}s(x) \, . \end{aligned}$$

We have  $G[p]_t v_t \ge 0$  a.e. and

$$f_R(x,p)_t v_t = \partial_p f(x,p) \bar{\mu}_R(x,p) |p_t|^2 \ge \frac{\mu^\flat}{\mu^\sharp} (1+R^2)^{-\gamma} \partial_p f_R(x,p) |p_t|^2 = \frac{\mu^\flat}{\mu^\sharp} \Gamma_R \partial_p f_R(x,p) |p_t|^2$$

according to (3.14), (4.1) and (4.28). On the other hand, by Hypothesis 3.1.1 (i)–(ii), we have  $|\nabla_x \bar{\mu}_R(x,p)| \leq C (\partial_p f_R(x,p))^{1/2}$ . In particular,

$$|\mu_0(x)p_t\nabla_x\bar{\mu}_R(x,p)\cdot\nabla v| \le C\left(f_R(x,p)_tp_t\right)^{1/2}|\nabla v|.$$
(4.53)

Furthermore, by (4.41) we have  $|\nabla v| \leq |\nabla_x M_R(x,p)| + \bar{\mu}_R(x,p)|\nabla p|$ , hence, by (4.26),

$$\int_{T}^{2T} \int_{\Omega} |\nabla v(x,t)|^2 \mathrm{d}x \mathrm{d}t \le C \left( 1 + \int_{T}^{2T} \int_{\Omega} \mu_R(x,p) |\nabla p|^2 \mathrm{d}x \mathrm{d}t \right) \le C.$$
(4.54)

Using (4.53)–(4.54), Hölder's inequality, and the inequality  $az^2 - bz \ge \frac{a}{2}z^2 - \frac{b^2}{2a}$  for a > 0, b > 0, and  $z \in \mathbb{R}$  we conclude that

$$\int_{T}^{2T} \int_{\Omega} \left( (f_R(x,p) + G[p])_t v_t + \mu_0(x) p_t \nabla_x \bar{\mu}_R(x,p) \cdot \nabla v \right) \mathrm{d}x \mathrm{d}t$$
  
$$\geq \frac{\mu^\flat}{\mu^\sharp \Gamma_R} \int_{T}^{2T} \int_{\Omega} \partial_p f_R(x,p) |p_t|^2 \mathrm{d}x \mathrm{d}t - C \left( \int_{T}^{2T} \int_{\Omega} \partial_p f_R(x,p) |p_t|^2 \mathrm{d}x \mathrm{d}t \right)^{1/2}$$

$$\geq \frac{\mu^{\flat}}{2\mu^{\sharp}\Gamma_R} \int_T^{2T} \int_{\Omega} \partial_p f_R(x,p) |p_t|^2 \mathrm{d}x \mathrm{d}t - \frac{C\mu^{\sharp}\Gamma_R}{2\mu^{\flat}}$$
(4.55)

with a constant C > 0 independent of  $R, \delta$ , and  $\varepsilon$ . The remaining terms in B(t) can be estimated using (4.26), (4.30), and (4.54) as follows.

$$\int_{T}^{2T} \int_{\Omega} |u_{tt}| |\nabla v| \mathrm{d}x \mathrm{d}t \leq C \Gamma_{R}^{1/2}$$
$$\int_{T}^{2T} \int_{\partial \Omega} \alpha(x) |\bar{p}_{t}| |v| \mathrm{d}s(x) \mathrm{d}t \leq C.$$

Hence,

$$\int_{T}^{2T} B(t) dt \ge \frac{c}{\Gamma_R} \int_{T}^{2T} \int_{\Omega} \partial_p f_R(x, p) |p_t|^2 dx dt - C\Gamma_R$$
(4.56)

with constants C > c > 0 independent of R,  $\delta$ , and  $\varepsilon$ . Using the relations  $0 \leq \hat{M}_R(x, v) = \int_0^p \bar{\mu}_R(x, p') p' dp' \leq (\mu^{\sharp}/2\mu^{\flat}) p^2$  and  $0 \leq M_R^*(x, p) \leq (\mu^{\sharp}/\mu^{\flat}) |p|^3$ , we similarly obtain from (4.26) that

$$\int_{T}^{2T} |A(t)| \mathrm{d}t \le C \left( 1 + \int_{T}^{2T} \int_{\Omega} |\nabla v|^2 \mathrm{d}x \mathrm{d}t + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) p^2 \mathrm{d}s(x) \mathrm{d}t \right) \le C.$$
(4.57)

We now choose arbitrary points 2T < r < 3T and T < s < r such that r - s < T. Then for each  $\varepsilon < \frac{r-s}{2}$  we set in (4.52)

$$\lambda(t) = \frac{1}{\varepsilon}(t-s) \text{ for } t \in (s,s+\varepsilon), \quad \lambda(t) = \frac{1}{\varepsilon}(r-t) \text{ for } t \in (r-\varepsilon,r),$$

choosing  $\lambda$  constant and continuous otherwise, *T*-periodically extended to the whole real line. All functions under the integrals in (4.52) are *T*-periodic. We thus can replace the integration domain [T, 2T] with [s, s + T], which yields that

$$\frac{1}{\varepsilon} \int_{r-\varepsilon}^{r} A(t) dt \le \frac{1}{\varepsilon} \int_{s}^{s+\varepsilon} A(t) dt - \int_{T}^{2T} B(t) dt.$$

Integrating the above inequality over s from r - T to  $r - 2\varepsilon$  we obtain

$$\frac{1}{\varepsilon} \int_{r-\varepsilon}^{r} A(t) \mathrm{d}t \le \frac{1}{T-2\varepsilon} \int_{T}^{2T} |A(t)| \mathrm{d}t - \int_{T}^{2T} B(t) \mathrm{d}t.$$
(4.58)

At each of its Lebesgue points t = r, the function A(t) admits therefore a pointwise bound given by the right hand side of (4.58), which is in turn estimated by (4.56)–(4.57), that is,

$$\frac{c}{\Gamma_R} \int_T^{2T} \int_{\Omega} \partial_p f_R(x, p) |p_t|^2 \mathrm{d}x \mathrm{d}t + \sup_{t \in [T, 2T]} \mathrm{ess} A(t) \le C\Gamma_R.$$
(4.59)

To estimate A(t) from below, we first notice that

$$\hat{M}_{R}(x,v) - \bar{p}v = \int_{0}^{p} (p' - \bar{p})\bar{\mu}_{R}(x,p')dp' \ge \frac{\mu^{\flat}}{2\mu^{\sharp}\Gamma_{R}}(p^{2} - C),$$

and

$$|\nabla v|^2 = \bar{\mu}_R^2(x, p) |\nabla p|^2 \ge \left(\frac{\mu^\flat}{\mu^\sharp \Gamma_R}\right)^2 |\nabla p|^2.$$

The above computations yield that

$$A(t) \ge c \left( \frac{1}{\Gamma_R^2} \int_{\Omega} |\nabla p|^2(x, t) \mathrm{d}x + \frac{1}{\Gamma_R} \int_{\partial \Omega} \alpha(x) p^2 \mathrm{d}s(x) \right) - C.$$

From (4.59) we thus obtain that

$$c\Gamma_R \int_T^{2T} \int_{\Omega} \partial_p f_R(x,p) |p_t|^2 \mathrm{d}x \mathrm{d}t + \sup_{t \in (T,2T)} \left( \int_{\Omega} |\nabla p|^2(x,t) \mathrm{d}x + \int_{\partial\Omega} \alpha(x) p^2 \mathrm{d}s(x) \right) \le C\Gamma_R^3.$$
(4.60)

### 4.4 Uniform estimates of p

To remove the cut-off parameter R, we proceed by Moser iterations and define Lipschitz continuous functions  $a_{R,k}(p)$  and their antiderivatives  $A_{R,k}(p)$  with indices  $k \ge 1$  and with R > 1 by the formula

$$a_{R,k}(p) = p(1 + \min\{p^2, R^2\})^k, \ A_{R,k}(p) = 1 + \int_0^p a_{R,k}(s) \mathrm{d}s.$$
 (4.61)

We have

$$a'_{R,k}(p) = \begin{cases} (1+p^2)^{k-1}(1+(2k+1)p^2) & \text{for } |p| < R,\\ (1+R^2)^k & \text{for } |p| \ge R, \end{cases}$$
(4.62)

and

$$A_{R,k}(p) = \begin{cases} \frac{1}{2k+2}(1+p^2)^{k+1} & \text{for } |p| < R, \\ \frac{1}{2k+2}(1+R^2)^{k+1} + \frac{1}{2}(p^2 - R^2)(1+R^2)^k \ge \frac{1}{2k+2}(1+p^2)(1+R^2)^k & \text{for } |p| \ge R. \end{cases}$$
(4.63)

Put

$$F_{k}(x,p) = \frac{f^{\flat}}{2k} + \int_{0}^{p} \partial_{p} f_{R}(x,s) a_{R,k}(s) \mathrm{d}s,$$

$$V_{k}[p] = \int_{0}^{\infty} \int_{0}^{\mathfrak{f}_{r}[p]} a_{R,k}(v) \psi(x,r,v) \mathrm{d}v \mathrm{d}r.$$
(4.64)

Then

$$G[p]_t a_{R,k}(p) - V_k[p]_t \ge 0 \quad a. e.$$
(4.65)

by Proposition 3.3.5, and testing (4.25) by  $a_{R,k}(p)$  we obtain

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (F_k(\cdot, p) + V_k[p])(x, t) \mathrm{d}x + \int_{\Omega} \mu_R(x, p) |\nabla p|^2 a'_{R,k}(p) \mathrm{d}x + \int_{\partial\Omega} \alpha(x)(p - \bar{p}) a_{R,k}(p) \mathrm{d}s(x) \\
\leq \int_{\Omega} (u_t \cdot \nabla p) a'_{R,k}(p) \mathrm{d}x \leq \frac{1}{2} \int_{\Omega} |u_t|^2 \frac{a'_{R,k}(p)}{\mu_R(x, p)} \mathrm{d}x + \int_{\Omega} \frac{\mu_R(x, p)}{2} |\nabla p|^2 a'_{R,k}(p) \mathrm{d}x.$$
(4.66)

We have omitted the positive  $\delta$ -term on the left-hand side which does not bring any relevant information. The boundary term will be estimated from below as follows. We first notice that the function  $a_{R,k}$  is increasing, hence

$$(p - \bar{p})a_{R,k}(p) \ge A_{R,k}(p) - A_{R,k}(\bar{p}).$$

Since the function  $\bar{p}$  is bounded, we may take

$$R > \sup \operatorname{ess} \bar{p} \tag{4.67}$$

and finally get the estimate

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (F_k(\cdot, p) + V_k[p]) \mathrm{d}x + c \int_{\Omega} |\nabla p|^2 \mu_R(x, p) a'_{R,k}(p) \mathrm{d}x + \int_{\partial \Omega} \alpha(x) A_{R,k}(p) \mathrm{d}s(x) \\
\leq C^{2k+2} + C \int_{\Omega} |u_t|^2 \frac{a'_{R,k}(p)}{\mu_R(x, p)} \mathrm{d}x$$
(4.68)

with constants c, C independent of  $\delta, \varepsilon, R$  and k. The last term on the right-hand side of (4.68) will again be estimated by Hölder's inequality and (4.40) as follows:

$$\int_{\Omega} |u_t|^2 \frac{a'_{R,k}(p)}{\mu_R(x,p)} \mathrm{d}x \le |u_t(t)|_6^2 \left( \int_{\Omega} \left( \frac{a'_{R,k}(p)}{\mu_R(x,p)} \right)^{3/2} \mathrm{d}x \right)^{2/3} \le C\Gamma_R \left( \int_{\Omega} \left( \frac{a'_{R,k}(p)}{\mu_R(x,p)} \right)^{3/2} \mathrm{d}x \right)^{2/3}$$

Then (4.68) can be reduced to

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (F_k(\cdot, p) + V_k[p]) \mathrm{d}x + c \int_{\Omega} |\nabla p|^2 \mu_R(x, p) a'_{R,k}(p) \mathrm{d}x + \int_{\partial \Omega} \alpha(x) A_{R,k}(p) \mathrm{d}s(x)$$

$$\leq C^{2k+2} + C\Gamma_R \left( \int_{\Omega} \left( \frac{a'_{R,k}(p)}{\mu_R(x,p)} \right)^{3/2} \mathrm{d}x \right)^{2/3}.$$
(4.69)

Put

$$w_{k} = \begin{cases} (1+p^{2})^{\frac{k+1-\eta}{2}} & \text{for } |p| < R, \\ (1+p^{2})^{\frac{1}{2}}(1+R^{2})^{\frac{k-\eta}{2}} & \text{for } |p| \ge R, \end{cases} \quad \hat{w}_{k} = \begin{cases} (1+p^{2})^{\frac{k+\gamma}{2}} & \text{for } |p| < R, \\ (1+R^{2})^{\frac{k+\gamma}{2}} & \text{for } |p| \ge R, \end{cases}$$

$$(4.70)$$

Using the hypothesis  $\gamma + \eta \leq 1$  from (3.13) we obtain the pointwise bounds almost everywhere

$$0 \leq \hat{w}_{k} \leq w_{k},$$

$$\frac{a'_{R,k}(p)}{\mu_{R}(x,p)} \leq \frac{1}{\mu^{\flat}} (2k+1) |\hat{w}_{k}|^{2},$$

$$|w_{k}|^{2} \leq (2k+2) A_{R,k}(p),$$

$$|\nabla w_{k}|^{2} \leq \frac{1}{\mu^{\flat}} (k+1) |\nabla p|^{2} \mu_{R}(x,p) a'_{R,k}(p).$$
(4.71)

From (4.69) we thus obtain

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega} (F_k(\cdot, p) + V_k[p]) \mathrm{d}x + \frac{c}{k+1} \left( \int_{\Omega} |\nabla w_k|^2 \mathrm{d}x + \int_{\partial \Omega} \alpha(x) |w_k|^2 \mathrm{d}s(x) \right) \\
\leq C^{2k+2} + C(k+1) \Gamma_R \left( \int_{\Omega} |\hat{w}_k|^3 \mathrm{d}x \right)^{2/3},$$
(4.72)

where c and C are similarly as before constants independent of  $\delta,~\varepsilon,~k$  and R.~ For simplicity, put

$$\mathbf{F}_{k}(t) = \int_{\Omega} F_{k}(x, p(x, t)) \mathrm{d}x, \quad \mathbf{V}_{k}(t) = \int_{\Omega} V_{k}[p](x, t) \mathrm{d}x.$$
(4.73)

Using the symbols  $|\cdot|_q$  and  $||\cdot||_{1,q}$  for  $1 \leq q \leq \infty$  to denote the norms in  $L^q(\Omega)$  and in  $W^{1,q}(\Omega)$ , respectively, we have the interpolation inequality

$$|\hat{w}_k(t)|_3 \le |\hat{w}_k(t)|_{3/2}^{1/3} |\hat{w}_k(t)|_6^{2/3} \le |\hat{w}_k(t)|_{3/2}^{1/3} |w_k(t)|_6^{2/3}, \tag{4.74}$$

and by embedding of  $W^{1,2}(\Omega)$  into  $L^6(\Omega)$ ,

$$|w_k(t)|_6 \le C ||w_k(t)||_{1,2}.$$

The right-hand side of (4.72) can thus be estimated from above by

$$C^{2k+2} + C(k+1)\Gamma_R |\hat{w}_k(t)|^{2/3}_{3/2} ||w_k(t)|^{4/3}_{1,2}.$$
(4.75)

We now use Young's inequality

$$a^{2/3}b^{4/3} \le \frac{1}{3\omega^3}a^2 + \frac{2\omega^{3/2}}{3}b^2$$

with  $a = (C(k+1)\Gamma_R)^{3/2} |\hat{w}_k(t)|_{3/2}$ ,  $b = ||w_k(t)||_{1,2}$ , and  $\omega = (\frac{c}{k+1})^{2/3}$ . This yields (note that  $k \ge 1$ )

$$\frac{\mathrm{d}}{\mathrm{d}t}(\mathbf{F}_k(t) + \mathbf{V}_k(t)) + \frac{c}{k} \|w_k(t)\|_{1,2}^2 \le C^{2k+2} + Ck^5 \Gamma_R^3 |\hat{w}_k(t)|_{3/2}^2.$$
(4.76)

By (3.28) we have

$$0 \le \mathbf{V}_k(t) \le CB^{2k+2},\tag{4.77}$$

for all  $t \ge 0$ , and from Hypothesis 3.1.1 we get the lower bound for  $\mathbf{F}_k$ 

$$\mathbf{F}_{k}(t) \ge \frac{f^{\flat}}{2k} |w_{k}(t)|_{2}^{2}.$$
(4.78)

An upper bound for  $F_k(x, p)$  for  $|p| \leq R$  is still straightforward, namely

$$F_k(x,p) \le \frac{f^\flat}{2k} + \int_0^p f^\sharp s(1+s^2)^k \mathrm{d}s \le \frac{f^\sharp}{2k} (1+p^2)^{k+1}.$$
(4.79)

The case p > R or p < -R is more delicate. For p > R we have

$$F_{k}(x,p) \leq \frac{f^{\flat}}{2k} + \int_{0}^{R} f^{\sharp} s(1+s^{2})^{k} ds + \int_{R}^{p} f^{\sharp} s(1+R^{2})^{k} ds$$
  
$$\leq \frac{f^{\sharp}}{2k} (1+R^{2})^{k} \left(1+R^{2}+k(p^{2}-R^{2})\right) \leq \frac{f^{\sharp}}{2} (1+p^{2})(1+R^{2})^{k}$$
  
$$\leq \frac{f^{\sharp}}{2} |w_{k}|^{q_{k}}$$
(4.80)

for a.e.  $(x,t) \in \Omega \times (T,2T)$  with

$$q_k = \frac{2(k+1)}{k+1-\eta}.$$
(4.81)

Indeed, (4.80) holds for p < -R as well by symmetry. By virtue of (4.70) and (4.78)–(4.80) we thus have

$$\frac{c}{k}|w_k(t)|_2^2 \le \mathbf{F}_k(t) \le C|w_k(t)|_{q_k}^{q_k}.$$
(4.82)

Note that  $q_k \leq 4$  for all  $k \geq 1$ . It follows from (4.82)–(4.82) and from the embedding  $W^{1,2}(\Omega)$  into  $L^6(\Omega)$  that

$$\frac{1}{k} \|w_k(t)\|_{1,2}^2 \ge \frac{c}{k} \mathbf{F}_k^{2/q_k}$$

with some constant c > 0. From (4.76) we thus obtain (note that  $2/q_k < 1$ ) that

$$\frac{\mathrm{d}}{\mathrm{d}t}(\mathbf{F}_{k}(t) + \mathbf{V}_{k}(t)) + \frac{c}{k}(\mathbf{F}_{k}(t) + \mathbf{V}_{k}(t))^{2/q_{k}} \leq C^{2k+2} + Ck^{5}\Gamma_{R}^{3}|\hat{w}_{k}(t)|_{3/2}^{2} \\
\leq Ck^{5}\Gamma_{R}^{3}\max\{C^{k}, |\hat{w}_{k}(t)|_{3/2}\}^{2} =: M_{R,k}(t).$$
(4.83)

Put  $M_{R,k}^* = \sup \operatorname{sup\,ess}_{t \in [T,2T]} M_{R,k}(t)$ . Then (4.83) is of the form

$$\dot{Y}(t) + bY^{\kappa}(t) \le M \tag{4.84}$$

with a *T*-periodic function  $Y = \mathbf{F}_k + \mathbf{V}_k$  and with constants b = c/k,  $\kappa = 2/q_k$ , and  $M = M_{R,k}^*$ . Let *H* be the Heaviside function H(s) = 1 for s > 0, H(s) = 0 for  $s \le 0$ . Multiplying (4.84) by  $H(Y^{\kappa}(t) - (M/b))$  we obtain

$$\frac{\mathrm{d}}{\mathrm{d}t} (Y(t) - (M/b)^{1/\kappa})^+ + (bY^{\kappa}(t) - M)^+ \le 0.$$

Integrating from T to 2T and using the fact that Y is periodic, we thus have

$$\int_{T}^{2T} (bY^{\kappa}(t) - M)^{+} \mathrm{d}t \le 0,$$

which is only possible if  $Y(t) \leq (M/b)^{1/\kappa}$  a.e. Hence, (4.83) yields that

$$\mathbf{F}_k(t) \le C(kM_{R,k}^*)^{q_k/2}.$$
 (4.85)

Referring again to (4.82) (note that  $q_k \leq 4$ ) we thus conclude that

$$\sup_{t \in [T,2T]} \exp \left| \hat{w}_k(t) \right|_2^2 \le \sup_{t \in [T,2T]} \exp \left| w_k(t) \right|_2^2 \le C k^{13} \Gamma_R^{3q_k/2} \max\{C^k, \sup_{t \in [T,2T]} \exp \left| \hat{w}_k(t) \right|_{3/2} \}^{q_k}.$$
(4.86)

Putting for  $q \ge 1$ 

$$\|\hat{w}_k\|_q^* = \sup_{t \in [T,2T]} |\hat{w}_k(t)|_q$$

we can reformulate (4.86) as

$$\left(\|\hat{w}_k\|_2^*\right)^2 \le Ck^{13} \Gamma_R^{3q_k/2} \left(\max\{C^k, \|\hat{w}_k\|_{3/2}^*\}\right)^{q_k} \tag{4.87}$$

with a constant C independent of k and R.

The Moser iteration technique will be applied to the new variable

$$w := 1 + \min\{p^2, R^2\}.$$
(4.88)

By (4.70), inequality (4.87) can be rewritten as

$$\left(\|w\|_{r_k}^*\right)^{r_k} \le Qk^{13}\Gamma_R^{3q_k/2} \max\{L^{r_k}, \left(\|w\|_{3r_k/4}^*\right)^{r_k/2}\}^{q_k}$$
(4.89)

with  $r_k = k + \gamma$  and with some constants  $Q \ge 1$ ,  $L \ge 1$  that we keep fixed from now on. We thus have

$$\|w\|_{r_k}^* \le (Qk^{13})^{1/r_k} \Gamma_R^{3q_k/2r_k} \max\{L, \|w\|_{3r_k/4}^*\}^{q_k/2}.$$
(4.90)

We are ready now to start the Moser iterations. We now choose k in (4.90) to be the sequence  $\{k_j\}$  for  $j \in \mathbb{N} \cup \{0\}$ 

$$k_j = 3\left(\frac{4}{3}\right)^j - \gamma,\tag{4.91}$$

and put

$$\rho_j = r_{k_j} = k_j + \gamma = 3\left(\frac{4}{3}\right)^j, \quad P_j = \max\{L, \|w\|_{\rho_j}^*\}, \quad \delta_j = \frac{q_{k_j}}{2} - 1 = \frac{\eta}{k_j + 1 - \eta}.$$

Then, by virtue of (4.81), the inequality (4.90) is of the form

$$P_j \le (\rho_j^{13}Q)^{1/\rho_j} \Gamma_R^{3(1+\delta_j)/\rho_j} P_{j-1}^{1+\delta_j} \text{ for } j = 1, 2, \dots$$
(4.92)

The logarithm applied to (4.92) yields

$$\log P_{j} - (1+\delta_{j}) \log P_{j-1} \leq \frac{1}{\rho_{j}} \left( \log(3^{13}Q) + 3(1+\delta_{j}) \log \Gamma_{R} + 13 \log\left(\frac{4}{3}\right)^{j} \right)$$

$$= \frac{1}{3} \left(\frac{3}{4}\right)^{j} \left( \log(3^{13}Q) + 3(1+\delta_{j}) \log \Gamma_{R} + 13j \log\frac{4}{3} \right),$$
(4.93)

or, equivalently, for  $j \in \mathbb{N}$ ,

$$\frac{\log P_j}{\prod_{i=0}^j (1+\delta_i)} - \frac{\log P_{j-1}}{\prod_{i=0}^{j-1} (1+\delta_i)} \le \frac{1}{3} \left(\frac{3}{4}\right)^j \frac{\log(3^{13}Q) + 3(1+\delta_j)\log\Gamma_R + 13j\log(4/3)}{\prod_{i=0}^j (1+\delta_i)} \le \frac{1}{3} \left(\frac{3}{4}\right)^j \frac{\log(3^{13}Q) + 3\log\Gamma_R + 13j\log(4/3)}{1+\delta_0}.$$
 (4.94)

The sequence on the right-hand side of (4.94) forms a convergent series, more precisely,

$$\sum_{j=1}^{\infty} \left(\frac{3}{4}\right)^j = 3, \quad \sum_{j=1}^{\infty} j \left(\frac{3}{4}\right)^j = 12.$$

Hence,

$$\sup_{j \ge 1} \frac{\log P_j}{\prod_{i=1}^j (1+\delta_i)} \le \log P_0 + \log(3^{13}Q) + 3\log \Gamma_R + 52\log\left(\frac{4}{3}\right).$$
(4.95)

We have by virtue of (4.60) that  $P_0 \leq C\Gamma_R^3$ . We thus have

$$\sup_{j \ge 1} \frac{\log P_j}{\prod_{i=1}^j (1+\delta_i)} \le 6 \log \Gamma_R + C \tag{4.96}$$

with a constant C > 0 independent of j and R. Note that

$$\delta_i = \frac{\eta}{k_i + 1 - \eta} \le \frac{1}{k_i + \gamma} = \frac{1}{3} \left(\frac{3}{4}\right)^i.$$

We thus have

$$\prod_{i=1}^{\infty} (1+\delta_i) = \exp\left(\sum_{i=1}^{\infty} \log(1+\delta_i)\right) \le \exp\left(\sum_{i=1}^{\infty} \delta_i\right) \le e,$$

hence, by definition (4.28) of  $\Gamma_R$ ,

$$\sup_{j \ge 1} \log P_j \le 6\gamma e \log(1 + R^2) + C^*$$
(4.97)

with a constant  $C^* > 0$  independent of j and R, that is,

$$\sup_{j \ge 1} P_j \le e^{C^*} (1 + R^2)^{\sigma} =: P^*(R).$$
(4.98)

with

 $\sigma = 6\gamma e < 1$ 

by Hypothesis (3.13). We see that the norms  $||w||_k^*$  are bounded by  $P^*(R)$  independent of k. We now easily prove that

$$w(x,t) \le P^*(R)$$
 for almost all  $(x,t) \in \Omega \times (T,2T)$ . (4.99)

Indeed, assume that there exist a > 0 and a set  $A \subset \Omega \times (T, 2T)$  such that  $w(x, t) \geq 0$ 

 $P^*(R) + a$  for a.e.  $(x, t) \in A$ . Then

$$(P^*(R) + a)^q |A| \le \iint_A w^q \mathrm{d}x \mathrm{d}t \le \int_T^{2T} \int_\Omega w^q \mathrm{d}x \mathrm{d}t \le T \sup_{t \in [T, 2T]} \int_\Omega w^q \mathrm{d}x \le TP^*(R)^q.$$
(4.100)

We conclude from (4.100) that

$$|A| \le T\left(\frac{P^*(R)}{P^*(R) + a}\right)^q$$

for all q > 1. This is only possible if |A| = 0. Hence,  $w(x, t) \leq P^*(R)$  a.e. in agreement with (4.99). We can summarize the above computations as follows.

**Proposition 4.4.1.** Let  $\bar{R} > 0$  be chosen such that  $1 + \bar{R}^2 > e^{C^*/(1-\sigma)}$  with  $C^*$  and  $\sigma$  as in (4.98). Let (u, p) be the solution to (4.24)–(4.25) given by Proposition 4.1.1 for  $R = \bar{R}$ . Then we have  $|p(x, t)| \leq \bar{R}$  for a. e.  $(x, t) \in \Omega \times (T, 2T)$ , and (u, p) is the solution to the system

$$\int_{T}^{2T} \int_{\Omega} \left( (\rho u_{tt} + \delta | u_t | u_t) \phi + (\mathbf{B}(x) \nabla_s u_t + \mathbf{A}(x) \nabla_s u + \varepsilon \mathbf{B}(x) \nabla_s u_{tt}) : \nabla_s \phi + \nabla p \ \phi \right) \lambda(t) dx dt + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) (\mathbf{C}(x) u + \mathbf{D}(x) u_t - g) \phi \lambda(t) ds(x) dt = 0,$$

$$\int_{T}^{2T} \int_{\Omega} \left( \left( (f(x, p) + G[p])_t + \delta | p | p \right) \psi + (\mu(x, p) \nabla p - u_t) \nabla \psi \right) \lambda(t) dx dt + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) (p - \bar{p}) \psi \lambda(t) ds(x) dt = 0$$

$$(4.102)$$

for all  $\phi \in X_3$ ,  $\psi \in X$ , and  $\lambda \in L^2_T$  with the regularity as in Proposition 4.1.1.

*Proof.* Assuming that for some (x,t) we have  $|p(x,t)| > \overline{R}$  would imply that  $w(x,t) = 1 + \min\{p^2(x,t), \overline{R}^2\} = 1 + \overline{R}^2$ . By virtue of (4.99) we have

$$1 + \bar{R}^2 = w(x,t) \le P^*(\bar{R}) = e^{C^*}(1 + \bar{R}^2)^{\sigma}$$

which contradicts the choice of  $\overline{R}$ . Hence, the solution satisfies the condition  $|p| \leq \overline{R}$  a.e. and the proof is complete.

We are now ready to finish the proof of Theorem 3.1.3.

#### Proof of Theorem 3.1.3.

Solutions to (4.101)-(4.102) satisfy, by virtue of (4.26), (4.29), (4.30), (4.36), (4.60) the

following estimates independent of  $\delta$  and  $\varepsilon$ 

$$\int_{T}^{2T} \int_{\Omega} \left( |u_{tt}|^{2} + \delta |u_{t}|^{3} + \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt} : \nabla_{s} u_{tt} + \mathbf{B}(x) \nabla_{s} u_{t} : \nabla_{s} u_{t} + \mathbf{A}(x) \nabla_{s} u : \nabla_{s} u \right) dxdt + \int_{T}^{2T} \int_{\partial\Omega} \beta(x) \left( \mathbf{D}(x) u_{t} \cdot u_{t} + \mathbf{C}(x) u \cdot u \right) ds(x) dt \leq C,$$
  
$$\sup_{(x,t)\in\Omega\times(T,2T)} |p(x,t)| + \int_{T}^{2T} \int_{\Omega} \left( \delta |p|^{3} + |p_{t}|^{2} + |\nabla p|^{2} \right) dxdt + \int_{T}^{2T} \int_{\partial\Omega} \alpha(x) p^{2} ds(x) \leq C.$$

The term  $\delta |p|p$  converges strongly to 0 in  $L^{\infty}(\Omega \times (T, 2T))$  as  $\delta \to 0$  due to the uniform upper bound for p. Similarly,  $u_t$  are bounded in  $L_T^4(L^4(\Omega))$  by Sobolev embeddings, hence  $\delta |u_t|u_t$  converge strongly to 0 in  $L_T^2(L^2(\Omega))$ . Furthermore, for  $\phi \in X_3$  and  $\lambda \in L_T^2$  we have

$$\begin{split} \int_{T}^{2T} &\int_{\Omega} \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt} \, \phi \lambda(t) \mathrm{d}x \mathrm{d}t \\ &\leq C \varepsilon^{1/2} \left( \int_{T}^{2T} \int_{\Omega} \varepsilon \mathbf{B}(x) \nabla_{s} u_{tt} : \nabla_{s} u_{tt} \mathrm{d}x \mathrm{d}t \right)^{1/2} \left( \int_{\Omega} |\phi(x)|^{2} \mathrm{d}x \right)^{1/2} \left( \int_{T}^{2T} |\lambda(t)|^{2} \mathrm{d}t \right)^{1/2} \\ &\leq C \varepsilon^{1/2} \left( \int_{\Omega} |\phi(x)|^{2} \mathrm{d}x \right)^{1/2} \left( \int_{T}^{2T} |\lambda(t)|^{2} \mathrm{d}t \right)^{1/2} . \end{split}$$

In the linear terms in (4.101)–(4.102), we pass to the limit as  $\delta \to 0$  and  $\varepsilon \to 0$  by weak convergence. In the nonlinear terms in (4.102), we use the compactness of the embedding of  $W_T^{1,2}(L^2(\Omega)) \cap L_T^2(X) \cap L^{\infty}(\Omega \times (T,2T))$  into  $L^q(\Omega; C_T)$  for every q > 1 to check that  $p = p^{\delta,\varepsilon}$  converge pointwise almost everywhere, and use the strong continuity of the operator G as well as the Lebesgue Dominated Convergence Theorem. Hence, for every fixed  $\phi \in X_3$ ,  $\psi \in X$ , and  $\lambda \in L_T^2$  we can pass to the limit as  $\delta \to 0$  and  $\varepsilon \to 0$  and conclude thus the proof of Theorem 3.1.3.

# Part III

# Obstacle problems applied to anisotropic materials

The aim of Part III is to focus on the analysis of higher differentiability results for solutions to a class of double phase obstacle problems in the scale of Besov spaces. In particular, this third part of the thesis contains the results published in [67, 68]. It is worth noticing that double phase functionals are a useful tool to study the behaviour of strongly anisotropic materials whose hardening properties are strongly dependent on the point and connected to the exponent ruling the growth of the gradient variable. The coefficient  $a(\cdot)$  (see (7.1)) regulates the mixture between two different materials, with pand q hardening, respectively (see, for instance, [121, 122]).

The obstacle problem appeared in the mathematical literature in the work of Stampacchia [111] in the special case  $\psi = \chi_E$  and related to the capacity of a subset  $E \Subset \Omega$ ; in an earlier independent work, Fichera [51] solved the first unilateral problem, the so-called *Signorini problem* in elastostatics.

It is usually observed that the regularity of solutions to the obstacle problems is influenced by the one of the obstacle; for example, for linear obstacle problems, obstacle and solutions have the same regularity [11, 17, 80]. This does not apply in the nonlinear setting, hence along the years, there have been intense research activities for the regularity of the obstacle problem in this direction. The regularity theory for obstacle problems driven by quasilinear operators of the *p*-Laplacian type started with the contributions of Duzaar and Fuchs [37], Duzaar [36], Chloe and Lewis [22] and Fuchs [57].

In the case of standard growth conditions, Eleuteri and Passarelli di Napoli [43] proved that an extra differentiability of integer or fractional order of the gradient of the obstacle transfers to the gradient of the solutions, provided the partial map  $x \mapsto D_{\xi} \tilde{F}(x,\xi)$  possesses a suitable differentiability property, where  $\tilde{F}$  is a general integrand independent of the w-variable.

Recently, it was proved in [61, 62] that the weak differentiability of integer order of the partial map  $x \mapsto D_{\xi} \tilde{F}(x,\xi)$  is a sufficient condition to prove that an extra differentiability of integer order of the gradient of the obstacle transfers to the gradient of the solutions to obstacle problems with p, q-growth conditions. This property was generalized also for fractional differentiability, connected to Besov spaces in [67].

The intermediate case of higher differentiability in the setting of variable exponents case has been carried out in the paper [54].

The regularity properties of local minimizers to double phase functionals have recently been investigated for unconstrained problems. In particular, we quote the work [26] by Colombo and Mingione where the functional H(x, Du) has been considered (see (7.3)), [8] by Baroni, Colombo and Mingione who studied the integrand defined in (7.2) and [27] by Coscia, who dealt with the functional defined by

$$\mathcal{F}(w,\Omega) := \int_{\Omega} b(x,w) [|Dw|^p + a(x)|Dw|^p \log(e+|Dw|)] \mathrm{d}x.$$

Furthermore, a higher fractional differentiability [120] and a Lipschitz continuity result [32] have been proved for solutions to double phase elliptic obstacle problems. We also recall that when referring to p, q-growth conditions, in order to ensure the regularity of minima, a smallness condition on the gap q/p > 1 is necessary (see, for instance, the counterexamples in [53, 63, 96]).

Part III is structured as follows. In Chapter 5 we recall some notation and preliminary results. In Chapter 6, we study of the higher differentiablity properties of solutions to (6.3) in case of p, q-growth conditions. We assume that both the gradient of the obstacle and the partial map  $x \mapsto \mathcal{A}(x,\xi)$  belong to a suitable Sobolev class of fractional order. The results contained in Chapter 6 are crucial to then prove the main result of Part III in Chapter 7, which is Theorem 7.0.1, namely, higher differentiability properties for a solution to a class of double phase functionals in the scale of Besov spaces.

## Chapter 5

### Notation and background

In what follows,  $B(x,r) = B_r(x) = \{y \in \mathbb{R}^n : |y-x| < r\}$  will denote the ball centered at x of radius r. We shall omit the dependence on the center and on the radius when no confusion arises. For a function  $u \in L^1(B)$ , the symbol

$$u_B := \int_B u(x) \mathrm{d}x = \frac{1}{|B|} \int_B u(x) \mathrm{d}x.$$

will denote the integral mean of the function u over the set B.

It is convenient to introduce an auxiliary function

$$V_d(\xi) = |\xi|^{\frac{d-2}{2}}\xi$$

defined for all  $\xi \in \mathbb{R}^n$ . One can easily check that

$$|\xi|^d = |V_d(\xi)|^2. \tag{5.1}$$

For the auxiliary function  $V_d$ , we recall the following estimate (see the proof of [64, Lemma 8.3]):

**Lemma 5.0.1.** Let  $1 < d < +\infty$ . There exists a constant c = c(n, d) > 0 such that

$$c^{-1}(|\xi|^2 + |\eta|^2)^{\frac{d-2}{2}} \le \frac{|V_d(\xi) - V_d(\eta)|^2}{|\xi - \eta|^2} \le c(|\xi|^2 + |\eta|^2)^{\frac{d-2}{2}}$$

for any  $\xi, \eta \in \mathbb{R}^n, \ \xi \neq \eta$ .

Now we state a well-known iteration lemma (see [64, Lemma 6.1] for the proof).

**Lemma 5.0.2.** Let  $\Phi : [\frac{R}{2}, R] \to \mathbb{R}$  be a bounded nonnegative function, where R > 0. Assume that for all  $\frac{R}{2} \leq r < s \leq R$  it holds

$$\Phi(r) \le \theta \Phi(s) + A + \frac{B}{(s-r)^2} + \frac{C}{(s-r)^{\gamma}}$$

where  $\theta \in (0,1)$ , A, B,  $C \ge 0$  and  $\gamma > 0$  are constants. Then there exists a constant  $c = c(\theta, \gamma)$  such that

$$\Phi\left(\frac{R}{2}\right) \le c\left(A + \frac{B}{R^2} + \frac{C}{R^{\gamma}}\right).$$

### 5.1 Besov-Lipschitz spaces

Let  $v : \mathbb{R}^n \to \mathbb{R}$  be a function. As in [70, Section 2.5.12], given  $0 < \alpha < 1$  and  $1 \le p, q < \infty$ , we say that v belongs to the Besov space  $B_{p,q}^{\alpha}(\mathbb{R}^n)$  if  $v \in L^p(\mathbb{R}^n)$  and

$$||v||_{B^{\alpha}_{p,q}(\mathbb{R}^n)} = ||v||_{L^p(\mathbb{R}^n)} + [v]_{B^{\alpha}_{p,q}(\mathbb{R}^n)} < \infty,$$

where

$$[v]_{B^{\alpha}_{p,q}(\mathbb{R}^n)} = \left(\int_{\mathbb{R}^n} \left(\int_{\mathbb{R}^n} \frac{|v(x+h) - v(x)|^p}{|h|^{\alpha p}} \mathrm{d}x\right)^{\frac{q}{p}} \frac{\mathrm{d}h}{|h|^n}\right)^{\frac{1}{q}} < \infty.$$

Equivalently, we could simply say that  $v \in L^p(\mathbb{R}^n)$  and  $\frac{\tau_h v}{|h|^{\alpha}} \in L^q(\frac{dh}{|h|^n}; L^p(\mathbb{R}^n))$ . As usual, if one simply integrates for  $h \in B(0, \delta)$  for a fixed  $\delta > 0$  then an equivalent norm is obtained, because

$$\left(\int_{\{|h|\geq\delta\}} \left(\int_{\mathbb{R}^n} \frac{|v(x+h)-v(x)|^p}{|h|^{\alpha p}} \mathrm{d}x\right)^{\frac{q}{p}} \frac{\mathrm{d}h}{|h|^n}\right)^{\frac{1}{q}} \leq c(n,\alpha,p,q,\delta) \|v\|_{L^p(\mathbb{R}^n)}.$$

Similarly, we say that  $v \in B_{p,\infty}^{\alpha}(\mathbb{R}^n)$  if  $v \in L^p(\mathbb{R}^n)$  and

$$[v]_{B^{\alpha}_{p,\infty}(\mathbb{R}^n)} = \sup_{h \in \mathbb{R}^n} \left( \int_{\mathbb{R}^n} \frac{|v(x+h) - v(x)|^p}{|h|^{\alpha p}} \mathrm{d}x \right)^{\frac{1}{p}} < \infty.$$

Again, one can simply take supremum over  $|h| \leq \delta$  and obtain an equivalent norm. By construction,  $B_{p,q}^{\alpha}(\mathbb{R}^n) \subset L^p(\mathbb{R}^n)$ . One also has the following version of Sobolev embeddings (a proof can be found at [70, Proposition 7.12]).

**Lemma 5.1.1.** Suppose that  $0 < \alpha < 1$ .

(a) If  $1 and <math>1 \le q \le p_{\alpha}^* = \frac{np}{n-\alpha p}$ , then there is a continuous embedding  $B_{p,q}^{\alpha}(\mathbb{R}^n) \subset L^p(\mathbb{R}^n)$ .

(b) If  $p = \frac{n}{\alpha}$  and  $1 \leq q \leq \infty$ , then there is a continuous embedding  $B_{p,q}^{\alpha}(\mathbb{R}^n) \subset BMO(\mathbb{R}^n)$ , where BMO denotes the space of functions with bounded mean oscillations [64, Chapter 2].

For further needs, we recall the following inclusions ([70, Proposition 7.10 and Formula (7.35)]).

**Lemma 5.1.2.** Suppose that  $0 < \beta < \alpha < 1$ . (a) If  $1 and <math>1 \le q \le r \le \infty$ , then  $B_{p,q}^{\alpha}(\mathbb{R}^n) \subset B_{p,r}^{\alpha}(\mathbb{R}^n)$ . (b) If  $1 and <math>1 \le q, r \le \infty$ , then  $B_{p,q}^{\alpha}(\mathbb{R}^n) \subset B_{p,r}^{\beta}(\mathbb{R}^n)$ . (c) If  $1 \le q \le \infty$ , then  $B_{\frac{n}{\alpha},q}^{\alpha}(\mathbb{R}^n) \subset B_{\frac{n}{\beta},q}^{\beta}(\mathbb{R}^n)$ .

Combining Lemmas 5.1.1 and 5.1.2, we get the following Sobolev type embedding theorem for Besov spaces  $B_{p,\infty}^{\alpha}(\mathbb{R}^n)$ .

**Lemma 5.1.3.** Suppose that  $0 < \alpha < 1$  and  $1 . There is a continuous embedding <math>B^{\alpha}_{p,\infty}(\mathbb{R}^n) \subset L^{p^*_{\beta}}(\mathbb{R}^n)$ , for every  $0 < \beta < \alpha$ . Moreover, the following local estimate

$$\|F\|_{L^{\frac{np}{n-\beta p}}(B_{\varrho})} \le c(R-\varrho)^{-\delta}(\|F\|_{L^{p}(B_{R})} + [F]_{B^{\alpha}_{p,q}(B_{R})})$$
(5.2)

holds for every ball  $B_{\varrho} \subset B_R$ , with  $c = c(n, p, q, \alpha, \beta)$  and  $\delta = \delta(n, p, q)$ .

Given a domain  $\Omega \subset \mathbb{R}^n$ , we say that v belongs to the local Besov space  $B^{\alpha}_{p,q,loc}$  if  $\varphi \ v \in B^{\alpha}_{p,q}(\mathbb{R}^n)$  whenever  $\varphi \in \mathcal{C}^{\infty}_c(\Omega)$ . It is worth noticing that one can prove suitable version of Lemma 5.1.1 and Lemma 5.1.2, by using local Besov spaces.

The following Lemma and its proof can be found in [6].

**Lemma 5.1.4.** A function  $v \in L^p_{loc}(\Omega)$  belongs to the local Besov space  $B^{\alpha}_{p,q,loc}$  if, and only if,

$$\left\|\frac{\tau_h v}{|h|^{\alpha}}\right\|_{L^q\left(\frac{dh}{|h|^n};L^p(B)\right)} < \infty$$

for any ball  $B \subset 2B \subset \Omega$  with radius  $r_B$ . Here the measure  $\frac{dh}{|h|^n}$  is restricted to the ball  $B(0, r_B)$  on the h-space.

It is known that Besov-Lipschitz spaces of fractional order  $\alpha \in (0, 1)$  can be characterized in pointwise terms. Given a measurable function  $v : \mathbb{R}^n \to \mathbb{R}$ , a fractional  $\alpha$ -Hajlasz gradient for v is a sequence  $\{g_k\}_k$  of measurable, non-negative functions  $g_k : \mathbb{R}^n \to \mathbb{R}$ , together with a null set  $N \subset \mathbb{R}^n$ , such that the inequality

$$|v(x) - v(y)| \le (g_k(x) + g_k(y))|x - y|^{\alpha}$$

holds whenever  $k \in \mathbb{Z}$  and  $x, y \in \mathbb{R}^n \setminus N$  are such that  $2^{-k} \leq |x - y| < 2^{-k+1}$ . We say that  $\{g_k\}_k \in l^q(\mathbb{Z}; L^p(\mathbb{R}^n))$  if

$$\|\{g_k\}_k\|_{l^q(L^p)} = \left(\sum_{k\in\mathbb{Z}} \|g_k\|_{L^p(\mathbb{R}^n)}^q\right)^{\frac{1}{q}} < \infty$$

The following result was proved in [81].

**Theorem 5.1.5.** Let  $0 < \alpha < 1$ ,  $1 \le p < \infty$  and  $1 \le q \le \infty$ . Let  $v \in L^p(\mathbb{R}^n)$ . One has  $v \in B_{p,q}^{\alpha}(\mathbb{R}^n)$  if, and only if, there exists a fractional  $\alpha$ -Hajlasz gradient  $\{g_k\}_k \in l^q(\mathbb{Z}; L^p(\mathbb{R}^n))$  for v. Moreover,

$$||v||_{B^{\alpha}_{p,q}(\mathbb{R}^n)} \simeq \inf ||\{g_k\}_k||_{l^q(L^p)},$$

where the infimum runs over all possible fractional  $\alpha$ -Hajlasz gradients for v.

### 5.2 Difference quotient

We recall some properties of the finite difference quotient operator that will be needed in the sequel. Let us recall that, for every function  $F : \mathbb{R}^n \to \mathbb{R}$  the finite difference operator is defined by

$$\tau_{s,h}F(x) = F(x + he_s) - F(x)$$

where  $h \in \mathbb{R}^n$ ,  $e_s$  is the unit vector in the  $x_s$  direction and  $s \in \{1, ..., n\}$ .

We start with the description of some elementary properties that can be found, for example, in [64].

**Proposition 5.2.1.** Let F and G be two functions such that  $F, G \in W^{1,p}(\Omega)$ , with  $p \ge 1$ , and let us consider the set

$$\Omega_{|h|} = \{ x \in \Omega : dist(x, \partial \Omega) > |h| \}.$$

Then

(i)  $\tau_h F \in W^{1,p}(\Omega_{|h|})$  and

$$D_i(\tau_h F) = \tau_h(D_i F).$$

(ii) If at least one of the functions F or G has support contained in  $\Omega_{|h|}$ , then

$$\int_{\Omega} F\tau_h G dx = \int_{\Omega} G\tau_{-h} F dx.$$

(iii) We have

$$\tau_h(FG)(x) = F(x+h)\tau_h G(x) + G(x)\tau_h F(x).$$

The next result about finite difference operator is a kind of integral version of Lagrange Theorem.

Lemma 5.2.2. If  $0 < \rho < R$ ,  $|h| < \frac{R-\rho}{2}$ , 1 and <math>F,  $DF \in L^p(B_R)$ , then  $\int_{B_\rho} |\tau_h F(x)|^p dx \le c(n,p) |h|^p \int_{B_R} |DF(x)|^p dx.$  Moreover,

$$\int_{B_{\rho}} |F(x+h)|^p dx \le \int_{B_R} |F(x)|^p dx.$$

We conclude this subsection recalling the following Lemma (see [85]), which can be seen as a consequence of Lemmas 5.1.1 and 5.1.2.

**Lemma 5.2.3.** Let  $F \in L^2(B_R)$ . Suppose that there exist  $\rho \in (0, R)$ ,  $0 < \alpha < 1$  and M > 0 such that

$$\sum_{s=1}^{n} \int_{B_{\rho}} |\tau_{s,h} F(x)|^2 dx \le M^2 |h|^{2\alpha},$$

for every h such that  $h < \frac{R-\rho}{2}$ . Then  $F \in L^{\frac{2n}{n-2\beta}}(B_{\rho})$  for every  $\beta \in (0, \alpha)$  and

$$\|F\|_{L^{\frac{2n}{n-2\beta}}(B_{\rho})} \le c(M + \|F\|_{L^{2}(B_{R})}),$$

with  $c = c(n, N, R, \rho, \alpha, \beta)$ .

### 5.3 Preliminary results on standard growth conditions

For sake of clarity, we would like to recall the following regularity result (see [43] for the proof), which will be used in order to prove Theorem 6.0.1.

In the case of a regularity of the type  $B_{p,\infty}^{\alpha}$ , which is the weakest one in the scale of Besov spaces, both on the coefficients and on the gradient of the obstacle, we have the following

**Theorem 5.3.1.** Assume that  $\mathcal{A}(x,\xi)$  satisfies (A1)-(A3) for an exponent  $2 \leq p = q < \frac{n}{\alpha}$ and let  $u \in \mathcal{K}_{\psi}(\Omega)$  be the solution to the obstacle problem (6.3). If there exists a nonnegative function  $k \in L^{\frac{n}{\alpha}}_{loc}(\Omega)$  such that

$$|\mathcal{A}(x,\xi) - \mathcal{A}(y,\xi)| \le |x - y|^{\alpha} (k(x) + k(y)) (\mu^2 + |\xi|^2)^{\frac{p-1}{2}},$$

for a.e.  $x, y \in \Omega$  and for every  $\xi \in \mathbb{R}^n$ , then the following implication

$$D\psi \in B^{\gamma}_{p,\infty,loc}(\Omega) \Rightarrow (\mu^2 + |Du|^2)^{\frac{p-2}{4}} Du \in B^{\alpha}_{2,\infty,loc}(\Omega),$$

holds, provided  $0 < \alpha < \gamma < 1$ .

## Chapter 6

# Higher differentiability in the case of lagrangians $\tilde{F}(x, Du)$

In this chapter we study of the higher fractional differentiability properties of the gradient of solutions  $u \in W^{1,p}(\Omega)$  to obstacle problems of the form

$$\min\left\{\int_{\Omega} \tilde{F}(x, Dw) dx : w \in \mathcal{K}_{\psi}(\Omega)\right\},\tag{6.1}$$

where  $\Omega$  is a bounded open set of  $\mathbb{R}^n$ ,  $n \geq 2$ . The function  $\psi : \Omega \to [-\infty, +\infty)$ , called *obstacle*, belongs to the Sobolev class  $W^{1,p}(\Omega)$  and the class  $\mathcal{K}_{\psi}(\Omega)$  is defined as follows

$$\mathcal{K}_{\psi}(\Omega) = \{ w \in W^{1,p}(\Omega) : w \ge \psi \text{ a.e. in } \Omega \}.$$
(6.2)

Note that the set  $\mathcal{K}_{\psi}(\Omega)$  is not empty since  $\psi \in \mathcal{K}_{\psi}(\Omega)$ .

We here assume that  $\tilde{F}: \Omega \times \mathbb{R}^n \to [0, +\infty)$  is a Carathéodory function such that there exists a function  $\bar{F}: \Omega \times [0, +\infty) \to [0, +\infty)$  satisfying the following equality

$$\tilde{F}(x,\xi) = \bar{F}(x,|\xi|) \tag{F1}$$

for a.e.  $x \in \Omega$  and every  $\xi \in \mathbb{R}^n$ .

Moreover, we also assume that there exist positive constants  $\tilde{\nu}$ ,  $\tilde{L}$ ,  $\tilde{l}$ , exponents  $2 \leq p < q < +\infty$  and a parameter  $\mu \in [0, 1]$ , that will allow us to consider in our analysis both the degenerate and the non-degenerate situation, such that the following assumptions are satisfied:

$$\frac{1}{\tilde{l}}(|\xi|^2 - \mu^2)^{\frac{p}{2}} \le \tilde{F}(x,\xi) \le \tilde{l}(\mu^2 + |\xi|^2)^{\frac{q}{2}}$$
(F2)

$$\langle D_{\xi\xi}\tilde{F}(x,\xi)\lambda,\lambda\rangle \ge \tilde{\nu}(\mu^2 + |\xi|^2)^{\frac{p-2}{2}}|\lambda|^2$$
 (F3)
$$D_{\xi\xi}\tilde{F}(x,\xi)| \le \tilde{L}(\mu^2 + |\xi|^2)^{\frac{q-2}{2}}$$
 (F4)

for a.e.  $x, y \in \Omega$  and every  $\xi \in \mathbb{R}^n$ .

Very recently, in [31] it has been proved that ( $\tilde{F}3$ ) and ( $\tilde{F}4$ ) imply ( $\tilde{F}2$ ), i.e. if p < q, the functional  $\tilde{F}$  has non-standard growth conditions of p, q-type, as initially defined and studied by Marcellini [94, 95, 96].

We remark that assumption (F1) is known in the literature as Uhlenbeck structure and it was showed in [114] that it prevents the irregularity phenomenon in problems with non-standard growth.

We say that function  $\tilde{F}$  satisfies assumption ( $\tilde{F}5$ ) if there exist a non-negative function  $k \in L^r_{loc}(\Omega)$ , with  $r > \frac{n}{\alpha}$  and  $0 < \alpha < 1$ , such that

$$|D_{\xi}\tilde{F}(x,\xi) - D_{\xi}\tilde{F}(y,\xi)| \le |x-y|^{\alpha}(k(x)+k(y))(\mu^{2}+|\xi|^{2})^{\frac{q-1}{2}}$$
(F5)

for a.e.  $x, y \in \Omega$  and every  $\xi \in \mathbb{R}^n$ .

On the other hand, we say that assumption (F6) is satisfied if there exists a sequence of measurable non-negative functions  $g_k \in L^r_{loc}(\Omega)$  such that

$$\sum_{k=1}^{\infty} \|g_k\|_{L^r(\Omega)}^{\sigma} < \infty,$$

and at the same time

$$|D_{\xi}\tilde{F}(x,\xi) - D_{\xi}\tilde{F}(y,\xi)| \le |x - y|^{\alpha}(g_k(x) + g_k(y))(\mu^2 + |\xi|^2)^{\frac{q-1}{2}}$$
(F6)

for a.e.  $x, y \in \Omega$  such that  $2^{-k} \operatorname{diam}(\Omega) \le |x - y| < 2^{-k+1} \operatorname{diam}(\Omega)$  and for every  $\xi \in \mathbb{R}^n$ .

It is worth observing that, in the case of standard growth conditions, i.e. p = q,  $u \in W^{1,p}(\Omega)$  is a solution to the obstacle problem in  $\mathcal{K}_{\psi}(\Omega)$  if, and only if,  $u \in \mathcal{K}_{\psi}(\Omega)$ solves the variational inequality

$$\int_{\Omega} \langle \mathcal{A}(x, Du), D(\varphi - u) \rangle dx \ge 0$$
(6.3)

for all  $\varphi \in \mathcal{K}_{\psi}(\Omega)$ , where we set

$$\mathcal{A}(x,\xi) = D_{\xi}\tilde{F}(x,\xi). \tag{6.4}$$

This equivalence has been proved successfully in the case non-standard growth conditions by Eleuteri and Passarelli di Napoli in [45]. From conditions ( $\tilde{F}2$ )–( $\tilde{F}4$ ), we deduce the existence of positive constants  $\nu$ , L, l such that the following *p*-ellipticity and *q*-growth conditions are satisfied by the map  $\mathcal{A}$ :

$$|\mathcal{A}(x,\xi)| \le l(\mu^2 + |\xi|^2)^{\frac{q-1}{2}}$$
(A1)

$$\langle \mathcal{A}(x,\xi) - \mathcal{A}(x,\eta), \xi - \eta \rangle \ge \nu |\xi - \eta|^2 (\mu^2 + |\xi|^2 + |\eta|^2)^{\frac{p-2}{2}}$$
 (A2)

$$|\mathcal{A}(x,\xi) - \mathcal{A}(x,\eta)| \le L|\xi - \eta|(\mu^2 + |\xi|^2 + |\eta|^2)^{\frac{q-2}{2}}$$
(A3)

for a.e.  $x, y \in \Omega$ , for every  $\xi, \eta \in \mathbb{R}^n$ , where we recall that  $0 < \alpha < 1$ .

Furthermore, if condition ( $\tilde{F}5$ ) or ( $\tilde{F}6$ ) holds, then  $\mathcal{A}$  satisfies assumptions (A4) or (A5), respectively, that is

$$|\mathcal{A}(x,\xi) - \mathcal{A}(y,\xi)| \le |x - y|^{\alpha} (k(x) + k(y))(\mu^2 + |\xi|^2)^{\frac{q-1}{2}}$$
(A4)

for a.e.  $x, y \in \Omega$  and every  $\xi \in \mathbb{R}^n$ , or

$$|\mathcal{A}(x,\xi) - \mathcal{A}(y,\xi)| \le |x-y|^{\alpha} (g_k(x) + g_k(y))(\mu^2 + |\xi|^2)^{\frac{q-1}{2}}$$
(A5)

for a.e.  $x, y \in \Omega$  such that  $2^{-k} \operatorname{diam}(\Omega) \le |x - y| < 2^{-k+1} \operatorname{diam}(\Omega)$  and for every  $\xi \in \mathbb{R}^n$ .

Our analysis comes from the fact that the regularity of the solutions to the obstacle problem (6.3) is strictly connected to the analysis of the regularity of the solutions to partial differential equations of the form

$$\operatorname{div} D_{\xi} \tilde{F}(x, Du) = \operatorname{div} D_{\xi} \tilde{F}(x, D\psi),$$

whose higher differentiability properties have been widely investigated (see for example [6, 24, 65, 106, 107]). We also notice that previous regularity results concerning local minimizers of integral functionals of the Calculus of Variations, under the assumption (A4), have been obtained by Kristensen and Mingione [85].

The main result of this chapter is a higher differentiability result, which can be seen as an extension of the [43] (see Theorem 5.3.1 in Section 5.3) to the case of functionals with p, q-growth.

**Theorem 6.0.1.** Let  $\mathcal{A}(x,\xi)$  satisfy (A1)-(A4) for exponents  $2 \leq p < \frac{n}{\alpha} < r$ , p < q such

that

$$\frac{q}{p} < 1 + \frac{\alpha}{n} - \frac{1}{r}.\tag{6.5}$$

Let  $u \in \mathcal{K}_{\psi}(\Omega)$  be the solution to the obstacle problem (6.3). Then we have

$$D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega) \Rightarrow (\mu^2 + |Du|^2)^{\frac{p-2}{4}} Du \in B^{\alpha}_{2,\infty,loc}(\Omega), \tag{6.6}$$

provided  $0 < \alpha < \gamma < 1$ .

This Chapter is structured as follows. In order to prove the main result, Theorem 6.0.1, the strategy is to establish the a priori estimate for an approximating solution and then pass to the limit in the approximating problem. Therefore, we present our approximation results in Section 6.1, namely we are able to prove the existence of a sequence of functions with p-growth conditions that monotonically converges to our initial problems. Then in Section 6.2, are able to prove our result. Namely, we derive the a priori estimates in Section 6.2.1 for an approximating problem satisfying standard growth conditions. Then, in Section 6.2.2 we exploit the results of Sections 6.1 and 6.2.1 and using compactness, strictly convexity and weak lower semi-continuity of functional  $\tilde{F}$ , we are able to prove Theorem 6.0.1.

# 6.1 Approximation results

We here collect some results which will be used to prove the passage to the limit in Theorems 6.0.1.

We first recall the following Theorem, whose complete version can be found in [30] and which will be used to prove Lemma 6.1.2.

**Theorem 6.1.1.** Let  $\tilde{F} : \Omega \times \mathbb{R}^n \to [0, +\infty), \tilde{F} = \tilde{F}(x, \xi)$ , be a Carathéodory function. Then, assumptions ( $\tilde{F}2$ ) and ( $\tilde{F}3$ ) imply that there exist  $c_0(p, q, \nu, R, l, L), c_1(p, \nu) > 0$ and a Carathéodory function  $g : \Omega \times \mathbb{R}^n \to [-c_0, +\infty)$  s.t. for a.e.  $x \in \Omega$  and every  $\xi \in \mathbb{R}^n$ ,

$$\tilde{F}(x,\xi) = c_1(\mu^2 + |\xi|^2)^{\frac{p}{2}} + g(x,\xi).$$

In the next lemma, we adapt a well known approximation result, which can be found in [30], to the case when the map  $x \mapsto D_{\xi} \tilde{F}(x,\xi)$  has a Besov regularity.

**Lemma 6.1.2.** Let  $\tilde{F} : \Omega \times \mathbb{R}^n \to [0, +\infty), \tilde{F} = \tilde{F}(x,\xi)$ , be a Carathéodory function, convex with respect to  $\xi$ , satisfying assumptions ( $\tilde{F}1$ ), ( $\tilde{F}2$ ), ( $\tilde{F}3$ ) and ( $\tilde{F}5$ ). Then there exists a sequence ( $\tilde{F}_j$ ) of Carathéodory functions  $\tilde{F}_j : \Omega \times \mathbb{R}^n \to [0, +\infty)$ , convex with respect to the last variable, monotonically convergent to  $\tilde{F}$ , such that

- (i) for a.e.  $x \in \Omega$  and every  $\xi \in \mathbb{R}^n$ ,  $\tilde{F}_j(x,\xi) = \bar{F}_j(x,|\xi|)$ ,
- (ii) for a.e.  $x \in \Omega$ , for every  $\xi \in \mathbb{R}^n$  and for every j,  $\tilde{F}_j(x,\xi) \leq \tilde{F}_{j+1}(x,\xi) \leq \tilde{F}(x,\xi)$ ,
- (iii) for a.e.  $x \in \Omega$  and every  $\xi \in \mathbb{R}^n$ , we have  $\langle D_{\xi\xi}\tilde{F}_j(x,\xi)\lambda,\lambda\rangle \geq \bar{\nu}(\mu^2 + |\xi|^2)^{\frac{p-2}{2}}|\lambda|^2$ , with  $\bar{\nu}$  depending only on p and  $\nu$ ,
- (iv) for a.e.  $x \in \Omega$  and for every  $\xi \in \mathbb{R}^n$ , there exist  $L_1$ , independent of j, and  $\overline{L}_1$ , depending on j, such that

$$1/L_1(\mu + |\xi|)^p \le \tilde{F}_j(x,\xi) \le L_1(\mu + |\xi|)^q,$$
  
$$\tilde{F}_j(x,\xi) \le \bar{L}_1(j)(\mu + |\xi|)^p,$$

(v) there exists a constant C(j) > 0 such that

$$\begin{aligned} |D_{\xi}\tilde{F}_{j}(x,\xi) - D_{\xi}\tilde{F}_{j}(y,\xi)| &\leq |x-y|^{\alpha}(k(x)+k(y))(\mu^{2}+|\xi|^{2})^{\frac{q-1}{2}},\\ |D_{\xi}\tilde{F}_{j}(x,\xi) - D_{\xi}\tilde{F}_{j}(y,\xi)| &\leq C(j)|x-y|^{\alpha}(k(x)+k(y))(\mu^{2}+|\xi|^{2})^{\frac{p-1}{2}} \end{aligned}$$

for a.e.  $x, y \in \Omega$  and for every  $\xi \in \mathbb{R}^n$ .

*Proof.* According to Theorem 6.1.1, which holds under hypotheses (F2) and (F3), there exist the positive constants  $c_0 = c_0(p, q, \nu, R, l, L)$  and  $c_1 = c_1(p, \nu)$  and a function  $g : \Omega \times \mathbb{R}^n \to [-c_0, +\infty)$  s.t.

$$\tilde{F}(x,\xi) = c_1(\mu^2 + |\xi|^2)^{\frac{p}{2}} + g(x,\xi)$$
(6.7)

with g convex. Moreover there exists  $\tilde{g}: \Omega \times [0, +\infty) \to [-c_0, +\infty)$  s.t.  $\tilde{g}(x, |\xi|) = g(x, \xi)$ for any  $\xi \in \mathbb{R}^n$ . Since  $n \ge 2$ , for a.e.  $x \in \Omega, t \mapsto \tilde{g}(x, t)$  is convex and increasing. For any  $j \in \mathbb{N}$ , we might then define  $\tilde{g}_j: \Omega \times [0, +\infty) \to [-c_0, +\infty)$  as

$$\begin{split} \tilde{g}_j(x,t) &= \tilde{g}(x,t) \qquad \forall (x,t) \in \Omega \times [0,j], \\ \tilde{g}_j(x,t) &= \tilde{g}(x,j) + D_t \tilde{g}(x,j)(t-j) \qquad \forall (x,t) \in \Omega \times (j,\infty) \end{split}$$

We notice that, by definition, for a.e.  $x \in \Omega, t \mapsto \tilde{g}_j(x,t)$  is convex and increasing in  $[0, +\infty)$  and  $\tilde{g}_j(x,t) \leq \tilde{g}_{j+1}(x,t) \leq \tilde{g}(x,t)$ . Combining assumption ( $\tilde{F}2$ ), the definition of  $\tilde{g}_j(x,t)$  and (6.7), we infer

$$\tilde{g}_j(x,t) \le l(\mu+t)^q, 
\tilde{g}_j(x,t) \le c(q,l,j)(\mu+t)^p.$$
(6.8)

We now want to show that  $D_t \tilde{g}_j$  has a ( $\tilde{F}5$ )-type growth. It is easy to see that  $D_t \tilde{g}_j(x,t) = D_t \tilde{g}(x,j)$  for  $t \ge j$ . In particular, assumption ( $\tilde{F}5$ ) yields  $|D_t \tilde{g}(x,j) - D_t \tilde{g}(y,j)| \le |x - y|^{\alpha} (k(x) + k(y))(\mu + j)^{q-1}$ . Hence, for a.e.  $x \in \Omega$  and every t > 0,

$$|D_t \tilde{g}(x,t) - D_t \tilde{g}(y,t)| \le |x - y|^{\alpha} (k(x) + k(y))(\mu + t)^{q-1}.$$
(6.9)

Moreover, for  $t \leq j$ , according to (6.7) and (6.9), we obtain

$$\begin{aligned} |D_t \tilde{g}(x,t) - D_t \tilde{g}(y,t)| &\leq |x-y|^{\alpha} (k(x) + k(y))(\mu+t)^{p-1} (\mu+t)^{q-p} \\ &\leq |x-y|^{\alpha} (k(x) + k(y))(\mu+t)^{p-1} (\mu+j)^{q-p} \\ &\leq c(j)|x-y|^{\alpha} (k(x) + k(y))(\mu+t)^{p-1}. \end{aligned}$$

On the other hand, in the same way, for t > j, we get

$$|D_t \tilde{g}(x,t) - D_t \tilde{g}(y,t)| \leq |x-y|^{\alpha} (k(x) + k(y))(\mu+j)^{p-1} (\mu+j)^{q-p}$$
  
$$\leq |x-y|^{\alpha} (k(x) + k(y))(\mu+t)^{p-1} (\mu+j)^{q-p}$$
  
$$\leq c(j)|x-y|^{\alpha} (k(x) + k(y))(\mu+t)^{p-1}.$$

Eventually, for any j, we define  $g_j: \Omega \times \mathbb{R}^n \to [-c_0, +\infty)$  as

$$g_j(x,\xi) = \tilde{g}_j(x,|\xi|).$$

Statements (i), (ii), (iii), (v) directly follow by setting  $\tilde{F}_j : \Omega \times \mathbb{R}^n \to [0, +\infty)$ 

$$\tilde{F}_j(x,\xi) := c_1(\mu^2 + |\xi|^2)^{\frac{p}{2}} + g_j(x,\xi).$$

Property (*iv*) is obtained combining (6.7) with (6.8) and the definition of  $\tilde{F}_j$ .

*Remark* 2. It is worth noting that an analogous version of Lemma 6.1.2 can be proved similarly, supposing ( $\tilde{F}6$ ) instead of ( $\tilde{F}5$ ). In particular, statement (v) would change as follows.

(v) There exists a constant C(j) > 0 such that

$$|D_{\xi}\tilde{F}_{j}(x,\xi) - D_{\xi}\tilde{F}_{j}(y,\xi)| \leq |x-y|^{\alpha}(g_{k}(x) + g_{k}(y))(\mu^{2} + |\xi|^{2})^{\frac{q-1}{2}}, |D_{\xi}\tilde{F}_{j}(x,\xi) - D_{\xi}\tilde{F}_{j}(y,\xi)| \leq C(j)|x-y|^{\alpha}(g_{k}(x) + g_{k}(y))(\mu^{2} + |\xi|^{2})^{\frac{p-1}{2}}$$

for a.e.  $x, y \in \Omega$  such that  $2^{-k} \operatorname{diam}(\Omega) \leq |x - y| < 2^{-k+1} \operatorname{diam}(\Omega)$  and for every  $\xi \in \mathbb{R}^n$ .

# 6.2 Proof of Theorem 6.0.1

In order to prove Theorem 6.0.1, we use the following strategy. In Section 6.2.1, we derive a suitable a priori estimate for minimizers of obstacle problems with p-growth conditions. Then, in Section 6.2.2, we conclude showing that the a priori estimate is preserved when passing to the limit.

## 6.2.1 A priori estimate

Let us consider

$$\min\left\{\int_{\Omega} \tilde{F}_j(x, Dw) dx : w \in \mathcal{K}_{\psi}(\Omega)\right\},\tag{6.10}$$

where  $\tilde{F}_j : \Omega \times \mathbb{R}^n \to [0, +\infty), \ \tilde{F}_j = \tilde{F}_j(x, \xi)$ , was set in Lemma 6.1.2 and  $\mathcal{K}_{\psi}(\Omega)$  was defined in (6.2).

Setting

$$\mathcal{A}_j(x,\xi) = D_\xi \tilde{F}_j(x,\xi),$$

one can easily check that  $\mathcal{A}_j$  satisfies (A1)–(A4) and the following assumptions:

$$|\mathcal{A}_j(x,\xi)| \le l_1(j)(\mu^2 + |\xi|^2)^{\frac{p-1}{2}}$$
(6.11)

$$|\mathcal{A}_j(x,\xi) - \mathcal{A}_j(x,\eta)| \le L_1(j)|\xi - \eta|(\mu^2 + |\xi|^2 + |\eta|^2)^{\frac{p-2}{2}}$$
(6.12)

$$|\mathcal{A}_{j}(x,\xi) - \mathcal{A}_{j}(y,\xi)| \le \Theta(j)|x - y|^{\alpha}(k(x) + k(y))(\mu^{2} + |\xi|^{2})^{\frac{p-1}{2}}$$
(6.13)

for a.e.  $x, y \in \Omega$ , for every  $\xi, \eta \in \mathbb{R}^n$ . It is well known that  $u_j \in \mathcal{K}_{\psi}(\Omega)$  is a minimizer of problem (6.10) if, and only if, the following variational inequality holds

$$\int_{\Omega} \langle \mathcal{A}_j(x, Du_j), D(\varphi - u_j) \rangle dx \ge 0, \quad \forall \varphi \in \mathcal{K}_{\psi}(\Omega).$$
(6.14)

The following result holds:

**Theorem 6.2.1.** Let  $\mathcal{A}_j(x,\xi)$  satisfy (A1)-(A4) and (6.12)-(6.13) for exponents  $2 \leq p < \frac{n}{\alpha} < r$ , p < q satisfying (6.5). Let  $u_j \in \mathcal{K}_{\psi}(\Omega)$  be the solution to the obstacle problem (6.14). Suppose that  $k \in L^r_{loc}(\Omega)$  and  $D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega)$ , for  $0 < \alpha < \gamma < 1$ . Then, the following estimate

$$\int_{B_{R/4}} |\tau_h V_p(Du_j)|^2 dx \le C|h|^{2\alpha} \left\{ \int_{B_R} (1+|Du_j|^p) dx + \|D\psi\|_{B_{2q-p,\infty}^{\gamma}(B_R)} \right\}^{\kappa}, \qquad (6.15)$$

holds for all balls  $B_{R/4} \subset B_R \Subset \Omega$ , for positive constants  $C := C(R, n, p, q, r, \beta)$ ,  $\kappa := \kappa(n, p, q, r, \beta)$ , both independent of j, and for some  $0 < \beta < \alpha$ .

*Proof.* We start by observing that, since p < 2q - p, we have

$$D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega) \Rightarrow D\psi \in B^{\gamma}_{p,\infty,loc}(\Omega),$$

thus an application of Theorem 5.3.1 implies

$$(\mu^2 + |Du_j|^2)^{\frac{p-2}{4}} Du_j \in B^{\alpha}_{2,\infty,loc}(\Omega),$$

which yields, by applying Lemma 5.2.3,

$$Du_j \in L_{loc}^{\frac{np}{n-2\beta}}(\Omega),$$

for all  $0 < \beta < \alpha$ . Thus, the integral

$$\int_{\Omega'} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx$$

is finite, for every  $\Omega' \Subset \Omega$  and  $\beta \in (0, \alpha)$ .

In the sequel we will profusely use the following inequality:

$$2q - p \le \frac{r(2q - p)}{r - 2} \le \frac{np}{n - 2\beta},\tag{6.16}$$

for  $\beta \in (\frac{\alpha nr}{nr+2(\alpha r-n)}, \alpha)$ . The first part of inequality (6.16) is trivial, while the second part comes from (6.5). Namely,

$$\frac{r(2q-p)}{r-2} \le \frac{np}{n-2\beta} \Leftrightarrow \frac{q}{p} \le \frac{nr-n-\beta r}{r(n-2\beta)}$$

and

$$1 + \frac{\alpha}{n} - \frac{1}{r} < \frac{nr - n - \beta r}{r(n - 2\beta)} \Leftrightarrow \beta > \frac{\alpha nr}{nr + 2(\alpha r - n)}$$

Fix  $0 < \frac{R}{4} < \rho < s < t < t' < \frac{R}{2}$  such that  $B_R \Subset \Omega$  and a cut-off function  $\eta \in \mathcal{C}_0^1(B_t)$ such that  $0 \le \eta \le 1$ ,  $\eta = 1$  on  $B_s$ ,  $|D\eta| \le \frac{C}{t-s}$ . Now, for  $|h| \le t' - t$ , we consider functions

$$v_1(x) = \eta^2(x)[(u_j - \psi)(x + h) - (u_j - \psi)(x)]$$

and

$$v_2(x) = \eta^2 (x-h) [(u_j - \psi)(x-h) - (u_j - \psi)(x)].$$

Then

$$\varphi_1(x) = u_j(x) + tv_1(x),$$
 (6.17)

$$\varphi_2(x) = u_j(x) + tv_2(x)$$
 (6.18)

are admissible test functions for all  $t \in [0, 1)$ .

Inserting (6.17) and (6.18) in (6.14), we obtain

$$\int_{\Omega} \langle \mathcal{A}_j(x, Du_j), D(\eta^2 \tau_h(u_j - \psi)) \rangle dx + \int_{\Omega} \langle \mathcal{A}_j(x, Du_j), D(\eta^2(x - h)\tau_{-h}(u_j - \psi)) \rangle dx \ge 0$$
(6.19)

By means of a simple change of variable, we can write the second integral on the left hand side of the previous inequality as follows

$$\int_{\Omega} \langle \mathcal{A}_j(x+h, Du_j(x+h)), D(-\eta^2 \tau_h(u_j - \psi)) \rangle dx$$
(6.20)

and so inequality (6.19) becomes

$$\int_{\Omega} \langle \mathcal{A}_j(x+h, Du_j(x+h)) - \mathcal{A}_j(x, Du_j(x)), D(\eta^2 \tau_h(u_j - \psi)) \rangle dx \le 0$$
(6.21)

We can write previous inequality as follows

$$0 \geq \int_{\Omega} \langle \mathcal{A}_{j}(x+h, Du_{j}(x+h)) - \mathcal{A}_{j}(x+h, Du_{j}(x)), \eta^{2} D\tau_{h} u_{j} \rangle dx$$
  

$$- \int_{\Omega} \langle \mathcal{A}_{j}(x+h, Du_{j}(x+h)) - \mathcal{A}_{j}(x+h, Du_{j}(x)), \eta^{2} D\tau_{h} \psi \rangle dx$$
  

$$+ \int_{\Omega} \langle \mathcal{A}_{j}(x+h, Du_{j}(x+h)) - \mathcal{A}_{j}(x, h, Du_{j}(x)), \eta^{2} D\tau_{h} u_{j} \rangle dx$$
  

$$+ \int_{\Omega} \langle \mathcal{A}_{j}(x+h, Du_{j}(x)) - \mathcal{A}_{j}(x, Du_{j}(x)), \eta^{2} D\tau_{h} \psi \rangle dx$$
  

$$- \int_{\Omega} \langle \mathcal{A}_{j}(x+h, Du_{j}(x)) - \mathcal{A}_{j}(x, Du_{j}(x)), \eta^{2} D\tau_{h} \psi \rangle dx$$
  

$$=: I_{1} + I_{2} + I_{3} + I_{4} + I_{5} + I_{6}, \qquad (6.22)$$

that yields

$$I_1 \le |I_2| + |I_3| + |I_4| + |I_5| + |I_6| \tag{6.23}$$

The ellipticity assumption (A2) implies

$$I_1 \ge \nu \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx$$
(6.24)

From the growth condition (A3), Young's and Hölder's inequalities and assumption on  $D\psi$ , we get

$$\begin{split} |I_2| &\leq L \int_{\Omega} \eta^2 |\tau_h D u_j| (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{q-2}{2}} |\tau_h D \psi| dx \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &+ C_{\varepsilon}(L) \int_{\Omega} \eta^2 |\tau_h D \psi|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &+ C_{\varepsilon}(L) \left( \int_{B_t} |\tau_h D \psi|^{2q-p} dx \right)^{\frac{2}{2q-p}} \left( \int_{B_{t'}} (1 + |D u_j|)^{2q-p} dx \right)^{\frac{2q-p-2}{2q-p}} \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &+ C_{\varepsilon}(L, n, p, q) |h|^{2\gamma} [D \psi]^{2}_{B_{2q-p,\infty}(B_R)} \left( \int_{B_{t'}} (1 + |D u_j|)^{2q-p} dx \right)^{\frac{2q-p-2}{2q-p}} \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &+ C_{\varepsilon}(L, n, p, q) |h|^{2\gamma} [D \psi]^{2q-p}_{B_{2q-p,\infty}(B_R)} + C_{\varepsilon}(L, n, p, q) |h|^{2\gamma} \int_{B_{t'}} (1 + |D u_j|)^{2q-p} dx. \end{split}$$

Therefore, from (6.16), we infer

$$|I_{2}| \leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx + C_{\varepsilon}(L,n,p,q) |h|^{2\gamma} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_{R})}^{2q-p} + C_{\varepsilon}(L,n,p,q) |h|^{2\gamma} \left( \int_{B_{t'}} (1+|D u_{j}|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{r-2}{r}}$$

$$(6.25)$$

Arguing analogously, we get

$$|I_3| \leq 2L \int_{\Omega} |D\eta| \eta |\tau_h Du_j| (1+|Du_j(x+h)|^2+|Du_j(x)|^2)^{\frac{q-2}{2}} |\tau_h(u_j-\psi)| dx$$

$$\begin{split} &\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ &+ \frac{C_{\varepsilon}(L)}{(t-s)^{2}} \int_{B_{t}} |\tau_{h}(u_{j}-\psi)|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{2q-p-2}{2}} dx \\ &\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ &+ \frac{C_{\varepsilon}(L)}{(t-s)^{2}} \left( \int_{B_{t}} |\tau_{h} \psi|^{2q-p} dx \right)^{\frac{2}{2q-p}} \left( \int_{B_{t'}} (1+|D u_{j}|)^{2q-p} dx \right)^{\frac{2q-p-2}{2q-p}} \\ &+ \frac{C_{\varepsilon}(L)}{(t-s)^{2}} \left( \int_{B_{t}} |\tau_{h} u_{j}|^{2q-p} dx \right)^{\frac{2}{2q-p}} \left( \int_{B_{t'}} (1+|D u_{j}|)^{2q-p} dx \right)^{\frac{2q-p-2}{2q-p}}. \end{split}$$

Using Young's inequality and Lemma 5.2.2, we obtain

$$\begin{aligned} |I_{3}| &\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ &+ \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}} |h|^{2} \left( \int_{B_{R}} |D\psi|^{2q-p} dx \right)^{\frac{2}{2q-p}} \left( \int_{B_{t'}} (1+|D u_{j}|)^{2q-p} dx \right)^{\frac{2q-p-2}{2q-p}} \\ &+ \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}} |h|^{2} \int_{B_{t'}} (1+|D u_{j}|)^{2q-p} dx \\ &\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ &+ \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}} |h|^{2} \int_{B_{R}} |D\psi|^{2q-p} dx \\ &+ \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}} |h|^{2} \int_{B_{t'}} (1+|D u_{j}|)^{2q-p} dx. \end{aligned}$$

$$(6.26)$$

Recalling the first inequality of (6.16), we can write

$$|I_{3}| \leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx + \frac{C_{\varepsilon}(L, n, p, q)}{(t-s)^{2}} |h|^{2} \int_{B_{R}} |D\psi|^{2q-p} dx + \frac{C_{\varepsilon}(L, n, p, q)}{(t-s)^{2}} |h|^{2} \left( \int_{B_{t'}} (1+|D u_{j}|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{r-2}{r}}.$$
(6.27)

In order to estimate the integral  $I_4$ , we use assumption (A4), and Young's and Hölder's inequalities as follows

$$|I_4| \leq \int_{\Omega} \eta^2 |\tau_h D u_j| |h|^{\alpha} (k(x+h) + k(x)) (1 + |D u_j(x)|)^{\frac{q-1}{2}} dx$$
$$\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx$$

$$+ C_{\varepsilon}|h|^{2\alpha} \int_{B_{t}} (k(x+h) + k(x))^{2} (1+|Du_{j}|)^{2q-p} dx$$
  

$$\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} Du_{j}|^{2} (\mu^{2}+|Du_{j}(x+h)|^{2}+|Du_{j}(x)|^{2})^{\frac{p-2}{2}} dx$$
  

$$+ C_{\varepsilon}|h|^{2\alpha} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{2}{r}} \left(\int_{B_{t}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{r-2}{r}}.$$
(6.28)

We now take care of  $I_5$ . Similarly as above, exploiting assumption (A4) and Hölder's inequality, we infer

$$\begin{aligned} |I_{5}| &\leq \int_{\Omega} \eta^{2} |\tau_{h} D\psi| |h|^{\alpha} \left(k(x+h) + k(x)\right) \left(1 + |Du_{j}|^{2}\right)^{\frac{q-1}{2}} dx \\ &\leq |h|^{\alpha} \left(\int_{B_{t'}} k^{r} dx\right)^{\frac{1}{r}} \left(\int_{B_{t}} |\tau_{h} D\psi|^{\frac{r}{r-1}} (1 + |Du_{j}|)^{\frac{r(q-1)}{r-1}} dx\right)^{\frac{r-1}{r}} \\ &\leq |h|^{\alpha} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{1}{r}} \left(\int_{B_{t}} |\tau_{h} D\psi|^{2q-p} dx\right)^{\frac{1}{2q-p}} \left(\int_{B_{t}} (1 + |Du_{j}|)^{\frac{r(q-1)(2q-p)}{(r-1)(2q-p)-r}} dx\right)^{\frac{(r-1)(2q-p)-r}{r(2q-p)}} \end{aligned}$$

Now, we observe

$$\frac{r(q-1)(2q-p)}{(r-1)(2q-p)-r} \le \frac{r(2q-p)}{r-2} \Leftrightarrow p-2+r(q-p) \ge 0,$$
(6.29)

which is true by assumption, that is  $p \ge 2$ ,  $r > \frac{n}{\alpha} > 2$  and q > p. Hence

$$|I_5| \le |h|^{\alpha+\gamma} \left( \int_{B_R} k^r dx \right)^{\frac{1}{r}} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_R)} \left( \int_{B_t} (1+|Du_j|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{(r-2)(q-1)}{r(2q-p)}}.$$
 (6.30)

From assumption (A4), hypothesis  $|D\eta| < \frac{C}{t-s}$  and Hölder's inequality, we infer the following estimate for  $I_6$ .

$$\begin{split} |I_{6}| &\leq \frac{C}{t-s} |h|^{\alpha} \int_{B_{t}} |\tau_{h}\psi| (k(x+h)+k(x))(1+|Du_{j}|^{2})^{\frac{q-1}{2}} dx \\ &+ \frac{C}{t-s} |h|^{\alpha} \int_{B_{t}} |\tau_{h}u_{j}| (k(x+h)+k(x))(1+|Du_{j}|^{2})^{\frac{q-1}{2}} dx \\ &\leq \frac{C}{t-s} |h|^{\alpha} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{t}} |\tau_{h}\psi|^{2q-p} dx \right)^{\frac{1}{2q-p}} \\ &\cdot \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{r(q-1)(2q-p)}{(r-1)(2q-p)-r}} dx \right)^{\frac{(r-1)(2q-p)-r}{r(2q-p)}} \\ &+ \frac{C}{t-s} |h|^{\alpha} \left( \int_{B_{t'}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{t}} |\tau_{h}u_{j}|^{\frac{r}{r-1}} (1+|Du_{j}|)^{\frac{r(q-1)}{r-1}} dx \right)^{\frac{r-1}{r}} \end{split}$$

Using once again Hölder's inequality, we have

$$\begin{split} |I_{6}| &\leq \frac{C}{t-s} |h|^{\alpha+1} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{t'}} |D\psi|^{2q-p} dx \right)^{\frac{1}{2q-p}} \\ & \cdot \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{r(q-1)(2q-p)}{(r-1)(2q-p)-r}} dx \right)^{\frac{(r-1)(2q-p)-r}{r(2q-p)}} \\ & + \frac{C}{t-s} |h|^{\alpha} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{t}} |\tau_{h} u_{j}|^{\frac{rq}{r-1}} dx \right)^{\frac{r-1}{rq}} \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{rq}{r-1}} dx \right)^{\frac{(r-1)(q-1)}{rq}} \end{split}$$

Using Lemma 5.2.2, we infer

$$\begin{aligned} |I_6| &\leq \frac{C}{t-s} |h|^{\alpha+1} \left( \int_{B_R} k^r dx \right)^{\frac{1}{r}} \left( \int_{B_R} |D\psi|^{2q-p} dx \right)^{\frac{1}{2q-p}} \\ &\cdot \left( \int_{B_t} (1+|Du_j|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{(q-1)(r-2)}{r(2q-p)}} \\ &+ \frac{C}{t-s} |h|^{\alpha+1} \left( \int_{B_R} k^r dx \right)^{\frac{1}{r}} \left( \int_{B_{t'}} (1+|Du_j|)^{\frac{rq}{r-1}} dx \right)^{\frac{r-1}{rq}}. \end{aligned}$$

We remark that

$$\frac{rq}{r-1} \le \frac{r(2q-p)}{r-2} \Leftrightarrow p + r(q-p) \ge 0, \tag{6.31}$$

which is true by assumption, that is  $p \ge 2$ ,  $r > \frac{n}{\alpha} > 2$  and q > p. Hence

$$\begin{aligned} |I_{6}| &\leq \frac{C}{t-s} |h|^{\alpha+1} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{R}} |D\psi|^{2q-p} dx \right)^{\frac{1}{2q-p}} \\ & \cdot \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{(q-1)(r-2)}{r(2q-p)}} \\ & + \frac{C}{t-s} |h|^{\alpha+1} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{1}{r}} \left( \int_{B_{t'}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{q(r-2)}{r(2q-p)}}. \end{aligned}$$
(6.32)

Inserting estimates (6.24), (6.25), (6.27), (6.28), (6.30) and (6.32) in (6.23), we infer

$$\begin{split} \nu \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ \leq & 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ & + C_{\varepsilon} (L, n, p, q) |h|^{2\gamma} [D \psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_{R})} + C_{\varepsilon} (L, n, p, q) |h|^{2\gamma} \left( \int_{B_{t'}} (1 + |D u_{j}|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{r-2}{r}} \end{split}$$

$$+ \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}}|h|^{2} \int_{B_{R}} |D\psi|^{2q-p} dx \\ + \frac{C_{\varepsilon}(L,n,p,q)}{(t-s)^{2}}|h|^{2} \left(\int_{B_{t'}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{r-2}{r}} \\ + C_{\varepsilon}|h|^{2\alpha} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{2}{r}} \left(\int_{B_{t}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{r-2}{r}} \\ + |h|^{\alpha+\gamma} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{1}{r}} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_{R})} \left(\int_{B_{t}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{(r-2)(q-1)}{r(2q-p)}} \\ + \frac{C}{t-s}|h|^{\alpha+1} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{1}{r}} \left(\int_{B_{R}} |D\psi|^{2q-p} dx\right)^{\frac{1}{2q-p}} \\ \cdot \left(\int_{B_{t}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{(q-1)(r-2)}{r(2q-p)}} \\ + \frac{C}{t-s}|h|^{\alpha+1} \left(\int_{B_{R}} k^{r} dx\right)^{\frac{1}{r}} \left(\int_{B_{t'}} (1+|Du_{j}|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{q(r-2)}{r(2q-p)}}.$$
(6.33)

We now introduce the following interpolation inequality

$$\|Dw\|_{\frac{r(2q-p)}{r-2}} \le \|Dw\|_p^{\delta} \|Dw\|_{\frac{np}{n-2\beta}}^{1-\delta},\tag{6.34}$$

where  $0 < \delta < 1$  is defined through the condition

$$\frac{r-2}{r(2q-p)} = \frac{\delta}{p} + \frac{(1-\delta)(n-2\beta)}{np}$$
(6.35)

which implies

$$\delta = \frac{nr(p-q) - np + \beta r(2q-p)}{\beta r(2q-p)}, \quad 1-\delta = \frac{n[r(q-p) + p]}{\beta r(2q-p)}.$$

Hence we get the following inequalities

$$\left(\int_{B_{t'}} (1+|Du_j|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{r-2}{r}} \leq \left(\int_{B_{t'}} (1+|Du_j|)^p dx\right)^{\frac{\delta(2q-p)}{p}} \cdot \left(\int_{B_{t'}} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx\right)^{\frac{(n-2\beta)[r(q-p)+p]}{\beta pr}}, \quad (6.36)$$
$$\left(\int_{B_t} (1+|Du_j|)^{\frac{r(2q-p)}{r-2}} dx\right)^{\frac{(r-2)(q-1)}{r(2q-p)}} \leq \left(\int_{B_t} (1+|Du_j|)^p dx\right)^{\frac{\delta(q-1)}{p}} \cdot \left(\int_{B_t} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx\right)^{\frac{(n-2\beta)(q-1)p'}{p}}, \quad (6.37)$$

$$\left( \int_{B_{t'}} (1+|Du_j|)^{\frac{r(2q-p)}{r-2}} dx \right)^{\frac{q(r-2)}{r(2q-p)}} \leq \left( \int_{B_{t'}} (1+|Du_j|)^p dx \right)^{\frac{\delta q}{p}} \cdot \left( \int_{B_{t'}} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{(n-2\beta)q[r(q-p)+p]}{\beta pr(2q-p)}}, \quad (6.38)$$

where  $p' = \frac{r(q-p)+p}{\beta r(2q-p)}$ . Inserting (6.36), (6.37) and (6.38) in (6.33), and exploiting the bounds

$$\frac{n[r(q-p)+p]}{\beta pr} < 1, \quad \frac{n(q-1)[r(q-p)+p]}{\beta rp(2q-p)} < 1, \quad \frac{nq[r(q-p)+p]}{\beta pr(2q-p)} < 1, \tag{6.39}$$

which hold by assumption (6.5) and for  $\beta \in (\frac{n[r(q-p)+p]}{pr}, \alpha)$ , from Young's inequality, we infer

$$\begin{split} \nu & \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ & \leq 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ & + C_{\varepsilon} (L, n, p, q) |h|^{2\gamma} [D \psi]_{B_{2q-p,\infty}^{2}(B_{R})}^{2q-p} + C_{\varepsilon,\theta} (L, n, p, q) |h|^{2\gamma} \left( \int_{B_{R}} (1 + |D u_{j}|)^{p} dx \right)^{\frac{\delta(2q-p)p}{p}} \\ & + \theta |h|^{2\gamma} \left( \int_{B_{t'}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ & + \frac{C_{\varepsilon} (L, n, p, q)}{(t-s)^{2}} |h|^{2} \int_{B_{R}} |D \psi|^{2q-p} dx \\ & + \theta |h|^{2} \left( \int_{B_{t'}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ & + \frac{C_{\varepsilon,\theta} (L, n, p, q)}{(t-s)^{2p}} |h|^{2} \left( \int_{B_{R}} (1 + |D u_{j}|)^{p} dx \right)^{\frac{p\delta(2q-p)}{p}} \\ & + C_{\varepsilon,\theta} (n, p, q) |h|^{2\alpha} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{2p}{r}} \\ & + \theta |h|^{2\alpha} \left( \int_{B_{R}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{p}} \\ & + \theta |h|^{2\alpha} \left( \int_{B_{R}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ & + C_{\theta} |h|^{\alpha+\gamma} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p''}{r}} \\ & + C_{\theta} |h|^{\alpha+\gamma} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p''}{r}} \end{split}$$

$$\cdot \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\delta(q-1)(2q-p)p''}{p}}$$

$$+ \theta |h|^{\alpha+\gamma} \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}}$$

$$+ \frac{C_{\theta}}{(t-s)^{p''}} |h|^{\alpha+1} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p''}{r}} \left( \int_{B_{R}} |D\psi|^{2q-p} dx \right)^{\frac{p''}{2q-p}}$$

$$\cdot \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\delta(q-1)p''}{p}}$$

$$+ \theta |h|^{\alpha+1} \left( \int_{B_{t}} (1+|Du_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}}$$

$$+ \frac{C_{\theta}}{(t-s)^{p^{*}}} |h|^{\alpha+1} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p^{*}}{r}} \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{p^{*} \delta q}{p}}$$

$$+ \theta |h|^{\alpha+1} \left( \int_{B_{t'}} (1+|Du_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} .$$

$$(6.40)$$

for some constant  $\theta \in (0, 1)$ , where we set  $\tilde{p} = \frac{\beta pr}{\beta pr - n[r(q-p)+p]}$ ,  $p'' = \frac{\beta rp(2q-p)}{\beta rp(2q-p) - (q-1)n[r(q-p)+p]}$ ,  $p^* = \frac{p}{p-(1-\delta)q}$ . For a better readability we now define

$$\begin{split} A = & C_{\varepsilon}(L,n,p,q) [D\psi]_{B_{2q-p,\infty}(B_{R})}^{2q-p} + C_{\varepsilon,\theta}(L,n,p,q) \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\delta(2q-p)\tilde{p}}{p}} \\ & + C_{\varepsilon,\theta}(n,p,q) \left( \int_{B_{R}} k^{r} dx \right)^{\frac{2\tilde{p}}{r}} \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\tilde{p}\delta(2q-p)}{p}} \\ & + C_{\theta} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p''}{r}} [D\psi]_{B_{2q-p,\infty}(B_{R})}^{p''} \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\delta(q-1)(2q-p)p''}{p}} \\ B_{1} = & C_{\varepsilon}(L,n,p,q) \int_{B_{R}} |D\psi|^{2q-p} dx, \\ B_{2} = & C_{\varepsilon,\theta}(L,n,p,q) \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\tilde{p}\delta(2q-p)}{p}} , \\ B_{3} = & C_{\theta} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p''}{r}} \left( \int_{B_{R}} |D\psi|^{2q-p} dx \right)^{\frac{p''}{2q-p}} \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{\delta(q-1)(p'')}{p}} , \\ B_{4} = & C_{\theta} \left( \int_{B_{R}} k^{r} dx \right)^{\frac{p^{*}}{r}} \left( \int_{B_{R}} (1+|Du_{j}|)^{p} dx \right)^{\frac{p^{*}\delta q}{p}} , \end{split}$$

so that we can rewrite the previous estimate as

$$\begin{split} \nu \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ \leq & 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} D u_{j}|^{2} (\mu^{2} + |D u_{j}(x+h)|^{2} + |D u_{j}(x)|^{2})^{\frac{p-2}{2}} dx \\ & + \theta (|h|^{2\alpha} + |h|^{\alpha+\gamma} + |h|^{\alpha+1}) \left( \int_{B_{t}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ & + \theta (|h|^{2} + |h|^{2\gamma} + |h|^{\alpha+1}) \left( \int_{B_{t'}} (1 + |D u_{j}|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ & + (|h|^{2\gamma} + |h|^{2\alpha} + |h|^{\alpha+\gamma}) A + |h|^{2} \frac{B_{1}}{(t-s)^{2}} + |h|^{2} \frac{B_{2}}{(t-s)^{2\tilde{p}}} \\ & + |h|^{\alpha+1} \frac{B_{3}}{(t-s)^{p''}} + |h|^{\alpha+1} \frac{B_{4}}{(t-s)^{p^{*}}}. \end{split}$$

Choosing  $\varepsilon = \frac{\nu}{6}$ , we can reabsorb the first integral in the right hand side of the previous estimate by the left hand side, thus getting

$$\begin{split} &\int_{\Omega} \eta^2 |\tau_h D u_j|^2 (\mu^2 + |D u_j(x+h)|^2 + |D u_j(x)|^2)^{\frac{p-2}{2}} dx \\ &\leq 3\theta |h|^{2\alpha} \left( \int_{B_t} (1+|D u_j|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} + 3\theta |h|^{2\alpha} \left( \int_{B_{t'}} (1+|D u_j|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} \\ &+ |h|^{2\alpha} A + |h|^2 \frac{B_1}{(t-s)^2} + |h|^2 \frac{B_2}{(t-s)^{\tilde{p}}} + |h|^{2\alpha} \frac{B_3}{(t-s)^{p''}} + |h|^{2\alpha} \frac{B_4}{(t-s)^{p^*}}, \end{split}$$

where we used the fact that  $\alpha < \gamma$ . Using Lemma 5.0.1 in the left hand side of the previous inequality, recalling that  $\eta = 1$  on  $B_s$ , we get

$$\int_{B_s} |\tau_h V_p(Du_j)|^2 dx \le |h|^{2\alpha} \left\{ 3\theta \left( \int_{B_t} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} + 3\theta \left( \int_{B_{t'}} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx \right)^{\frac{n-2\beta}{n}} + A + \frac{B_1}{(t-s)^2} + \frac{B_2}{(t-s)^{\tilde{p}}} + \frac{B_3}{(t-s)^{p''}} + \frac{B_4}{(t-s)^{p^*}} \right\}.$$
(6.41)

Lemma 5.2.3 and inequality (5.1) imply

$$\left(\int_{B_s} |Du_j|^{\frac{np}{n-2\beta}} dx\right)^{\frac{n-2\beta}{n}} \leq 3\theta \left(\int_{B_t} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx\right)^{\frac{n-2\beta}{n}} + 3\theta \left(\int_{B_{t'}} (1+|Du_j|)^{\frac{np}{n-2\beta}} dx\right)^{\frac{n-2\beta}{n}} + A + \frac{B_1}{(t-s)^2} + \frac{B_2}{(t-s)^{2\tilde{p}}} + \frac{B_3}{(t-s)^{p''}} + \frac{B_4}{(t-s)^{p^*}}, \quad (6.42)$$

for all  $\beta \in (0, \alpha)$ . Setting

$$\Phi(r) = \left(\int_{B_r} |Du_j|^{\frac{np}{n-2\beta}} dx\right)^{\frac{n-2\beta}{n}},$$

we can write inequality (6.42) as

$$\Phi(s) \le 3\theta\Phi(t) + 3\theta\Phi(t') + A + \frac{B_1}{(t-s)^2} + \frac{B_2}{(t-s)^{2\tilde{p}}} + \frac{B_3}{(t-s)^{p''}} + \frac{B_4}{(t-s)^{p^*}}.$$
 (6.43)

By virtue of Lemma 5.0.2, choosing  $0 < \theta < 1/3$ , we obtain

$$\Phi(\varrho) \le c \left( 3\theta \Phi(t') + A + \frac{B_1}{R^2} + \frac{B_2}{R^{2\tilde{\rho}}} + \frac{B_3}{R^{p''}} + \frac{B_4}{R^{p^*}} \right), \tag{6.44}$$

for some constant  $c := c(n, p, q, r, \beta, \theta)$ . Then, applying Lemma 5.0.2 again, we get

$$\Phi\left(\frac{R}{4}\right) \le \tilde{c}\left(A + \frac{B_1}{R^2} + \frac{B_2}{R^{2\tilde{p}}} + \frac{B_3}{R^{p''}} + \frac{B_4}{R^{p^*}}\right),\tag{6.45}$$

with  $\tilde{c} := \tilde{c}(n, p, q, r, \beta, \theta).$ 

Now, recalling the definition of  $\Phi$ , we obtain

$$\left(\int_{B_{R/4}} |Du_j|^{\frac{np}{n-2\beta}} dx\right)^{\frac{n-2\beta}{n}} \le \tilde{c} \left\{\int_{B_R} (1+|Du_j|^p) dx + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(B_R)}\right\}^{\kappa}, \tag{6.46}$$

thus, using Lemma 5.2.3, from inequalities (6.46) and (6.41), we deduce the a priori estimate

$$\int_{B_{R/4}} |\tau_h V_p(Du_j)|^2 dx \le C |h|^{2\alpha} \left\{ \int_{B_R} (1 + |Du_j|^p) dx + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(B_R)} \right\}^{\kappa}, \qquad (6.47)$$

for some  $\beta < \alpha$ , where  $C := C(R, n, p, q, r, \beta)$  and  $\kappa := \kappa(n, p, q, r, \beta)$ .

### 6.2.2 Passage to the limit

Let  $u \in \mathcal{K}_{\psi}(\Omega)$  be a solution to (6.3), and let  $\tilde{F}_j$  be defined as in Lemma 6.1.2. From Theorem 6.1.1, there exists  $c_1 > 0$  such that

$$|\xi|^p \le c_1(1 + \tilde{F}_j(x,\xi)), \quad \forall j \in \mathbb{N}.$$
(6.48)

Fixed  $B_R \Subset \Omega$ , let  $u_j$  be the solution of the problem

$$\min\left\{\int_{B_R} \tilde{F}_j(x, Dw) dx : w \ge \psi \text{ a.e. in } B_R, \ w \in u + W_0^{1, p}(B_R)\right\}.$$

From (6.48), the minimality of  $u_j$  implies

$$\int_{B_R} |Du_j|^p dx \leq c_1 \int_{B_R} (1 + \tilde{F}_j(x, Du_j)) dx$$
$$\leq c_1 \int_{B_R} (1 + \tilde{F}_j(x, Du)) dx$$
$$\leq c_1 \int_{B_R} (1 + \tilde{F}(x, Du)) dx, \qquad (6.49)$$

where in the last inequality we used Lemma 6.1.2 (ii). Thus, up to subsequences,

$$u_j \rightharpoonup \tilde{u} \text{ in } u + W_0^{1,p}(B_R) \tag{6.50}$$

and

$$u_j \to \tilde{u} \text{ in } L^p(B_R).$$
 (6.51)

For any j,  $\tilde{F}_j$  satisfies the assumptions of Theorem 6.2.1. Combining (6.46) and (6.49) we get

$$\|Du_{j}\|_{L^{\frac{np}{n-2\beta}}(B_{R/4})} \leq \tilde{c} \left\{ \int_{B_{R}} (1 + \tilde{F}(x, Du)) dx + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(B_{R})} \right\}^{\tilde{\kappa}},$$
(6.52)

thus, by (6.50), (6.52) and weak lower semicontinuity, we infer

$$\begin{split} \|D\tilde{u}\|_{L^{\frac{np}{n-2\beta}}(B_{R/4})} &\leq \liminf_{j \to \infty} \|Du_j\|_{L^{\frac{np}{n-2\beta}}(B_{R/4})} \\ &\leq \tilde{c} \bigg\{ \int_{B_R} (1 + \tilde{F}(x, Du)) dx + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(B_R)} \bigg\}^{\tilde{\kappa}}. \end{split}$$
(6.53)

Let  $j_0, j \in \mathbb{N}$  s.t.  $j_0 < j$ . Then, by Lemma 6.1.2 (ii) and the fact that  $u_j$  is a minimum for  $\tilde{F}_j$ , we might write

$$\int_{B_R} \tilde{F}_{j_0}(x, Du_j) dx \le \int_{B_R} \tilde{F}_j(x, Du_j) dx$$
$$\le \int_{B_R} \tilde{F}_j(x, Du) dx \le \int_{B_R} \tilde{F}(x, Du) dx.$$

Now from weak lower semicontinuity of  $\tilde{F}_{j_0}$  and (6.50), it holds, for every  $j_0 \in \mathbb{N}$ ,

$$\int_{B_R} \tilde{F}_{j_0}(x, D\tilde{u}) dx \le \liminf_{j \to +\infty} \int_{B_R} \tilde{F}_{j_0}(x, Du_j) dx \le \int_{B_R} \tilde{F}(x, Du) dx.$$

Combining these last inequalities, we get

$$\int_{B_R} \tilde{F}(x, D\tilde{u}) dx = \lim_{j_0 \to +\infty} \int_{B_R} \tilde{F}_{j_0}(x, D\tilde{u}) dx \le \int_{B_R} \tilde{F}(x, Du) dx, \tag{6.54}$$

where we also applied the monotone convergence theorem, according to Lemma 6.1.2 (ii). Moreover, by the weak convergence (6.50), the limit function  $\tilde{u}$  still belongs to  $\mathcal{K}_{\psi}(B_R)$ , since this set is convex and closed. Thus, we can conclude that

$$\tilde{u} = u$$
 a.e. in  $B_R$  (6.55)

by strict convexity of  $\tilde{F}$ , and, recalling estimate (6.53),

$$\|Du\|_{L^{\frac{np}{n-2\beta}}(B_{R/4})} \le \tilde{c} \left\{ \int_{B_R} (1 + \tilde{F}(x, Du)) dx + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(B_R)} \right\}^{\tilde{\kappa}}.$$
 (6.56)

Finally, we can repeat the proof of Theorem 6.2.1 obtaining  $V_p(Du) \in B^{\alpha}_{2,\infty,loc}(\Omega)$ .

# Chapter 7

# Higher differentiability in the case of double-phase lagrangians F(x, u, Du)

According to the results presented in Chapter 6, we here study the higher fractional differentiability properties of the gradient of the solutions  $u \in W^{1,p}(\Omega)$  to variational obstacle problems of the form

$$\min\left\{\int_{\Omega} F(x, w, Dw) \mathrm{d}x : w \in \mathcal{K}_{\psi}(\Omega)\right\},\tag{7.1}$$

where the energy density  $F: \Omega \times \mathbb{R} \times \mathbb{R}^n \to \mathbb{R}$  is defined by

$$F(x, w, z) = b(x, w)H(x, z),$$
 (7.2)

being

$$H(x,z) = |z|^{p} + a(x)|z|^{q},$$
(7.3)

where  $2 \leq p < q$ .

Here  $\Omega$  is a bounded open set of  $\mathbb{R}^n$ ,  $n \geq 2$ , the obstacle function  $\psi$  is the one defined in Chapter 6, as well as the set  $\mathcal{K}_{\psi}(\Omega)$  (see (6.2)).

We assume that the coefficients a(x) and b(x, w) satisfy the following assumptions:

#### Assumption 1

(i)  $a: \Omega \to [0, +\infty)$  is a bounded and measurable function such that

$$|a(x) - a(y)| \le \omega_a(|x - y|),$$

for all  $x, y \in \Omega$ , where  $\omega_a : \mathbb{R}^+ \to [0, 1]$  is defined by  $\omega_a(\rho) = \min\{\rho^{\alpha}, 1\}$ , for some  $\alpha \in (0, 1)$ ;

(ii) the function  $b: \Omega \times \mathbb{R} \to (0, +\infty)$  is a bounded Carathéodory function, i.e. there

exist  $0 < \nu \leq L$  such that

$$0 < \nu \le b(x, w) \le L < \infty.$$

### Assumption 2

(i) there exists a function  $\omega_b : \mathbb{R}^+ \to [0,1]$  defined by  $\omega_b(\rho) = \min\{\rho^\beta, 1\}$ , for some  $\beta \in (0,1)$ , such that

$$|b(x, u) - b(y, v)| \le \omega_b(|x - y| + |u - v|),$$

for all  $x, y \in \Omega$  and every  $u, v \in \mathbb{R}$ .

The energy density given by (7.2) is a model case of functions F satisfying the following set of conditions

$$\nu_1 |z|^p \le F(x, w, z) \le L_1(1 + |z|^q) \tag{F1}$$

$$\nu_2 |z|^{p-2} |\lambda|^2 \le \langle D_{zz} F(x, w, z) \lambda, \lambda \rangle \le L_2 (1+|z|^{q-2}) |\lambda|^2$$
(F2)

$$|F(x_1, w_1, z) - F(x_2, w_2, z)| \le l_1 \omega_\delta(|x_1 - x_2| + |w_1 - w_2|)(1 + |z|^q)$$
(F3)

for all  $x, x_1, x_2 \in \Omega$ ,  $w, w_1, w_2 \in \mathbb{R}$  and every  $z, \lambda \in \mathbb{R}^n$ , where  $0 < \nu_1 \leq L_1, 0 < \nu_2 \leq L_2$ ,  $l_1 \geq 1$  are fixed constants and  $\omega_{\delta} : \mathbb{R}^+ \to [0, 1]$  is a function defined by  $\omega_{\delta}(\rho) = \min\{\rho^{\delta}, 1\}$ , for some  $\delta \in (0, 1)$  depending on  $\alpha$  and  $\beta$  introduced in Assumption 1 and 2 respectively. We point out that the choice of stating Assumption 1 and 2 separately is due to the fact that they are needed independently.

The main difficulty here is the dependence of our double phase functional both on the x-variable and the w-variable, where the map  $w \mapsto b(x, w)H(x, z)$  is non-differentiable. In order to deal with this issue, we follow the strategy proposed in [85] and later used in [44]. Namely, we introduce the so-called "freezed" functional defined in (7.4) and the solution to the corresponding obstacle problem (see (7.5)) for which we prove a higher differentiability result in the scale of Besov spaces following the argument in [67]. The idea is to compare the solution u to the original obstacle problem (7.1) and the solution vto the "freezed" one (7.5). More precisely, we estimate the fractional difference quotients of u and v, in an integral sense, gaining a Besov regularity for u. In order to do so, we also have to derive some ad hoc higher integrability results, both at the interior and up to boundary, that is for the solution u of the original obstacle problem (7.1) and the solution v to the freezed one (7.5) respectively. The first one is obtained adapting the argument in [38], while the second one generalizes the result by Cupini, Fusco and Petti in [29]. Eventually, we use a boot-strapping argument to get the maximal higher fractional differentiability.

The main result of this Chapter is the following.

**Theorem 7.0.1.** Let  $u \in W^{1,p}(\Omega)$  be the solution to the obstacle problem (7.1), with F defined by (7.2), under Assumptions 1 and 2, for exponents  $2 \le p < \frac{n}{\alpha}$ , p < q verifying

$$\frac{q}{p} < 1 + \frac{\alpha}{n}.$$

If  $D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega)$ , for  $0 < \alpha < \gamma < 1$ , then there exists  $\tilde{\sigma} := \tilde{\sigma}(p,q,n,\alpha,\beta) \in (0,1)$ s.t.

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^t_{2,\infty,loc}(\Omega), \quad \forall t \in (0,\tilde{\sigma}).$$

The chapter is organized as follows. In Section 7.1, we show that the solution to the freezed obstacle problem (7.5) satisfies a variational inequality and moreover we present interior and up to the boundary higher integrability properties, which will be crucial for the comparison argument, as already mentioned. In Section 7.2, we exploit the results obtained in Chapter 6 in order to prove the higher fractional differentiability of the solution to the freezed obstacle problem (7.5). We remark that the procedure used in order to do so requires the assumption  $p \ge 2$ . The comparison argument is presented in Section 7.3. Finally, in Section 7.4, we show that a suitable fractional differentiability property on the gradient of the obstacle transfers to a higher fractional differentiability for the gradient of the minimizer, so that we are eventually able to prove Theorem 7.0.1.

We point out that, in order to prove the higher integrability of the solution to the original obstacle problem (see Theorem 7.1.2) and the higher differentiability of the solution to the freezed obstacle problem in Section 7.2, Assumption 1 (*ii*) is the only one needed on the function b(x, w). On the other hand, in order to prove the comparison lemma (see Lemma 7.3.2), we require Assumption 2 on the coefficient b(x, w).

## 7.1 Higher integrability

The results contained in this section will be crucial for the comparison argument presented in Section 7.3.

Let u be a solution to the obstacle problem (7.1) and fix a ball  $B = B_{\frac{R}{2}}(x_0) \Subset \Omega$ , for a given radius R > 0 and  $x_0 \in \Omega$ . Let us consider the so-called "freezed" functional

$$\int_{B} \tilde{F}(x, Dw) \mathrm{d}x = \int_{B} b(x_0, u_B) H(x, Dw) \mathrm{d}x,$$
(7.4)

where H was defined in (7.3), and let  $v \in u + W_0^{1,p}(B)$  be the solution to

$$\min\left\{\int_{B} \tilde{F}(x, Dw) \mathrm{d}x : w \in \mathcal{K}_{\psi}(\Omega), w = u \text{ on } \partial B\right\}.$$
(7.5)

Now, we show that a local minimizer of functional (7.4) satisfies a variational inequality. More precisely, we have

**Proposition 7.1.1.** A function  $v \in u + W_0^{1,p}(B)$  is a solution to (7.5) if and only if it satisfies the following variational inequality

$$\int_{B} \langle D_z H(x, Dv), D(\varphi - v) \rangle dx \ge 0,$$
(7.6)

for every  $\varphi \in u + W_0^{1,p}(B) \cap \mathcal{K}_{\psi}(\Omega)$  such that  $H(x, D\varphi) \in L^1(B)$ .

*Proof.* We set  $g = v + \varepsilon(\varphi - v)$  for  $\varepsilon \in (0, 1)$ , which belongs to the obstacle class, indeed

$$g = v + \varepsilon(\varphi - v) = \varepsilon\varphi + (1 - \varepsilon)v \ge \psi.$$

We first notice that  $H(x, D(v + \varepsilon(\varphi - v))) \in L^1$ . Moreover,

$$\int_{B} H(x, Dv) \mathrm{d}x \leq \int_{B} H(x, Dv + \varepsilon D(\varphi - v)) \mathrm{d}x,$$

which leads to

$$\int_{B} H(x, Dv + \varepsilon D(\varphi - v)) dx - \int_{B} H(x, Dv) dx \ge 0.$$

From Lagrange's theorem, for  $\theta \in (0, 1)$  it holds

$$\int_{B} \langle D_{z}H(x, Dv + \varepsilon \theta D(\varphi - v)), \varepsilon D(\varphi - v) \rangle \mathrm{d}x \ge 0.$$

Since  $\varepsilon > 0$ ,

$$\int_{B} \langle D_{z}H(x, Dv + \varepsilon \theta D(\varphi - v)), D(\varphi - v) \rangle \mathrm{d}x \ge 0.$$
(7.7)

According to [27, Lemma 2.2], it holds

$$|\langle D_z H(x,z),\lambda\rangle| \le C \left(H(x,z) + H(x,\lambda)\right).$$

Therefore,

$$\begin{aligned} |\langle D_z H(x, Dv + \varepsilon \theta D(\varphi - v)), D(\varphi - v) \rangle| \\ \leq C \left( H(x, Dv + \varepsilon \theta D(\varphi - v)) + H(x, D(\varphi - v)) \right) \\ \leq C \left( H(x, Dv) + H(x, \varepsilon \theta D(\varphi - v)) + H(x, D\varphi) + H(x, Dv) \right) \\ \leq C \left( H(x, Dv) + (\varepsilon \theta)^p H(x, D\varphi) + (\varepsilon \theta)^p H(x, Dv) + H(x, D\varphi) \right), \end{aligned}$$
(7.8)

where in the last passage we also used the direct property  $H(x, \varepsilon \theta z) \leq (\varepsilon \theta)^p H(x, z)$ . For  $\varepsilon \to 0$ , the second and the third term on the right hand side of (7.8) go to zero. Hence, the right hand side tends to  $C(H(x, Dv) + H(x, D\varphi))$  in  $L^1$ . Then, we can pass to the limit for  $\varepsilon \to 0$  in (7.7) applying the Dominated convergence theorem, which concludes the proof.

If u is a solution to (7.1), then we are able to establish for u a higher integrability result.

**Theorem 7.1.2.** Let u be a solution to the obstacle problem (7.1) where the integrand satisfies Assumption 1, for exponents  $2 \le p < q$  verifying

$$\frac{q}{p} \le 1 + \frac{\alpha}{n}.$$

If the function  $\psi$  is s.t.  $H(x, D\psi) \in L^{m_1}_{loc}(\Omega)$ , for some  $m_1 > 1$ , then there exist an exponent  $m_1 > m_2 > 1$  and a positive constant C s.t. it holds

$$\left(\int_{B_{\frac{R}{2}}} (H(x, Du))^{m_2} dx\right)^{\frac{1}{m_2}} \le C \left[\int_{B_R} H(x, Du) dx + \left(\int_{B_R} (H(x, D\psi))^{m_1} dx\right)^{\frac{1}{m_1}}\right].$$

for all balls  $B_{\frac{R}{2}} \subset B_R \Subset \Omega$ .

Proof. Let  $\frac{R}{2} \leq t < s \leq R \leq 1$  and let  $\eta \in C_0^{\infty}(B_R)$  be a cut-off function s.t.  $0 \leq \eta \leq 1, \eta \equiv 1$  on  $B_t, \eta \equiv 0$  outside  $B_s, |D\eta| \leq \frac{2}{s-t}$ . We set  $\varphi = \eta(x)(u(x)-u_{B_R})-\eta(x)(\psi(x)-\psi_{B_R}))$  and  $g = u - \varphi \in \mathcal{K}_{\psi}(\Omega)$ . We observe that g = u on  $\partial B_s$  and  $g = \psi - \psi_{B_R} + u_{B_R}$  on  $B_t$ , therefore  $Dg = D\psi$  on  $B_t$ . Using Assumption 1 (ii) and the fact that u is a local minimizer, we have

$$\int_{B_t} H(x, Du(x)) dx$$
  
$$\leq C \int_{B_t} F(x, u(x), Du(x)) dx$$

$$\begin{split} &\leq C \int_{B_s} F(x,g(x),Dg(x))\mathrm{d}x \\ &\leq C \int_{B_s} |Dg(x)|^p + a(x)|Dg(x)|^q \mathrm{d}x \\ &\leq C \int_{B_s} [|D\eta(x)|(\psi(x) - \psi_{B_R}) + \eta(x)|D\psi(x)| + |D\eta(x)|(u(x) - u_{B_R}) + (1 - \eta(x))|Du(x)|]^p \\ &\quad + a(x) \left[ |D\eta(x)|(\psi(x) - \psi_{B_R}) + \eta(x)|D\psi(x)| + |D\eta(x)|(u(x) - u_{B_R}) + (1 - \eta(x))|Du(x)| \right]^q \mathrm{d}x \\ &\leq C \int_{B_s} (1 - \eta(x))^p (|Du|^p + a(x)|Du|^q) \mathrm{d}x \\ &\quad + C \int_{B_s} \left[ \left| \frac{u(x) - u_{B_R}}{s - t} \right|^p + a(x) \left| \frac{u(x) - u_{B_R}}{s - t} \right|^q \right] \mathrm{d}x \\ &\quad + C \int_{B_s} \left[ \left| \frac{\psi(x) - \psi_{B_R}}{s - t} \right|^p + a(x) \left| \frac{\psi(x) - \psi_{B_R}}{s - t} \right|^q \right] \mathrm{d}x \\ &\quad + C \int_{B_s} (|D\psi(x)|^p + a(x)|D\psi(x)|^q) \mathrm{d}x \\ &\leq C \int_{B_s \setminus B_t} H(x, Du(x)) \mathrm{d}x \\ &\quad + \frac{C}{|s - t|^p} \int_{B_R} |u(x) - u_{B_R}|^p \mathrm{d}x + \frac{C}{|s - t|^q} \int_{B_R} a(x)|u(x) - u_{B_R}|^q \mathrm{d}x \\ &\quad + \frac{C}{|s - t|^p} \int_{B_R} |\psi(x) - \psi_{B_R}|^p \mathrm{d}x + \frac{C}{|s - t|^q} \int_{B_R} a(x)|\psi(x) - \psi_{B_R}|^q \mathrm{d}x \\ &\quad + C \int_{B_R} H(x, D\psi(x)) \mathrm{d}x. \end{split}$$

Adding the quantity  $C \int_{B_t} H(x, Du(x)) dx$  to both sides of the previous estimate, by Lemma 5.0.2 we get

$$\begin{split} \int_{B_{\frac{R}{2}}} H(x, Du(x)) \mathrm{d}x &\leq C \left[ \frac{1}{R^{p}} \int_{B_{R}} |u(x) - u_{B_{R}}|^{p} \mathrm{d}x + \frac{1}{R^{q}} \int_{B_{R}} a(x) |u(x) - u_{B_{R}}|^{q} \mathrm{d}x \\ &+ \frac{1}{R^{p}} \int_{B_{R}} |\psi(x) - \psi_{B_{R}}|^{p} \mathrm{d}x + \frac{1}{R^{q}} \int_{B_{R}} a(x) |\psi(x) - \psi_{B_{R}}|^{q} \mathrm{d}x \\ &+ \int_{B_{R}} H(x, D\psi(x)) \mathrm{d}x \right]. \end{split}$$

Setting  $\tilde{H}(x, u(x)) := |u(x)|^p + a(x)|u(x)|^q$  and  $\tilde{H}(x, \psi(x)) := |\psi(x)|^p + a(x)|\psi(x)|^q$ , we can write the previous inequality as

$$\int_{B_{\frac{R}{2}}} H(x, Du(x)) \mathrm{d}x$$

$$\leq \int_{B_R} \tilde{H}\left(x, \frac{u(x) - u_{B_R}}{R}\right) \mathrm{d}x + \int_{B_R} \tilde{H}\left(x, \frac{\psi(x) - \psi_{B_R}}{R}\right) \mathrm{d}x + \int_{B_R} H(x, D\psi(x)) \mathrm{d}x.$$
(7.9)

According to [105, Theorem 2.13] and Hölder's inequality, it holds

$$\int_{B_R} \tilde{H}\left(x, \frac{u(x) - u_{B_R}}{R}\right) \mathrm{d}x \leq \left(\int_{B_R} \left(\tilde{H}\left(x, \frac{u(x) - u_{B_R}}{R}\right)\right)^{d_1} \mathrm{d}x\right)^{\frac{1}{d_1}}$$
$$\leq \left(\int_{B_R} \left(H(x, Du(x))\right)^{d_2} \mathrm{d}x\right)^{\frac{1}{d_2}}, \tag{7.10}$$

where  $d_2 < 1 < d_1$  depend on  $n, p, q, \alpha$ . Analogously,

$$\int_{B_R} \tilde{H}\left(x, \frac{\psi(x) - \psi_{B_R}}{R}\right) \mathrm{d}x \le \left(\int_{B_R} \left(\tilde{H}\left(x, \frac{\psi(x) - \psi_{B_R}}{R}\right)\right)^{d_1} \mathrm{d}x\right)^{\frac{1}{d_1}} \le \left(\int_{B_R} \left(H(x, D\psi(x))\right)^{d_2} \mathrm{d}x\right)^{\frac{1}{d_2}}.$$
(7.11)

Inserting (7.10) and (7.11) in (7.9) and exploiting Hölder's inequality, we infer

$$\int_{B_{\frac{R}{2}}} H(x, Du(x)) \mathrm{d}x \le C \left[ \left( \int_{B_{R}} (H(x, Du(x)))^{d_{2}} \right)^{\frac{1}{d_{2}}} + \int_{B_{R}} H(x, D\psi(x)) \mathrm{d}x \right].$$
(7.12)

Since  $H(x, D\psi(x)) \in L^{m_1}$ , for  $m_1 > 1$ , from Gehring's lemma proved in [64] it follows that there exists  $m_1 > m_2 > 1$  s.t.  $H(x, Du(x)) \in L^{m_2}$ . Then, holding to  $d_2 < 1$ , we might write

$$\int_{B_{\frac{R}{2}}} (H(x, Du(x)))^{m_2} dx \le C \left[ \left( \int_{B_R} H(x, Du(x)) dx \right)^{m_2} + \int_{B_R} (H(x, D\psi(x)))^{m_2} dx \right] \\ \le C \left[ \left( \int_{B_R} H(x, Du(x)) dx \right)^{m_2} + \left( \int_{B_R} (H(x, D\psi(x))^{m_1} dx \right)^{\frac{m_2}{m_1}} \right].$$

Hence,

$$\left(\int_{B_{\frac{R}{2}}} (H(x, Du(x)))^{m_2} \mathrm{d}x\right)^{\frac{1}{m_2}} \le C \left[\int_{B_R} H(x, Du(x)) \mathrm{d}x + \left(\int_{B_R} (H(x, D\psi(x))^{m_1} \mathrm{d}x\right)^{\frac{1}{m_1}}\right].$$

The higher integrability of the minimizer u stated in Theorem 7.1.2 allows us to prove

the following higher integrability up to the boundary result for the solution to the freezed obstacle problem (7.5).

**Theorem 7.1.3.** Let  $v \in u + W_0^{1,p}(B_{\frac{R}{2}})$  be a solution to the obstacle problem (7.5) where the integrand  $\tilde{F}$  satisfies Assumption 1, for exponents  $2 \leq p < q$  verifying

$$\frac{q}{p} \le 1 + \frac{\alpha}{n}.$$

If the function  $\psi$  is s.t.  $H(x, D\psi) \in L^{m_1}_{loc}(\Omega)$ , for some  $m_1 > 1$ , then  $H(x, Du) \in L^{m_2}_{loc}(\Omega)$ , for some  $m_1 > m_2 > 1$ , and there exist a constant C and an exponent  $m_3$ , with  $m_1 > m_2 > m_3 > 1$ , s.t.  $H(x, Dv) \in L^{m_3}_{loc}(\Omega)$  and

$$\left(\int_{B_{\frac{R}{2}}} \left(H(x,Dv)\right)^{m_3} dx\right)^{\frac{1}{m_3}} \le C\left[\left(\int_{B_R} (H(x,Du))^{m_2} dx\right)^{\frac{1}{m_2}} + \left(\int_{B_R} (H(x,D\psi))^{m_2} dx\right)^{\frac{1}{m_2}}\right]$$

*Proof.* We start setting

$$w(x) := \begin{cases} v(x) & \text{if } x \in B_{\frac{R}{2}}, \\ u(x) & \text{if } x \in B_R \setminus B_{\frac{R}{2}} \end{cases}$$
(7.13)

We first consider  $B_{\rho}(x_1) \subset B_{\frac{R}{2}}$ . In this case the Caccioppoli inequality (7.12) holds, namely

$$\int_{B_{\frac{\rho}{2}}} H(x, Dv) dx \le C \left[ \left( \int_{B_{\rho}} (H(x, Dv))^{d_2} dx \right)^{\frac{1}{d_2}} + \int_{B_{\rho}} H(x, D\psi) dx \right].$$
(7.14)

Let us now focus on the case  $B_{\rho}(x_1) \subset B_R$ , with  $x_1 \in \partial B_{\frac{R}{2}}$ . Fix  $\frac{\rho}{2} \leq t < s \leq \rho$ and a cut-off function  $\eta$  between  $B_s(x_1)$  and  $B_t(x_1)$ , with  $|D\eta| \leq \frac{2}{t-s}$ . Let us set  $g(x) := (1 - \eta(x))v + \eta(x)u(x)$ . It is straightforward that  $g \in u + W_0^{1,p}$  and  $g(x) \geq \psi(x)$ . Since v is a minimizer, according to the definition of H and Assumption 1 (ii), we have

$$\begin{split} \int_{B_t \cap B_{\frac{R}{2}}} H(x, Dv) \mathrm{d}x &\leq C \int_{B_t \cap B_{\frac{R}{2}}} \tilde{F}(x, Dv) \mathrm{d}x \\ &\leq C \int_{B_s \cap B_{\frac{R}{2}}} \tilde{F}(x, Dg) \mathrm{d}x. \end{split}$$

Therefore, from the definitions of g and  $\eta$ , we get

$$\begin{split} &\int_{B_t \cap B_{\frac{R}{2}}} H(x, Dv) \mathrm{d}x \\ \leq & C \left[ \int_{B_s \cap B_{\frac{R}{2}}} \left( \frac{1}{(t-s)^p} |u-v|^p + a(x) \frac{1}{(t-s)^q} |u-v|^q \right) \mathrm{d}x \right. \\ & \left. + \int_{(B_s \setminus B_t) \cap B_{\frac{R}{2}}} H(x, Dv) \mathrm{d}x \right. \\ & \left. + \int_{B_s} H(x, Du) \mathrm{d}x \right]. \end{split}$$

As before, adding the quantity  $C \int_{B_t \cap B_{\frac{R}{2}}} H(x, Dv) dx$  to both sides of the previous inequality, by Lemma 5.0.2 we get

$$\int_{B_{\frac{\rho}{2}}\cap B_{\frac{R}{2}}} H(x, Dv) \mathrm{d}x$$

$$\leq C \left[ \int_{B_{\rho}\cap B_{\frac{R}{2}}} \left( \frac{1}{\rho^{p}} |u-v|^{p} + a(x) \frac{1}{\rho^{q}} |u-v|^{q} \right) \mathrm{d}x + \int_{B_{\rho}} H(x, Du) \mathrm{d}x \right].$$
(7.15)

We set

$$\tilde{H}\left(x,\frac{u-v}{\rho}\right) := \frac{1}{\rho^p}|u-v|^p + a(x)\frac{1}{\rho^q}|u-v|^q.$$

Adapting the argument in [26, Remark 2] and [105, Theorem 2.13] and exploiting Hölder's inequality, we have

$$\int_{B_{\rho}\cap B_{\frac{R}{2}}} \tilde{H}\left(x, \frac{u-v}{\rho}\right) \mathrm{d}x \leq \left(\int_{B_{\rho}\cap B_{\frac{R}{2}}} \left(\tilde{H}\left(x, \frac{u-v}{\rho}\right)\right)^{d_{1}} \mathrm{d}x\right)^{\frac{1}{d_{1}}}$$
$$\leq \left(\int_{B_{\rho}\cap B_{\frac{R}{2}}} \left(H(x, Du - Dv)\right)^{d_{2}} \mathrm{d}x\right)^{\frac{1}{d_{2}}}, \qquad (7.16)$$

where  $d_2 < 1 < d_1$  depend on  $n, p, q, \alpha$ . Inserting (7.16) in (7.15), it yields

$$\begin{split} \int_{B_{\frac{\rho}{2}}\cap B_{\frac{R}{2}}} H(x,Dv) \mathrm{d}x &\leq C \left[ \left( \int_{B_{\rho}} (H(x,Du(x))^{d_{2}} \mathrm{d}x \right)^{\frac{1}{d_{2}}} \\ &+ \left( \int_{B_{\rho}\cap B_{\frac{R}{2}}} (H(x,Dv))^{d_{2}} \mathrm{d}x \right)^{\frac{1}{d_{2}}} \\ &+ \int_{B_{\rho}} H(x,Du(x)) \mathrm{d}x \right] \\ &\leq C \left[ \left( \int_{B_{\rho}\cap B_{\frac{R}{2}}} (H(x,Dv))^{d_{2}} \mathrm{d}x \right)^{\frac{1}{d_{2}}} + \int_{B_{\rho}} H(x,Du(x)) \mathrm{d}x \right]. \end{split}$$

Therefore, from the definition of w in (7.13), we infer

$$\int_{B_{\frac{\rho}{2}}} H(x, Dw(x)) dx \leq C \left[ \left( \int_{B_{\rho}} \left( H(x, Dw(x)) \right)^{d_{2}} dx \right)^{\frac{1}{d_{2}}} + \int_{B_{\rho}} H(x, Du(x)) dx + \int_{B_{\rho}} H(x, D\psi(x)) dx \right].$$
(7.17)

Hence, by (7.14) it follows that (7.17) holds not only if  $B_{\rho}(x_1) \subset B_{\frac{R}{2}}$  or  $B_{\rho}(x_1) \cap B_{\frac{R}{2}} \neq \emptyset$ , but also when  $B_{\rho}(x_1) \subset B_R$  and  $x_1 \in \partial B_{\frac{R}{2}}$ .

We now take care of the case  $B_{\rho}(x_1) \cap \partial B_{\frac{R}{2}} \neq \emptyset$  and  $B_{4\rho} \subset B_R$ . We fix  $x_2 \in B_{\rho}(x_1) \cap \partial B_{\frac{R}{2}}$ .

$$\begin{split} &\int_{B_{\frac{\rho}{2}}(x_1)} H(x, Dw) \mathrm{d}x \\ &\leq 3^N \int_{B_{\frac{3\rho}{2}}(x_2)} H(x, Dw) \mathrm{d}x \\ &\leq C \left[ \left( \int_{B_{3\rho(x_2)}} (H(x, Dw))^{d_2} \mathrm{d}x \right)^{\frac{1}{d_2}} + \int_{B_{3\rho}(x_2)} H(x, Du) \mathrm{d}x + \int_{B_{3\rho}(x_2)} H(x, D\psi) \mathrm{d}x \right] \\ &\leq C \left[ \left( \int_{B_{4\rho(x_1)}} (H(x, Dw))^{d_2} \mathrm{d}x \right)^{\frac{1}{d_2}} + \int_{B_{4\rho}(x_1)} H(x, Du) \mathrm{d}x + \int_{B_{4\rho}(x_1)} H(x, D\psi) \mathrm{d}x \right]. \end{split}$$

Since this estimate holds for every  $B_{\frac{\rho}{2}}$  such that  $B_{4\rho} \subset B_R$ , by a covering argument it follows that inequality (7.17) holds for every  $B_{\frac{\rho}{2}}$  such that  $B_{\rho} \subset B_R$ . Now, since  $H(x, D\psi) \in L^{m_1}, m_1 > 1$ , Theorem 7.1.2 yields that there exists  $m_2$ , with  $1 < m_2 < m_1$ , s.t.  $H(x, Du) \in L^{m_2}$ . Therefore, according to Gehring's lemma, there exists  $m_3$ , with  $1 < m_3 < m_2 < m_1$ , such that,

$$\left( \int_{B_{\frac{\rho}{2}}(x_1)} (H(x, Dw))^{m_3} dx \right)^{\frac{1}{m_3}}$$
  
 
$$\leq C \left[ \int_{B_{\rho}(x_1)} H(x, Dw(x)) dx + \left( \int_{B_{\rho}(x_1)} (H(x, Du(x)))^{m_2} dx \right)^{\frac{1}{m_2}} + \left( \int_{B_{\rho}(x_1)} (H(x, D\psi(x)))^{m_2} dx \right)^{\frac{1}{m_2}} \right].$$

In particular, for  $\rho \equiv R$  and  $x_1 = x_0$ , recalling the definition of w we have

$$\begin{split} & \left( \int_{B_{\frac{R}{2}}} (H(x,Dv))^{m_{3}} dx \right)^{\frac{1}{m_{3}}} \\ \leq & C \left[ \int_{B_{\frac{R}{2}}} H(x,Dv) dx + \int_{B_{R} \setminus B_{\frac{R}{2}}} H(x,Du(x)) dx \\ & + \left( \int_{B_{R}} (H(x,Du(x)))^{m_{2}} dx \right)^{\frac{1}{m_{2}}} + \left( \int_{B_{R}} (H(x,D\psi(x)))^{m_{2}} dx \right)^{\frac{1}{m_{2}}} \right] \\ \leq & C \left[ \int_{B_{\frac{R}{2}}} H(x,Dv) dx \\ & + \left( \int_{B_{R}} (H(x,Du(x)))^{m_{2}} dx \right)^{\frac{1}{m_{2}}} + \left( \int_{B_{R}} (H(x,D\psi(x)))^{m_{2}} dx \right)^{\frac{1}{m_{2}}} \right]. \end{split}$$

Since v is a minimizer and recalling that  $m_2 > 1$ , it holds

$$\left( \int_{B_{\frac{R}{2}}} (H(x, Dv))^{m_3} dx \right)^{\frac{1}{m_3}}$$
  
  $\leq C \left[ \left( \int_{B_R} (H(x, Du(x)))^{m_2} dx \right)^{\frac{1}{m_2}} + \left( \int_{B_R} (H(x, D\psi(x)))^{m_2} dx \right)^{\frac{1}{m_2}} \right],$ 

i.e. the conclusion.

*Remark* 3. We point out that Theorems 7.1.2 and 7.1.3 hold true also under the more general hypothesis q > p > 1. However, they are stated for  $q > p \ge 2$  for later purpose in Section 7.4.

# 7.2 Higher differentiability for comparison maps

In Chapter 6 we established in Theorem 6.0.1 the higher differentiability of the solution v to (7.5) under more general assumptions on the coefficients. Indeed, the result presented in this Section recalls Theorem 6.0.1 with  $r = \infty$ . Here, we only give the proof of the a priori bounds, focusing on the differences with respect to the proof carried out in Section 6.2.1. This is crucial in order to establish precise estimates on the difference quotient that will be pivotal for the comparison argument. On the other hand, the approximation procedure is achieved using the same arguments in Section 6.2.2, therefore it will not be presented.

Before stating the result, it is worth noticing that Assumption 1 implies that there exist positive constants  $\tilde{l}, \tilde{\nu}, \tilde{L}$  such that the following conditions are satisfied:

$$|D_{\xi}\tilde{F}(x,\xi)| \le \tilde{l}(|\xi|^{p-1} + a(x)|\xi|^{q-1})$$
(Ā1)

$$\langle D_{\xi}\tilde{F}(x,\xi) - D_{\xi}\tilde{F}(x,\eta), \xi - \eta \rangle \ge \tilde{\nu}(|\xi - \eta|^2 (|\xi|^2 + |\eta|^2)^{\frac{p-2}{2}} + a(x)|\xi - \eta|^2 (|\xi|^2 + |\eta|^2)^{\frac{q-2}{2}}) \quad (\bar{A}2)$$

$$|D_{\xi}\tilde{F}(x,\xi) - D_{\xi}\tilde{F}(x,\eta)| \le \tilde{L}(|\xi - \eta|(|\xi|^{2} + |\eta|^{2})^{\frac{p-2}{2}} + a(x)|\xi - \eta|(|\xi|^{2} + |\eta|^{2})^{\frac{q-2}{2}}) \quad (\bar{A}3)$$

$$|D_{\xi}\tilde{F}(x,\xi) - D_{\xi}\tilde{F}(y,\xi)| \le |x - y|^{\alpha} |\xi|^{q-1}$$
 (Ā4)

for every  $x, y \in \Omega$  and every  $\xi, \eta \in \mathbb{R}^n$ .

The following lemma holds:

**Lemma 7.2.1.** Let  $v \in u + W_0^{1,p}(B)$  be the solution to (7.5) under Assumption 1, for exponents  $2 \leq p < \frac{n}{\alpha}$ , p < q satisfying

$$\frac{q}{p} < 1 + \frac{\alpha}{n}.\tag{7.18}$$

If

$$D\psi \in B^{\gamma}_{2q-p,\infty}(B),$$

for  $0 < \alpha < \gamma < 1$ , then

$$V_p(Dv) \in B^{\alpha}_{2,\infty,loc}(B)$$

and the following estimate

$$\int_{B_{r/4}} (|\tau_h V_p(Dv)|^2 + a(x+h)|\tau_h V_q(Dv)|^2) dx \\
\leq C|h|^{2\alpha} \left\{ \frac{1}{r^{2\tilde{p}}} \left( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) dx \right)^{\kappa} + [D\psi]^{2q-p}_{B_{2q-p,\infty}^{\gamma}(B_r)} \right\},$$
(7.19)

holds for all balls  $B_{r/4} \subset B_r \Subset B$ , for some  $\mu < \alpha$ , with  $C := C(n, p, q, \mu, \|a\|_{\infty}, \|D\psi\|_{B^{\gamma}_{2q-n,\infty}})$ ,  $\tilde{p}:=\tilde{p}(n,p,q,\mu)>1 \ and \ \kappa:=\kappa(n,p,q,\mu)<\tilde{p}.$ 

*Proof.* We a priori assume that  $Dv \in L^{\frac{np}{n-2\mu}}_{\text{loc}}(B)$ , for all  $\frac{\alpha n}{n+2\alpha} < \mu < \alpha$ . In the sequel we will profusely use the following inequality:

$$2q - p \le \frac{np}{n - 2\mu},\tag{7.20}$$

for  $\mu \in \left[\frac{\alpha n}{n+2\alpha}, \alpha\right)$ . Indeed,

$$2q - p \le \frac{np}{n - 2\mu} \Leftrightarrow \frac{q}{p} \le \frac{n - \mu}{n - 2\mu}$$

and

$$1 + \frac{\alpha}{n} \le \frac{n-\mu}{n-2\mu} \Leftrightarrow \mu \ge \frac{\alpha n}{n+2\alpha}$$

Fix  $0 < \frac{r}{4} < \rho < s < t < t' < \frac{r}{2}$  such that  $B_r \Subset B$  and a cut-off function  $\eta \in \mathcal{C}_0^1(B_t)$  such that  $0 \le \eta \le 1$ ,  $\eta = 1$  on  $B_s$ ,  $|D\eta| \le \frac{C}{t-s}$ .

Now, for  $|h| < \frac{r}{4}$ , we consider functions

$$w_1(x) = \eta^2(x)[(v - \psi)(x + h) - (v - \psi)(x)]$$

and

$$w_2(x) = \eta^2 (x-h)[(v-\psi)(x-h) - (v-\psi)(x)].$$

Then

$$\varphi_1(x) = v(x) + tw_1(x),$$
(7.21)

$$\varphi_2(x) = v(x) + tw_2(x)$$
 (7.22)

are admissible test functions for all  $t \in [0, 1)$ .

Arguing analogously as in the proof of Theorem 6.2.1, we obtain the following estimate

$$0 \ge \int_{\Omega} \langle D_{\xi} H(x+h, Dv(x+h)) - D_{\xi} H(x+h, Dv(x)), \eta^2 D\tau_h v \rangle \mathrm{d}x$$

$$-\int_{\Omega} \langle D_{\xi} H(x+h, Dv(x+h)) - D_{\xi} H(x+h, Dv(x)), \eta^{2} D\tau_{h} \psi \rangle dx + \int_{\Omega} \langle D_{\xi} H(x+h, Dv(x+h)) - D_{\xi} H(x+h, Dv(x)), 2\eta D\eta \tau_{h} (v-\psi) \rangle dx + \int_{\Omega} \langle D_{\xi} H(x+h, Dv(x)) - D_{\xi} H(x, Dv(x)), \eta^{2} D\tau_{h} v \rangle dx - \int_{\Omega} \langle D_{\xi} H(x+h, Dv(x)) - D_{\xi} H(x, Dv(x)), \eta^{2} D\tau_{h} \psi \rangle dx + \int_{\Omega} \langle D_{\xi} H(x+h, Dv(x)) - D_{\xi} H(x, Dv(x)), 2\eta D\eta \tau_{h} (v-\psi) \rangle dx =: I_{1} + I_{2} + I_{3} + I_{4} + I_{5} + I_{6},$$
(7.23)

that yields

$$I_1 \le |I_2| + |I_3| + |I_4| + |I_5| + |I_6|.$$
(7.24)

The ellipticity assumption  $(\bar{A}2)$  and the properties of a(x) imply

$$I_{1} \geq \tilde{\nu} \int_{\Omega} \eta^{2} |\tau_{h} Dv|^{2} (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{p-2}{2}} dx + \tilde{\nu} \int_{\Omega} \eta^{2} a(x+h) |\tau_{h} Dv|^{2} (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{q-2}{2}} dx \geq \tilde{\nu} \int_{\Omega} \eta^{2} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h) |\tau_{h} V_{q}(Dv)|^{2}) dx.$$
(7.25)

From the growth condition ( $\overline{A}3$ ), the boundedness of a(x) and Young's inequality, we get

$$\begin{split} |I_{2}| &\leq \tilde{L} \int_{\Omega} \eta^{2} |\tau_{h} Dv| (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{p-2}{2}} |\tau_{h} D\psi| dx \\ &+ \tilde{L} \int_{\Omega} \eta^{2} a(x+h) |\tau_{h} Dv| (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{q-2}{2}} |\tau_{h} D\psi| dx \\ &\leq \tilde{L} \int_{\Omega} \eta^{2} |\tau_{h} Dv| (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{p-2}{2}} |\tau_{h} D\psi| dx \\ &+ \tilde{L} ||a||_{\infty} \int_{\Omega} \eta^{2} |\tau_{h} Dv| (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{q-2}{2}} |\tau_{h} D\psi| dx \\ &\leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} Dv|^{2} (|Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{p-2}{2}} dx \\ &+ C_{\varepsilon} (\tilde{L}, ||a||_{\infty}) \int_{\Omega} \eta^{2} |\tau_{h} D\psi|^{2} (1 + |Dv(x+h)|^{2} + |Dv(x)|^{2})^{\frac{2q-p-2}{2}} dx. \end{split}$$

The calculations performed in Theorem 6.2.1 and Lemma 5.0.1 lead us to the following

estimate for the integral  ${\cal I}_2$ 

$$|I_{2}| \leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} V_{p}(Dv)|^{2} dx + C_{\varepsilon}(\tilde{L}, p, q, ||a||_{\infty}) |h|^{2\gamma} [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_{r})} + C_{\varepsilon}(\tilde{L}, p, q, ||a||_{\infty}) |h|^{2\gamma} \int_{B_{t'}} (1 + |Dv|)^{2q-p} dx.$$
(7.26)

Now, we consider the integral  $I_3$ . From assumption ( $\overline{A}3$ ), hypothesis  $|D\eta| \leq \frac{C}{t-s}$  and Young's inequality, we get

$$\begin{aligned} |I_3| &\leq 2\tilde{L} \int_{\Omega} |D\eta| \eta |\tau_h Dv| (|Dv(x+h)|^2 + |Dv(x)|^2)^{\frac{p-2}{2}} |\tau_h(v-\psi)| \mathrm{d}x \\ &+ 2\tilde{L} \|a\|_{\infty} \int_{\Omega} |D\eta| \eta |\tau_h Dv| (1 + |Dv(x+h)|^2 + |Dv(x)|^2)^{\frac{q-2}{2}} |\tau_h(v-\psi)| \mathrm{d}x \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h Dv|^2 (|Dv(x+h)|^2 + |Dv(x)|^2)^{\frac{p-2}{2}} \mathrm{d}x \\ &+ \frac{C_{\varepsilon} (L, \|a\|_{\infty})}{(t-s)^2} \int_{B_t} |\tau_h(v-\psi)|^2 (|Dv(x+h)|^2 + |Dv(x)|^2)^{\frac{2q-p-2}{2}} \mathrm{d}x, \end{aligned}$$

where we also used the boundedness of function a(x).

Arguing analogously as in the proof of Theorem 6.2.1, we can estimate the integral  $I_3$  as follows

$$|I_{3}| \leq \varepsilon \int_{\Omega} \eta^{2} |\tau_{h} V_{p}(Dv)|^{2} \mathrm{d}x + \frac{C_{\varepsilon}(\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2}} |h|^{2} \int_{B_{r}} |D\psi|^{2q-p} \mathrm{d}x + \frac{C_{\varepsilon}(\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2}} |h|^{2} \int_{B_{t'}} (1+|Dv|)^{2q-p} \mathrm{d}x.$$
(7.27)

In order to estimate the integral  $I_4$ , we use assumption ( $\overline{A}4$ ), Young's inequality and Lemma 5.0.1 as follows

$$\begin{split} |I_4| &\leq \int_{\Omega} \eta^2 |\tau_h Dv| |h|^{\alpha} |Dv|^{q-1} \mathrm{d}x \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h Dv|^2 (|Dv(x+h)|^2 + |Dv(x)|^2)^{\frac{p-2}{2}} \mathrm{d}x \\ &+ C_{\varepsilon} |h|^{2\alpha} \int_{B_t} |Dv|^{2q-p} \mathrm{d}x \\ &\leq \varepsilon \int_{\Omega} \eta^2 |\tau_h V_p(Dv)|^2 \mathrm{d}x \end{split}$$

$$+ C_{\varepsilon} |h|^{2\alpha} \int_{B_t} |Dv|^{2q-p} \mathrm{d}x.$$
(7.28)

We now take care of  $I_5$ . Similarly as above, exploiting assumption ( $\overline{A}4$ ) and Hölder's inequality, we infer

$$I_{5}| \leq \int_{\Omega} \eta^{2} |\tau_{h} D\psi| |h|^{\alpha} |Dv|^{q-1} \mathrm{d}x$$
  
$$\leq |h|^{\alpha} \left( \int_{B_{t}} |\tau_{h} D\psi|^{2q-p} \mathrm{d}x \right)^{\frac{1}{2q-p}} \left( \int_{B_{t}} |Dv|^{\frac{(q-1)(2q-p)}{2q-p-1}} \mathrm{d}x \right)^{\frac{2q-p-1}{2q-p}}$$

Now, we observe

$$\frac{(q-1)(2q-p)}{2q-p-1} < 2q-p \Leftrightarrow p < q.$$

$$(7.29)$$

Hence

$$|I_{5}| \leq |h|^{\alpha+\gamma} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_{r})} \left( \int_{B_{t}} |Dv|^{2q-p} \mathrm{d}x \right)^{\frac{q-1}{2q-p}} \\ \leq C(q) |h|^{\alpha+\gamma} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_{r})}^{q} \\ + C(q) |h|^{\alpha+\gamma} \left( \int_{B_{t}} |Dv|^{2q-p} \mathrm{d}x \right)^{\frac{q}{2q-p}}.$$
(7.30)

From assumption (Ā4), hypothesis  $|D\eta| \leq \frac{C}{t-s}$  and Hölder's inequality, we infer the following estimate for  $I_6$ .

$$\begin{split} |I_{6}| &\leq \frac{C}{t-s} |h|^{\alpha} \int_{B_{t}} |\tau_{h}\psi| |Dv|^{q-1} \mathrm{d}x \\ &+ \frac{C}{t-s} |h|^{\alpha} \int_{B_{t}} |\tau_{h}v| |Dv|^{q-1} \mathrm{d}x \\ &\leq \frac{C}{t-s} |h|^{\alpha} \left( \int_{B_{t}} |\tau_{h}\psi|^{2q-p} \mathrm{d}x \right)^{\frac{1}{2q-p}} \left( \int_{B_{t}} |Dv|^{\frac{(q-1)(2q-p)}{2q-p-1}} \mathrm{d}x \right)^{\frac{2q-p-1}{2q-p}} \\ &+ \frac{C}{t-s} |h|^{\alpha} \left( \int_{B_{t}} |\tau_{h}v|^{2q-p} \mathrm{d}x \right)^{\frac{1}{2q-p}} \left( \int_{B_{t}} |Dv|^{\frac{(q-1)(2q-p)}{2q-p-1}} \mathrm{d}x \right)^{\frac{2q-p-1}{2q-p}}. \end{split}$$

Using Lemma 5.2.2, (7.29) and Hölder's and Young's inequality, we have

$$|I_6| \leq \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_{t'}} |D\psi|^{2q-p} \mathrm{d}x \right)^{\frac{1}{2q-p}} \left( \int_{B_t} |Dv|^{\frac{(q-1)(2q-p)}{2q-p-1}} \mathrm{d}x \right)^{\frac{2q-p-1}{2q-p}}$$

$$+ \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_{t'}} |Dv|^{2q-p} dx \right)^{\frac{1}{2q-p}} \left( \int_{B_t} |Dv|^{\frac{(q-1)(2q-p)}{2q-p-1}} dx \right)^{\frac{2q-p-1}{2q-p}} \\ \leq \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_r} |D\psi|^{2q-p} dx \right)^{\frac{1}{2q-p}} \left( \int_{B_t} |Dv|^{2q-p} dx \right)^{\frac{q-1}{2q-p}} \\ + \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_{t'}} |Dv|^{2q-p} dx \right)^{\frac{q}{2q-p}} \\ \leq \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_r} |D\psi|^{2q-p} dx \right)^{\frac{q}{2q-p}} \\ + \frac{C(n,p,q)}{t-s} |h|^{\alpha+1} \left( \int_{B_{t'}} |Dv|^{2q-p} dx \right)^{\frac{q}{2q-p}}.$$
(7.31)

Inserting estimates (7.25), (7.26), (7.27), (7.28), (7.30) and (7.31) in (7.24), we infer

$$\begin{split} \nu \int_{\Omega} \eta^{2} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h) |\tau_{h} V_{p}(Dv)|^{2}) dx \\ \leq & 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} V_{p}(Dv)|^{2} dx \\ &+ C_{\varepsilon}(\tilde{L}, p, q, ||a||_{\infty}) |h|^{2\gamma} [D\psi]_{B_{2q-p,\infty}^{2q-p}}^{2q-p} \\ &+ C_{\varepsilon}(\tilde{L}, p, q, ||a||_{\infty}) |h|^{2\gamma} \int_{B_{t'}} (1+|Dv|)^{2q-p} dx \\ &+ \frac{C_{\varepsilon}(\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2}} |h|^{2} \int_{B_{r}} |D\psi|^{2q-p} dx \\ &+ \frac{C_{\varepsilon}(\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2}} |h|^{2} \int_{B_{t'}} (1+|Dv|)^{2q-p} dx \\ &+ C_{\varepsilon} |h|^{2\alpha} \int_{B_{t}} |Dv|^{2q-p} dx \\ &+ C(q) |h|^{\alpha+\gamma} [D\psi]_{B_{2q-p,\infty}^{q}(B_{r})}^{q} + C(q) |h|^{\alpha+\gamma} \left( \int_{B_{t}} |Dv|^{2q-p} dx \right)^{\frac{q}{2q-p}} \\ &+ \frac{C(n, p, q)}{t-s} |h|^{\alpha+1} \left( \int_{B_{r}} |D\psi|^{2q-p} dx \right)^{\frac{q}{2q-p}}. \end{split}$$
(7.32)

We now introduce the following interpolation inequality

$$\|Dw\|_{2q-p} \le \|Dw\|_p^{\delta} \|Dw\|_{\frac{np}{n-2\mu}}^{1-\delta}, \tag{7.33}$$
where  $0 < \delta < 1$  is defined through the condition

$$\frac{1}{(2q-p)} = \frac{\delta}{p} + \frac{(1-\delta)(n-2\mu)}{np}$$
(7.34)

which implies

$$\delta = \frac{n(p-q) + \mu(2q-p)}{\mu(2q-p)}, \quad 1-\delta = \frac{n(q-p)}{\mu(2q-p)}.$$

Hence we get the following inequalities

$$\int_{B_{t'}} (1+|Dv|)^{2q-p} \mathrm{d}x \le \left( \int_{B_{t'}} (1+|Dv|)^p \mathrm{d}x \right)^{\frac{\delta(2q-p)}{p}} \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{(n-2\mu)(q-p)}{\mu p}},$$
(7.35)

$$\left(\int_{B_{t'}} |Dv|^{2q-p} \mathrm{d}x\right)^{\frac{q}{2q-p}} \leq \left(\int_{B_{t'}} |Dv|^p \mathrm{d}x\right)^{\frac{\delta q}{p}} \cdot \left(\int_{B_{t'}} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{(n-2\mu)qp'}{p}},\tag{7.36}$$

where  $p' = \frac{q-p}{\mu(2q-p)}$ . Inserting (7.35) and (7.36) in (7.32), and exploiting the bounds

$$\frac{n(q-p)}{\mu p} < 1, \quad \frac{nq(q-p)}{\mu p(2q-p)} < 1, \tag{7.37}$$

which hold by assumption (7.18) and for  $\mu \in (\frac{n(q-p)}{p}, \alpha)$ , from Young's inequality, we infer

$$\begin{split} \nu & \int_{\Omega} \eta^{2} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h) |\tau_{h} V_{p}(Dv)|^{2}) \mathrm{d}x \\ \leq & 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} V_{p}(Dv)|^{2} \mathrm{d}x \\ & + C_{\varepsilon} (\tilde{L}, p, q, ||a||_{\infty}) |h|^{2\gamma} [D\psi]_{B_{2q-p,\infty}^{2}(B_{r})}^{2q-p} \\ & + C_{\varepsilon,\theta} (\tilde{L}, n, p, q, ||a||_{\infty}) |h|^{2\gamma} \left( \int_{B_{r}} (1+|Dv|)^{p} \mathrm{d}x \right)^{\frac{\delta(2q-p)\tilde{p}}{p}} \\ & + \theta |h|^{2\gamma} \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} \\ & + \frac{C_{\varepsilon} (\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2}} |h|^{2} \int_{B_{r}} |D\psi|^{2q-p} \mathrm{d}x \\ & + \theta |h|^{2} \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} + \frac{C_{\varepsilon,\theta} (\tilde{L}, n, p, q, ||a||_{\infty})}{(t-s)^{2\tilde{p}}} |h|^{2} \left( \int_{B_{r}} (1+|Dv|)^{p} \mathrm{d}x \right)^{\frac{\tilde{p}\delta(2q-p)}{p}} \\ & + C_{\varepsilon,\theta} |h|^{2\alpha} \left( \int_{B_{r}} |Dv|^{p} \mathrm{d}x \right)^{\frac{\tilde{p}\delta(2q-p)}{p}} + \theta |h|^{2\alpha} \left( \int_{B_{t}} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} \end{split}$$

$$+ C_{\theta}(q)|h|^{\alpha+\gamma} [D\psi]_{B_{2q-p,\infty}^{\gamma}(B_{r})}^{q} + C_{\theta}(n,p,q)|h|^{\alpha+\gamma} \left(\int_{B_{r}} |Dv|^{p} \mathrm{d}x\right)^{\frac{p^{*}\delta q}{p}} \\ + \theta|h|^{\alpha+\gamma} \left(\int_{B_{t}} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}} \\ + \frac{C_{\theta}(n,p,q)}{t-s}|h|^{\alpha+1} \left(\int_{B_{r}} |D\psi|^{2q-p} \mathrm{d}x\right)^{\frac{q}{2q-p}} \\ + \frac{C_{\theta}(n,p,q)}{(t-s)^{p^{*}}}|h|^{\alpha+1} \left(\int_{B_{r}} |Dv|^{p} \mathrm{d}x\right)^{\frac{p^{*}\delta q}{p}} + \theta|h|^{\alpha+1} \left(\int_{B_{t'}} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}}.$$
(7.38)

for some constant  $\theta \in (0, 1)$ , where we set  $\tilde{p} = \frac{\mu p}{\mu p - n(q-p)}$ ,  $p^* = \frac{\mu p(2q-p)}{\mu p(2q-p) - n(q-p)q}$ . For a better readability we now define

$$\begin{split} A = & C_{\varepsilon}(\tilde{L}, p, q, \|a\|_{\infty}) [D\psi]_{B_{2q-p,\infty}(B_{r})}^{2q-p} + C_{\varepsilon,\theta}(\tilde{L}, n, p, q, \|a\|_{\infty}) \left( \int_{B_{r}} (1+|Dv|)^{p} dx \right)^{\frac{\delta(2q-p)\tilde{p}}{p}} \\ & + C_{\varepsilon,\theta} \left( \int_{B_{r}} |Dv|^{p} dx \right)^{\frac{\tilde{p}\delta(2q-p)}{p}} \\ & + C_{\theta}(q) |h|^{\alpha+\gamma} [D\psi]_{B_{2q-p,\infty}(B_{r})}^{q} + C_{\theta}(n, p, q) |h|^{\alpha+\gamma} \left( \int_{B_{r}} |Dv|^{p} dx \right)^{\frac{p^{*}\delta q}{p}} \\ B_{1} = & C_{\varepsilon}(\tilde{L}, n, p, q, \|a\|_{\infty}) \int_{B_{r}} |D\psi|^{2q-p} dx, \\ B_{2} = & C_{\varepsilon,\theta}(\tilde{L}, n, p, q, \|a\|_{\infty}) \left( \int_{B_{r}} (1+|Dv|)^{p} dx \right)^{\frac{\tilde{p}\delta(2q-p)}{p}}, \\ B_{3} = & C_{\theta}(n, p, q) \left( \int_{B_{r}} |D\psi|^{2q-p} dx \right)^{\frac{q}{2q-p}}, \\ B_{4} = & C_{\theta}(n, p, q) \left( \int_{B_{r}} |Dv|^{p} dx \right)^{\frac{p^{*}\delta q}{p}}, \end{split}$$

so that we can rewrite the previous estimate as

$$\begin{split} \nu & \int_{\Omega} \eta^{2} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h) |\tau_{h} V_{p}(Dv)|^{2}) \mathrm{d}x \\ \leq & 3\varepsilon \int_{\Omega} \eta^{2} |\tau_{h} V_{p}(Dv)|^{2} \mathrm{d}x \\ & + \theta (|h|^{2\alpha} + |h|^{\alpha+\gamma}) \left( \int_{B_{t}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} \\ & + \theta (|h|^{2} + |h|^{2\gamma} + |h|^{\alpha+1}) \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} \end{split}$$

$$+ (|h|^{2\gamma} + |h|^{2\alpha} + |h|^{\alpha+\gamma})A + |h|^2 \frac{B_1}{(t-s)^2} + |h|^2 \frac{B_2}{(t-s)^{2\tilde{p}}} + |h|^{\alpha+1} \frac{B_3}{(t-s)^{p''}} + |h|^{\alpha+1} \frac{B_4}{(t-s)^{p^*}}.$$

Choosing  $\varepsilon = \frac{\nu}{6}$ , we can reabsorb the first integral in the right hand side of the previous estimate by the left hand side, thus getting

$$\int_{\Omega} \eta^{2} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h)|\tau_{h} V_{q}(Dv)|^{2}) \mathrm{d}x$$

$$\leq 3\theta |h|^{2\alpha} \left( \int_{B_{t}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} + 3\theta |h|^{2\alpha} \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x \right)^{\frac{n-2\mu}{n}} + |h|^{2\alpha} A + |h|^{2} \frac{B_{1}}{(t-s)^{2}} + |h|^{2} \frac{B_{2}}{(t-s)^{2\tilde{p}}} + |h|^{2\alpha} \frac{B_{3}}{t-s} + |h|^{2\alpha} \frac{B_{4}}{(t-s)^{p^{*}}}, \quad (7.39)$$

where we used the fact that  $\alpha < \gamma$ .

Since the right hand side of (7.39) depends on the integrability of Dv, in order to exploit inequality (7.19), we need to derive an a priori estimate for the gradient of the minimzer v. First, we bound (7.39) from below as follows

$$\int_{B_{s}} |\tau_{h} V_{p}(Dv)|^{2} dx 
\leq \int_{B_{s}} (|\tau_{h} V_{p}(Dv)|^{2} + a(x+h)|\tau_{h} V_{q}(Dv)|^{2}) dx 
\leq |h|^{2\alpha} \left\{ 2\theta \left( \int_{B_{t}} (1+|Dv|)^{\frac{np}{n-2\mu}} dx \right)^{\frac{n-2\mu}{n}} + 3\theta \left( \int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} dx \right)^{\frac{n-2\mu}{n}} 
+ A + \frac{B_{1}}{(t-s)^{2}} + \frac{B_{2}}{(t-s)^{2\tilde{p}}} + \frac{B_{3}}{t-s} + \frac{B_{4}}{(t-s)^{p^{*}}} \right\},$$
(7.40)

where we also used that  $\eta = 1$  on  $B_s$ . Then, Lemma 5.1.3 and equality (5.1) imply

$$\left(\int_{B_s} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}} \leq 2\theta \left(\int_{B_t} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}} + 3\theta \left(\int_{B_{t'}} (1+|Dv|)^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}} + A + \frac{B_1}{(t-s)^2} + \frac{B_2}{(t-s)^{2\tilde{p}}} + \frac{B_3}{t-s} + \frac{B_4}{(t-s)^{p^*}}, \quad (7.41)$$

for all  $\mu \in (\frac{n(q-p)}{p}, \alpha)$ .

Now, applaying the iteration Lemma 5.0.2 twice, we obtain

$$\left(\int_{B_{r/4}} |Dv|^{\frac{np}{n-2\mu}} \mathrm{d}x\right)^{\frac{n-2\mu}{n}} \le C|h|^{2\alpha} \left\{ \frac{1}{r^{2\tilde{p}}} \left( \int_{B_r} (1+|Dv|^p+|D\psi|^{2q-p}) \mathrm{d}x \right)^{\kappa} + [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_r)} \right\}$$
(7.42)

thus, using Lemma 5.1.3, from inequalities (7.42) and (7.40), we deduce the a priori estimate

$$\int_{B_{r/4}} (|\tau_h V_p(Dv)|^2 + a(x+h)|\tau_h V_q(Dv)|^2) \mathrm{d}x$$
  
$$\leq C|h|^{2\alpha} \left\{ \frac{1}{r^{2\tilde{p}}} \left( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) \mathrm{d}x \right)^{\kappa} + [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_r)} \right\},$$
(7.43)

for some  $\mu < \alpha$ , where  $C := C(n, p, q, \mu, ||a||_{\infty})$  and  $\kappa := \frac{\delta(2q-p)\tilde{p}}{p} < \tilde{p}$ .

According to the previous result, we state the following remarks, which will be crucial for the proof of Theorem 7.4.1.

Remark 4. From Proposition 5.2.1 (iii), it follows that

$$|\tau_h(\sqrt{a(x)}V_q(Dv))|^2 \le Ca(x+h)|\tau_h V_q(Dv)|^2 + C|V_q(Dv)|^2|\tau_h a(x)|.$$
(7.44)

Combining (7.43) and (7.44), we obtain

$$\begin{split} &\int_{B_{r/4}} |\tau_h(\sqrt{a(x)}V_q(Dv))|^2 \mathrm{d}x \\ &\leq \int_{B_{r/4}} (|\tau_h V_p(Dv)|^2 + |\tau_h(\sqrt{a(x)}V_q(Dv))|^2 \mathrm{d}x \\ &\leq C \int_{B_{r/4}} (|\tau_h V_p(Dv)|^2 + a(x+h)|\tau_h V_q(Dv)|^2 + |V_q(Dv)|^2 |\tau_h a(x)|) \mathrm{d}x \\ &\leq C |h|^{2\alpha} \bigg\{ \frac{1}{r^{2\tilde{p}}} \bigg( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) \mathrm{d}x \bigg)^{\kappa} + [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_r)} \bigg\} + C|h|^{\alpha} \|Dv\|^q_{L^q(B_r)} \\ &\leq C |h|^{\alpha} \bigg\{ \frac{1}{r^{2\tilde{p}}} \bigg( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) \mathrm{d}x \bigg)^{\kappa} + \|Dv\|^q_{L^q(B_r)} + [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_r)} \bigg\}, \quad (7.45) \end{split}$$

which is finite by Theorem 7.2.1. Therefore,

$$\sqrt{a(x)}V_q(Dv) \in B^{\frac{\alpha}{2}}_{2,\infty,\mathrm{loc}}(B)$$

Lemma (5.1.3) yields

$$a(x)|Dv|^q \in L^{\frac{n}{n-2\beta}}_{\text{loc}}(B), \quad \forall \beta < \frac{\alpha}{2}.$$

Remark 5. Choosing  $\mu < \alpha$  s.t.  $q = \frac{np}{n-2\mu}$ , estimates (7.42) and (7.45) yield

$$\int_{B_{r/4}} (|\tau_h V_p(Dv)|^2 + |\tau_h(\sqrt{a(x)}V_q(Dv))|^2 \mathrm{d}x$$

$$\leq C|h|^{\alpha} \left\{ \frac{1}{r^{2\tilde{p}_1}} \left( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) \mathrm{d}x \right)^{\kappa} + [D\psi]^{2q-p}_{B^{\gamma}_{2q-p,\infty}(B_r)} \right\}^{\frac{q}{p}}$$

$$\leq C|h|^{\alpha} \left\{ \frac{1}{r^{2\tilde{p}_1}} \left( \int_{B_r} (1+|Dv|^p + |D\psi|^{2q-p}) \mathrm{d}x \right)^{\kappa_1} + [D\psi]^{q_1}_{B^{\gamma}_{2q-p,\infty}(B_r)} + 1 \right\}, \quad (7.46)$$

where  $\frac{\tilde{p}q}{p} = \tilde{p}_1 > 1$ ,  $\frac{\kappa q}{p} = \kappa_1 < \tilde{p}_1$  and  $\frac{(2q-p)q}{p} = q_1 < \tilde{p}_1$ , with  $\tilde{p}$  and  $\kappa$  introduced in (7.38) and (7.43) respectively.

### 7.3 Comparison

In this section we prove a comparison Lemma (see Lemma 7.3.2 below), where we estimate the distance between the solution u to the problem (7.1) and the solution v to the problem (7.5). In order to do so, we first need the following lemma.

**Lemma 7.3.1.** Let  $\tilde{F} : \Omega \times \mathbb{R}^n \to \mathbb{R}$  be the function defined in (7.4) under Assumption 1. Then there exists a positive constant  $c = c(r, n, \nu)$  such that the following inequality holds for every  $x \in \Omega$  and every  $z_1, z_2 \in \mathbb{R}^n$ 

$$c(|V_p(z_1) - V_p(z_2)|^2 + a(x)|V_q(z_1) - V_q(z_2)|^2) \leq \tilde{F}(x, z_1) - \tilde{F}(x, z_2) - \langle D_{\xi}\tilde{F}(x, z_2), z_1 - z_2 \rangle.$$
(7.47)

*Proof.* We start proving that for every  $r \ge 2$  there exists a constant c = c(r, n) such that

$$c(r,n)|V_r(z_1) - V_r(z_2)|^2 \le g_r(z_1) - g_r(z_2) - \langle D_{\xi}g_r(z_2), z_1 - z_2 \rangle,$$
(7.48)

where we denote  $g_r(z) := |z|^r$ .

Let us consider the function  $G_r : [0,1] \to \mathbb{R}$  defined by  $G_r(t) := g_r(tz_1 + (1-t)z_2)$ . Since  $G_r \in \mathcal{C}^2([0,1])$ , by using Taylor's formula with integral remainder, we obtain

$$G_r(1) = G_r(0) + G'_r(0) + \int_0^1 (1-s)G''_r(s)ds.$$
(7.49)

Since

$$G'_{r}(t) = \langle D_{\xi}g_{r}(tz_{1} + (1-t)z_{2}), z_{1} - z_{2} \rangle,$$
  

$$G''_{r}(t) = \langle D_{\xi\xi}g_{r}(tz_{1} + (1-t)z_{2})(z_{1} - z_{2}), z_{1} - z_{2} \rangle,$$

from (7.49) we get

$$g_r(z_1) - g_r(z_2) - \langle D_{\xi}g_r(z_2), z_1 - z_2 \rangle$$
  
=  $\int_0^1 (1-s) \langle D_{\xi\xi}g_r(sz_1 + (1-s)z_2)(z_1 - z_2), z_1 - z_2 \rangle ds$   
 $\ge c(r) |z_1 - z_2|^2 \int_0^1 (1-s) |sz_1 + (1-s)z_2|^{r-2} ds.$  (7.50)

Now, we want to estimate from below  $|sz_1 + (1 - s)z_2|^{r-2}$ . If  $|z_1| \le |z_2|$  and  $s \in [3/4, 1]$ , then  $-1/4 \le s - 1 \le 0$  and

$$|sz_1 + (1-s)z_2| \ge s|z_1| + (s-1)|z_2| \ge \frac{3}{4}|z_1| - \frac{1}{4}|z_2| \ge \frac{1}{4}(|z_1| + |z_2|),$$

while, if  $|z_2| > |z_2|$  and  $s \in [0, 1/4]$ , then  $3/4 \le 1 - s \le 1$  and

$$|sz_1 + (1-s)z_2| \ge -s|z_1| + (1-s)|z_2| \ge -\frac{1}{4}|z_1| + \frac{3}{4}|z_2| \ge \frac{1}{4}(|z_1| + |z_2|).$$

Therefore

$$|sz_1 + (1-s)z_2|^{r-2} \ge 4^{2-r}(|z_1| + |z_2|)^{r-2}$$
(7.51)

holds on a suitable subinterval of [0, 1]. Eventually, inserting (7.51) in (7.50) we obtain

$$g_r(z_1) - g_r(z_2) - \langle D_{\xi}g_r(z_2), z_1 - z_2 \rangle \ge c(r)(|z_1| + |z_2|)^{r-2}|z_1 - z_2|^2$$
$$\ge c(r, n)|V_r(z_1) - V_r(z_2)|^2,$$

where in the last inequality we used Lemma 5.0.1.

At this point, using the bound from below on b in Assumption 1 and estimate (7.47) we deduce

$$\begin{split} \tilde{F}(x,z_1) &- \tilde{F}(x,z_2) - \langle D_{\xi} \tilde{F}(x,z_2), z_1 - z_2 \rangle \\ &= b(x_0,u_B) [g_p(z_1) - g_p(z_2) - \langle D_{\xi} g_p(z_2), z_1 - z_2 \rangle \\ &+ a(x) (g_q(z_1) - g_q(z_2) - \langle D_{\xi} g_q(z_2), z_1 - z_2 \rangle)] \\ &\geq c(r,n,\nu) (|V_p(z_1) - V_p(z_2)|^2 + a(x)|V_q(z_1) - V_q(z_2)|^2). \end{split}$$

*Remark* 6. In the proof of Lemma 7.3.2 we will take advantage of the higher integrability results established in Section 7.1, in particular in the case  $\frac{q}{p} < 1 + \frac{\alpha}{n}$ .

Indeed, the assumption  $D\psi \in B^{\gamma}_{2q-p,\infty,\text{loc}}(\Omega)$  and Lemma 5.1.3 imply that  $D\psi \in L^{\frac{n(2q-p)}{n-\mu(2q-p)}}$ , for every  $0 < \mu < \gamma$ . Therefore,  $H(x, D\psi)$  belong to some  $L^m$ , with m > 1.

**Lemma 7.3.2.** Let u be a solution to (7.1) and  $v \in u + W_0^{1,p}(B)$  be the solution to (7.5), under Assumptions 1 and 2, for exponents  $2 \le p < q$  verifying

$$\frac{q}{p} < 1 + \frac{\alpha}{n}$$

If

$$D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega),$$

for  $0 < \alpha < \gamma < 1$ , then

$$\int_{B} |V_{p}(Du) - V_{p}(Dv)|^{2} + a(x)|V_{q}(Du) - V_{q}(Dv)|^{2}dx$$
$$\leq CR^{\sigma} \int_{2B} (1 + (H(x, Du))^{m} + (H(x, D\psi))^{m})dx, \quad (7.52)$$

with  $\sigma = \min\{\beta, m-1\}$ , where  $\beta$  is the exponent appearing in the Assumption 2 and where m is the minimum of the two higher integrability exponents of Theorems 7.1.2 and 7.1.3.

*Proof.* Assumption 1, the definition of  $\tilde{F}$  and the minimality of v imply

$$\int_{B} H(x, Dv) \mathrm{d}x \le C \int_{B} \tilde{F}(x, Dv) \mathrm{d}x \le \int_{B} \tilde{F}(x, Du) \mathrm{d}x \le C \int_{B} H(x, Du) \mathrm{d}x, \qquad (7.53)$$

on the other hand, Theorem 7.1.3 yields

$$\int_{B} (H(x, Dv))^{m} \mathrm{d}x \le \int_{B} [(H(x, Du))^{m} + (H(x, D\psi))^{m}] \mathrm{d}x,$$
(7.54)

for some m > 1. From inequality (7.47) we get

$$\begin{split} \int_{B} |V_p(Du) - V_p(Dv)|^2 + a(x)|V_q(Du) - V_q(Dv)|^2 \mathrm{d}x \\ &\leq C \int_{B} \tilde{F}(x, Du) - \tilde{F}(x, Dv) - \langle D_{\xi} \tilde{F}(x, Dv), Du - Dv \rangle \mathrm{d}x, \end{split}$$

moreover, recalling inequality (7.6), i.e.

$$\int_{B} \langle D_{\xi} H(x, Dv), Du - Dv \rangle \mathrm{d}x \ge 0, \tag{7.55}$$

and that  $b(x_0, u_B) \ge \nu > 0$ , we deduce

$$\int_{B} |V_p(Du) - V_p(Dv)|^2 + a(x)|V_q(Du) - V_q(Dv)|^2 \mathrm{d}x \le C \int_{B} \tilde{F}(x, Du) - \tilde{F}(x, Dv) \mathrm{d}x.$$

Hence, we can write the previous estimate as follows

$$\begin{split} \int_{B} |V_{p}(Du) - V_{p}(Dv)|^{2} + a(x)|V_{q}(Du) - V_{q}(Dv)|^{2} dx \\ &\leq C \int_{B} \tilde{F}(x, Du) - \tilde{F}(x, Dv) dx \\ &= C \int_{B} [b(x_{0}, u_{B})H(x, Du) - b(x_{0}, u_{B})H(x, Dv)] dx \\ &= C \int_{B} [b(x_{0}, u_{B})H(x, Du) - b(x, u_{B})H(x, Du)] dx \\ &+ C \int_{B} [b(x, u_{B})H(x, Du) - b(x, u)H(x, Du)] dx \\ &+ C \int_{B} [b(x, u)H(x, Du) - b(x, v)H(x, Dv)] dx \\ &+ C \int_{B} [b(x, v)H(x, Dv) - b(x, v_{B})H(x, Dv)] dx \\ &+ C \int_{B} [b(x, v_{B})H(x, Dv) - b(x, u_{B})H(x, Dv)] dx \\ &+ C \int_{B} [b(x, u_{B})H(x, Dv) - b(x, u_{B})H(x, Dv)] dx \\ &+ C \int_{B}$$

We proceed estimating the various pieces arising up from (7.56).

By Assumption 2 and estimate (7.53), we get

$$I_{1} + I_{6} \leq \int_{B} \omega_{b}(|x - x_{0}|)H(x, Du)dx + \int_{B} \omega_{b}(|x - x_{0}|)H(x, Dv)dx$$
  
$$\leq \int_{B} |x - x_{0}|^{\beta}(H(x, Du) + H(x, Dv))dx$$
  
$$\leq CR^{\beta} \int_{B} H(x, Du)dx$$
  
$$\leq CR^{\beta} \int_{B} [1 + (H(x, Du))^{m}]dx.$$
(7.57)

Now, we take care of the integral  $I_2$ . From Assumption 2, Young's and Poincaré's inequalities, we infer

$$I_{2} \leq \int_{B} \omega_{b}(|u - u_{B}|)H(x, Du)dx$$
  
= 
$$\int_{B} \frac{1}{R^{\frac{\sigma}{1+\sigma}}} \omega_{b}(|u - u_{B}|)R^{\frac{\sigma}{1+\sigma}}H(x, Du)dx$$
  
$$\leq C \int_{B} \frac{1}{R} \omega_{b}(|u - u_{B}|)^{\frac{1+\sigma}{\sigma}} + R^{\sigma}(H(x, Du))^{1+\sigma}dx$$

$$\leq CR^{\sigma} \int_{B} \frac{1}{R^{1+\sigma}} |u - u_{B}|^{1+\sigma} + (H(x, Du))^{1+\sigma} dx$$
  
$$\leq CR^{\sigma} \int_{B} |Du|^{1+\sigma} + (H(x, Du))^{1+\sigma} dx$$
  
$$\leq CR^{\sigma} \int_{B} (1 + |Du|^{p(1+\sigma)}) + (H(x, Du))^{1+\sigma} dx$$
  
$$\leq CR^{\sigma} \int_{B} [1 + (H(x, Du))^{m}] dx, \qquad (7.58)$$

where  $\sigma := \min\{\beta, m-1\}.$ 

The minimality of u yields that

$$I_3 \le 0. \tag{7.59}$$

Arguing analogously as for the integral  $I_2$ , we obtain

$$I_{4} \leq \int_{B} \omega_{b}(|v-v_{B}|)H(x,Dv)dx$$
  

$$\leq CR^{\sigma} \int_{B} \frac{1}{R^{1+\sigma}}|v-v_{B}|^{1+\sigma} + (H(x,Dv))^{1+\sigma}dx$$
  

$$\leq CR^{\sigma} \int_{B} |Dv|^{1+\sigma} + (H(x,Dv))^{1+\sigma}dx$$
  

$$\leq CR^{\sigma} \int_{B} [1 + (H(x,Dv))^{m}]dx$$
  

$$\leq CR^{\sigma} \int_{B} [1 + (H(x,Du))^{m} + (H(x,D\psi))^{m}]dx, \qquad (7.60)$$

where in the last inequality we used (7.54).

Since u = v on  $\partial B$ , using Poicaré inequality for the function u - v, we infer the following estimate for  $I_5$ .

$$I_{5} \leq \int_{B} \omega_{b} (|u_{B} - v_{B}|) H(x, Dv) dx$$
  

$$\leq CR^{\sigma} \int_{B} \frac{1}{R^{1+\sigma}} \omega_{b} (|u_{B} - v_{B}|)^{\frac{1+\sigma}{\sigma}} + (H(x, Du))^{1+\sigma} dx$$
  

$$\leq CR^{\sigma} \int_{B} \frac{1}{R^{1+\sigma}} |u - v|^{1+\sigma} + (H(x, Dv))^{1+\sigma} dx$$
  

$$\leq CR^{\sigma} \int_{B} |Du|^{1+\sigma} + |Dv|^{1+\sigma} + (H(x, Dv))^{1+\sigma} dx$$
  

$$\leq CR^{\sigma} \int_{B} [1 + (H(x, Du))^{m} + (H(x, Dv))^{m}] dx$$
  

$$\leq CR^{\sigma} \int_{B} [1 + (H(x, Du))^{m} + (H(x, D\psi))^{m}] dx, \qquad (7.61)$$

where in the inequality we used estimate (7.53). Finally, inserting estimates (7.57)–(7.61) in (7.56), we get the desired estimate.  $\Box$ 

#### 7.4 Main result

In order to prove Theorem 7.0.1 we follow the strategy first proposed in [85].

Before proving our main result, in Section 7.4.1, we fix some further notation and derive a preliminary regularity theorem for solutions to (7.1).

For a ball  $\mathcal{B} \subseteq \Omega$  of radius R, we will denote by  $\mathcal{Q}_1 = \mathcal{Q}_1(\mathcal{B})$  and  $\mathcal{Q}_2 = \mathcal{Q}_2(\mathcal{B})$  the largest and the smallest cubes, concentric to  $\mathcal{B}$  and with sides parallel to the coordinate axes, contained in  $\mathcal{B}$  and containing  $\mathcal{B}$  respectively. It is easy to verify that  $|\mathcal{B}| \approx |\mathcal{Q}_1| \approx |\mathcal{Q}_2| \approx$  $R^n$ . We also denote the enlarged ball by  $\hat{\mathcal{B}} = 4\mathcal{B}$ . We set

$$\mathcal{Q}_1 = \mathcal{Q}_1(\mathcal{B}) \quad \hat{\mathcal{Q}}_2 = \mathcal{Q}_2(\hat{\mathcal{B}})$$

so that we have the following chain of inclusions

$$\mathcal{Q}_1 \subset \mathcal{B} \Subset 2\mathcal{B} \Subset \mathcal{Q}_1(\hat{\mathcal{B}}) \subset \hat{\mathcal{B}} \subset \hat{\mathcal{Q}}_2.$$

In what follows, we shall always take  $\mathcal{B}$  such that  $\mathcal{Q}_2(\hat{\mathcal{B}}) \subseteq \Omega$ .

Our next result shows that a fractional differentiability property on the gradient of the obstacle transfers to a higher fractional differentiability for the gradient of the minimizer.

**Theorem 7.4.1.** Let u be a solution to (7.1) under Assumptions 1 and 2, for exponents  $2 \le p < \frac{n}{\alpha}$ , p < q verifying

$$\frac{q}{p} < 1 + \frac{\alpha}{n}.$$

Then the following implication

$$D\psi \in B^{\gamma}_{2q-p,\infty,loc}(\Omega) \Rightarrow V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^{\sigma_{\alpha}}_{2,\infty,loc}(\Omega)$$

holds provided  $0 < \alpha < \gamma < 1$ , with  $\sigma_{\alpha} = \sigma_{\alpha}(p,q,n,\alpha,\beta,m)$ , where  $\beta$  is the exponent appearing in the Assumption 2 and where m is the minimum of the two higher integrability exponents of Theorems 7.1.2 and 7.1.3.

Proof. Let us fix arbitrary open subsets  $\Omega' \Subset \Omega'' \Subset \Omega$  and choose  $x_0 \in \Omega'$ . We recall the definition of  $\tilde{p}_1$  from (7.46). Let  $\delta \in \left(0, \frac{\alpha}{2\tilde{p}_1}\right)$  be chosen later and consider the ball  $\mathcal{B} = \mathcal{B}(x_0, |h|^{\delta})$  with |h| sufficiently small, depending on the dimension n, the parameter  $\delta$  and the distance between  $\Omega'$  and the boundary of  $\Omega''$  such that  $\hat{\mathcal{Q}}_2 \Subset \Omega''$ . Furthermore, let  $v \in u + W_0^{1,p}(B)$  be the solution to (7.5) with  $B = \hat{\mathcal{B}}$ . We estimate the difference quotient for  $V_p(Du)$  and  $\sqrt{a(x)}V_q(Du)$  as follows

$$\int_{\mathcal{B}} |\tau_{h} V_{p}(Du)|^{2} + |\tau_{h}(\sqrt{a(x)} V_{q}(Du))|^{2} dx \\
= \int_{\mathcal{B}} |V_{p}(Du(x+h)) - V_{p}(Du(x))|^{2} dx \\
+ \int_{\mathcal{B}} |\sqrt{a(x+h)} V_{q}(Du(x+h)) - \sqrt{a(x)} V_{q}(Du(x))|^{2} dx \\
\leq C \int_{\mathcal{B}} |V_{p}(Du(x+h)) - V_{p}(Dv(x+h))|^{2} dx \\
+ C \int_{\mathcal{B}} |V_{p}(Dv(x+h)) - V_{p}(Dv(x))|^{2} dx \\
+ C \int_{\mathcal{B}} |V_{p}(Dv(x)) - V_{p}(Du(x))|^{2} dx \\
+ C \int_{\mathcal{B}} |\sqrt{a(x+h)} V_{q}(Du(x+h)) - \sqrt{a(x+h)} V_{q}(Dv(x+h))|^{2} dx \\
+ C \int_{\mathcal{B}} |\sqrt{a(x+h)} V_{q}(Dv(x+h)) - \sqrt{a(x)} V_{q}(Dv(x))|^{2} dx \\
+ C \int_{\mathcal{B}} |\sqrt{a(x+h)} V_{q}(Dv(x+h)) - \sqrt{a(x)} V_{q}(Dv(x))|^{2} dx \\
+ C \int_{\mathcal{B}} |\sqrt{a(x+h)} V_{q}(Dv(x)) - \sqrt{a(x)} V_{q}(Dv(x))|^{2} dx \\$$
(7.62)

Notice that if  $x \in \mathcal{B}$ , then  $x + h \in \hat{\mathcal{B}}$ , for  $|h| \leq 1$ . Thus, we get

$$\int_{\mathcal{B}} |V_p(Du(x+h)) - V_p(Dv(x+h))|^2 dx + \int_{\mathcal{B}} |\sqrt{a(x+h)}V_q(Du(x+h)) - \sqrt{a(x+h)}V_q(Dv(x+h))|^2 dx \leq \int_{\hat{\mathcal{B}}} |V_p(Du) - V_p(Dv)|^2 + a(x)|V_q(Du) - V_q(Dv)|^2 dx.$$
(7.63)

Inserting inequality (7.63) in (7.62), we obtain

$$\int_{\mathcal{B}} |\tau_h V_p(Du)|^2 + |\tau_h(\sqrt{a(x)}V_q(Du))|^2 dx$$

$$\leq C \int_{\hat{\mathcal{B}}} |V_p(Du) - V_p(Dv)|^2 + a(x)|V_q(Du) - V_q(Dv)|^2 dx$$

$$+ C \int_{\mathcal{B}} |\tau_h V_p(Dv)|^2 + |\tau_h(\sqrt{a(x)}V_q(Dv))|^2 dx$$

$$=:J_1 + J_2.$$
(7.64)

From estimate (7.52) applied over the ball  $\hat{\mathcal{B}}$ , we infer

$$J_1 \le C|h|^{\sigma\delta} \int_{\hat{\mathcal{Q}}_2} (1 + (H(x, Du))^m + (H(x, D\psi))^m) \mathrm{d}x,$$
(7.65)

where we used that the radius of  $\hat{\mathcal{B}}$  is proportional to  $|h|^{\delta}$ . Now estimate (7.46) (see Remark 5) applied over the ball  $\mathcal{B}$  yields

$$J_{2} \leq C|h|^{\alpha - 2\delta\tilde{p}_{1}} \left( \int_{\hat{\mathcal{B}}} (1 + |Dv|^{p} + |D\psi|^{2q-p}) \mathrm{d}x \right)^{\kappa_{1}} + C|h|^{\alpha} [D\psi]^{q_{1}}_{B^{\gamma}_{2q-p,\infty}(\hat{\mathcal{B}})} + C|h|^{\alpha}, \quad (7.66)$$

recalling that the radius of  $\mathcal{B}$  is  $|h|^{\delta}$ . Inserting (7.65) and (7.66) in (7.64), we get

$$\int_{\mathcal{B}} |\tau_{h}V_{p}(Du)|^{2} + |\tau_{h}(\sqrt{a(x)}V_{q}(Du))|^{2} dx$$

$$\leq C|h|^{\sigma\delta} \int_{\hat{Q}_{2}} (1 + (H(x, Du))^{m} + (H(x, D\psi))^{m}) dx$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1}} \left( \int_{\hat{B}} (1 + |Dv|^{p} + |D\psi|^{2q-p}) dx \right)^{\kappa_{1}} + C|h|^{\alpha} [D\psi]^{q_{1}}_{B^{\gamma}_{2q-p,\infty}(\hat{\mathcal{B}})} + C|h|^{\alpha}$$

$$\leq C|h|^{\sigma\delta} \int_{\hat{Q}_{2}} (1 + (H(x, Du))^{m} + (H(x, D\psi))^{m}) dx$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1}} \left( \int_{\hat{Q}_{2}} (1 + H(x, Du)) dx \right)^{\kappa_{1}} + C|h|^{\alpha} [D\psi]^{q_{1}}_{B^{\gamma}_{2q-p,\infty}(\hat{Q}_{2})}$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1}} \left( \int_{\hat{Q}_{2}} |D\psi|^{2q-p} dx \right)^{\kappa_{1}} + C|h|^{\alpha},$$
(7.67)

where in the last inequality we used (7.53).

Now we choose  $\delta$  in order to minimize the right hand side of the previous estimate. It is easy to check that the best possible estimate is given by the choice

$$\delta = \frac{\alpha}{\sigma + 2\tilde{p}_1} \in \left(0, \frac{\alpha}{2\tilde{p}_1}\right).$$

With such a choice of  $\delta$  estimate (7.67) becomes

$$\int_{\mathcal{B}} |\tau_h V_p(Du)|^2 + |\tau_h(\sqrt{a(x)}V_q(Du))|^2 \mathrm{d}x$$
  
$$\leq C|h|^{\frac{\alpha\sigma}{\sigma+2\tilde{p}_1}} \left\{ \int_{\hat{\mathcal{Q}}_2} (1 + (H(x,Du))^m + (H(x,D\psi))^m) \mathrm{d}x + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(\hat{\mathcal{Q}}_2)} + 1 \right\}^{\kappa_2}, \quad (7.68)$$

where  $\kappa_2 := \kappa_2(n, p, q, \mu)$ , for some  $\mu < \alpha$ .

At this point, arguing as in [85, Lemma 4.5], a covering argument allows us to replace the cubes  $Q_1$  and  $\hat{Q}_2$  with the fixed open subsets  $\Omega'$  and  $\Omega''$ , respectively. Indeed for each  $|h| \in \mathbb{R}^n$  sufficiently small we can find balls  $\mathcal{B}_1 = \mathcal{B}_1(x_1, |h|^{\sigma}), ..., \mathcal{B}_K = \mathcal{B}_K(x_K, |h|^{\sigma}),$ being  $K = K(h) \in \mathbb{N}$ , such that the corresponding inner cubes  $Q_1(\mathcal{B}_1), ..., Q_1(\mathcal{B}_K)$  are disjoint and satisfy

$$\left|\Omega' \setminus \bigcup_{k=1}^{K} \mathcal{Q}_1(\mathcal{B}_k)\right| = 0.$$

By our assumption we have that  $\mathcal{Q}_2(\hat{\mathcal{B}}_k) \subset \Omega''$ , for every  $k \leq K$  and each of the dilated outer cubes  $\mathcal{Q}_2(\hat{\mathcal{B}}_k)$  intersects at most  $(16\sqrt{n})$  of the other cubes  $\mathcal{Q}_2(\hat{\mathcal{B}}_j)$ , with  $j \neq k$ . Hence, after summing up (7.68) over the inner cubes  $\mathcal{Q}_1 \in {\mathcal{Q}_1(\mathcal{B}_1), ..., \mathcal{Q}_1(\mathcal{B}_K)}$ , and enlarging the constant by a fixed factor only depending on n and p (in particular independent of h), we arrive at

$$\int_{\Omega'} |\tau_h V_p(Du)|^2 + |\tau_h(\sqrt{a(x)}V_q(Du))|^2 \mathrm{d}x$$
  
$$\leq C|h|^{\frac{\alpha\sigma}{\sigma+2\tilde{p}_1}} \left\{ \int_{\Omega''} (1 + (H(x,Du))^m + (H(x,D\psi))^m) \mathrm{d}x + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(\Omega'')} + 1 \right\}^{\kappa_2}.$$
(7.69)

Since the right hand side of the previous estimate is finite by our assumptions, it follows by arbitrariness of  $\Omega'$  that

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B_{2,\infty}^{\frac{\alpha\sigma}{2(\sigma+2\tilde{p}_1)}}(\Omega)$$
 locally.

Setting

$$\sigma_{\alpha} := \frac{\alpha \sigma}{2(\sigma + 2\tilde{p}_1)},\tag{7.70}$$

it follows the conclusion.

#### 7.4.1 Proof of Theorem 7.0.1

We are now able to give the proof of the main result of this work.

Let us consider the function

$$A(t) = \frac{\alpha\sigma}{2[2(\tilde{p}_1 - \kappa_1 t) + \sigma]}, \qquad \forall t \in \left(0, \frac{\sigma + 2\tilde{p}_1 - \sqrt{(\sigma + 2\tilde{p}_1)^2 - 4\kappa_1\alpha\sigma}}{4\kappa_1}\right) =: (0, \tilde{\sigma}),$$

$$(7.71)$$

where  $\tilde{p}_1, \kappa_1$  are defined in (7.46),  $\sigma$  is defined in Lemma 7.3.2 and  $\alpha$  is the exponent appearing in Assumption 1.

It is easy to see that  $t \mapsto A(t)$  is increasing and that

$$t < A(t) < \tilde{\sigma},\tag{7.72}$$

$$A(\tilde{\sigma}) = \tilde{\sigma}.\tag{7.73}$$

It is worth noticing that

$$\sigma_{\alpha} < \tilde{\sigma} < \frac{\alpha \sigma}{2\tilde{p}_1},\tag{7.74}$$

where  $\sigma_{\alpha}$  was introduced in (7.70). Indeed, owing to (7.70), the first part of inequality (7.74) holds if, and only if,

$$(\sigma + 2\tilde{p}_1)\sqrt{(\sigma + 2\tilde{p}_1)^2 - 4\kappa_1\alpha\sigma} < (\sigma + 2\tilde{p}_1)^2 - 2\kappa_1\alpha\sigma.$$

The last inequality is satisfied if, and only if,

$$(\sigma + 2\tilde{p}_1)^4 - 4\alpha\kappa_1\sigma(\sigma + 2\tilde{p}_1)^2 < (\sigma + 2\tilde{p}_1)^4 + 4\kappa_1^2\alpha^2\sigma^2 - 4\kappa_1\alpha\sigma(\sigma + 2\tilde{p}_1)^2,$$

that is equivalent to

$$0 < 4\kappa_1^2 \alpha^2 \sigma^2.$$

On the other hand, the second part of inequality (7.74) is valid if, and only if,

$$\tilde{p}_1(\sigma + 2\tilde{p}_1) - 2\kappa_1\alpha\sigma < \tilde{p}_1\sqrt{(\sigma + 2\tilde{p}_1)^2 - 4\kappa_1\alpha\sigma}$$

or, equivalently,

$$\tilde{p}_{1}^{2}(\sigma+2\tilde{p}_{1})^{2}+4\kappa_{1}^{2}\alpha^{2}\sigma^{2}-4\tilde{p}_{1}\kappa_{1}\alpha\sigma(\sigma+2\tilde{p}_{1})<\tilde{p}_{1}^{2}(\sigma+2\tilde{p}_{1})^{2}-4\tilde{p}_{1}^{2}\kappa_{1}\alpha\sigma.$$

The previous inequality can be written as

$$\kappa_1 \alpha \sigma - \tilde{p}_1 \sigma < \tilde{p}_1^2,$$

that holds true since  $1 < \kappa_1 < \tilde{p}_1$  and  $\alpha, \sigma \in (0, 1)$ . Let us now fix

$$\theta_0 \in \left(0, \frac{\alpha \sigma}{2(\sigma + 2\tilde{p}_1)}\right)$$

and denote

$$\theta_j = A(\theta_{j-1}), \quad \forall j \in \mathbb{N}, \ j \ge 1.$$

Hence, the sequence  $(\theta_j)_j$  is increasing and

$$\lim_{j} \theta_j = \tilde{\sigma}. \tag{7.75}$$

Now we define the sequence  $(\iota_j)_j$  inductively as follows:

$$\iota_0 = \frac{\theta_0}{2} + \frac{\alpha\sigma}{4(\sigma + 2\tilde{p}_1)} < \frac{\alpha\sigma}{2(\sigma + 2\tilde{p}_1)},$$

$$\iota_j = \frac{\theta_j + A(\iota_{j-1})}{2}.$$

Using the fact that A is increasing and (7.72), (7.73), we obtain

$$\theta_j < \iota_j < \tilde{\sigma}, \qquad \forall j \in \mathbb{N},$$

$$(7.76)$$

and therefore, from (7.75), it follows that

$$\lim_{i} \iota_j = \tilde{\sigma}.\tag{7.77}$$

Arguing by induction, we shall prove that

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^{\iota_j}_{2,\infty,\text{loc}}(\Omega) \quad \forall j \in \mathbb{N}.$$

The case j = 0 follows from Theorem 7.4.1 and our choice of  $\iota_0$ . Now, let us prove the implication

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^{\iota_{j-1}}_{2,\infty,\text{loc}}(\Omega) \Rightarrow V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^{\iota_j}_{2,\infty,\text{loc}}(\Omega).$$
(7.78)

By virtue of Lemma 5.1.3, the assumptions  $V_p(Du), \sqrt{a(x)}V_q(Du) \in B_{2,\infty,\text{loc}}^{\iota_{j-1}}(\Omega)$  imply

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in L^{\frac{2n}{n-2\lambda}}(\hat{\mathcal{Q}}_2),$$

for every  $0 < \lambda < \iota_{j-1}$  and so, recalling equality (5.1), we have that

$$|Du|^p$$
,  $a(x)|Du|^q \in L^{\frac{n}{n-2\lambda}}(\hat{\mathcal{Q}}_2)$ .

In particular, it follows

$$H(x, Du) \in L^{\frac{n}{n-2\lambda}}(\hat{\mathcal{Q}}_2),$$

for every  $0 < \lambda < \iota_{j-1}$ . Moreover, the assumption  $D\psi \in B^{\gamma}_{2q-p,\infty,\text{loc}}(\Omega)$  and Lemma 5.1.3 imply that  $D\psi \in L^{\frac{n(2q-p)}{n-\pi(2q-p)}}(\hat{Q}_2)$ , for every  $0 < \pi < \gamma$ . Therefore, using Hölder's inequality in estimate (7.67) we infer

$$\begin{split} &\int_{\mathcal{B}} |\tau_h V_p(Du)|^2 + |\tau_h(\sqrt{a(x)}V_q(Du))|^2 \mathrm{d}x \\ \leq & C|h|^{\sigma\delta} \int_{\hat{\mathcal{Q}}_2} (1 + (H(x,Du))^m + (H(x,D\psi))^m) \mathrm{d}x \\ &+ C|h|^{\alpha - 2\delta \tilde{p}_1 + 2\delta\kappa_1\lambda} \bigg( \int_{\hat{\mathcal{Q}}_2} (1 + H(x,Du))^{\frac{n}{n-2\lambda}} \mathrm{d}x \bigg)^{\frac{(n-2\lambda)\kappa_1}{n}} + C|h|^{\alpha} [D\psi]_{B^{\gamma}_{2q-p,\infty}(\hat{\mathcal{Q}}_2)}^{q_1} \end{split}$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1} + (2q-p)\delta\kappa_{1}\pi} \left( \int_{\hat{Q}_{2}} |D\psi|^{\frac{n(2q-p)}{n - \pi(2q-p)}} dx \right)^{\frac{(n - \pi(2q-p))\kappa_{1}}{n}} + C|h|^{\alpha}$$

$$\leq C|h|^{\sigma\delta} \int_{\hat{Q}_{2}} (1 + (H(x, Du))^{m} + (H(x, D\psi))^{m}) dx$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1} + 2\delta\kappa_{1}\lambda} \left( \int_{\hat{Q}_{2}} (1 + H(x, Du))^{\frac{n}{n-2\lambda}} dx \right)^{\frac{(n - 2\lambda)\kappa_{1}}{n}} + C|h|^{\alpha} [D\psi]^{q_{1}}_{B^{\gamma}_{2q-p,\infty}(\hat{Q}_{2})}$$

$$+ C|h|^{\alpha - 2\delta\tilde{p}_{1} + 2\delta\kappa_{1}\lambda} \left( \int_{\hat{Q}_{2}} |D\psi|^{\frac{n(2q-p)}{n - \pi(2q-p)}} dx \right)^{\frac{(n - \pi(2q-p))\kappa_{1}}{n}} + C|h|^{\alpha}, \qquad (7.79)$$

for some  $\pi \geq \frac{2\lambda}{2q-p}$ , where we used the fact that the radius of the cube  $\hat{Q}_2$  is proportional to  $|h|^{\delta}$ . Therefore, choosing  $\delta$  in order to maximize the right hand side of (7.79), namely

$$\delta = \frac{\alpha}{\sigma + 2(\tilde{p}_1 - k_1\lambda)},$$

we have

$$\int_{\mathcal{B}} |\tau_h V_p(Du)|^2 + |\tau_h(\sqrt{a(x)}V_q(Du))|^2 \mathrm{d}x$$

$$\leq C|h|^{\frac{\alpha\sigma}{\sigma+2(\hat{p}_1-k_1\lambda)}} \left\{ \int_{\hat{\mathcal{Q}}_2} (1+(H(x,Du))^m + (H(x,D\psi))^m) \mathrm{d}x + \int_{\hat{\mathcal{Q}}_2} (1+H(x,Du))^{\frac{n}{n-2\lambda}} \mathrm{d}x + \|D\psi\|_{B^{\gamma}_{2q-p,\infty}(\hat{\mathcal{Q}}_2)} + 1 \right\}^{\kappa^*},$$
(7.80)

where  $\kappa^* := \kappa^*(n, p, q, \mu, \lambda)$ . Thus, again through a covering argument, we deduce that

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B_{2,\infty,\text{loc}}^{\frac{\alpha\sigma}{2[\sigma+2(\tilde{p}_1-k_1\lambda)]}}(\Omega) = B_{2,\infty,\text{loc}}^{A(\lambda)}(\Omega), \quad \forall \lambda < \iota_{j-1}$$

We have just proved the following implication

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^{\iota_{j-1}}_{2,\infty,\text{loc}}(\Omega) \Rightarrow V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^t_{2,\infty,\text{loc}}(\Omega),$$
(7.81)

for all  $t < A(\iota_{j-1})$ . Since A is increasing, it follows from (7.76) that  $\theta_j < A(\iota_{j-1})$ . Moreover, the definition of  $\iota_j$  implies  $\iota_j < A(\iota_{j-1})$ . Therefore, (7.78) follows from (7.81). Besides, from (7.76) and (7.77), we infer

$$V_p(Du), \ \sqrt{a(x)}V_q(Du) \in B^t_{2,\infty,\text{loc}}(\Omega), \quad \forall t \in (0, \tilde{\sigma}).$$

It is worth noting that the exponent  $\tilde{\sigma}$  defined in (7.71) is bigger than  $\sigma_{\alpha}$ . Therefore, Theorem 7.0.1 improves the higher fractional differentiability result established in Theorem 7.4.1. Appendix

## Appendix A

# Sobolev embeddings and interpolation theory

### A.1 2D results

In the sequel we will also frequently use the following 2D interpolation inequalities:

$$\|v\|_{L^4(\Omega)} \le c \, \|v\|_V^{1/2} \, \|v\|^{1/2} \,, \tag{A.1a}$$

$$\|v\|_{L^{\infty}(\Omega)} \le c \|v\|_{H^{2}(\Omega)}^{1/2} \|v\|^{1/2}, \qquad (A.1b)$$

$$\|v\|_{L^{r}(\Omega)} \le c \|v\|_{V}^{\frac{2}{r}} \|v\|_{V}^{1-\frac{2}{r}}, \qquad r \in [2,\infty),$$
(A.1c)

$$\|v\|_{L^{r}(\Omega)} \leq c \|v\|_{L^{s}(\Omega)}^{1-\alpha} \|v\|_{L^{\infty}(\Omega)}^{\alpha}, \qquad \alpha = 1 - \frac{s}{r}, \qquad r \geq 1$$
(A.1d)

$$\|v\|_{H^{s}(\Omega)} \le c \|v\|_{H^{s_{1}}(\Omega)}^{1-\theta} \|v\|_{H^{s_{2}}(\Omega)}^{\theta}, \quad \theta = \frac{s-s_{1}}{s_{2}-s_{1}},$$
(A.1e)

$$\|v - v_{\Omega}\| \le c \|\nabla v\|, \qquad (A.1f)$$

holding for any sufficiently smooth function v and for suitable embedding constants, all denoted by the same symbol c > 0 for brevity. We will also use the following non linear Poincaré inequality (see [66])

$$\left\| v^{p/2} \right\|_{V}^{2} \le c_{p} \left( \left\| v \right\|_{L^{1}(\Omega)}^{p} + \left\| \nabla v^{p/2} \right\|^{2} \right), \tag{A.2}$$

holding for all non-negative  $v \in L^1(\Omega)$  such that  $\nabla v^{p/2} \in L^2(\Omega)$ , and for all  $p \in [2, \infty)$ . We also recall that

$$\|v\| \le c \|\nabla v\|^{1/2} \|v\|_{V'}^{1/2}, \quad \forall v \in V_0,$$
(A.3)

as one can prove simply combining the standard interpolation inequality  $||v|| \leq c ||v||_V^{1/2} ||v||_{V'}^{1/2}$ with the Poincaré-Wirtinger inequality (A.1f).

### A.2 General results

We first recall here a general interpolation result for  $L^{p}$ - spaces.

**Proposition A.2.1.** If  $f \in L^p \cap L^s$ , then  $f \in L^r$ , with r s.t.  $\frac{1}{r} = \frac{\gamma}{p} + \frac{1-\gamma}{s}$ .

The following interpolation in Sobolev spaces can be found in [90].

**Theorem A.2.2.** Let  $\Omega$  be an open and bounded subset of  $\mathbb{R}$ . Let X, Y be Hilbert spaces. If  $s_1, s_2 \geq 0$ ,

(i) then we have

$$[H^{s_1}(\Omega; X), H^{s_2}(\Omega, Y)]_{\theta} = H^{(1-\theta)s_1+\theta s_2}(\Omega; [X, Y]_{\theta});$$
(A.4)

(ii) if  $s_2 \neq \mu + \frac{1}{2}$  (integer  $\mu \ge 0$ ), then we have

$$[H^{s_1}(\Omega), H^{-s_2}(\Omega)]_{\theta} = H^{(1-\theta)s_1 - \theta s_2}(\Omega)$$
(A.5)

if 
$$(1-\theta)s_1 - \theta s_2 \neq -\frac{1}{2} - \nu$$
 (integer  $\nu \ge 0$ ).

The next theorem can be found in [49].

**Theorem A.2.3.** Suppose  $u \in L^2(0,T; H^1_0(U))$  with  $u' \in L^2(0,T; H^{-1}(U))$ .

(i) Then

$$u \in C([0,T]; L^2(U)).$$
 (A.6)

(after possibly being redefined on a set of measure zero).

(ii) The mapping

$$t \mapsto ||u(t)||^2_{L^2(U)}$$

is absolutele continuous, with

$$\frac{d}{dt} \|u(t)\|_{L^{2}(U)}^{2} = 2\langle u'(t), u(t) \rangle$$

for a.e.  $0 \le t \le T$ .

(iii) Furthermore, we have the estimate

$$\max_{0 \le t \le T} \|u(t)\|_{L^2(U)}^2 \le C \left( \|u\|_{L^2(0,T;H_0^1(U))}^2 + \|u\|_{L^2(0,T;H^{-1}(U))}^2 \right),$$

the constant C depending only on T.

**Lemma A.2.4** (Aubin-Lions). Let  $X_0, X$  and  $X_1$  be Banach spaces with  $X_0 \subset X \subset X_1$ . Suppose that  $X_0 \hookrightarrow X$  and  $X \hookrightarrow X_1$ . For  $1 \leq p, q \leq +\infty$ , let

$$W = \{ u \in L^p([0,T]; X_0) \ s.t. \ u' \in L^q([0,T]; X_1) \}.$$

- (i) If  $p < +\infty$ , then  $W \hookrightarrow L^p([0,T];X)$ .
- *ii)* If  $p = +\infty$  and q > 1, then  $W \hookrightarrow C([0, T]; X)$ .

The following compactness embeddings can be found in [12].

**Theorem A.2.5.** Let  $0 \le r \le s$  and let  $\Omega \subset \mathbb{R}^n$  be bounded. For  $1 \le p \le q \le \infty$ , if  $r - \frac{n}{q} < s - \frac{n}{p}$ , then

$$W^{s,p}(\Omega) \hookrightarrow \to W^{r,q}(\Omega).$$

**Theorem A.2.6.** Let  $0 \le r \le s$ ,  $1 \le p \le q \le \infty$ ,  $V \hookrightarrow W$  and  $r - \frac{1}{q} \le s - \frac{1}{p}$ . Then

$$W^{s,p}(0,T;V) \hookrightarrow W^{r,q}(0,T;W).$$

In particular, the inclusion is compact if  $r - \frac{1}{q} < s - \frac{1}{p}$  and  $V \hookrightarrow W$ .

## Appendix B

### **Topological degree**

The content of this Appendix can be found in [119].

Let  $\mathcal{A} = \{(f, \Omega, p) \text{ s.t. } \Omega \subseteq \mathbb{R}^N \text{ open and bounded}, f : \overline{\Omega} \to \mathbb{R}^N \text{ continuous}, p \notin \mathbb{R}^N \setminus f(\partial\Omega)\}.$ 

**Definition B.1.** The topological degree is a function  $d : \mathcal{A} \to \mathbb{Z}$  s.t.

- d1. Normalization property  $d(\mathrm{id},\Omega,p)=1, \quad p\in\Omega.$
- d2. DOMAIN ADDITIVITY Let  $(f, \Omega, p) \in \mathcal{A}$  be fixed and  $\Omega_1, \Omega_2 \subset \Omega$  open be given s.t.  $\Omega_1 \cup \Omega_2 = \emptyset$  and  $p \notin f(\overline{\Omega} \setminus (\Omega_1 \cup \Omega_2))$  $\Rightarrow d(f, \Omega, p) = d(f, \Omega_1, p) + d(f, \Omega_2, p).$
- d3. HOMOTOPY INVARIANCE  $h: [0,1] \times \overline{\Omega} \to \mathbb{R}^N$  continuous  $p: [0,1] \to \mathbb{R}^N$  continuous and s.t.  $p(t) \notin h(t, \partial \Omega)$  $\Rightarrow d(h(t, \cdot), \Omega, p(t)) = const$  w.r.t. t.

**Proposition B.0.1.** The topological degree has the following properties:

P0.  $d(f, \emptyset, p) = 0.$ 

P1. EXCISION

 $\forall (f, \Omega, p) \in \mathcal{A}, \forall \Omega_1 \subset \Omega \text{ open}, p \notin f(\bar{\Omega} \setminus \Omega_1)$  $\Rightarrow d(f, \Omega, p) = d(f, \Omega_1, p).$ 

$$\begin{split} P\mathcal{Z}. \ \forall (f,\Omega,p) \in \mathcal{A}, \forall p \not\in f(\bar{\Omega}) \\ \Rightarrow d(f,\Omega,p) = 0. \end{split}$$

P3. EXISTENCE  $\forall (f, \Omega, p) \in \mathcal{A}, d(f, \Omega, p) \neq 0$   $\Rightarrow \exists x \in \Omega : f(x) = p \text{ with } p \notin f(\partial \Omega).$ Namely, the equation has at least one solution.

We also report here the following result, contained in ([119]).

**Theorem B.0.2** (Borsuk). Let  $\Omega \subset \mathbb{R}^n$  be opened bounded and symmetric with respect to  $0 \in \Omega$ . Let  $f \in C(\overline{\Omega})$  be odd and  $0 \notin f(\partial \Omega)$ . Then the topological degree  $d(f, \Omega, 0)$  is odd.

## Acknowledgements

- Prof. Michela Eleuteri, my supervisor.
   I could never be enough grateful for all the trust I have received during these years and for having guided me with contagious enthusiasm, but also pragmatism, since my Bachelor's studies.
- Prof. Pavel Krejčí.

It has been an honor to work and learn during these years. His spirit and cordiality truly make mathematical difficulties look less scary.

- Dr. Chiara Gavioli. A colleague and a friend, truly a mentor.
- All the colleagues met during these years. They all have been an insipiration and an interesting source of formative discussion.
- Emanuele. Thank you for being the cornerstone of a lifetime.
- Anna.

My companion and my best friend, we are finally able to say *cat*.

• My family.

Thank you for not having laughed in my face when in middle school I said that I wanted to be a mathematician.

## Bibliography

- N.D. Alikakos, L<sup>p</sup>-bounds of solutions of reaction-diffusion equations, Communications in Partial Differential Equations, 4(8) (1979), 827–868.
- [2] N. D. Alikakos, P. W. Bates and X. Chen, The convergence of the solution of the Cahn-Hilliard equation to the solution of Hele-Shaw model, Arch. Ration. Mech. Anal., 128 (1994), 165–205.
- [3] M.B.P. Amato, C.S.V. Barbas, D.M. Medeiros, R.B. Magaldi, G.P. Schettino, G. Lorenzi-Filho, R.A. Kairalla, D. Deheinzelin, C. Munoz, R. Oliveira, T.Y. Takagaki and C. R. Ribeiro Carvalho, Effect of a protective-ventilation strategy on mortality in the acute respiratory distress syndrome, *New England Journal of Medicine*, **338** (1998), 347–354.
- [4] R. Amini, K. Creeden and U. Narusawa, A mechanistic model for quasistatic pulmonary pressure-volume curves for inflation, *Journal of Biomechanical Engineering*, 127 (2005), 619–629.
- [5] R.P. Araujo and D.L.S. McElwain, A History of the Study of Solid Tumour Growth: The Contribution of Mathematical Modelling, *Bull. Math. Biol.* 66 (2004), 1039– 1091.
- [6] A.L. Baison, A. Clop, R. Giova, J. Orbitg and A. Passarelli di Napoli, Fractional differentiability for solutions of nonlinear elliptic equations, *Potential Anal.* 46(3) (2017) 403–430.
- [7] G. Bankoff, S.H. Davis and A. Oron, Long-scale evolution of thin liquid films, *Rev. Mod. Phys.*, 69 (1997), 931–980.
- [8] P. Baroni, M. Colombo and G. Mingione, Regularity for general functionals with double phase, Calculus of Variations 57 (2018).
- [9] N. Bellomo, N.K. Li, and P.K. Maini, On the foundations of cancer modelling: selected topics, speculations, and perspectives, *Math. Models Methods Appl. Sci.*, 18(4) (2008), 593–646.

- [10] H. Brezis, Functional analysis, Sobolev spaces and partial differential equations, Universitext. Springer, New York, 2011.
- [11] H. Brezis and D. Kinderlehrer, The smoothness of solutions to nonlinear variational inequalities, *Indiana Univ. Math. J.* 23 (1973–1974), 831–844.
- [12] F. Brezzi and G. Gilardi, Functional Analysis, Functional Spaces, Partial Differential Equations in: H. Kardestuncer and Norrie eds., Finite Element Handbook, McGraw-Hill Book Co., New York, (1987). Chapters 1-3, pp. 1-121 of Part 1.
- [13] M. Brokate and J. Sprekels, *Hysteresis and phase separation*, Vol. 121. App. Math. Sci. Chapter 4. Springer, 1996.
- [14] M. Bulíček, E. Feireisl and J. Máalek, A Navier-Stokes-Fourier system for incompressible fluids with temperature dependent material coefficients, *Nonlinear Anal. Real World Appl.*, **10** (2009), 992-1015.
- [15] J.M. Bull, Whole body hyperthermia as an anticancer agent, CA Cancer J Clin., 32(2) (1982), 123–128.
- [16] H.M. Byrne and M.A.J. Chaplain, Growth of Nonnecrotic Tumors in the Presence and Absence of Inhibitors, *Mathematical Biosciences*, **130** (1995), 151–181.
- [17] L.A. Caffarelli and D. Kinderlehrer, Potential methods in variational inequalities, J. Anal. Math. 37 (1980), 285–295.
- [18] J. W. Cahn, On the spinodal decomposition, Acta Metall., 9 (1961), 795–801.
- [19] J. W. Cahn and J. Hilliard, Free energy of a non uniform system. I. Interfacial free energy, J. Chem. Phys., 28 (1958), 258–267.
- [20] C. Cavaterra, S. Frigeri, M. Grasselli, Nonlocal Cahn-Hilliard-Hele-Shaw Systems with Singular Potential and Degenerate Mobility, J. Math. Fluid Mech. 24 (13) (2022).
- [21] L. Cherfils, S. Gatti, A. Miranville, R. Guillevin, Analysis of a model for tumor growth and lactate exchanges in a glioma, *Discrete and Continuous Dynamical Sys*tems - S, doi: 10.3934/dcdss.2020457.
- [22] H.J. Choe and J.L. Lewis, On the obstacle problem for quasilinear elliptic equations of p-Laplace type, SIAM J. Math. Anal. 22(3) (1991), 623–638.

- [23] M. Cloetta, Untersuchungen über die Elastizität der Lunge und deren Bedeutung für die Zirkulation, Pflüger's Archiv für die gesamte Physiologie des Menschen und der Tiere 152 (1913), 339–364.
- [24] A. Clop, R. Giova, A. Passarelli di Napoli, Besov regularity for solutions of pharmonic equations, Adv. Nonlinear Anal. 8(1) (2019), 762–778.
- [25] D. Cohen and J. M. Murray, A generalized diffusion model for growth and dispersion in population, J. Math. Biol., 12 (1981), 237–248.
- [26] M. Colombo and G. Mingione, Regularity for double phase variational problems, Arch. Ration. Mech. Anal. 215 (2015), 443–496.
- [27] A. Coscia, Regularity for minimizers of double phase functionals with mild transition and regular coefficients, J. Math. Anal. Appl. (2021).
- [28] V. Cristini and J. Lowengrub, Multiscale modeling of cancer. An Integrated Experimental and Mathematical Modeling Approach, Cambridge Univ. Press, Cambridge, (2010).
- [29] G. Cupini, N. Fusco and R. Petti, Hölder continuity of local minimizers, J. Math. Anal. Appl. 235 (1999), 578–597.
- [30] G. Cupini, M. Guidorzi and E. Mascolo, Regularity of minimizers of vectorial integrals with p, q-growth, Nonlinear Analysis, 54 (2003), 591–616.
- [31] G. Cupini, P. Marcellini, E. Mascolo and A. Passarelli di Napoli, Lipschitz regularity for degenerate elliptic integrals with p, q-growth, Adv. Calc. Var. 16(2) (2023), 443– 465.
- [32] C. De Filippis and G. Mingione, Lipschitz bounds and nonautonomous integrals, Arch. Rational Mech. Anal. 242 (2021), 973–1057.
- [33] D. Demory, J.M. Arnal, M. Wysocki, S. Donati, I. Granier, G. Corno and J. Durand-Gasselin, Recruitability of the lung estimated by the pressure volume curve hysteresis in ARDS patients, *Intensive Care Med* 34 (2008), 2019–2025.
- [34] J. Dockery and I. Klapper, Role of cohesion in the material description of biofilms, *Phys. Rev. E* 74 (2006), 0319021–0319028.
- [35] C. Domb, *The Critical Point*, Taylor and Francis, 1996.
- [36] F. Duzaar, Variational inequalities and harmonic mappings, J. reine angew. Math. 374 (1987), 39–60.

- [37] F. Duzaar and M. Fuchs, Optimal regularity theorems for variational problems with obstacles, *manuscripta math.* **56** (1986), 209–234.
- [38] M. Eleuteri, Hölder continuity results for a class of functionals with non standard growth, Boll. Unione Mat. Ital. 8, 7-B (2004), 129–157.
- [39] M. Eleuteri, S. Gatti and G. Schimperna, Regularity and long-time behavior for a thermodynamically consistent model for complex fluids in two space dimensions, *Indiana Univ. Math. J.* 68 (2019), 1465–1518.
- [40] M. Eleuteri, E. Ipocoana, J. Kopfová and P. Krejčí, Periodic solutions of a hysteresis model for breathing, *ESAIM: Math. Model. Num. Anal.* 54 (1) (2020), 255–271.
- [41] M. Eleuteri, J. Kopfová and P. Krejčí, Fatigue accumulation in a thermo-viscoelastoplastic plate, Discrete Cont. Dynam. Syst, Ser. B, 19 (2014), 2091–2109.
- [42] M. Eleuteri, L. Lussardi and U. Stefanelli, Thermal control of the Souza-Auricchio model for shape memory alloys, *Discrete Cont. Dynam. Syst.*, Ser. S, 6 (2013), 369–386.
- [43] M. Eleuteri and A. Passarelli di Napoli, Higher differentiability for solutions to a class of obstacle problems, *Calc. Var.* 57, 115 (2018).
- [44] M. Eleuteri and A. Passarelli di Napoli, Regularity results for a class of nondifferentiable obstacle problems *Nonlinear Analysis*, **194**, 111434 (2020).
- [45] M. Eleuteri and A. Passarelli di Napoli, On the validity of variational inequalities for obstacle problems with non-standard growth, to appear in *Annales Academiae Scientiarum Fennicae Mathematica* (2021).
- [46] M. Eleuteri, E. Rocca and G. Schimperna, On a non-isothermal diffuse interface model for two phase flows of incompressible fluids, *DCDS*, **35.6** (2015), 2497–2522.
- [47] M. Eleuteri, E. Rocca and G. Schimperna, Existence of solutions to a two-dimensional model for nonisothermal two-phase flows of incompressible fluids, Ann. Inst. H. Poincare Anal. Non Lineaire, 33 (2016), 1431–1454.
- [48] C. M. Elliot and Z. Songmu, On the Cahn-Hilliard equation, Arch. Ration. Mech. Anal., 96.4 (1986), 339–357.
- [49] L.C. Evans, Partial Differential Equations. Providence, R.I.: American Mathematical Society, (1998).

- [50] E. Feireisl, E. Rocca, G. Schimperna, and A. Zarnescu, Nonisothermal nematic liquid crystal flows with the Ball-Majumdar free energy, Ann. Mat. Pura Appl., 194 (2015), 1269–1299.
- [51] G. Fichera, Problemi elastostatici con vincoli unilaterali: il problema di Signorini con ambigue condizioni al contorno, Atti Accad. Naz. Lincei Mem. Cl. Sci. Fis. Mat. Nat. Sez. Ia 7 (8) (1963-1964), 91–140.
- [52] D. Flynn, A survey of hysteresis models of mammalian lungs, Rend. Sem. Mat. Univ. Politec. Torino 72 (1-2) (2014), 17–36.
- [53] I. Fonseca, J. Maly and G. Mingione, Scalar minimizers with fractal singular sets, Arch. Ration. Mech. Anal. 72 (2) (2004), 295–307.
- [54] N. Foralli and G. Giliberti, Higher differentiability of solutions for a class of obstacle problems with variable exponents, **313** (2022), 244–268.
- [55] M. Frémond, Non-smooth thermomechanics, Springer, 2002.
- [56] S. Frigeri, M. Grasselli and E. Rocca, On a diffuse interface model of tumor growth, *Eur. J. Appl. Math.*, 26 (2015), 215–243.
- [57] M. Fuchs, Hölder continuity of the gradient for degenerate variational inequalities, Nonlinear Anal. 15 (1990), 85–100.
- [58] H. Garcke and K.F. Lam, Well-posedness of a Cahn-Hilliard system modelling tumour growth with chemotaxis and active transport, *Eur. J. Appl. Math.* 28 (2017), 284–316.
- [59] H. Garcke, K.F. Lam, R. Nürnberg and E. Sitka, A multiphase Cahn-Hilliard-Darcy model for tumour growth with necrosis, *Math. Models Methods Appl. Sci.* 28 (2018), 525–577.
- [60] H. Garcke, K.F. Lam, E. Sitka and V. Styles, A Cahn-Hilliard-Darcy model for tumour growth with chemotaxis and active transport, *Math. Models Methods Appl. Sci.* 26(6) (2016), 1095–1148.
- [61] C. Gavioli, Higher differentiability of solutions to a class of obstacle problems under non-standard growth conditions, *Forum Math.* **31**(6) (2019), 1501–1516.
- [62] C. Gavioli, A priori estimates for solutions to a class of obstacle problems under p, qgrowth conditions Journal of Elliptic and Parabolic Equations, 5 (2019), 325–347.

- [63] M. Giaquinta, Growth conditions and regularity, a counterexample, Manuscripta Math. 59 (2) (1987), 245–248.
- [64] E. Giusti, Direct methods in the calculus of variations, World scientific publishing Co., Singapore, 2003.
- [65] R. Giova, A. Passarelli di Napoli, Regularity results for a priori bounded minimizers of non-autonomous functionals with discontinuous coefficients, Adv. Calc. Var. 12(1) (2019), 85–110.
- [66] M. Grasselli et al., Analysis of the Cahn-Hilliard equation with the chemical potential dependent mobility, *Commun. Partial Differ. Equ.*, 36 (2011), 1193–1238.
- [67] A.G. Grimaldi and E. Ipocoana, Higher fractional differentiability for solutions to a class of obstacle problems with non-standard growth conditions, Adv. Calc. Var. (2022).
- [68] A.G. Grimaldi and E. Ipocoana, Higher differentiability results in the scale of Besov spaces to a class of double-phase obstacle problems, *ESAIM: COCV* **28** (2022), 51.
- [69] M.E. Gurtin, Generalized Ginzburg-Landau and Cahn-Hilliard equations based on a microforce balance, *Physica D* 92 (1996), 178–192.
- [70] D. Haroske, Envelopes and sharp embeddings of function spaces, Chapman and Hall CRC, Boca Raton (2006).
- [71] R.S. Harris, Pressure-volume curves of the respiratory system, *Respir. Care* 50 (1) (2005), 78–99.
- [72] A. Hawkins-Daarud, S. Prudhomme, K.G. van der Zee and J.T. Oden, Bayesian calibration, validation, and uncertainty quantitation of diffuse interface models of tumor growth, *Journal of Mathematical Biology* 67 (2013), 1457–1485.
- [73] J. Hildebrandt, Pressure-volume data of cat lung interpreted by a plastoelastic, linear viscoelastic model, *Journal of Applied Physiology* 28 (1970), 365–372.
- [74] A. Hönig, B. Niethammer and F. Otto, On first-order corrections to LSW theory I: Infinite systems, J. Stat. Phys., 119 (2005), 61–122.
- [75] E. Ipocoana, On a non-isothermal Cahn-Hilliard model for tumor growth, J. Math. Anal. Appl 506(2) 125665 (2022).
- [76] E. Ipocoana and P. Krejčí, A model for assisted periodic breathing with degenerate permeability, submitted (2022).

- [77] E. Ipocoana and A. Zafferi, Further regularity and uniqueness results for a nonisothermal Cahn-Hilliard equation, *Comm. Pure Appl. Anal.* **20**(2) (2021), 763–782.
- [78] J. Jiang, H. Wu and S. Zheng, Well-posedness and long-time behavior of a nonautonomous Cahn-Hilliard-Darcy system with mass source modeling tumor growth, *J. Differ. Equ.* 259 (2015), 3032–3077.
- [79] M. Kasprowicz, M. Czosnyka, Z. Czosnyka, S. Momjian, P. Smielewski, H. Juniewicz and J.D. Pickard, Hysteresis of the cerebrospinal pressure-volume curve in hydrocephalus, *Acta Neurochir. Supp.* 86 (2003), 529–532.
- [80] D. Kinderlehrer and G. Stampacchia, An Introduction to Variational Inequalities and Their Applications, *Academic Press, Cambridge* (1980).
- [81] P. Koskela, D. Yang and Y. Zhou, Pointwise characterizations of Besov and Triebel-Lizorkin spaces and quasiconformal mappings, Adv. Math 226 (4) (2011) 3579–3621.
- [82] E. Knobloch and U. Thiele, Thin liquid films on a slightly inclined heated plate, *Physica D*, **190** (2004), 213–248.
- [83] M. A. Krasnosel'skii and A. V. Pokrovskii, Systems with Hysteresis, Moscow: Nauka, (1983).
- [84] P. Krejčí, *Hysteresis, Convexity and Dissipation in hyperbolic equations*, Tokyo: Gakkotosho, (1996).
- [85] J. Kristensen and G. Mingione, Boundary regularity in variational problems, Arch. Ration. Mech. Anal. 180 (2006), 331–398.
- [86] S.J. Lai-Fook and R.E. Hyatt, Effects of age on elastic moduli of human lungs, J. Appl. Physiol. 89 (2000), 163–168.
- [87] Ph. Laurençot, Solutions to a Penrose-Fife model of phase-field type, J. Math. Anal. Appl., 185 (1994), 262–274.
- [88] E. Lee, T. Chung, K. Kim, B. Bae, B. Kim, S. Kim, D. Ryu, S. Bae and K. Ha, Macrophage Stimulated by Low Ambient Temperature Hasten Tumor Growth via Glutamine Production, *Biomedicines*, 8(2020), 381.
- [89] I.M. Lifshitz and V.V. Slyozov, The kinetics of precipitation for supersaturated solid solutions, J. Phys. Chem. Solids, 19 (1961), 35–50.
- [90] J.L. Lions and E. Magenes Non-Homogeneous Boundary Value Problems and Applications: Vol. 1, Springer (1971).

- [91] S. Liu, S.S. Margulies and T.A. Wilson, Deformation of the dog lung in the chest wall, J. Appl. Physiol. 68 (1990), 1979–1987.
- [92] J. Lowengrub, E. Titi and K. Zhao, Analysis of a mixture model of tumor growth, European J. Appl. Math. 24 (2013), 1–44.
- [93] S. K. Ma, *Statistical Mechanics*, World Scientific, 1985.
- [94] P. Marcellini, On the definition and the lower semicontinuity of certain quasiconvex integrals, Ann. Inst. H. Poincaré Anal. Non Linéaire 3 (1986), no.5, 391–409.
- [95] P. Marcellini, Regularity of minimizers of integrals of the calculus of variations with non-standard growth conditions, Arch. Ration. Mech. Anal. 105(3) (1989), 267-284.
- [96] P. Marcellini, Regularity and existence of solutions of elliptic equations with p, q-growth conditions, J. Differential Equations **90**(1) (1991), 1-30.
- [97] A. Marveggio and G. Schimperna, On a non-isothermal Cahn-Hilliard model based on a microforce balance, J. Differ. Equ., 274 (2020).
- [98] T. Matsuda, Hyperthermia in the treatment of cancer, Gan No Rinsho, 32(10) (1986), 1095–1099.
- [99] J. Mead, Mechanical properties of lungs, *Physiological Reviews*, **41** (1961), 281–330.
- [100] A. Miranville, Generalized Cahn-Hilliard equations based on a microforce balance, (2002).
- [101] A. Miranville, The Cahn-Hilliard Equation: Recent Advances and Applications, Society for Industrial and Applied Mathematics, U.S., 2019.
- [102] A. Miranville, E. Rocca and G. Schimperna, On the long time behavior of a tumor growth model, J. Differential Equations, 267 (2019), 2616–2642.
- [103] M.L. Mogensen, A physiological mathematical model of the respiratory system, PhD thesis, Center for Model-based Medical Decision Support, Department of Health Science and Technology, Aalborg University, Denmark, (2011).
- [104] J.T. Oden, A. Hawkins and S. Prudhomme, General diffuse-interface theories and an approach to predictive tumor growth modeling, *Math. Models Methods Appl. Sci.* 20 (2010), 477–517.
- [105] J. Ok, Regularity of ω-minimizers for a class of functionals with non-standard growth, Calc. Var. & PDE, 56(48) (2017).

- [106] A. Passarelli di Napoli, Higher differentiability of minimizers of variational integrals with sobolev coefficients, Adv. Calc. Var. 7(1) (2014), 59–89.
- [107] A. Passarelli di Napoli, Regularity results for non-autonomous variational integrals with discontinuous coefficients, Atti Accad. Naz. Lincei, Rend. Lincei Mat. Appl. 26 (4) (2015), 475–496.
- [108] Y.P. Raizer and Y.B. Zeldovich, Physics of Shock Waves and High-temperature Hydrodynamic Phenomena, Academic Press, 1967.
- [109] E.A. Repasky, S.S. Evans and M.W. Dewhirst, Temperature Matters! And Why it Should Matter to Tumor Immunologists, *Cancer Immunol Res.* 1(4) (2013), 210–216.
- [110] G. L. Rohdenburg, Fluctuations in the Growth Energy of Malignant Tumors in Man, with Especial Reference to Spontaneous Recession, J. Cancer Research, 3 193, (1918).
- [111] G. Stampacchia, Formes bilineaires coercivitives sur les ensembles convexes, C.R. Acad. Sci. Paris 258 (1964), 4413–4416.
- [112] R. Temam, Infinite-Dimensional Dynamical Systems in Mechanics and Physics, Vol. 68. AMS. Springer, New York, NY, 1988.
- [113] S. Tremaine, On the origin of irregular Saturn's rings, Astron. J., 125 (2003), 894–901.
- [114] K. Uhlenbeck, Regularity for a class of non-linear elliptic systems, Acta Math. 138 (3–4) (1977), 219–240.
- [115] J. Van der Zee, Heating the patient: A promising approach?, Annals of Oncology, 13 (2002), 1173–1184.
- [116] A. Visintin, *Differential Models of Hysteresis*, Berlin-Heidelberg: Springer, (1994).
- [117] T.V. Wilson, Relations among recoil pressure, surface area, and surface tension in the lung, J. Appl. Physiol. Respirat. Environ. Exercise Physiol. 50 (1981), 921–926.
- [118] X. Wu, G.J. van Zwieten and K.G. van der Zee, Stabilized second-order convex splitting schemes for Cahn-Hilliard models with applications to diffuse-interface tumor-growth models, *Int. J. Numer. Meth. Biomed. Engng.* **30** (2014), 180–203.
- [119] E. Zeidler, Nonlinear Functional Analysis and its Applications: I. Fixed Point Theorems. New York: Springer-Verlag, (1986).

- [120] X. Zhang and S. Zheng, Besov regularity for the gradients of solutions to nonuniformly elliptic obstacle problems, *Journal of Mathematical Analysis and Applications* 505(2) (2021).
- [121] V.V. Zhikov, Averaging of functionals of the calculus of variations and elasticity theory, *Izv. Akad. Nauk SSSR Ser. Mat.* **50** (1986), 675–710.
- [122] V.V. Zhikov, On some variational problems, Russian J. Math. Phys. 5 (1997), 105–116.